

Ι.

П.

Ш.

IV.

V.

VI.

VII.

Oneida Business Committee

Regular Meeting 9 a.m. Wednesday, March 11, 2015 BC Conference Room, 2nd floor, Norbert Hill Center

Agenda



VIII. Continuing Resolution Reports

- A. Accept Oneida Gaming Commission continuing resolution closeout report Sponsor: Mark A. Powless, Sr., Chairman
- B. Accept Oneida Gaming Division continuing resolution closeout report Sponsor: Louise Cornelius, Gaming General Manager

- C. Accept Department of Public Works continuing resolution closeout report Sponsor: Bruce Danforth, Asst. Division Director/Development Operations
- D. Accept Final Audit Memo and Delete the continuing resolution closeout reports Sponsor: Lisa Summers, Tribal Secretary

IX. Standing Committees

A. Legislative Operating Committee

Sponsor: Councilman Brandon Stevens, Chair

1. Accept Legislative Operating Committee meeting minutes of February 18, 2015

B. Finance Committee

Sponsor: Treasurer Trish King, Chair

1. Accept Finance Committee minutes of March 2, 2015

20

- C. <u>Community Development</u> a ning Committee (No Requested Action)
- D. Querry of Lie (No Request of acti
- X. General Cou
- XI. Unfinished Busines (No K juested Action)
- XII. Tabled Business (No Require

XIII. New Business

- A. Approve Limited Waiver of Sovereign mmulity Anthem Live Closs slue Shield Contract #2014-1170
 Sponsor: Debbie Danforth, Division Director/Coretations-Compresensive Heat
- B. Approve Limited Waiver of Sovereign Immunity United Healthcare Insurance Contract #2014-0648

Sponsor: Debbie Danforth, Division Director Operations-Comprehensive health

- C. Approve Limited Waiver of Sovereign Immunity rise Health Fun Insurance Contract #2015-0018
 Sponsor: Debbie Danforth, Division Director/Operation - Complete Health
- D. Approve Valley Forge Lobbying Gifts Liaison: Brandon Stevens, Councilman
- E. Accept Oneida Nation School Board's SOP for Contract Personnel Salaries and Benefits Liaison: Fawn Billie, Councilwoman
- F. Accept quarterly reporting update and direct appropriate follow-up Sponsor: Lisa Summers, Tribal Secretary
- **G.** Approve Kelly L. Skenandore to represent the Tribe as a member of the TribalNet Advisory Board Sponsor: Debbie Danforth, Division Director/Operations

- H. Support Dissertation Research Review Sponsor: Jo Anne House, Chief Counsel
- I. Accept Self-Funded Health Insurance Rate Financial Impact 1st Quarter Report Sponsor: Larry Barton, Chief Financial Officer

XIV. Travel

A. Travel Reports

1. Accept travel report for Councilman Brandon Stevens – NCAI – Oct. 27-31, 2014 Sponsor: Brandon Stevens, Councilman

B. Travel Requests

1. Tribal Secretary Lisa Summers – Crimes Against Children in Indian Country Conference – Carlton MN, Apr. 27-30, 2015

Sponsor: Lisa Summers, And al Secretary

XV. Reports This section of the active answer the luled to begin at 1:30 p.m.)

- A. Operational eports
 - 1. Organizational Development Let nie Bekhart, Supervisor
 - 2. Self-C ... that is Chris Johns, Coordinat

B. Corporate Reports

- 1. Oneida Airport Hot Cortoration Jan ce Skenandor Hirth, Agent
- 2. Oneida Golf Enternine Corporation Janice Skenandore-Hith, Chairwoman
- **3.** Bay Bancorporation In _ Jeff Bowman, President
- 4. Oneida Total Integrated Emerprises When "Butch" Rentmenster Chairman
- 5. Oneida Engineering Science & Onstructic Group, LLC Jokie Jalim, Chairwoman

C. Boards Committees and Commissions

- Land Claims Commission Amelia Conclius, Charwoman Liaison: Brandon Stevens, Councilman
 Excerpt from February 25, 2015: (1) Motion by Jennifer Woster to halt stipends to the Land Claims Commission until the Land Claims Commission's quarteringepoint, submitted to and accepted by the Business Committee, seconded by Lisa Summers. Motion carried unanimously.
 (2) Motion by Lisa Summers to defer the concern regarding deimque Caporting to the Land Claims Commission's liaisons, Brandon Stevens and Jennifer Wesster, freuollow up, seconded by Melinda J. Danforth. Motion carried unanimously.
- Land Commission Amelia Cornelius, Chairwoman Liaison: Tehassi Hill, Councilman
 Excerpt from February 25, 2015: Motion by Melinda J. Danforth to defer the Oneida Land Commission quarterly report to the March 11, 2015 Business Committee meeting, seconded by Lisa Summers. Motion carried unanimously.

XVI. Executive Session

A. Executive Session meeting minutes of March 10, 2015 (No Requested Action)

B. Reports

- 1. Oneida Seven Generations Corporation Gene Keluche, Sagestone Management LLC
- 2. Oneida Airport Hotel Corporation Janice Skenandore-Hirth, Agent
- 3. Oneida Golf Enterprise Corporation Janice Skenandore-Hirth, Chairwoman
- 4. Bay Bancorporation Inc. Jeff Bowman, President
- 5. Oneida Total Integrated Enterprises William "Butch" Rentmeester, Chairman
- 6. Oneida Engineering Science & Construction Group, LLC Jackie Zalim, Chairwoman
- 7. Chief Counsel report Jo Anne House, Chief Counsel
- 8. Officers' report Melinder Danforth, Tribal Vice-Chairwoman

C. <u>Audit mittee</u>

- Sporsor: Councilman Topussi Lui, Av dit Committee Chairman
- Accept the Slot Compliance Auditano if the confidentiality requirement allowing Tribal members to view the audit.
 Species: Tribalsi Hill, Councillian
 Excert to make the intervention of the second secon

D. Unfinished Busine s

1. Approve Limited Wriver of Sovereign Immunity for risits Oneida, LLC, Contract #2015-0110 Sponsor: Tehassi Hill Courtained

Excerpt from Marc. 03. 015: (1) Notion by Melada J. Dar orth to defer this item to the regular Business Committee meeting of March 11, 2015, in order or us unduct further due diligence, seconded by Brandon Stavens. Motion arried unanime sly. (1) Notion by Brandon Stavens for the Chairwoman's Office set up concerns the between the Land Commission and the Business Committee before March 11, 2015, including a memorandum identifying the tail of concerns, seconded by Tehassi Hill. Motion carried unanimously.

Excerpt from February 25, 2015: Notion by Melinda J. Danforth to rove this arenda item to the special Business Committee meeting that will be scheduled on or beinre March 94, 2015, and to direct Law Office to provide the necessary follow-up formation, seconded by Lisa Summers. Motion carried unanimously.

E. <u>Tabled Business</u> (No Requested Action)

F. <u>New Business</u>

- 1. Ratify e-poll: approve procedural exception for offer to purchase 2 poperties Sponsor: Lisa Summers, Tribal Secretary
- 2. Approve State Lobbyist contract Sponsor: Lisa Summers, Tribal Secretary
- **3.** Approve Attorney Sweeney contract #2015-0096 Liaison: Lisa Summers, Tribal Secretary
- 4. Approve amendment to Attorney contract #2015-0188 Sponsor: Jo Anne House, Chief Counsel
- 5. Approve 36 new enrollments

Liaison: Brandon Stevens, Councilman

- 6. Approve 1 new enrollment Liaison: Brandon Stevens, Councilman
- **7.** Safety concern Sponsor: Gerald L. Hill, Chief Judge-Appellate and Denise Beans, Chief Judge-Trial
- 8. Update regarding Gaming impact from Tier IV upgrade (Scheduled at 1:30 p.m. on 03/10/15) Sponsor: Louise Cornelius, Gaming General Manager

XVII. Adjourn

Posted on the Tribe's official website, <u>vert.oneida-nsn.gov</u>, at 01:00 p.m. Friday, March 06, 2015, pursuant to the Open Records and Open Meetinge 27 v, section 7.17-1. For additional information, please call the Business Committee Support Office at (920) 369 4364.

The meeting packet of the open section material methis meeting is available to Tribal members by going to the Members-only section of the Tribe's official weighte at www.oneida-nsn.gov/MembersOnly

For information about this meeting, please cell the Business Committee Support Office at (920) 869-4364 or (800) 236-2214.

Page 6 of 340

Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: 🛛 Open 🛛 Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one):
	Agenda item title (see instructions):
	Oath of Office for the Pardon and Forsiveness Screening Committee
	Action requested (choose one)
	Action - please escribe:
	Administer Oathon Once to Gene Reshail
3.	Justification
	Why BC action is required seconstructions):
	Comprehensive Policy Governing Boards, Committees and Commissions requirement Article VI. Appointed Positions 6-3.
4.	Supporting Materials
	Memo of explanation with required information (see instructions)
	□ Report □ Resolution □ Contract (check the box below if stypetine required)
	Other - please list (Note: multi-media presentation, due to Trik) Clerk 2 days prior to meeting)
	13
	2. 4.
	Business Committee signature required
5.	Submission Authorization
	Authorized sponsor (choose one): Lisa Summers, Tribal Secretary
	Requestor (if different from above):
	Name, Title / Dept. or Tribal Member Additional signature (as needed):
	Name, Title / Dept.
	Additional signature (as needed): Name, Title / Dept.

- Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org



I also sent an e-mail to Don White on January 9, 2015 advising nimes make a recommendation for an SSB representative and an alternate for the Pardon and Forgatoness Screening Committee for a 3 year term. He responded concessing he will forward this to kinds Torres for the recommendation. The recommendation was made on January 12, 701 with Gene Redhail being the primary person and Evangeline Datforth being the alternate. The Charwonian made her appointment at the February 25, 2015 Business committee meeting with Gene Redhail being the primary person and Evangeline Danforth being the alternate.

Page 8 of 340

Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: Open Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one):
	Agenda item title (see instructions):
	Oath of Office for Environmental Resource Board
	Action requested (choose one)
	 ☑ Information Coly ☑ Action - please pescribe:
	Administer oathor office to Shawn Stenandore
•	
3.	Justification
	Why BC action is required seconstructions):
4.	Supporting Materials Instructions
	□ Report □ Resolution □ Contract (check the box below if storative required)
	Other - please list (Note: multi-media presentation, due to Trible Clerk 2 days prior to meeting)
	1. 3.
	2. 4.
	Business Committee signature required
5	Submission Authorization
υ.	
	Authorized sponsor (choose one): Lisa Summers, Tribal Secretary
	Requestor (if different from above): Name, Title / Dept. or Tribal Member
	Additional signature (as needed):
	Name, Title / Dept. Additional signature (as needed):
	Name, Title / Dept.

- Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org



Oneida Business Committee Meeting Agenda Request Form Page 10 of 340

1. Meeting Date Requested: 03 / 11 / 15 2. Nature of request Session: 🖂 Open Executive - justification required. See instructions for the applicable laws that define what is considered "executive" information, then choose from the list: Agenda Header (choose one): New Business/Request Agenda item title (see instructions): Approve February 25, 2015 regular receting minutes Action requested (choose one) Information of ☑ Action please escribe: Approve February 25, 2015 regular meeting nutes 3. Justification Why BC action is required instructions): se BC minutes require BC approval 4. Supporting Materials Instructions Memo of explanation with required information (see instructions) Contract check fe box below if stratifie required) Report □ Resolution Other - please list (Note: multi-media presentation due to Trible Clerk 2 days prior to meeting) 1. February 25, 2015 regular meeting minutes 3. 2. 4. Business Committee signature required 5. Submission Authorization Lisa Summers, Tribal Secretary Authorized sponsor (choose one): Requestor (if different from above): Name, Title / Dept. or Tribal Member Additional signature (as needed): Name, Title / Dept. Additional signature (as needed): Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org



Oneida Business Committee

Regular Meeting 9 a.m. Wednesday, Feb. 25, 2015 BC Conference Room, 2nd floor, Norbert Hill Center

Minutes - DRAFT

Present: Chairwoman Tina Danforth, Treasurer Trish King, Council members: Fawn Billie, Tehassi Hill, Brandon Stevens, Jennifer Webster;

Not Present: :

Arrived at: Secretary Lisa Summers 9:20 a.m.; Vice-Chairwoman Melinda J. Danforth 9:49 a.m. Others present: Daniel King, Brad Char am, Bill Graham, Ravinder Vir, Laura Manthe, Debbie Danforth, Michelle Mays, Lisa Aho, Geraldine Danfort, Ly in Franzmeier, Shawn Skenandore, Michele Doxtator, Sheila Huff, Cheryl Skolaski, Chernowens, Pri Dessart, Traeper Jams, Leah Dodge, Dianne Mclester-Heim, Joanie Buckley

rder and Roll Call by Chairwor an ina Danforth at 9:02 a.m. Ι. Call to

П. Opening by Council an Tel si Hill

A. Moment of Sile ce ch irwoman Tina D. nforth A Moment of Silence for the family of Jan Tenuta, State Lobbyist, wh as passed away.

Ш. Adopt the agenda

Motion by Tehassi Hill to adopt the alenda with one change (to move agental em VII.C. Appoint Gene Redhail, Evangeline Danforth, Eric Boulanger, and Lyle Mooxe, to the Pardon are Forgiveness Screening Committee to the beginning of the agenda), seconded by Trice Ying. Mation carried chanim ush Ayes: Trish King, Fawn Balie, Brandon Stevens, Thorasi Hill Innifer Webster Not Present: Melinda J. Dalforth, Lisa dummers

Oaths of Office administered by Tribal Chairwom, Tina Punforth IV.

A. Pardon and Forgiveness Screening Committee - Cangeline Danforth, Eric Boulanger, and Lyle Metoxen (Gene Redhail not present)

V. **Minutes**

A. Approve February 11, 2015 regular meeting minutes

Motion by Jennifer Webster to approve the February 11, 2015 regular meeting minutes seconded by Tehassi Hill. Motion carried unanimously:

Ayes: Not Present:

Trish King, Fawn Billie, Brandon Stevens, Teneralill, Jennifer Webster Melinda J. Danforth, Lisa Summers

VI. **Resolutions**

A. Adopt resolution titled Indian Child Welfare Act Policy (No Update Submitted) Sponsor: Jo Anne House, Chief Counsel Excerpt from February 11, 2015: Motion by Lisa Summers to defer the resolution titled Indian Child Welfare Act Policy to the next regular Business Committee meeting, seconded by Jennifer Webster. Motion carried unanimously.

Motion by Jennifer Webster to defer for 30 days the resolution titled Indian Child Welfare Act Policy, seconded by Trish King. Motion carried unanimously:

> Ayes: Trish King, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Not Present: Melinda J. Danforth, Lisa Summers

B. Adopt resolution titled Administration for Children and Families – Administration for Native Americans, Native American Language Preservation and Maintenance/Esther Martinez Immersion Don White, Division Director/GSD Sponsor:

Motion by Brandon Stevens to adopt resolution 02-25-15-A Administration for Children and Families -Administration for Native Americans, Native American Language Preservation and Maintenance/Esther Martinez Immersion, seconded by Tehassi Hill. Motion carried unanimously:

> Ayes: Trish King, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Not Present: Melinda J. Danforth, Lisa Summers

C. Adopt resolution titled Administrative Procedures Act Amendments Emergency Adoption

Sponsor: Councilman Brazin n Stevens, LOC Chairman Motion by Tehassi Hill to adopt resolution 02-25-15-B Administrative Procedures Act Amendments Emergency Adoption, seconded by Trish King not on carried unanimously:

rrish King of m Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Mel da J, Fanfor Lisa Summers es: ot Present:

D. Adupt resolution titled Adoption or Amondments to the following to remove references to the Oneida Append Commission purchant to GTC desolution 07-01-13-A: Attorney Contract Policy, Condominate Organance, Emergence Management and Homeland Security, Employee Protection Policy, Local La d User regulation Rembrase nent Folicy, Notary Act, Oneida Election Law, Oneida Food Salvice Code, Oneida Natio, Law Enforcement Ordinance, Oneida Vendor Licensing, Deel Protection Home Policy, Toncing Law Enforcement Ordinance, Oneida Vendor Licensing, Real Property Law Social Media Policy, Tabooing and Birly Piercing Law and Tribal Environmental Response Sponsor: Councilmental

Sponsor: Councilmental Response Sponsor: Councilmentation Servens, LOC Chainnan Motion by Tehassi Hill to adopt resolution 02-25 15-C Adoption of Americanento to the following to remove references to the Oneida Appeals Commission pursuant to GTC Resolution of 01-13-A: Attorney Contract Policy, Condominium Ordinance, Emergency Management and Homeland Secure y, Employee Protection Policy, Local Land Use Regulation Reimbursement Policy, Norty Act, Oneida Election Lay, Oraida Food Service Code, Oneida Nation Law Enforcement Ordinance, oneice, Vendor Licensing, Real Property Low, Social Media Policy, Tattooing and Body Piercing Law and Tribal Environmental Response, seconded by Jennife, Webster, Motion carried unanimously:

Trish King, Fawn Billin, Brandon Stevens, Tehas, Hill, Lennifer Webster Melinda J. Danforth, isa Summurs,

VII. **Appointments**

A. Appoint Shawn Skenandore to the Environmental Resource Board

Sponsor: Tina Danforth, Tribal Chairwoman

Ayes: Not Present:

Excerpt from February 11, 2015: Motion by Lisa Summers Lodefer this to the next regular Business Committee meeting and have ERB bring back their information on how they can come up some solutions to the training aspect, seconded by Brandon Stevens. Motion carried unanimusly.

Motion by Brandon Stevens to appoint Shawn Skenandore to the Environmental Produce Board, seconded by Jennifer Webster. Motion carried with one abstention:

Trish King, Fawn Billie, Jennifer Webster Ayes: Tehassi Hill Abstained: Not Present: Melinda J. Danforth, Lisa Summers

B. Appoint Daniel King, Safety Coordinator, to the Department of Energy – Nuclear Energy Tribal Working Group

Sponsor: Pat Pelky, Division Director/EH&S Motion by Jennifer Webster to appoint Daniel King, Safety Coordinator, to the Department of Energy - Nuclear Energy Tribal Working Group, seconded by Tehassi Hill. Motion carried unanimously:

Ayes: Trish King, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Not Present: Melinda J. Danforth, Lisa Summers

C. Appoint Gene Redhail, Evangeline Danforth, Eric Boulanger, and Lyle Metoxen to the Pardon and Forgiveness Screening Committee

Sponsor: Tina Danforth, Tribal Chairwoman

Motion by Jennifer Webster to appoint Gene Redhail, Evangeline Danforth, Eric Boulanger, and Lyle Metoxen to the Pardon and Forgiveness Screening Committee, seconded by Brandon Stevens. Motion carried with one abstention:

> Ayes: Trish King, Fawn Billie, Jennifer Webster Abstained: Tehassi Hill Not Present: Melinda J. Danforth, Lisa Summers

VIII. **Continuing Resolution Reports**

 A. Environmental, Health, and Safety Division continuing resolution closeout report
 Sportor: Pat Pelky, Division Pare tor/EH&S
 Excerpt from February 11, 1015: Lotion 2, Lisa Summers to defer the Environmental, Health, and Safety Division continuing resolution close out report to the next regular Business Committee meeting so we can have a representative from Fu&S present, seconded by Melinda J. Danforth. Motion carried unanimously

Motion by Jennife, we step to approve the Environment I, Health, and Safety Division continuing resolution closeout report, seconder by Fact Billie. Motion carried transmously: Ayes: Tash King, Fawn Brie, Brandon Stryens, Tehassi Hill, Jennifer Webster

Not Present: eling, J. Danforth, La Summers

B. Comprehensive Health Livisian Continuing resolution closeour report
 Sponsor: Dr. Ravi Vir, Drision Director/Medical & Debra Jupanforth, Division Director/Comp. Health
 Excerpt from February 11, 0015: Motion by Lisa Summers in defense & Comprehensive Health Division
 continuing resolution close observer to the met regular business for mittee meeting so we can have a
 representative from Comp. Health present, seconded by Jennier Weister Motion carried unanimously.

 Motion by Jennifer Webster to approve the Comprehensive Health Division continuing resolution closeout report,
 seconded by Fawn Billie. Motion carried unanimously.

Trish King, Faw, Billie Grandon Stevens, Tehas Hill, Jenner Webster Ayes: Melinda J. Danforth Asa Summer Not Present:

C. Trust/Enrollment Committee continuing resolution closeout report

Brandon Stevens, Councilman Liaison:

uing Motion by Fawn Billie to approve the Trust/Enrollment Commisse cont solution closeout report, seconded by Jennifer Webster. Motion carried unanimously:

Aves: Not Present:

Trish King, Fawn Billie, Brandon Styens, Chastellil, Jennifer Webster Melinda J. Danforth, Lisa Summers

D. Organization Development continuing resolution closeout report

Sponsor: Melanie Burkhart, Organization Development Supervise

Motion by Jennifer Webster to approve the Organization Development continuing resolution closeout report, seconded by Tehassi Hill. Motion carried unanimously:

Trish King, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Aves: Not Present: Melinda J. Danforth, Lisa Summers

E. Human Resources Department continuing resolution closeout report

Geraldine Danforth, Area Manager/HRD Sponsor:

Motion by Jennifer Webster to approve the Human Resources Department continuing resolution closeout report, seconded by Brandon Stevens. Motion carried unanimously:

Ayes: Trish King, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Not Present: Melinda J. Danforth, Lisa Summers

F. Personnel Commission continuing resolution closeout report

Lisa Summers, Tribal Secretary Liaison:

Motion by Jennifer Webster to approve the Personnel Commission continuing resolution closeout report, seconded by Tehassi Hill. Motion carried unanimously:

Trish King, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Aves: Not Present: Melinda J. Danforth, Lisa Summers

Secretary Lisa Summers arrives at 9:35 a.m.

G. Retail Enterprise continuing resolution closeout report

Michele Doxtator, Retail Area Profit Manager Sponsor:

Motion by Jennifer Webster to approve the Retail Enterprise continuing resolution closeout report, seconded by Tehassi Hill. Motion carried with one abountion: Ayes: Trist vilg, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Abstained: Lifa Simmers

Melir Ja J JA ent:

IX. Standi Committees

orth

 A. Legislauve Crutat up Committee Sponsor. Sourcellman Brandon Stovens, Luc Charman Accept Legislat e Operating Committee creeking minutes of February 4, 2015
 Motion by Brandon Stevens to accept the Legislative Operating Committee meeting minutes of February 4, 2015, seconded by Jennifer Webger. Notion farried unanimaryly:

ris King Lisa Summers, Fawn Ilie, Dandon Stevens, Tehassi Hill, Jennifer Ayes: abs*

Not Present:

Meliy da J. Dan orth

B. Finance Committee

LEIEC on item in the Feb 16 J15 Finance Committee 1. Ratify e-poll: Approval of Secret minutes

Sponsor: Lisa Summers, Tribal Scretary

Motion by Trish King to ratify e-poll for the approved of the Secretarial Election item in the Feb 16, 2015 Finance Committee minutes, seconded by Fawn Billie. Motion carried unaphously:

Trish King, Lisa Sumhers, Fay-Billie, Brandon Street, Tehassi Hill, Jennifer Ayes: Webster

Not Present:

Melinda J. Danforth

2. Approve Finance Committee meeting minutes of Furuar, 16, 2015

Sponsor: Tribal Treasurer Trish King, Finance Commit e Charwor

Motion by Jennifer Webster to approve the Finance Committee meeting minutes of Ferruary 16, 2015, seconded by Fawn Billie. Motion carried unanimously:

Trish King, Lisa Summers, Fawn Billie, Bradon Sterns, Tehassi Hill, Jennifer Ayes: Webster

Aves: Melinda J. Danforth

C. Community Development Planning Committee

Sponsor: Tribal Vice-Chairwoman Melinda J. Danforth, CDPC Chairwoman

Accept Community Development Planning Committee meeting notes of February 5, 2015

Motion by Fawn Billie to accept the Community Development Planning Committee meeting notes of February 5, 2015, seconded by Trish King. Motion carried unanimously:

Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Ayes: Webster

Not Present: Melinda J. Danforth

D. Quality of Life (No Requested Action)

X. General Tribal Council

A. Petitioner Madelyn Genskow: Request Special GTC meeting to address 6 resolutions

Sponsor: Lisa Summers, Tribal Secretary

Motion by Tehassi Hill to accept the verified petitions from Petitioner Madelyn Genskow: Request Special GTC meeting to address 6 resolutions seconded by Jennifer Webster. Motion carried unanimously:

Ayes: Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster

Not Present: Melinda J. Danforth

Motion by Tehassi Hill to send the verific ppetitions to the Law, Finance, Legislative Reference and Direct Report offices for the legal, financial, legislative and administrative analyses to be completed, seconded by Fawn Billie. Motion carried unanimously:

 Ares:
 Yrish King, etc. Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Welster

 Not Present:
 Melindar, Darrorth

Motion by Jenny Websith to direct Law Office. Finance, and egislative Reference Offices to submit the analyses to the Tribal Secretary office within 60 days and that an rogress report be submitted in 45 days, seconded by Lisa Summers. Motion confied up himously:

Tulsh King, Lisa Suumers, Lawn Buie, Brandon Stevens, Tehassi Hill, Jennifer

Not Present:

Ayes:

le^r.da <u>J. Da</u>nforth

Motion by Lisa Summers to direct our Direct Report offices to Submit appropriate administrative analyses to the Tribal Secretary's office within 30 days, seconded by Fawn Billie. Motion cancel unanimously: Ayes: Trish King, Lisa Sumpers, Fawn Bule, Plancin Stevens, Tehassi Hill, Jennifer Webster

Not Present:

Vice-Chairwoman Melinda J. Danforth arrives at 1:49 a

B. Approve meeting materials for March 28, 2, 15 specie GTC meeting

Melinda J. D. nfor

Sponsor: Lisa Summers, Tribal Secretary Motion by Jennifer Webster to approve meeting materials for the March 20,2015 special GTC meeting, seconded by Lisa Summers. Motion carried unanimously: Aves: Trish King, Lisa Summers, Fawi Bine, Velinda J. Danforth, Brandon Stevens,

Trish King, Lisa Summers, Fawn Bine, Melinda J. Danforth, Brandon Stevens, Tehassi Hill, Jennifer Webster

- XI. Unfinished Business (No Requested Action)
- XII. Tabled Business (No Requested Action)

XIII. New Business

- A. Approve Oneida Head Start/Early Head Start Policy Council documents Sponsor: Jennifer Webster, Councilwoman
 - 1. Oneida Head Start/Early Head Start Policy Council By-laws

Motion by Lisa Summers to approve the Oneida Head Start/Early Head Start Policy Council By-laws, seconded by Fawn Billie. Motion carried unanimously:

Trish King, Lisa Summers, Fawn Billie, Melinda J. Danforth, Brandon Stevens, Tehassi Hill, Jennifer Webster

2. Oneida Head Start/Early Head Start Policy Council Impasse Resolution Agreement with the **Oneida Business Committee**

Motion by Melinda J. Danforth to approve the Oneida Head Start/Early Head Start Policy Council Impasse Resolution Agreement with the Oneida Business Committee, seconded by Tehassi Hill. Motion carried with one opposed:

Ayes:

Ayes:

Trish King, Lisa Fawn Billie, Melinda J. Danforth, Brandon Stevens, Tehassi Hill, Jennifer Webster

Opposed: Lisa Summers For the record:

Lisa Summers stated I do appreciate the way that the agreement is laid out, and I think that it's a good one. I just don't support it because I think that we should first locities internal resources before we go to outside resources. I just want to make site that it's clear, that it's not that I don't support what your intent is here; I think it is a good intent, I just think we look internally first.

Oneida Head Start/Earl nead start Nection Criteria – Eligibility 3.

elinda J. Janforth to approve me Creida Read Start/Early Head Start Selection Criteria - Eligibility, Motion by seconded by Lisa Summers Motion carried manapously. Aver Trish King, isa Summers Tawn Billie, Melinda J. Danforth, Brandon Stevens,

Tehassi Hill, Jonnifer Wooste

4. Oneida Hear StandEurly Head Start Engibility Recruitment, Selection, Enrollment and Attendance (FISE/ Plan

Motion by Melinda J. Danforth to approve the Oneida Head Start/Fully Head Start Eligibility, Recruitment, Selection, Enrollment and Attendice (TASEA) Plan, seconder by Fawn Filie. Motion carried unanimously: Ayes: Trist King, Lisa Summers, Fawn Billie Lielingh J. Danforth, Brandon Stevens, Tehessi Hill, Jennifor Webster

XIV. Travel

A. Travel Reports

1. Accept travel report for Councilwo pan from Billie – MBK Community Charlenge National Convening – Washington D.C, Februar, 11-13, 201 (originally scheduled for January 29, 2015) Sponsor: Fawn Billie, Councilwoman

Motion by Brandon Stevens to accept the travel report for Control woman Fawn Billie - MBK Community Challenge National Convening – Washington D.C, February 1-13, 2015 Signally scheduled for January 29, 2015), seconded by Lisa Summers. Motion carried unanimous

Trish King, Lisa Summers, Fawn Pine, Nelinda J. Danforth, Brandon Stevens, Aves: Tehassi Hill, Jennifer Webster

Chairwoman Tina Danforth departs at 10:27 a.m. Vice-Chairwoman Meindar J. Danfor assumes the responsibilities of the Chair.

2. Accept travel report for Councilman Brandon Stevens – MBK Community Challenge National Convening – Washington D.C. January 28, 2015

Sponsor: Brandon Stevens, Councilman

Motion by Jennifer Webster to accept the travel report for Councilman Brandon Stevens – MBK Community Challenge National Convening – Washington D.C, January 28, 2015, seconded by Trish King. Motion carried unanimously:

Trish King, Lisa Summers, Fawn Billie, Tehassi Hill, Brandon Stevens, Jennifer Ayes: Webster **Tina Danforth** Not Present:

B. <u>Travel Requests</u>

- 1. Councilman Brandon Stevens NIGA Tradeshow & Convention San Diego, CA, March 30-April 2, 2015
 - Sponsor: Brandon Stevens, Councilman

Motion by Fawn Billie to approve the travel request for Councilman Brandon Stevens – NIGA Tradeshow & Convention – San Diego, CA, March 30-April 2, 2015, seconded by Lisa Summers. Motion withdrawn.

Motion by Fawn Billie to defer this item to 1:30 p.m., seconded by Tehassi Hill. Motion carried unanimously: Ayes: Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster

Not Present:

Tina Danforth

Motion by Trish King to approve the travel request for Councilman Brandon Stevens – NIGA Tradeshow & Convention – San Diego, CA, March 30-April 2, 2015, seconded by Fawn Billie. Motion carried unanimously:

Aves: Abstained: Not Present: Trish Kon, Lisa Summers, Fawn Billie, Tehassi Hill, Jennifer Webster Bran, o Stevens Ti a Dunforth

a Summers to recess a 10:51 to resume at 1:30 p.m., seconded by Fawn Billie. Motion Motion by L .m. • carried unanimously: Ayes:

Not R ___e

mmers, Fawn Billie, Tehassi Hill, Brandon Stevens, Jennifer Trish King, / Ja 、 Webster Tina Danforth

age da is schedul d to begin a XV. 30 p.m.) Reports (This sector of th

le p eting back into session Chairwoman Tina Danforth calle at 1:29 p

Councilman Tehassi Hill is excused for the remainder of the meeting Councilman Brandon Stevens is not pesent. Councilman Brandon Stevens arrives at 1.32 p

A. Operational Reports

1. Internal Services Division report Abanic Buckley, Division Directo Motion by Jennifer Webster to accept the Internal Services Division quarterly repo , seconded by Trish King. Motion carried unanimously:

Melinda J. Danforth, Trish Ki J, Lisa Summers, Fawn Billie, Brandon Stevens. Ayes: Jennifer Webster Not Present: Tehassi Hill

Councilman Brandon Stevens departs at 2:01 p.m. Councilman Brandon Stevens arrives at 2:02 p.m.

Office of the Ombudsman report – Dianne McLester-Heim, Cobudsman

Motion by Melinda J. Danforth to accept the Office of the Ombudsman guarter report and have a meeting with the Ombudsman to start identifying the request for the roles, responsibilities and expectations, seconded by Fawn Billie. Motion carried unanimously: Ayes:

Not Present:

Melinda J. Danforth, Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Jennifer Webster

Tehassi Hill

B. Corporate Reports (No Requested Action)

C. Boards Committees and Commissions

1. Oneida Nation School Board – Debbie Danforth, Chairwoman

Liaison: Fawn Billie, Councilwoman

Motion by Jennifer Webster to accept the Oneida Nation School Board's guarterly report, seconded by Fawn Billie. Motion carried unanimously:

Ayes:	Melinda J. Danforth, Trish King, Lisa Summers, Fawn Billie, Brandon Stevens,
	Jennifer Webster
Not Present:	Tehassi Hill

Land Claims Commission – Amelia Cornelius, Chairwoman (No Report Submitted)

Liaison: Brandon Stevens, Councilman

Motion by Jennifer Webster to halt stipends to the Land Claims Commission until the Land Claims Commission's guarterly report is submitted to and accepted by the Business Committee, seconded by Lisa Summers. Motion carried unanimously:

Aves:

s:

Melinda J. Danforth, Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Jennifer Webster Tehas

Not Present:

Motion by Lisa Summers to defer the concern regarding delinquent reporting to the Land Claims Commission's liaisons, Brandon Suyens and Jennifor Webser, for follow-up, seconded by Melinda J. Danforth. Motion carried unanimous Ay

Melinda , Dar orth, Nish King, Lisa Summers, Fawn Billie, Brandon Stevens, Jennifer Wei ster Tehassi Hin

3. Land Comm sion melia Cornelius,

Liaison: Thas Ai Councilman

Motion by Melinda J. Danfrom to refer the Oneida Lan Commission prarterly report to the March 11, 2015 Business Committee meeting, seconded by Lisa Summers. Motion carried unanimously: Ayes: Calindro. Danforth, Trish Kinge Lisa Summers, Fawn Billie, Brandon Stevens, Jenn er Webst r Not Present: Teh ssi Hill

Aak vomak

Not reser

XVI. **Executive Session**

A. <u>Executive Session meeting minutes of Pubruary 23, 2015</u> Present: Chairwoman Tina Danforth, "ice-Chairwoman Melinda J. Lunforth, Treasurer Secretary Lisa Summers, Council members Faren Billie, Tehassi Hill, Jen fer Websur; infor a, Treast er Trish King, Not Present: :

Arrived At: Council member Brandon Stevene 9:09 a.m. Others Present: Louise Cornelius, Chad Fuss, Amelia Cornelius, Mary Jo Nasn, Pat Pelky, Larry Barton, Nate King, Jo Anne House. Priscilla Leverence, De bie Danforth Arandon Stevens, Wes Martin, Dave Larsen, Kelly McAndrews, Nathan King, Lois Strong, Vorence, etri, NaryAnn Kruckeberg;

1. Call to order by Chairwoman Tina Danforth at 8:59 a.

2. Adopt the agenda

Motion by Lisa Summers to adopt the agenda with noted times (agenda ite XVI.D.1 Discussion regarding acreage in Brown County at 9:00 a.m. and agenda item XVI.D.2. Family Care protonentation at 3:00 p.m.), seconded by Tehassi Hill. Motion carried unanimously:

> Ayes: Trish King, Lisa Summers, Melinda J. Danforth, Fawn Billie, Tehassi Hill, Jennifer Webster

Brandon Stevens Not Present:

3. Executive discussion

Motion by Lisa Summers to go into executive session at 9:01 a.m., seconded by Fawn Billie. Motion carried unanimously:

Ayes: Trish King, Lisa Summers, Melinda J. Danforth, Fawn Billie, Tehassi Hill, Jennifer Webster

Not Present: **Brandon Stevens**

Council member Brandon Stevens arrives at 9:09 a.m.

Consensus to break at 12:30 p.m. and to reconvene at 2:00 p.m.

Reconvened meeting called to order at 2:00 p.m. by Chairwoman Tina Danforth.

4. Adjourn

Motion by Tehassi Hill to come out of executive session at 6:55 p.m., seconded by Jennifer Webster. Motion carried unanimously:

Ayes:

Trish King, Lisa Summers, Melinda J. Danforth, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster

Motion by Lisa Summers to adjourn 2005 p.m., seconded by Fawn Billie. Motion carried unanimously: Tring, Lisa Summers, Melinda J. Danforth, Fawn Billie, Brandon Stevens, Aves: eha si Hill, Conifer Webster

B. Reports

1. Chief Financial Officer report a Law Barton, Chief Financial Officer Motion by Jennier Websier to accept the Chief Financial Officer report, seconded by Trish King. Motion carried unanimously:

Fawir Billie, Brandon Stevens, Tehassi Hill, Jennifer sh King, Lisa Sur 1êr Ayes: ebster Not Present:

na Dinforth

2. Chief Counsel report – Journne Jouse, Chief Counsel Motion by Lisa Summers to accept the Chief Counsel report including the requested action within the report, seconded by Jennifer Webster. Motion carried manimously:

Trish Lisa 2 m. ers, Fawn Bule, Plancen Stevens, Tehassi Hill, Jennifer Aves: Webster Tina Danfor

Not Present:

3. Officers' report – Melinda J. Danfold, Triker Vice-Chairwoman (No requested Action)

4. Intergovernmental Affairs and Communications Nathan King, Leg. Lative Affairs Director Motion by Lisa Summers to accept the Intergovernmenta Affers and Communications report, seconded by Jennifer Webster. Motion carried unanimously:

Trish King, Lisa Summers, Fayn Billi Brand of Stevens, Tehassi Hill, Jennifer Ayes: Webster Tina Danforth Not Present:

C. Audit Committee

Councilman Tehassi Hill, Audit Committee Chairman, Sponsor:

1. Accept Audit Committee meeting minutes of January 15, 201

Motion by Tehassi Hill to accept the Audit Committee meeting minutes of January 15, 2015, seconded by Fawn Billie. Motion carried unanimously:

Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Aves: Webster Tina Danforth

Not Present:

2. Accept Internal Audit report for January 2015

Motion by Jennifer Webster to accept the Internal Audit report for January 2015, seconded by Trish King. Motion carried unanimously:

Ayes: Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster

Not Present: Tina Danforth

3. Accept the Slot Compliance Audit and lift the confidentiality requirement allowing Tribal members to view the audit

Motion by Tehassi Hill to defer this agenda item the next regular Business Committee meeting, seconded by Lisa Summer. Motion carried unanimously:

Ayes:

Not Present:

Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster **Tina Danforth**

4. Accept the Craps Rules of Play Audit and lift the confidentiality requirement allowing Tribal members to view the audit

Motion by Lisa Summer to accept the Craps Rules of Play Audit and lift the confidentiality requirement allowing Tribal members to view the audit, second d by Fawn Billie. Motion carried unanimously: Ayes: Trish of g, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Wost r

5. Review Levelopment Division or cern.

vina Japf

na Dinforth

Motion by L a Summer to accept the Review Development Division concerns report as information and have the appropriate parties were conther to follow up with any regreed action, seconded by Jenny Webster. Motion carried unanimou 1.

> Fawir Billie, Brandon Stevens, Tehassi Hill, Jennifer sh King, Lisa Sur 1êr ebster

Not Present:

Ayes:

The ont:

- **D.** Unfinished Business
 - 1. Discussion regarding a reage in Brown County (Scheol red a 9:00 a.m. on 2/23/15) Meli da J. Danorth Tribal Vice-Chai wom Sponsor:

Excerpt from the February 11 2015 Motion by Melinea J. pant inth arcept the item as information and request that, at the next Busines. Committee meeting, the Inergenernmental Affairs and Communications Department are the Coming Division provide the Usines. Committee with an update on the five (5) acres, seconded by List Summers. Motion carried unanticously.

- i., **Gaming Division**

Sponsor: Louise Cornelius, Gaming general Manager Motion by Trish King to accept the Gaming Division and Intergovernmental Affairs and Communications repor as information and request Gaming Division and Intergovernmental Affairs and Communications to follow-up, munications reports seconded by Jennifer Webster. Motion carried unanimously

> Trish King, Lisa Summers, Fayn Billi Brancin Stevens, Tehassi Hill, Jennifer Ayes: Webster Tina Danforth

Not Present:

Motion by Lisa Summers to schedule a special Business Committee meeting on or being March 04, 2015 to address the follow-up on this this item, seconded by Trish King. Motion cannot unanipously: Ayes:

Trish King, Lisa Summers, Fawn Billie, Brander Stevens, Tehassi Hill, Jennifer Webster **Tina Danforth**

Not Present:

ii. Intergovernmental Affairs and Communications

Nathan King, Legislative Affairs Director Sponsor:

Motion by Trish King to accept the Gaming Division and Intergovernmental Affairs and Communications reports as information and request Gaming Division and Intergovernmental Affairs and Communications to follow-up, seconded by Jennifer Webster. Motion carried unanimously:

Ayes: Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster

Not Present: Tina Danforth

Motion by Lisa Summers to schedule a special Business Committee meeting on or before March 04, 2015, to address the follow-up on this this item, seconded by Trish King. Motion carried unanimously: Aves:

Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Tina Danforth

Not Present:

2. Family Care Implementation (Scheduled at 3:00 p.m. on 2/23/15)

Sponsor: Melinda J. Danforth, Tribal Vice-Chairwoman

Excerpt from February 11, 2015: Trish King to accept the update and direct the Tribal Secretary to schedule a debriefing for next week following the conference call with the State that is being held on

Friday, seconded by Tehassi Hill. Motion carried unanimously.¹ Motion by Tehassi Hill to accept the upper on the Family Care Implementation report and request the capacity numbers and additional information the flusiness Committee had requested be brought back, seconded by Brandon Stevens. Motion carried unanimously:

rish King ar Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer es: We Not Present: Tina Dar Jrth

Amendment to the main not on by Lisa Summers that the family Care Implementation report and update be placed on the age as or the special Business Committee me ting that will be scheduled on or before March 04, 2015, seconded by Jenn r Web er. Motion carried na imous Ayes:

sh King, Lisa Su, mers, Tawn Blie, Brandon Stevens, Tehassi Hill, Jennifer ebst

Not Present:

E. Tabled Business (No Negu sted Act

F. New Business

nnel mmission 1. Review Complaint re: Onelua Pe

⊿anforth

n)

inz

Sponsor: Lisa Summers, Trichl Secretary Motion by Lisa Summers to accept the completed re: Oneir Personnel Com Motion by Lisa Summers to accept the complete re: Oneich Personnel Completion and deter the complaint to the Legislative Operating Committee and Law Once for additional follow-up to be prought back to the March 25, 2015 Business Committee meeting, seconded by Far a Billie. Motion carried unar mousing

Trish King, Lisa Sum, ers, Faw, Billie, Brandon Streng, Tehassi Hill, Jennifer Ayes: Webster

Not Present:

Tina Danforth

2. Update re: Internal Services Complaint

Lisa Summers, Tribal Secretary Sponsor:

Motion by Lisa Summers to accept the recommendation presented to the One of a Bus ness Committee regarding the Update re: Internal Services Complaint, seconded by Trish King. Motion Parried un nimously: Ayes: Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer

Webster

Not Present:

3. Approve temporary wage adjustment

Sponsor: Melinda J. Danforth, Tribal Vice-Chairwoman

Tina Danforth

Motion by Tehassi Hill to approve temporary wage adjustment with the effective date of February 25, 2015, seconded by Trish King. Motion carried unanimously:

¹ At the February 11, 2015, regular Business Committee meeting, Vice-Chairwoman Melinda J. Danforth noted that this item should remain on the Business Committee agenda

Ayes: Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Webster Tina Danforth

Not Present:

4. Approve Limited Waiver of Sovereign Immunity for Vision Oneida, LLC, Contract #2015-0110 Sponsor: Tehassi Hill, Councilman

Motion by Lisa Summers to defer the Limited Waiver of Sovereign Immunity for Vision Oneida, LLC, Contract #2015-0110 to the next Business Committee meeting and have the Law office work with the Division of Land Management to address the identified concerns, seconded by Fawn Billie. Motion carried unanimously:

Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Tehassi Hill, Jennifer Ayes: Webster Tina Danforth Not Present:

Motion by Lisa Summers to go into executive session at 2:55 p.m., seconded by Brandon Stevens. Motion carried unanimously:

Melind Danforth, Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Ayes: Jenr e Webster as Hill Not Present: T andon Servens to come an of recut Motion by P session at 3:25 p.m., seconded by Fawn Billie. Motion carried una himously: Ayes: Melinda J. Duffs th, Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Jennifer Whater Tehassi Hill Not R Motion by Melinda J. Day orthogothout his agendation to the special Business Committee meeting that will be scheduled on or before March 04, 2015 and to direct the Office to provide the necessary follow-up information, seconded by Lisa Summers. Moton Arried unanimously zsa Sumpers, Fawn Billie, Brandon Stevens, Aves: enn er Webst Teh ssi Hill Not Present: 5. Approve procedural exception for one (1, offer to purchase Sponsor: Pat Pelky, Division Director/DOLM Motion by Lisa Summers to approve procedured exception for one (1) offer to parch Je, nou o exceed the appraised value, seconded by Jennifer Webster. Motion carried unanimously: Ayes: Trish King, Lisa Surpriers, Fawn Pillie, Brandon Tenassi Hill, Jennifer teve. Webster Not Present: Tina Danforth 6. Approve OBC responses to four (4) petitions Sponsor: Lisa Summers, Tribal Secretary Motion by Lisa Summers to accept the approved OBC responses of our Appetities as information, seconded by Tehassi Hill. Motion carried unanimously: Trish King, Lisa Summers, Fawn Billie, Brag Ayes: on Stevens, Tehassi Hill, Jennifer Webster Not Present: Tina Danforth

XVII. **Adiourn**

Motion by Melinda J. Danforth to adjourn at 2:55 p.m., seconded by Fawn Billie. Motion withdrawn.

Motion by Melinda J. Danforth to adjourn at 3:27 p.m., seconded by Fawn Billie. Motion carried unanimously: Melinda J. Danforth, Trish King, Lisa Summers, Fawn Billie, Brandon Stevens, Ayes: Jennifer Webster Tehassi Hill Not Present:

Minutes prepared by Chad Wilson, Project Manager Minutes approved as presented/corrected on _____



Oneida Business Committee Meeting Agenda Request Form Page 24 of 340

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: Open Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one): New Business/Request
	Agenda Header (choose one). New Business/Request
	Agenda item title (see instructions):
	Approve March 3, 2015 special meeting minutes
	Action requested (choose one)
	Action - please escribe:
	Approve Materia, 2015 special meeting minutes
3.	Justification
	Why BC action is required sequestructions):
	BC minutes require BC approval
4.	Supporting Materials
	Memo of explanation with required information (see instructions)
	□ Report □ Resolution □ Contract (check the box below if standard required)
	☑ Other - please list (Note: multi-media presentation due to Trike) Clerk 2 days prior to meeting)
	1. March 3, 2015 special meeting minutes 3.
	2. 4.
	Business Committee signature required
5.	Submission Authorization
	Requestor (if different from above): Name, Title / Dept. or Tribal Member
	Additional signature (as needed):
	Name, Title / Dept.
	Additional signature (as needed): Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org



Oneida Business Committee

Special Meeting 8:15 a.m. Tuesday, Mar. 3, 2015 BC Conference Room, 2nd floor, Norbert Hill Center

Minutes – Draft

Present: Vice-Chairwoman Melinda J. Danforth, Treasurer Trish King, Secretary Lisa Summers, Council members: Fawn Billie, Tehassi Hill, Brandon Stevens, Jennifer Webster; **Not Present:** :

Arrived at: Chairwoman Tina Danforth 18:31 a.m.; Others present: ;

I. Call to order and Roll Call by Ce-Charwork. Melinda J. Danforth at 8:17 a.m.

anforth

II. Opening by Count and Tehassi Hill

III. Adopt the agenda

Motion by Lisa Summers treadop the agenda, seconde Lby Brandom, evens. Motion carried unanimously: Ayes: Aye

Not Present:

IV. New Business

A. Approve April 11, 2015 date for the Special Election

í ina

Liaison: Melinda J. Danforth, Thiel Vice-Charwoman Motion by Brandon Stevens to approve the recommendation of the Election Board to set the late of the Special Election to fill the Business Committee vacancy for Audi 11,2015, recorded by Favn Bine, Motion carried unanimously:

> Ayes: Not Present:

Fawn Billie, Tehassi Hill, Trig, King, Brandon Stevens, Lisa Summers, Jennifer Webster Ting Depforth

Tina Danforth

Chairwoman Tina Danforth arrives at 8:32 a.m. and assumes the reponsibilities the Chair.

V. Executive Session

Motion by Lisa Summers to go into executive session at 8:33 a.m., seconded the King. Motion carried unanimously:

Ayes:

Fawn Billie, Melinda J. Danforth, Tehassi Hill, Trish King, Brandon Stevens, Lisa Summers, Jennifer Webster

Motion by Melinda J. Danforth to come out of executive session at 10:50 a.m., seconded by Tehassi Hill. Motion carried unanimously:

Ayes:

Fawn Billie, Melinda J. Danforth, Tehassi Hill, Trish King, Brandon Stevens, Lisa Summers, Jennifer Webster

A. Unfinished Business

1. Discussion regarding acreage in Brown County

Sponsor: Melinda J. Danforth, Tribal Vice-Chairwoman

Excerpt from February 25, 2015: (1) Motion by Trish King to accept the Gaming Division and Intergovernmental Affairs and Communications reports as information and request Gaming Division and Intergovernmental Affairs and Communications to follow-up, seconded by Jennifer Webster. Motion carried unanimously. (2) Motion by Lisa Summers to schedule a special Business Committee meeting on or before March 04, 2015 to address the follow-up on this this item, seconded by Trish King. Motion carried unanimously.

Excerpt from the February 11, 2015: Motion by Melinda J. Danforth accept the item as information and request that, at the next Business Committee meeting, the Intergovernmental Affairs and Communications Department and the Gaming Division provide the Business Committee with an update on the five (5) acres, seconded by Lisa Summers. Motion carried unanimously.

Motion by Brandon Stevens to accept the discussion regarding the acreage in Brown County as information, seconded by Jennifer Webster. Motion carried unanimously:

Fawn Fie, Melinda J. Danforth, Tehassi Hill, Trish King, Brandon Stevens, Lisa Aves: Sum le s, Jennifer Webster

2. 🖬

Aves:

Sponsor: Melinda J. Dataorth Inbalance-Chairwoman

Excerpt from February 25, 15: 1 Mon by Tehassi Hill to accept the update on the Family Care implementation report and require the capacity numbers and additional information the Business Committee made equested be brought back, seconded by Brandon Stevens. Motion carried unanchously. (1) Amendment to the main region section sections state the Family Care Implementation report and update be placed on the agenda for the special Business Committee meeting that will be scheduled on or before March 04, 2015, seconded by Jennifer Webster. Motion carried unanim usly

 Carried unanimalsiy
 Excerpt from February 1, 2015: Trish King to accerpt the update and direct the Tribal Secretary to schedule a debriefing or nealweek following the conterence of with the State that is being held on Friday, seconded by Tehrssi Hill. Notion carried unanimour J.
 Motion by Melinda J. Danforth to accept the verbal update from Complehent and Health Division and Intergovernmental Affairs and Communications are request that these parties o continue to work to develop a plan of action for the state budget process as some as pensible, second d by Lawr Billie. Motion carried unanimously. unanimously:

Fawn Billie, Malinda J. Danorth, Tehassi Hing Iriş' King, Bandon Stevens, Lisa Aves: Summers, Jennier Weister

3. Approve Limited Waiver of Sovereign Aumunity of Vision Oneida, LC contract #2015-0110 Sponsor: Tehassi Hill, Councilman

Excerpt from February 25, 2015: Motion by Mulinda J. Dar both to move this agenda item to the special Business Committee meeting that will be scheduler on or before March 04, 2015, and to direct Law Office to provide the necessary follow-up information, seconded by Lisa Summers. Motion carried unanimously.

Motion by Melinda J. Danforth to defer this item to the regular Business Computed peeting of March 11, 2015, in order for us conduct further due diligence, seconded by Brandon Stevers. Motion parried unanimously: Ayes: Fawn Billie, Melinda J. Danforth, Tehassi I. J., Trish Kog, Brandon Stevens, Lisa

Summers, Jennifer Webster

Motion by Brandon Stevens for the Chairwoman's Office set up a meeting between the Land Commission and the Business Committee before March 11, 2015, including a memorandum identifying the stated concerns, seconded by Tehassi Hill. Motion carried unanimously:

> Fawn Billie, Melinda J. Danforth, Tehassi Hill, Trish King, Brandon Stevens, Lisa Summers, Jennifer Webster

VI. Adjourn

Motion by Brandon to adjourn at 10:53 a.m., seconded by Tehassi Hill. Motion carried unanimously: Ayes: Fawn Billie, Melinda J. Danforth, Tehassi Hill, Trish King, Brandon Stevens, Lisa Summers, Jennifer Webster

Minutes prepared by Lisa Liggins, Executive Assistant Minutes approved as presented/corrected on _____



Oneida Business Committee Meeting Agenda Request Form

1.	1. Meeting Date Requested: 03 / 11 / 15	
2.	2. Nature of request Session: ⊠ Open □ Executive - justification required. See instructions for	or the applicable laws that
	define what is considered "executive" information, the	en choose from the list:
	Agenda Header (choose one): Resolution	
	Agondo item title (and instructions):	
	Agenda item title (see instructions) Take your Son/Daughter to Work 2/ y	
	Action requested (choose one)	
	Information only	
	Action - please describe:	
	Adopt Take your Sin/Daughter to Work Day Resolution	
3.	3. Justification	
	Why BC action is required true in the tions):	
÷,		
	Resolution needs to be adopted a OBC	
4.	4. Supporting Materials	Instructions
	Memo of explanation with required information (see instructions)	
	□ Report ⊠ Resolution □ Contract check the box below if sign	
	□ Other - please list (Note: multi-media presentation raue to Tribal Clerk 2 d	ays pror to meeting)
	1.	
	2. 4.	
	☐ Business Committee signature required	
5.	5. Submission Authorization	
	Authorized sponsor (choose one): Fawn Billie, Council Member	· · · · · · · · · · · · · · · · · · ·
	Requestor (if different from above):	
	Name, Title / Dept. or Tribal Member Additional signature (as needed):	
	Name, Title / Dept.	
	Additional signature (as needed):	

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Memorandum

To: Oneida Business Committee

From: Councilwoman Fawn Billie

Date: February 17, 2015

Subject: Take your Son/Daughter to Work Day Resolution

Purpose: Lum respectfully requesting the Oneida Business Committee to officially recognize April 24th 2015 as Take Your Set/Daughter to Work Day by adopting the attached resolution. No statement of offect needed per Local Resource Office.

Background: I would like to the establish this day for the Caneda Tribe. The Take Your Son/Daughter to Work Dien's important because it allows our youth to see different perspectives of responsibilities of the work lace. It angre with our Strategic Directions of Committing to Building a Responsible Nation, Advancian Onkyote?a.ka Principles Interementing "Good Governance" processes and Creating a positive organizational Culture. This is coneficial to both parties involved by encouraging and giving exposure to a positive work experience. By providing this opportunity for our youth is the key in building a responsible nation. This also supports the My Brother's Keeper Initiative by creating youth le dership and neutorship.

"Exposing girls and boys to what a parent or mentor in one lives does during the work day is important, but showing them the value of their education, he ping them discover the power and possibilities associated with a balanced work and family life, providing them an opportunity to share how they envision the future, and allowing them to begin steps to ward their end goals in a hands-on and interactive environment is key to their achieving success."http://www.daughtersandsonstowork.org

Requested Action: Motion to adopt Take Your Son/Daughter to Work Day Resolution.

Page 30 of 340 Oneida Tribe of Indians of Wisconsin Used Devices of the help of this Oneida Chief in cementing a friendship bundred bags of corn to

UGWA DEMOLUM YATEHE Because of the help of this Oneida Chief in cementing a friendship between the six nations and the colony of Pennsylvania, a new nation, the United States was made possble.

BC Resolution _____ Take your Son/Daughter to Work Day

Washington's starving army

at Valley Forge, after the

colonists had consistently

fire.

refused to aid them.

WHEREAS, the Oneida Tribe of Indians of Wisconsin is a federally recognized Indian government and ed by the laws of the United States of America; and a treaty tribe recogn WHEREAS. the Oneida General Tribal Council is the governing body of the Oneida Tribe of Indians of Visconsin; and WHER AS. e Oneida Business Computtee has been delegated the authority of Article IV of the Oneida Tribal Constitution of the Oneida General Tribal Council; and WHEREAS, Or ida Tribe of Indians has a side of being a Nation of strong families built on Tsi ukwali ot and a strong er n my; a arace the growth and levelopment strong families by promoting quality WHEREAS, we enc education and a sitive work ethic, nd the Oneida ribe of Indians has a work force of near 3000 employees that could have a WHEREAS. tremendous impact on the buth in our community and of We consin operate multiple rograms that support the the Oneida Tribe WHEREAS, youth, elders, education rade ship development, and of error community wellbeing; and WHEREAS, the Oneida Business Comhittee har defined its Strategic Directions which include: Implementing Good Governance Process Committing to build a responsible nation Advancing Onvyote?a ka principles Creating a positive organizational Cature ful outh the foundation for sustaining the encouragement and support we provide for WHEREAS, a successful, healthy and growing community with good mind good hearts and a strong

NOW THEREFORE BE IT RESOLVED, that the Oneida Business committee leartily endorses Take Your Son/Daughter to Work Day on the fourth Thursday of April each very. The Oneida Business Committee encourages parents, aunts, uncles and grandparents to bring their child to work on Thursday, April 23, 2015.

BE IT FURTHER RESOLVED, the Directors, Supervisors and Managers shall work in the best and most safe interest of the Oneida Tribe to accommodate the requests of employees to bring their children to work on this day.

Page 31 of 340

BC Resolution _____ Take your Son/Daughter to Work Day Page 2 of 2

BE IT FINALLY RESOLVED, the Oneida School System shall be encouraged to participate and support their students to be engaged in this positive learning experience as long as the students are in good attendance and academic standing.



Oneida Business Committee Meeting Agenda Request Form

Page 32 of 340

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: 🛛 Open 🛛 Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one): Legislative Operating Committee
	Agenda item title (see instructions):
	Resolution: Rules of Appellate Procedure Amendments
	Action requested (choose one)
	Action - please escribe:
	Adopt the Reservation amending the rolles of Aprimete Procedure
3.	Justification
	Why BC action is required seconstructions):
Л	Supporting Materials
4.	Supporting Materials Instructions Instructions
	□ Report
	☑ Other - please list (Note: multi-media presentation, due to Trik) Clerk 2 days prior to meeting)
	1. Fiscal Impact Statement 3. Drate Law with legislative analysis
	2. Statement of Effect 4.
	Business Committee signature required
5.	Submission Authorization
	Authorized sponsor (choose one): Brandon Stevens, Council Member
	Requestor (if different from above):
	Name, Title / Dept. or Tribal Member
	Additional signature (as needed):
	Name, Title / Dept. Additional signature (as needed):
	Name, Title / Dept.

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Oneida Tribe of Indians of Wisconsin

Legislative Reference Office P.O. Box 365 Oneida, WI 54155 (920) 869-4376 (800) 236-2214 http://oneida-nsn.gov/LOC



Committee Members Brandon Stevens, Chairperson Tehassi Hill, Vice Chairperson Fawn Billie, Councilmember Jennifer Webster, Councilmember

Memorandum

To:	Oneida Business Committee
From:	Brandon Stevens, LOC Chairperson
Date:	March 4, 2015
Re:	Rules of Appellate P of Edure Amendments

Please find *plached* he following for the onsideration:

- Fiscal Impact Rules of Appellate Procedure Amendments
 Resolution: Rules of Appellate Procedure Amendments
- 3. Statement of Effect Rules of App late Noced Amendments
- 4. Rules of appellat Procedure (redlin
- 5. Rules of Appelle e Provinure (clean)

Overview

On December 19, 2014, the Creada Plasmess Committee adapted emergency amendments to the Rules of Appellate Procedure (Rules). The attacked resolution would adopt hose amendments, with some minor formatting updates, on a permanent basis. The amendments y of a create new definitions in the Rules and correct inadvertent omissioner. This meades updating the definition of "attorney" to remove references to advocates; and defining "ad occite" "original hearing had and "record." The terms "advocate" and "original hearing body" as also shoor brated through a throw ules, where appropriate. "Initial Review" would also be defined under the mendments and a section is added to the Rules to require an Initial Review be conducted by three members of the Court when a Notice of Appeal or Perfected Notice of Appeal is filed.

Additional amendments would allow tracked U.S. or privite mail to be sed for service and clarify Clerk responsibilities regarding the certification of records. The approximates also increase the time for the Court to complete a case from one hundred and twenty days to one hundred and eighty days. Finally, the amendments give the Court the flexibility to allow parties to the case, during oral argument, even if those cases have not been cited in a brief.

Additional, minor revisions were made that do not affect the content of the Kules. A public meeting was held on February 5, 2015 in accordance with the Legislative Procedures Act and no comments were received on the proposed changes.

Requested Action

Approve the Resolution: Rules of Appellate Procedure Amendments.



Oneidas bringing several hundred bags of corn to Washington's starving army at Valley Forge, after the colonists had consistently refused to aid them.

ONEIDA TRIBE OF INDIANS

ONEIDA FINANCE OFFICE Office: (920) 869-4325 • Toll Free: 1-800-236-2214 FAX # (920) 869-4024

MEMORANDUM



UGWA DEMOLUM YATEHE Because of the help of this Oneida Chief in cementing a friendship between the six nations and the colony of Pennsylvania, a new nation, the United States was made possble.



I. Background

The Oneida Tribal Judicial System was created by GTC Reputitor (01-07-13-B. BC Resolution 04-25-14-B adopted the Oneida Judiciary Rules of appendic Diocedure to be effective when the Judiciary goes into arte t November 1, 2014. The Family Court Judge has requested that amendments be made to the Law social the Court or appends caloring in a more effective and efficient manner. The amendment include the following:

- Definitions were expanded.
- Processes and procedures were added or claused.

II. Executive Summary of Findings

The operational costs of the Judiciary were budgeted beginning with the iscal year 2014 budget. There are no additional costs to the proposed amendments.

III. Financial Impact

No fiscal impact

IV. Recommendation

The Finance Department does not make a recommendation in regards to course of action in this matter. Rather, it is the purpose of this report to disclose potential financial impact of an action, so that General Tribal Council has full information with which to render a decision.

Page 35 of 340 Oneida Tribe of Indians of Wisconsin



UGWA DEMOLUM YATEHE Because of the help of this Onelda Chief in cementing a friendship between the six nations and the colony of Pennsylvania, a new nation, the United States was made possble.

1822 3/1

WHEREAS, the Oneida Tribe of Indians of Wisconsin is a federally recognized Indian government and a treaty tribe recogneed by the laws of the United States of America; and WHEREAS. the Oneida General Tribal Council is the governing body of the Oneida Tribe of Indians of Visconsin; and WHER AS, e Oneida Business com attee has been delegated the authority of Article IV of the Oneida Tribal Constrution of the Oneida General Tribal Council; and WHEREAS, Jan ary 7, 2013 the General Trips, Council adopted the Judiciary Law; and WHEREAS, with ne 2 o lion of the Judic try La General Tribal Council directed that Rules of cedue (Rules) be adopted by he Oneida Business Committee or by , Pr Appella General Tribel Cancil; and the Oneida using as Compittee adopted rules on April 25, 2014; and WHEREAS, Court of Appends (Chefpludge) requested amendments WHEREAS, the Chief Appelle Judge of definitions and to include provisions that were be made to the clarn JS inadvertently omitted; and the Oneida Business Compittee adapted those amendment on an emergency basis on WHEREAS, December 19, 2014 to ensure they would be implemente begins accepting cases on January 5, 2015; and before the new Judiciary WHEREAS, Legislative Operating Committee processed the appledments for permanent adoption, including presenting them at a public meeting on rebustry 5, 2015, in accordance with the Legislative Procedures Act.; and

NOW THEREFORE BE IT RESOLVED, that the attached mendment to the Rules of Appellate Procedure are hereby adopted and shall be effective in ten (10) bus nessedays.

BC Resolution _____ Rules of Appellate Procedure Amendments

Oneidas bringing several hundred bags of corn to Washington's starving army at Valley Forge, after the colonists had consistently refused to aid them.



Oneida Tribe of Indians of Wisconsin Legislative Reference Office

Lynn A. Franzmeier, Staff Attorney Taniquelle J. Thurner, Legislative Analyst Candice E. Skenandore, Legislative Analyst



P.O. Box 365 Oneida, WI 54155 (920) 869-4376 (800) 236-2214 https://oneida-nsn.gov/Laws

Statement of Effect

Rules of Appellate Procedure Amendments

Summary

This Resolution adopts amendments to the Rules of Appellate Procedure to clarify definitions and to include provisions that wave indivertently omitted when the Rules were originally adopted.

Submitted, Lynn ... ranzmeier, Souff Atto ney

alysis by the Leastanive Reference Office

On January 7, 2019 the G heral Tribal Coursil (GTC) advected the Judiciary Law and directed that Rules of Appenate Froctaure be adopted by the Cheice Business Committee (OBC) or GTC. The OBC adopted bales of Appellate Procedure on April 25, 2014. The Judiciary began accepting cases on January 5, 2015 and equested emergency amendments to the Rules be made before that date in order to enable the Judiciary to operate unterconsistent rules. The OBC adopted emergency amendments to the Rules on December 11, 2014 in order to clarify definitions and include provisions the were inadvertently omitted used the Fules.

After the emergency amendments were adopted, the Legislative Operating Complittee processed the amendments for permanent adoption in accordance with the Legislative Procedures Act, including presenting them at a public meeting on February 5, 2015.

Conclusion

The adoption of this Resolution does not conflict with an current Tribal Law or Policy.
Page 37 of 340

For OBC consideration (redline) 03/11/15

Chapter 154 Rules of Appellate Procedure

- 154.1. Purpose and Policy
 154.2. Adoption, Amendment, Repeal
 154.3. Definitions
 154.4. General Provisions
 154.5. Initiating the Appeal
 154.6. Appeal by Permission
- 154.7. Joint, Consolidated and Cross Appeals
- 154.8. Service, Filing and Certification
- 154.9. Time Computation

154.10. Motions
154.11. Briefs
154.12. Oral Argument
154.13. Entry and Form of Judgment
154.14. Interest of Judgments
154.15. Penalties
154.16. Substitution of Parties
154.17. Costs

Analysis by the Legislative Reference Office										
Title	Rules of Appellator rocedures (Law)									
Requester	Chief Appellat sy lge Drafter Lynn Franzmeier Analyst Candice Skenandore									
Reason for	The Chief Appel ate Jud has requested the Law be amended for clarification									
Reg .st										
Purpose	The purpose of this haw is to go tern the procedures in all actions and proceedings in the Tribe's Court of Appea?									
Authorized Affected Entities	Coart of Appeals, Could Staff, person publicing the Court of Appeals									
Due Process	Cente Appeals									
Related Legislation	Rules of Ci il Pocedure and the Feleral Rule of oppellate Procedure can be used as a guide when this Leanues not address an inde; however, those rules must be consistent with existing Orada Rule of Procedure, Tribal lands or customs of the Tribe									
Policy Mechanism	The Court of Appeals can assume not penalties for friction of the easy and non- compliance with the rules									
Enforcement	The Court of Appeals can use penalties which may include among other things, court costs, attorney fees, double costs, interest on the award coount came as, dismissal of the appeal, summary reveal of the original hearing body dousion and prother actions as the Court of Appeals consider appropriate.									

1 2

Overvie

This Law governs the procedures in all actions and proceedings of the Court of Appeals (Court) and can be used in conjunction with the Reles of Sivil Procedure *[See 154.1-1 and 154.4-1]*. The Oneida Business Committee approved concretely amondments to this Law on December 19, 2014, and will now be considering these an endments on a permanent basis. If these amendments are not permanently adopted or are not extended, these emergency amendments will expire on June 19, 2015.

- 9
- 10 11

Proposed Amendments

The proposed amendments include the following:

- The definition section has been expanded to include definitions for "initial review",
 "original hearing body" and "record" as well as separates the definitions of "advocate" and "attorney" [See 154.3-1 (a), (o), (s) and (v)].
- The current Law does not address what happens when the Court denies a request for stay;
 therefore, language was added that requires the Court to state the reasons for denying an
 appeal or request for stay within 30 days of the receipt of Notice of Appeal. *[See 154.4-1* (b)].

- 19 The Law will now require three Appellate Judges be assigned to perform an initial review 20 of the Notice of Appeal within ten days of filing the Notice of Appeal or the Perfected Notice of Appeal [See 154.5-2 (b)]. The current Law does not require an initial review. 21
- 22 A party can now file required papers to the Court by using private mail so long as it has a delivery tracking feature. In accordance with the current Law, a party filing by mail must 23 do so by using certified mail with a return receipt [See 154.8-1 (c)]. 24
- If a party or Clerk demands, the party filing documents must provide, among other 25 things, proof of delivery of the filing in question. Proof of delivery is not specifically 26 required in the current Law [See 154.8-2 (a)]. 27
- When accepting an appeal, the Clerk must now notify the Trial Court clerk or original 28 hearing body that an appeal has been filed and request that the Trial Court clerk or 29 original hearing body prefere and file with the Court all papers comprising the record of 30 the appealed case with 0 days. When the Clerk certifies the record, it must be served 31 to all parties. The finite Judge can extend this 30 day timeline for filing and certifying the record for good cause april a written request from the Trial Court clerk. Currently 32 33 the Law requires the Approache court Clerk, not the Trial Court Clerk, to prepare, to certify and file all the papers concerning the record of appealed cases with the Court. In addition, the corrent Law does not address extending the filing and certifying the record 34 35 36 30 day the eli e [See 154.8-4]. 37
- Audio recordings and be considered a record of the case [See 154.8-4 (a)]. The Court vill tave 180 days to complete a case, instead of 120 days which is currently 39 required, not including extensions [See 14.9-3]. 40
- The Appellant has 2 day now when the Certification of the Record is accepted to serve 41 a brief to the Respondent and fill the brief with the Clerk. Corrently the Appellant has 20 42 days from when the <u>Notice of Append</u> is filed to <u>serve</u> file the brief [See 154.11-1] 43 44 (d)].
- The Court can now permit parties to cite or discuss a case at an oral argument that was 45 not cited in one of the briefs *we 154.12-2*. The current **w** of Ty allo vs parties to cite 46 or discuss a case if the case has been cited in one of the briefs. 47
- 48 49

38

Consideratio

This Law will require three Appellate Judges be assigned to perform an initial review of 50 the Notice of Appeal within ten days of the Notice Appeal or Perfected Notice of Appeal 51 are filed [See 154.5-2 (b)]. There may be times when are Appellate Judges are not available to 52 perform an initial review. In order to avoid burdening the Curt, A Legislative Operating 53 Committee may want to consider eliminating the number of Appellar Judges required to 54 perform the initial review which would allow the Law to have me e flexibility. 55 56

Miscellaneous

57 A public meeting was held on February 5, 2015. Additional, minor revisions were made that do not affect the content of this Law. 58

59 60

61 **154.1.** Purpose and Policy

154.1-1. Purpose. The purpose of this Law is to govern the procedure in all actions and 62 63 proceedings in the divisions that make up the Court of Appeals within the Judiciary that fall

under the jurisdiction of the Tribe. 64

65 154.1-2. *Policy*. It is the policy of the Tribe that these rules are to be liberally construed to 66 ensure a speedy, fair, and inexpensive determination of every appeal.

67

81

68 154.2. Adoption, Amendment, Repeal

- 154.2-1. This Law was adopted by the Oneida Business Committee by resolution BC-04-25-14-69 70 Β.
- 154.2-2. This Law may be amended or repealed pursuant to the procedures set out in the Oneida 71
- Legislative Procedures Act by the Oneida Business Committee or the Oneida General Tribal 72
- 73 Council.
- 154.2-3. Should a provision of this Law or the application thereof to any person or 74 circumstances be held as invalid, such invalidity shall not affect other provisions of this Law 75 which are considered to have legatoric without the invalid portions. 76
- 154.2-4. In the event of a conditional to between a provision of this Law and a provision of another 77 law, the provisions of this *V* w hall control. 78
- 154.2-5 This haw is adopted up to 7 shority of the Constitution of the Oneida Tribe of Indians 79 of Wigconsin. 80

154.3. Demition 82

- 154.3-1. This section shall govern the definitions of words and phrases used within this Law. 83 All words not defin d here a shall be used ire new ordinary and everyday sense: (a) "Advoc te" and mean an Oneida non-adorney dvocate as provided by law or other 84
- 85 advocate who is presered to the court as the representative or advisor to a party. 86
- (a)(b) "Agent" shall near a person authorized that on b half of another. 87
- (b)(c) "Amicus Curiae" shall man (literally, friend c, the purt) a person who is not a 88 party to a case, nor solutied by any f the parties, y to fills a brief to assist the Court by 89 furnishing information of relative garding questions of raw or fact. 90
- 91
- (c)(d) "Answer" shall mean a vertue response in opposition to a brief or petition. (d)(e) "Appeal" shall mean a vertue of the Court of Appends by appeal or writ of error 92 authorized by law of a judgment worder of the Trial Court or <u>a Iginal her ing body</u>. 93
- (e)(f) "Appellant" shall mean a person who files photice of appeal. 94
- (f)(g) "Attorney" shall mean an Onere non-storney advocate a r svided by law and 95 other advocatea person who is admitted to ractice law and is presented to the court as 96 the representative or advisor to a party. 97
- (g)(h) "Brief" shall mean a written legal document which aids in the Court's decision by 98 reciting the facts of the case, the arguments being raised on peal, and the applicable 99 100 law.
- (h)(i) "Clerk" shall mean the Clerk of the Court of Appeal. 101
- (i)(i) "Court" shall mean the Court of Appeals of the Tribe. 102
- (i)(k) "Cross-Appeal" shall mean an appeal brought by the Respondent against the 103 Appellant after the Appellant has already filed an appeal. 104
- (k)(1) "Days" shall mean calendar days, unless otherwise specifically stated. 105
- (+)(m) "Docketed" shall mean an appeal that has been filed and assigned a docket 106 107 number.
- (m)(n) "Electronic" shall mean an electronic communication system, including, but is not 108 109 limited to E-mail, used for filing papers with the Court or serving papers on any other party. 110

111 (o) "Initial Review" shall mean review of the Notice of Appeal to determine if the case is acceptable for appellate review. 112 (n)(p) "Interlocutory" shall mean an order or appeal that occurs before the Trial Court or 113 original hearing body issues a final ruling on a case. 114 (o)(q) "Joinder" shall mean the joining together of several claims or several parties all in 115 one (1) hearing, provided that the legal issues and the factual situation are the same for all 116 Appellants and Respondents. 117 (p)(r) "Judiciary" shall mean the Oneida Tribal Judicial System. 118 (s) "Original hearing body" shall mean the administrative agency decision-making panel 119 which heard a contested case under the Administrative Procedures Act (or similar law) 120 and from which appeal is permitted by law. 121 (q)(t) "Petitioner" shall mon a person filing a petition. 122 (r)(u) "Pro se" shall m at advocating on one's own behalf before the Court, rather than 123 124 125 $\frac{(s)(w)}{(s)}$ "Leply Brief" shall y can const of a party to a legal action in answer to points of law raised in an opponent's brin out not in his or her own. 126 127 "Region ent" shall meen a perion adverse to the Appellant. 128 (u)(s_ Reles shall mean the Court of Appear these Rules of Appellate Procedure. 129 (v)(z) "Stay shall mean a suspension of a case or a suspension of a particular proceeding including orders, within case that presents enforcement pending appeal or 130 131 other circumstance 132 (w)(aa) "Trial Court" shall mean the Trial Court of the Tribe. (x)(bb) "Tribal" or "Tribe" shall mean the Oneida Tribe of Italians of Wisconsin. 133 134 (y)(cc) "Tribal law" shall mean a code, act, statute rule collection, policy or ordinance 135 enacted by the Oneida Original Typal Original or the Original Busidess Committee. 136 137 **154.4.** General Provisions 138 154.4-1. These Rules may be used in aging fon with the Rules of civil Procedure. Matters 139 and proceedings not specifically set forth herein shall be handled in accordance with reasonable 140 justice, as determined by the Court. Where the Rule fail to address an and, the Federal Rules 141 of Appellate Procedure may be used as a guide, second as those rules are not inconsistent with 142 existing Oneida Rules of Procedure, Tribal laws, or the custor is on the Tribe. 143 154.4-2. On its own or by a party's motion; the Court man, to expedite its decision or for other 144 good cause, suspend any provision of these Rules in a particular case of order proceedings as it 145 directs. 146 154.4-3. The Chief JusticeJudge of the Court shall, when hearing a case have the authority to 147 compel the production of documents where such is deemed necessary to endition of the Court's 148 opinion. There shall not be a new trial in the Court. The Court may review both the factual 149 findings and conclusions of law of the Trial Court or original hearing body. 150 151 152 **154.5.** Initiating the Appeal 153 154.5-1. *Right of Appeal*. Any party to a civil action, who is aggrieved by a final judgment or order of the Trial Court or original hearing body, may appeal to the Court of Appeals. 154 155 (a) In any case brought on appeal, the Appellant may petition the Court for an order staying the judgment or order. A stay shall be granted in all cases in which it is requested 156 157

- render a stay conditioned upon execution of a bond to guarantee performance of the 158 judgment or order when deemed necessary. 159 (b) In the event the appeal or request for stay is denied, the Court shall state the reasons 160 for the refusal within thirty (30) days of the receipt of the Notice of Appeal. 161 154.5-2. *Notice of Appeal*. Any party who is appealing shall appeal in the manner prescribed by 162 this Rule. 163 (a) Such party shall file with the Clerk a Notice of Appeal from such judgment or order, 164 together with a filing fee, as set by the Court, within thirty (30) days after the day such 165 judgment or order was rendered. A Notice of Appeal shall not be filed by electronic 166 means. 167 (b) Within ten (10) days of the filing of the Notice of Appeal or the Perfected Notice of 168 Appeal as provided und 154.5-3, three (3) Appellate Judges shall be assigned to 169 perform an Initial Review of the Notice of Appeal. 170 (1) Waiver f(x). The Chief Judge of the Court may waive the filing fee upon notion for a fet wary by the Appellant where the Chief Judge of the Court is 171 172 stisfied the Appell at larks be means to pay the filing fee. The motion shall include an affidavit domenstrating inability to pay and shall accompany the 173 174 f Appeal. 175 (b)(c_in vdc tion to the Notice of Apr al and filing fee, the following information shall 176 be provide upor q e filing of the prace. 177 (1) A cupy of the written decision of the Tric Court<u>or original hearing body;</u> 178 (2) A short tatement explaining that relies sought by the Appellant; 179 (3) A short rate on xplaining the local groups for seeking the appeal and 180 justification for ne relief equested; 181 (4) Name, address and phone numbers of all partice, including respondent; and 182 (5) Name, addressed plone numbers of all party atteneys or advocates, if 183 known. 184 (c)(d) A cash deposit or bong in an amount equal to the mount of any judgment, plus 185 costs assessed by the Trial Courter or <u>rainal hearing body</u>, or a motion for waiver of this 186 requirement, shall accompany the Narce of Appel. The deposit/bold requirement may 187 be waived only when, in the judgment of the Court, such depression of the court, such depression of the court is not in the 188 interest of justice and such waiver does not interessarily harm the judgment holder. The 189 motion for waiver of the deposit/bond requirement stall to requested with notice to all 190 parties. If the motion for waiver is denied, the deposit ond shall be submitted within ten 191 (10) days of the denial. The appeal shall be dismissed in the posit/bond is not paid or 192 waived. 193 (1) Exception. The Tribe, or an officer or agen of the ribe shall be exempt 194 from the requirement of providing any cash deposit or fond. The exemption 195 196 under this section shall be automatic and shall not require a motion or waiver. (d)(e) An appeal shall not be dismissed for informality of form or title of the notice of 197 appeal, or for failure to name a party whose intent to appeal is otherwise clear from the 198 notice. 199 200 154.5-3. *Perfection of Notice*. If the appellant fails to provide a completed Notice of Appeal Form, the filing fee or waiver form, or any required documents or materials, the Appellant shall 201 202 be notified of any filing deficiencies by the Clerk within five (5) business days and shall have five (5) business days from receipt of this notice to perfect the filing. Failure to perfect the filing 203
- within five (5) business days may result in the non-acceptance of the appeal.

205

154.6. Appeal by Permission 206

154.6-1. Appeal by Permission. An appeal from an interlocutory order may be sought by filing 207 208 a Petition for Permission to Appeal with the Clerk within ten (10) business days after the entry of 209 such order with proof of service on all other parties to the action. Within ten (10) business days after service of the petition, an adverse party may file an Answer in opposition. A decision shall 210 be issued in a reasonable time, but no longer than thirty (30) days from the first deliberation 211 unless good cause to extend the deadline is found by the Court. This extension shall be in 212 writing. The petition shall contain: 213

- (a) a statement of the facts necessary to develop an understanding of the question of law 214 determined by the order of the Trial Court or original hearing body; and 215
- (b) a statement of the que ion itself; and 216
- (c) a statement of the *t* alons why substantial basis exists for a difference of opinion on 217
- the question; and 218
- the Nief sought: ar 219
- 220
- (e) why in immediate appendman(1) materially advance are termination of the litigation;
 - (2) no ect the petitioner from substantial or irreparable injury; or
 - (3, c) fify an issue of general intervalue in the administration of justice; and
- (f) The petition shall include or have a copy of the order of the Trial Court <u>or original</u> <u>hearing boy</u> at ach d thereto. 224 225
- 226

221

222

223

154.7. Joint, Consolidated and cross Appeals 227

154.7-1. Joint or Consolidated Appeal. When two (2) or core arties are entitled to appeal 228 from a Trial Court or original hearing body judgment or galer. A their interests make joinder 229 practicable, the parties may file icent nonce Cappeal. The partie may then proceed on appeal 230 as a single Appellant. 231

- (a) When the parties have find separate mely notices appear, the appeals may be 232 joined or consolidated by the Court 233
- (b) If the persons do not file a joint repeal or electo proceed a a single Appellant, or if 234
- their interests are such as to make inder inpractical, the proceed as 235 Appellant and co-Appellant, with each co-appellant to have the same procedural rights 236 and obligations as the Appellant. The Appellant shall be the person who filed first. 237
- Cross Appeal. A Respondent who seeks medication of the judgment or order 154.7-2. 238 appealed from or of another judgment or order entered in the same action or proceeding shall file 239 a notice of cross-appeal within the time established for the hing of notic of appeal or ten (10) 240 business days after the receipt of the notice of appeal, whichever is later. he Respondent shall 241 be listed as the cross-Appellant. A cross-Appellant has the same right and obligations as an 242 243 Appellant under these Rules.
- 244

245 **154.8.** Service, Filing and Certification

154.8-1. A paper required or permitted to be filed in the Court shall be filed with the Clerk. The 246 filing party shall supply the Clerk with the original papers and three (3) copies. The filing party 247 shall also provide one (1) copy of the papers for each opposing party or party's attorney or 248 249 advocate. Filing shall be complete by the close of business on the day which the filing is due.

The following methods of filing shall be used, in order of preference: 250

251 (a) In Person: A party to a pending case, or the party's attorney, advocate or authorized 252 Agent may file papers in person before the Clerk. (b) *Electronic*: A party to a pending case may file papers electronically to the electronic 253 254 address, designated for such filings, of the Clerk. A paper filed by electronic means shall constitute a written paper for the purpose of applying these Rules. Upon receipt by the 255 Clerk, any paper filed electronically shall be deemed filed, signed and verified by the 256 filing party. 257 (c) By Mail: A party to a pending case may file papers by certified U.S. or private mail 258 with return receipt the ability to track the delivery, with cover documents to be addressed 259 to the Clerk. Filing shall not be completed upon mailing, but only upon receipt. 260 (1) Certified mail shall include the filing Filing of papers is also permitted 261 through the Tribalertified interoffice mail system. 262 154.8-2. Proof of Service. <u>In a demand by a party or the Appellate Clerk, a party filing documents shall provide op 11 of the following:</u>
<u>A prover presented for fame shall contain either of the following:</u>
<u>(a) Proof of delivery of the same story estion;</u>
(a) <u>(b)</u> an acknowledgment of revice by the person served; or 263 264 265 266 267 f service consisting of a statement by the person who made service Dre Dr 268 certif, mg 269 (1) t e dat and manner of se ace, 270 (2) the names of the persons s rved; 271 (3) the main or dectronic addresses, facsing the numbers of the persons served, or 272 the address of the process of delivery as appropriate for the manner of service; 273 274 and (4) if served exctronically, writing by the per of being served consenting to 275 service by electric in mean. 276 154.8-3. Service of All Papers Required. A party shall, at or be present time of filing a paper, 277 serve a copy on all other parties to the appear. A party may be rive by electronic means, if 278 such party consents in writing to service by electronic means. Service on a party represented by 279 an attorney or advocate shall be made on the arty's attorney or advoca 280 154.8-4. Certification of the Record. Upon receiptace of the Not. Appeal and Proof 281 of Service, the Clerk shall, notify the Trial Court Lerk or original hearing body that an appeal 282 has been filed and request, the Trial Court clerk or riginal Larin, body to prepare, certify and 283 file with the Appellate Court all papers comprising the reast of the case appealed. The within 284 thirty (30) days. Upon Certification of the Record by the Verk nishalle, served on all parties as 285 provided for in 154.8-3. The time for filing and certifying he recent may be extended for good 286 cause by the Chief Judge of the Court upon a written request from the Trial Court clerk or 287 original hearing body. 288 (a) The record of the case shall consist of all papers filed with the Trial Court or original 289 hearing body, exhibits, thea transcript/ or audio recording of the proceedings, and the 290 final decision of the Trial Court<u>or original hearing body</u>. 291 292 154.9. Time Computation 293

154.9-1. *Deadline Computation*. Time lines are determined by designating the day after notice
is received as day one. Computation involving calendar days shall include intermediate Tribally
observed holidays and weekend days, provided that if the last day of the period falls on a
Saturday, Sunday or Tribally observed holiday, then the next business day shall be the due date.

Computation involving business days shall not include intermediate weekend days or Tribally
observed holidays. All papers due to be filed with the Clerk are due prior to the close of business
on the last day of the time period.

301 (a) If notice is mailed, then three (3) days shall be added to the time line in order to302 determine the due date.

154.9-2. *Extension of Time*. For good cause, the Court may extend the time prescribed by these
Rules or by its order to perform any act, or may permit an act to be done after that time expires.
But the Court shall not extend the time to file:

- 306 (a) a notice of appeal; or
 - (b) a petition for permission to appeal.

154.9-3. *Time to Complete*. Unless time is extended by the Court with the knowledge of the parties, the time from the filing of the Notice of Appeal to the completion and entry of the final written decision shall not exceed the hundred and twenty (120eighty (180)) days.

311 312 **154.10** Motio.

307

154.101. Application for Renef. on arguination for an order or other relief in a docketed case
shall be made by motion unless these laws prescribe another form. A motion shall be in writing
unless the court of the otherwise. The moving party shall file all motions with the Clerk and
serve opposing parties as provided in 1.4.8.

- 317 154.10-2. Content of car otion. A motion shall state with particularity the grounds for the
 318 motion, the relief cought, and the legal argument necessary in support it.
- (a) Any afficavit c other paper necessary to support a motion shall be served and filed
 with the motion. A afficavit shall contain only factual information, not legal argument.
 A motion seeking subpartive relief shall include a constrained of the Trial Court's or original
 hearing body's opinion as a separate exhibit.

154.10-3. *Response to a Motion*. Any party may file a response to a motion, in accordance with 154.11-2. The response shall be filed within ten (10) days after service of the motion unless the

325 Court shortens or extends the time.

154.10-4. *Motion for a Procedural Order*. The Court may act on a motion for a procedural
order at any time without awaiting a response. A party diversely affected by the Court's action
may file a motion to reconsider, vacate, or modify the action within five (f) days of receipt of
notice of the decision.

154.10-5. *Motion for Voluntary Dismissal*. An appellant may dismiss an appeal by filing a
 motion to dismiss. If not yet docketed in the Court, menute motion shall be filed in the Trial

motion to dismiss. If not yet docketed in the Court, merche motion shall be filed in the Trial
Court<u>or original hearing body</u>. The dismissal of an appeal shall not shall be status of a crossappeal or the right of a respondent to file a cross appeal.

- 154.10-6. *Form.* Motions shall be typed, legible and include the case crotion. Every motion shall:
- (a) Contain a caption heading, the name Judiciary- Court of Appeals, the title of the
 action, the docket number (if known) and a designation as to the purpose or type of
 motion.
- (b) Contain the names of all parties to the action.
- 340 (c) Be organized in sections containing a clear designation, which shall include, but is341 not limited to:
- 342 (1) The facts, events or occurrences which make a specific motion for relief343 necessary;
- 344 (2) The specific relief requested by the moving party;

For OBC consideration (redline) 03/11/15

- 345 (3) The applicable law or laws to the motion at hand, including citations; and 346 (4) The legal reasons the relief should be granted. (d) Be on $8\frac{1}{2}$ by 11 inch paper. The text shall be double-spaced, but quotations more 347 than two (2) lines may be indented and single-spaced. Headings and footings may be 348 single-spaced. Margins must be at least one (1) inch on all four (4) sides. Page numbers 349 may be placed in the margins, but no other text shall appear there. 350 (e) Be typed in a plain, roman style, although italics or boldface may be used for 351 emphasis. Case names shall be italicized or underlined. 352 (f) Not exceed twenty (20) pages, unless the Court permits or directs otherwise. 353 354 355 154.11. Briefs Briefs Generally. Ariefs shall be used by the Court to aid the Court in its 154.11-1. 356 consideration of the issues press ed. 357 (a) *Form.* The brief shall be 1.5 line spaced, typed, 1 inch margins, and on 8.5 x 11 inch 358 aper, a d shall be signed by the party or the party's attorney or advocate, if represented. 359 The fron cover of a brief shall craat. 360 (1) the number of the graventer d at the top; 361 (2) and the court; 362 (3, the title of the case; 363 (4) the nation of the proceeding (e x., Appeal, Petition for Review) and the name 364 of the chart elow 365 (5) the title of t' brief, identifying the part of parties for whom the brief is filed; 366 and 367 (6) the name, whice address, and telephone pumber of the attorney or advocate 368 representing the party for whom the brief is field, if presented. 369 (b) Length. The brief shall be no more than twenty (2) piges, the (1) sided, in length, 370 not including any addendumer appendices, attachments, it the tables of contents and 371 authorities. 372 (c) *Filing*. When a party is represented by an attorney <u>or advecate</u>, only the attorney <u>or</u> <u>advocate</u> shall file briefs and pleadings. The individual shall bot file on his or her own 373 374 unless he or she is pro se. Three (3) covies of uch brief shall be the with the Clerk and 375 one (1) copy to all parties to the appeal. 376 (d) Time to Serve and File a Brief. The Applant shares on the Respondent and file 377 with the Clerk a brief within twenty (20) dry's after the Notice of Appeal is filed.acceptance of the Certification of the Record. The Responsent's brief shall be filed 378 379 with the Clerk within twenty (20) days of receipt of the Aperlant brief. A reply brief, 380 if necessary, shall be filed within fourteen (14) days of receipt d Respondent's brief. 381 The Court may, on its own, order different time lines for an part s time to file a brief. 382 383 (e) Consequence of Failure to File. If an Appellant fails to file a brief within the time provided by this Rule, or within an extended time, a Respondent may move to dismiss the 384 appeal. A Respondent who fails to file a brief shall not be heard at oral argument unless 385 386 the Court grants permission. 387 154.11-2. Appellant's Brief. The Appellant's brief shall contain, under appropriate headings and in the order indicated: 388 389 (a) Content:
- 390
- (1) a table of contents, with page references;

204	(2) - table of antibacities are calculated at all summer d) statutes and other
391	(2) a table of authorities-cases (alphabetically arranged), statutes, and other
392	authorities-with references to the pages of the brief where they are cited;
393	(3) a jurisdictional statement, including:
394	(A) the basis for <u>the</u> Trial Court's <u>or original hearing body's</u> subject-matter
395	jurisdiction;
396	(B) the basis for the Court of Appeals' jurisdiction;
397	(C) the filing dates establishing the timeliness of the appeal; and
398	(D) an assertion that the appeal is from a final order or judgment that
399	disposes of all parties' claims, or information establishing the Court of
400	Appeals' jurisdiction on some other basis;
401	(4) a statement of the issues presented for review;
402	(5) a statement of the case briefly indicating the nature of the case, the course of
403	proceedings, are the disposition below;
404	(6) a statement of fact relevant to the issues submitted for review with
405	appropriate references the record;
406	() a summary of the argument which shall contain a succinct, clear, and accurate
407	statement of the arguments made in the body of the brief, and which shall not
408	meany epeat the argument heading
409	(8) the argument, which shall contain:
410	(A) pellant's contendors and the reasons for them, with citations to the
411	ath rities and parts of the record on which the Appellant relies; and
412	(B) or ruch issue, a concre statement of the applicable standard of review
413	(where my appear in the discussion of the issue or under a separate
414	heading placed before the discussion of the issues);
415	(9) a short concusion stating the precise relies sort a
416	(10) a short appending to include
417	(A) relevant doct et entries in the Trial Court or <u>aginal hearing body</u> ;
418	(B) limited portions of the record essentiant of a understanding of the
419	issues raised;
420	(C) the judgment, order, or decision in question; and
421	(D) other parts of the record to prich the parties with a direct the Court's
422	attention; and
423	(11) where the record is required by law to be confidential, reference to
424	individuals shall be by initials rather than by names.
425	154.11-3. Respondent's Brief. The Respondent's brief shall conformer, the same requirements
426	as 154.11-2 (Appellant's Brief).
427	(a) The Respondent's brief shall address each issue and argument presented by the
428	Appellant's brief.
429	(b) The Respondent's brief may present additional issues, with the Respondent's
430	positions and arguments on such issues.
431	154.11-4. <i>Reply Brief.</i> The Appellant may file a brief in reply to the Respondent's brief. Unless
432	the Court permits, no further briefs may be filed. A reply brief shall conform to the requirements
433	of 154.11-3 (Respondent's Brief), except that a reply brief shall be no more than fifteen (15)
434	pages, one (1) sided, in length.
435	154.11-5. Amicus Curiae Brief. A person who is not a party to a case but has some interest in
436	the outcome of the case may, upon timely motion and with permission of the Court, submit an
	-

437 amicus curiae brief in support of a party to the action. The Court may, on its own motion, 438 request amicus participation from appropriate individuals or organizations.

- (a) Amicus curiae briefs shall conform to the requirements of 154.11-2 (Appellant's 439 Brief), except as provided in the following: 440
- (1) Amicus curiae shall file his or her brief no later than seven (7) days after the 441 brief of the party being supported is filed. Amicus curiae that do not support 442 either party shall file his or her brief no later than seven (7) days after the 443 Appellant's or Respondent's brief is filed. The Court may grant leave for later 444 filing, specifying the time within which an opposing party shall answer. 445
- 154.11-6. Briefs in a Case Involving Multiple Appellants or Respondent. In a case involving 446 more than one (1) Appellant or Respondent, including consolidated cases, any number of 447 Appellants or Respondents may i in in a brief, and any party may adopt, by reference, a part of another's brief. Parties may all of oin in reply briefs. 448 449

154.12 Oral Argument 451

154.17.1. Oral Arguments. The court may order oral argument when issues of fact or law remain unclear and/or the positions or an parties on an issue are unclear or otherwise not fully 452 453 developed. The Coart shall direct that an appeal be submitted on briefs only, if: 454

- (a) The appendix frivolous; 455
- 456
- (b) The distribution such as the second authoritatively decided; or(c) The facts and legal arguments are adequately presented in the briefs and record, and 457 the decisional process y ould not be sign scantly a ready oral argument. 458
- 154.12-2. Notice. The Charter provide notice, of a feast ten (10) business days, to all parties 459 when oral arguments are sche uled. The notice shan list the location of the oral argument and 460 the time allowed for each side. The Court shall determine the amount of time for oral arguments. 461 A motion to postpone the argument or to extend the argument time frame shall be filed at least 462 five (5) business days before the hearing date. 463
- 154.12-3. Citation of Authorities a Oral Argument. Parties ess similar by the Court, 464 parties may not cite or discuss a case at a oral gument unless the case has been cited in one (1) 465 of the briefs. 466
- 467

450

154.13. Entry and Form of Judgment 468

- 154.13-1. Entry. A judgment is entered when it is noted on the docket. The Clerk shall prepare, 469 sign, and enter the judgment after receiving the Court's opinion 470
- (a) The decision and opinion of the Court shall be y a majority of the court shall be 471
- (b) The Court may: 472
- (1) Reverse, affirm, or modify the judgment or or as to approve all parties; 473
- (2) Remand the matter to the Trial Court or original hearing body and order a new 474 trial on any or all issues presented; the order remanding a case shall contain 475 specific instructions for the Trial Court or original hearing body; 476
- (3) If the appeal is from a part of a judgment or order, the Court may reverse, 477 affirm or modify as to the part which is appealed; 478
- (4) Direct the entry of an appropriate judgment or order; or 479
- (5) Require such other action or further proceeding as may be appropriate to each 480 481 individual action.
- (c) On the date when judgment is entered, the Clerk shall serve all parties with a copy of 482 the decision and opinion as entered. 483

Page 48 of 340

For OBC consideration (redline) 03/11/15

484 154.13-2. *Form.* All decisions of the Court shall be in writing and accompanied by an opinion
485 stating the legal issues and the basis for the decision. Decisions of the Court shall be issued no
486 later than sixty (60) days after the conclusion of oral argument or after the expiration of time to
487 file a *Reply Brief* or *Response Brief* if no oral argument is held.

- (a) The time for issuing a decision and opinion may be extended provided all parties are
 notified of the extension in writing. The notice of extension shall include the cause for
 and length of such extension.
- 491

492 **154.14. Interest of Judgments**

493 154.14-1. Unless the law provides otherwise, if a money judgment in a civil case is affirmed,
494 whatever interest is allowed by law is payable from the date when the Trial Court's <u>or original</u>
495 <u>hearing body's</u> judgment was entered. If the Court modifies or reverses a judgment with a
496 direction that a money judgment of e entered in the Trial Court<u>or by the original hearing body</u>, the
497 mandate shall contain instructions about the allowance of interest.

498 499 **154.1: Penalti s**

154.15.1. *Frivolous Appeals*. If an appeal or closs-appeal is found by the Court to be frivolous,
the Court may awrea to the successive party costs and attorney's <u>or advocate's</u> fees.

- (a) Costs may be assessed against the Appellant or cross-Appellant, the (cross)Appellant's attorned or advocate, or both the (cross)-Appellant and his/her attorned or
 advocate jointh
- 505 (1) Court costs shall be based on ctual cost of befined by the Court.
- 506(b) A finding of a photocompletal or cross-appeal shall be made if one (1) or more of the507following elements are found by the Court:
- 508(1) The appeal or cross appeal was filed, used, or continued in bad faith, solely509for purposes of other than smeet or injuring the opposing party; or
- 510 (2) The party or party's atterney <u>or advocate</u> knew or should have known, that 511 the appeal or cross-app al was without any reason are bears in two or equity and 512 could not be supported by a good raith argument for an extension, modification or 513 reversal of existing law.
- 514 154.15-2. *Delay*. If the Court finds that an appeal or cross-appeal was about for the purpose of delay, it may award one (1) or more of the following to the opposing party:
- 516 (a) Double costs;
- 517 (b) A penalty of additional interest not exceeding to precent (10%) on the award amount 518 affirmed;
- 519 (c) Damages caused by the delay; and/or
- 520 (d) Attorney's <u>or advocate's</u> fees.
- 521 154.15-3. *Non-Compliance with Rules*. Failure of a party to comply with a requirement of these
 522 Rules or an order of the Court, does not affect the jurisdiction of the Court over the appeal but
 523 may be grounds for one (1) or more of the following:
- 524 (a) Dismissal of the appeal;
- 525 (b) Summary reversal of the Trial Court<u>or original hearing body</u>;
- 526 (c) Striking of a paper, document or memorandum submitted by a party;
- 527 (d) Imposition of a penalty or costs on a party or party's attorney <u>or advocate</u>; and/or
- 528 (e) Other action as the Court considers appropriate.
- 529

530 154.16. Substitution of Parties

531 154.16-1. *Death of a Party*. Death of a party does not automatically end a party's right to appeal.

- (a) *After Notice of Appeal Is Filed.* If a party dies after a notice of appeal has been filed
 or while a proceeding is pending in the Court, the decedent's personal representative may
 be substituted as a party on motion filed with the Clerk by the representative or by any
 party. A party's motion shall be served on the representative. If the Decedent has no
 representative, any party may suggest the death on the record, and the Court may then
 direct appropriate proceedings.
- (b) *Before Notice of Appeal Is Filed-Potential Appellant.* If a party entitled to appeal dies before filing a notice of appeal, the decedent's personal representative, or if there is no personal representative, the decedent's attorney <u>or advocate</u> of record, may file a notice of appeal within the time prescribed by these Rules. After the notice of appeal is filed substitution shall be in accordance with 154.16-1(a).
- 544 (*)* Before Notice of Append 1 Filed-Potential Respondent. If a party against whom an 545 appeal may be taken dies after energy of a judgment or order in the Trial Court<u>or original</u> 546 bearing body, but before a notice of appeal is filed, an Appellant may proceed as if the 547 deam had not occurred. After the notice of appeal is filed, substitution shall be in 548 accordance with 154.16-1(a).
- 549 154.16-2. Substitution for a Reason Other Tran Leath. If a party needs to be substituted for any
 550 reason other than ceath the procedure set in 54.16-1(a) applies.
 551

552 154.17. Costs

- 553 154.17-1. *Costs.* Costs in an opeal shall be as follows unless other vise ordered by the Court:
- (a) Against the appellant when the appeal is dismissed or dejudgment or order affirmed;
 (b) Against the respondent when the judgment or order is refersed.
- 556 154.17-2. Allowable Costs. Allowable costs shart include:
- (a) Cost of printing and assembling the number of copie and orien and appendices
 required by the Rules;
- (b) Fees charged by the Court and/or elerk;
- 560 (c) Cost of the preparation of the transcript of testimony of the received of appeal; and
- 561 (d) Other costs as ordered by the Court.
- 154.17-3. *Recovery of Costs.* A party seeking to recover costs in the Court shall file a statement
 ofhteof the costs within fourteen (14) days of the riling on the decision of the Court. An
 opposing party may file, within eleven (11) days after service of the statement, a motion
 objection to the statement of costs.
- 567 *End*.
- 568 Adopted BC-04-25-14-B
- 569 Emergency Amended BC-12-19-14-A

For OBC consideration (clean) 03/11/15

Chapter 154 **Rules of Appellate Procedure**

154.1. Purpose and Policy 154.10. Motions 154.2. Adoption, Amendment, Repeal 154.11. Briefs 154.3. Definitions 154.12. Oral Argument 154.13. Entry and Form of Judgment 154.4. General Provisions 154.5. Initiating the Appeal 154.14. Interest of Judgments 154.6. Appeal by Permission 154.15 Penalties 154.7. Joint, Consolidated and Cross Appeals 154.16. Substitution of Parties 154.8. Service, Filing and Certification 154.17. Costs 154.9. Time Computation

1 2

154.1. Purpose and Policy

154.1-1. Purpose. The purpose of this Law is to govern the procedure in all actions and 3 proceedings in the divisions that make up the Court of Appeals within the Judiciary that fall 4 under the jurisdiction of the file. 5

- 154.1-2 *from* v. It is the policy and e Tribe that these rules are to be liberally construed to ensure a speedy fair, and inexpensive determination of every appeal. 6 7
- 8

9

- **154.2.** A leption, A soundment, Receal 154.2-1. This 2-w yas adopted by the Oneida Jushess Committee by resolution BC-04-25-14-10 solut
- B and amended by 11
- 154.2-2. This Law may be mended or reputed pursuant to the procedures set out in the Oneida 12
- Legislative Procedurys Ac by the Oneida Boliness Copulatee or the Oneida General Tribal 13 14 Council.
- Should a provision of the Law or the application thereof to any person or 154.2-3. 15
- circumstances be held as invelid, such invalidity shall not affect ther provisions of this Law 16 which are considered to have level force y and ut the invalid portion 17
- 154.2-4. In the event of a conflict between a provision of this Law a 18 d a provision of another law, the provisions of this Law shall ontrol. 19
- 154.2-5. This Law is adopted under autority of the Constitution of the Oneida Tribe of Indians 20 of Wisconsin. 21
- 22

23 154.3. Definitions

- 154.3-1. This section shall govern the definitions of words and phrases used within this Law. All words not defined herein shall be used in their order ary on lever day sense: 24 25
- (a) "Advocate" shall mean an Oneida non-attorned advocate as provided by law or other 26 advocate who is presented to the court as the representative or advocate to a party. 27
- 28 (b) "Agent" shall mean a person authorized to act on beha¹ of another.
- (c) "Amicus Curiae" shall mean (literally, friend of the court) a person who is not a party 29
- to a case, nor solicited by any of the parties, who files a one to assist the Court by 30 furnishing information or advice regarding questions of law or fact. 31
- 32 (d) "Answer" shall mean a written response in opposition to a brief or petition.
- (e) "Appeal" shall mean a review in the Court of Appeals by appeal or writ of error 33 authorized by law of a judgment or order of the Trial Court or original hearing body. 34
- (f) "Appellant" shall mean a person who files a notice of appeal. 35
- (g) "Attorney" shall mean a person who is admitted to practice law. 36
- (h) "Brief" shall mean a written legal document which aids in the Court's decision by 37 reciting the facts of the case, the arguments being raised on appeal, and the applicable 38 39 law.

For OBC consideration (clean) 03/11/15

- (i) "Clerk" shall mean the Clerk of the Court of Appeals. 40 (j) "Court" shall mean the Court of Appeals of the Tribe. 41 (k) "Cross-Appeal" shall mean an appeal brought by the Respondent against the 42 Appellant after the Appellant has already filed an appeal. 43 (1) "Days" shall mean calendar days, unless otherwise specifically stated. 44 (m) "Docketed" shall mean an appeal that has been filed and assigned a docket number. 45 (n) "Electronic" shall mean an electronic communication system, including, but is not 46 limited to E-mail, used for filing papers with the Court or serving papers on any other 47 party. 48 (o) "Initial Review" shall mean review of the Notice of Appeal to determine if the case is 49 50 acceptable for appellate review. (p) "Interlocutory" shall the an order or appeal that occurs before the Trial Court or 51 original hearing body jost is a final ruling on a case. 52 (q) "Joinder" shall r can the joining together of several claims or several parties all in one 53 hearing, provided that me egal issues and the factual situation are the same for all 54 Appellar s and Respondent 55 (r) "Judiciary" shall mean the an ida Th bal Judicial System. 56 (s) Original faring body" sall main the dministrative agency decision-making panel 57 which need, contested case a der the Administrative Procedures Act (or similar law) 58 and from which are all is permitted by law 59 (t) "Petitioner" nal mean a person thing a petition. 60 (u) "Pro se" shall nege advocating on ne's own be alf before the Court, rather than 61 being represented an corner or advocate. 62 (v) "Record" shall mean all materials identified in 154, -4(a) of these Rules. 63 (w) "Reply Brief" shall mean a brief a party to a gal ft n in answer to points of law 64 raised in an opponent's chief but yet in is or her own. 65 (x) "Respondent" shall mean apers in adverse to the Appel int 66 (y) "Rules" shall mean these Rules of Apprate Procedure 67 a particular proceeding. (z) "Stay" shall mean a suspension of a case or a suspension 68 including orders, within a case the prevents inforcement ending appeal or other 69 circumstances. 70 (aa) "Trial Court" shall mean the Trial Court of the Tribe 71 (bb) "Tribal" or "Tribe" shall mean the Onen Tribe. Incluse of Wisconsin. 72 (cc) "Tribal law" shall mean a code, act, statute run regulation, policy or ordinance 73 enacted by the Oneida General Tribal Council or the Oneida By Less Committee. 74 75 **154.4.** General Provisions 76 154.4-1. These Rules may be used in conjunction with the Rule of C'al Procedure. Matters 77 and proceedings not specifically set forth herein shall be handled in accordance with reasonable 78 79 justice, as determined by the Court. Where these Rules fail to address an issue, the Federal Rules of Appellate Procedure may be used as a guide, so long as those rules are not inconsistent with 80 81 existing Oneida Rules of Procedure, Tribal laws, or the customs of the Tribe. 82 154.4-2. On its own or by a party's motion; the Court may, to expedite its decision or for other good cause, suspend any provision of these Rules in a particular case and order proceedings as it 83
- 84 directs.

154.4-3. The Chief Judge of the Court shall, when hearing a case, have the authority to compel
the production of documents where such is deemed necessary to rendition of the Court's opinion.

For OBC consideration (clean) 03/11/15

87 There shall not be a new trial in the Court. The Court may review both the factual findings and 88 conclusions of law of the Trial Court or original hearing body.

89

117

121

122

90 **154.5.** Initiating the Appeal

- 154.5-1. Right of Appeal. Any party to a civil action, who is aggrieved by a final judgment or 91 order of the Trial Court or original hearing body, may appeal to the Court of Appeals. 92
- 93 (a) In any case brought on appeal, the Appellant may petition the Court for an order 94 staying the judgment or order. A stay shall be granted in all cases in which it is requested unless plain and obvious injustice would result from granting the stay. The Court may 95 render a stay conditioned upon execution of a bond to guarantee performance of the 96 judgment or order when deemed necessary. 97
- (b) In the event the appeal or request for stay is denied, the Court shall state the reasons 98 for the refusal within that (30) days of the receipt of the Notice of Appeal. 99
- 154.5-2. Notice of Appeal. An party who is appealing shall appeal in the manner prescribed by 100 this Rul 101
- (a) Such party shall file with the clear a Notice of Appeal from such judgment or order, together with a filing fee, as only the Sourt, within thirty (30) days after the day such judgment of order was rendired. A Notice of Appeal shall not be filed by electronic 102 103 104 mean. 105
- (b) Within the (19 days of the filing of us Notice of Appeal or the Perfected Notice of Appeal as provide, under 154.5-3, three (3) Appellate Judges shall be assigned to 106 107 perform an Initial Levi w of the Notice & Appeal 108
- (1) Waiver Ferrar Chief Judge of the Court may waive the filing fee upon 109 motion for a fer waiver by the Appenant where the Chief Judge of the Court is 110 satisfied the Appellant lacks the means to by the ling fee. The motion shall 111 include an afficient demonstrating inability of pay and shall accompany the 112 Notice of Appeal. 113
- (c) In addition to the Notice of Appeal and filing fee, the proving internation shall be 114 provided upon the filing of the indice: 115
- (1) A copy of the written decision of the Tral Court or original hearing body; 116
 - (2) A short statement explaining what rulef is sought by the Appellant;
- (3) A short statement explaining the legal grounds for seeking the appeal and 118 justification for the relief requested; 119
- (4) Name, address and phone numbers of all parties, including respondent; and (5) Name, address and phone numbers of all party corneys or advocates, if 120
 - known.
- (d) A cash deposit or bond in an amount equal to the amount of any judgment, plus costs 123 assessed by the Trial Court or original hearing body, on a motion for waiver of this 124 requirement, shall accompany the Notice of Appeal. The deposit/bond requirement may 125 be waived only when, in the judgment of the Court, such deposit/bond is not in the 126 interest of justice and such waiver does not unnecessarily harm the judgment holder. The 127 motion for waiver of the deposit/bond requirement shall be requested with notice to all 128 parties. If the motion for waiver is denied, the deposit/bond shall be submitted within ten 129 (10) days of the denial. The appeal shall be dismissed if the deposit/bond is not paid or 130 131 waived.
- (1) Exception. The Tribe, or an officer or agency of the Tribe shall be exempt 132 from the requirement of providing any cash deposit or bond. The exemption 133 under this section shall be automatic and shall not require a motion or waiver. 134

Page 53 of 340

For OBC consideration (clean) 03/11/15

(e) An appeal shall not be dismissed for informality of form or title of the notice of 135 136 appeal, or for failure to name a party whose intent to appeal is otherwise clear from the 137 notice.

138 154.5-3. Perfection of Notice. If the appellant fails to provide a completed Notice of Appeal Form, the filing fee or waiver form, or any required documents or materials, the Appellant shall 139 be notified of any filing deficiencies by the Clerk within five (5) business days and shall have 140 five (5) business days from receipt of this notice to perfect the filing. Failure to perfect the filing 141 within five (5) business days may result in the non-acceptance of the appeal. 142

143

154.6. Appeal by Permission 144

154.6-1. Appeal by Permission. An appeal from an interlocutory order may be sought by filing 145 a Petition for Permission to Appel with the Clerk within ten (10) business days after the entry of such order with proof of service in all other parties to the action. Within ten (10) business days after service of the petition an edverse party may file an Answer in opposition. A decision shall be issued in a peasonable time, but to longer than thirty (30) days from the first deliberation unless good cause to extend the readline refound by the Court. This extension shall be in 146 147 148 149 150 writing. The petition shall contain: 151

- (a) a statement of the facts in cessar, to develop an understanding of the question of law 152 determined by the order of the Nial Court or original hearing body; and 153
- 154
- (b) a statement of a question itself and(c) a statement of the reasons why substantial basis exists for a difference of opinion on 155 the question; and 156
- (d) the relief sough and 157
- (e) why an immediate opeal ma 158
- 159
- (1) materially a lyance the termination of the atige of 160
 - (2) protect the politicitier from substantial or irregarable inferry; or
- (3) clarify an issue of general importance in the administration of justice; and(f) The petition shall include is have a copy of the ordered the trial Sourt or original 161 162 hearing body attached thereto. 163
- 164

154.7. Joint, Consolidated and Cross Appea 165

154.7-1. Joint or Consolidated Appeals. When the (2) or more parties are entitled to appeal 166 from a Trial Court or original hearing body judgment or order, and their interests make joinder 167 practicable, the parties may file a joint notice of appeal. The prties may then proceed on appeal 168 as a single Appellant. 169

- (a) When the parties have filed separate timely notices of appear the appeals may be 170 joined or consolidated by the Court. 171
- (b) If the persons do not file a joint appeal or elect to proceed as single Appellant, or if 172 their interests are such as to make joinder impractical, the person shall proceed as 173 Appellant and co-Appellant, with each co-Appellant to have the same procedural rights 174 and obligations as the Appellant. The Appellant shall be the person who filed first. 175
- 154.7-2. Cross Appeal. A Respondent who seeks modification of the judgment or order 176 appealed from or of another judgment or order entered in the same action or proceeding shall file 177 a notice of cross-appeal within the time established for the filing of a notice of appeal or ten (10) 178 179 business days after the receipt of the notice of appeal, whichever is later. The Respondent shall be listed as the cross-Appellant. A cross-Appellant has the same rights and obligations as an 180 Appellant under these Rules. 181
- 182

For OBC consideration (clean) 03/11/15

154.8. Service, Filing and Certification 183

184 154.8-1. A paper required or permitted to be filed in the Court shall be filed with the Clerk. The filing party shall supply the Clerk with the original papers and three (3) copies. The filing party 185 shall also provide one (1) copy of the papers for each opposing party or party's attorney or 186 advocate. Filing shall be complete by the close of business on the day which the filing is due. 187 The following methods of filing shall be used, in order of preference: 188

- (a) In Person: A party to a pending case, or the party's attorney, advocate or authorized 189 Agent may file papers in person before the Clerk. 190
- (b) *Electronic*: A party to a pending case may file papers electronically to the electronic 191 address, designated for such filings, of the Clerk. A paper filed by electronic means shall 192 constitute a written paper for the purpose of applying these Rules. Upon receipt by the 193 Clerk, any paper filed electronically shall be deemed filed, signed and verified by the 194 filing party. 195
- (c) *By Mail*: A party to a pending case may file papers by U.S. or private mail with the offity to track the delivery, with cover documents to be addressed to the Clerk. Filing shall not be completed upor nailing, but only upon receipt. 196 197 198
- (1) Filing of papers is a permitted through the Tribal certified interoffice mail 199 200

154.8-2. Proof fervice. Upon de and be a party or the Appellate Clerk, a party filing 201 documents shall provide and (1) of the following. (a) Proof a deliver of the filing in question; 202

- 203
- (b) an acknowledg her of service by the person served or 204
- (c) proof of service considering of a statement by the person who made service certifying: 205
- (1) the date and manner of service; 206
 - (2) the names the persons erved;
- (3) the mail or experimentation addresses, facsimile, admiters of the persons served, or 208 the addresses of the places of derivery, as appropriate or the manner of service; 209 and 210
- (4) if served electronical, a writing by the person bung server consenting to 211 service by electronic means. 212

154.8-3. Service of All Papers Required. A party shall, at or before the side of filing a paper, 213 serve a copy on all other parties to the appeal. Appearty may be served by electronic means, if 214 such party consents in writing to service by electronic means. Service on a party represented by 215 an attorney or advocate shall be made on the party's attorney of advocate. 216

154.8-4. Certification of the Record. Upon acceptance on the Appeal ... Clerk shall, notify the 217 Trial Court clerk or original hearing body that an appeal he bee filed and request, the Trial 218 Court clerk or original hearing body to prepare, and file with the Appendix Court all papers 219 comprising the record of the case appealed within thirty (30) day. Usin Certification of the 220 Record by the Clerk it shall be served on all parties as provided for in 154.8-3. The time for 221 filing and certifying the record may be extended for good cause by the Chief Judge of the Court 222 upon a written request from the Trial Court clerk or original hearing body. 223

- (a) The record of the case shall consist of all papers filed with the Trial Court or original 224 hearing body, exhibits, a transcript or audio recording of the proceedings, and the final 225 decision of the Trial Court or original hearing body. 226
- 227

207

228 154.9. Time Computation

154.9-1. Deadline Computation. Time lines are determined by designating the day after notice 229 230 is received as day one. Computation involving calendar days shall include intermediate Tribally

For OBC consideration (clean) 03/11/15

observed holidays and weekend days, provided that if the last day of the period falls on a
Saturday, Sunday or Tribally observed holiday, then the next business day shall be the due date.

233 Computation involving business days shall not include intermediate weekend days or Tribally

- observed holidays. All papers due to be filed with the Clerk are due prior to the close of business
- on the last day of the time period.
- (a) If notice is mailed, then three (3) days shall be added to the time line in order to determine the due date.
- 238 154.9-2. *Extension of Time*. For good cause, the Court may extend the time prescribed by these
 239 Rules or by its order to perform any act, or may permit an act to be done after that time expires.
 240 But the Court shall not extend the time to file:
- 241 (a) a notice of appeal; or
 - (b) a petition for permission to appeal.

154.9-3. *Time to Complete. of less time is extended by the Court with the knowledge of the parties, the time from the fring of the Notice of Appeal to the completion and entry of the final written accision shall not excerd an elandred and eighty (180) days.*

247 154.10 Motions

242

- 154.10-1. Application for Relief. An application for an order or other relief in a docketed case
 shall be made over no non-unless these K less prescribe nother form. A motion shall be in writing
 unless the Court permits or erwise. The maxing party shall file all motions with the Clerk and
 serve opposing parties as provided in 154.8.
- 252 154.10-2. *Contents of a Motion*. A motion shall state with particularity the grounds for the motion, the relief sought, for the egal argument necessary to support it.
- (a) Any affidavit or other paper necessary to support a motion shall be served and filed
 with the motion. An affidavit shall contain only factual internation, not legal argument.
 A motion seeking substantive reliaf shall include a cony of the Teial Court's or original
 hearing body's opinion as a securate exhibit.
- 154.10-3. *Response to a Motion*. Any party may be a response to a motion, in accordance with
 154.11-2. The response shall be filed within ter (10) days after service of the motion unless the
 Court shortens or extends the time.
- 261 154.10-4. *Motion for a Procedural Order*. The Court may act on a hor in for a procedural
 - order at any time without awaiting a response. A party adversely affected by the Court's action may file a motion to reconsider, vacate, or modify that action where five (5) days of receipt of notice of the decision.
 - 154.10-5. *Motion for Voluntary Dismissal*. An appellant may dispute an appeal by filing a
 motion to dismiss. If not yet docketed in the Court, then the motion shall be filed in the Trial
 Court or original hearing body. The dismissal of an appeal shall not affect the status of a crossappeal or the right of a respondent to file a cross appeal.
 - 154.10-6. *Form.* Motions shall be typed, legible and include the case caption. Every motionshall:
 - (a) Contain a caption heading, the name Judiciary- Court of Appeals, the title of the
 action, the docket number (if known) and a designation as to the purpose or type of
 motion.
 - (b) Contain the names of all parties to the action.
 - (c) Be organized in sections containing a clear designation, which shall include, but is not limited to:
 - 277 (1) The facts, events or occurrences which make a specific motion for relief278 necessary;

For OBC consideration (clean) 03/11/15

- 279 (2) The specific relief requested by the moving party: 280 (3) The applicable law or laws to the motion at hand, including citations; and (4) The legal reasons the relief should be granted. 281 (d) Be on $8\frac{1}{2}$ by 11 inch paper. The text shall be double-spaced, but quotations more 282 than two (2) lines may be indented and single-spaced. Headings and footings may be 283 single-spaced. Margins must be at least one (1) inch on all four (4) sides. Page numbers 284 may be placed in the margins, but no other text shall appear there. 285 (e) Be typed in a plain, roman style, although italics or boldface may be used for 286 emphasis. Case names shall be italicized or underlined. 287 (f) Not exceed twenty (20) pages, unless the Court permits or directs otherwise. 288 289 154.11. Briefs 290 Briefs Generally priefs shall be used by the Court to aid the Court in its 291 154.11-1. consideration of the issues resented. 292 The brief skall to line spaced, typed, 1 inch margins, and on 8.5 x 11 inch 293 s Fon 294 paper, and shall be signed by the arc, or the party's attorney or advocate, if represented. The front cover of a brief shall contain: 295 (1) ne number of the ase centered t the top; 296 (2) the name of the court 297 (3) the title the case; 298 (4) he dtu e of the proceeding (e.g., Appel Petition for Review) and the name 299 of the court bel w; 300 (5) the title *c* the men identifying the carty or paties for whom the brief is filed; 301 and 302 (6) the name, fice address and telephone number of the attorney or advocate 303 representing the pert for non the brief is file, if presented. 304 (b) Length. The brief shall be no pore man twenty (20) pige, one (1) sided, in length, 305 not including any addendum, appendice, attachments, the tables of contents and 306 authorities. 307 (c) Filing. When a party is represented by an attrney or advocate, ally the attorney or 308 advocate shall file briefs and pleading. The dividual shall not for his or her own 309 unless he or she is pro se. Three (3) copies a each brief shall be filed with the Clerk and 310 311 one (1) copy to all parties to the appeal. (d) *Time to Serve and File a Brief*. The Appellant may serve on the Respondent and file 312 with the Clerk a brief within twenty (20) days after acceptance the Certification of the 313 Record. The Respondent's brief shall be filed with the Clerk with twenty (20) days of 314 receipt of the Appellant's brief. A reply brief, if necessar, shall be filed within fourteen 315 (14) days of receipt of Respondent's brief. The Court may on s own, order different 316 317 time lines for any party's time to file a brief. (e) Consequence of Failure to File. If an Appellant fails to file a brief within the time 318 provided by this Rule, or within an extended time, a Respondent may move to dismiss the 319 appeal. A Respondent who fails to file a brief shall not be heard at oral argument unless 320 the Court grants permission. 321 154.11-2. Appellant's Brief. The Appellant's brief shall contain, under appropriate headings and 322 323 in the order indicated: 324 (a) Content:
- 325

Page 57 of 340

For OBC consideration (clean) 03/11/15

326	(2) a table of authorities-cases (alphabetically arranged), statutes, and other
327	authorities-with references to the pages of the brief where they are cited;
328	(3) a jurisdictional statement, including:
329	(A) the basis for the Trial Court's or original hearing body's subject-matter
330	jurisdiction;
331	(B) the basis for the Court of Appeals' jurisdiction;
332	(C) the filing dates establishing the timeliness of the appeal; and
333	(D) an assertion that the appeal is from a final order or judgment that
334	disposes of all parties' claims, or information establishing the Court of
335	Appeals' jurisdiction on some other basis.
336	(4) a statement of the issues presented for review;
337	(5) a statement of the case briefly indicating the nature of the case, the course of
338	proceedings, and the disposition below;
339	(6) a statement of facts relevant to the issues submitted for review with
340	appropriate references the record;
341	() a summary of the argument which shall contain a succinct, clear, and accurate
342	statement of the arguments made in the body of the brief, and which shall not
343	mental epeat the argument heading the
344	(8) the argument, which hall comain:
345	(A) opellant's contendors and the reasons for them, with citations to the
346	atherities and parts of the record on which the Appellant relies; and
347	(B) or each issue, a concre statement of the applicable standard of review
348	(where my appear in the discussion of the issue or under a separate
349	heading slaced be ore the discussion of the is ties);
350	(9) a short concusion stating the precise relies sound
351	(10) a short appendint to include
352	(A) relevant dogreet entries in the Trial Court or original hearing body;
353	(B) limited portions or the record essentianto or understanding of the
354	issues raised;
355	(C) the judgment, order, or decision in question; and
356	(D) other parts of the record to price the parties where direct the Court's
357	attention; and
358	(11) where the record is required by law to be confidential, reference to
359	individuals shall be by initials rather than by names.
360	154.11-3. <i>Respondent's Brief.</i> The Respondent's brief shall conformer, the same requirements
361	as 154.11-2 (Appellant's Brief).
362	(a) The Respondent's brief shall address each issue and argument presented by the
363	Appellant's brief.
364	(b) The Respondent's brief may present additional issues, with the Respondent's
365	positions and arguments on such issues.
366	154.11-4. <i>Reply Brief</i> . The Appellant may file a brief in reply to the Respondent's brief. Unless
367	the Court permits, no further briefs may be filed. A reply brief shall conform to the requirements
368	of 154.11-3 (Respondent's Brief), except that a reply brief shall be no more than fifteen (15)
369	pages, one (1) sided, in length.
370	154.11-5. <i>Amicus Curiae Brief.</i> A person who is not a party to a case but has some interest in
371	the outcome of the case may, upon timely motion and with permission of the Court, submit an
372	amicus curiae brief in support of a party to the action. The Court may, on its own motion,
373	request amicus participation from appropriate individuals or organizations.
575	request anneas participation nom appropriate merviduals of organizations.

For OBC consideration (clean) 03/11/15

either party shall file his or her brief no later than seven (7) days after the 378 Appellant's or Respondent's brief is filed. The Court may grant leave for later 379 filing, specifying the time within which an opposing party shall answer. 380 154.11-6. Briefs in a Case Involving Multiple Appellants or Respondent. In a case involving 381 more than one (1) Appellant or Respondent, including consolidated cases, any number of 382 Appellants or Respondents may join in a brief, and any party may adopt, by reference, a part of 383 another's brief. Parties may also join in reply briefs. 384 385 154.12. Oral Argument 386 154.12-1. *Oral Argument* The Court may order oral argument when issues of fact or law remain inclear and/or the positions of the parties on an issue are unclear or otherwise not fully 387 388 developed. The Court shall direct that ar appeal be submitted on briefs only, if: 389 (a) The appeal is frivolous; 390 (b) the dipositive issue or issues have be authoritatively decided; or 391 (c) The firsts and legal arguments are a equally presented in the briefs and record, and 392 the decision a process would not be egnineantly aided by oral argument. 154.12-2. *Notice*. The clerk shall provide notice, of at least ten (10) business days, to all parties 393 394 when oral arguments are the aled. The notice shall list the location of the oral argument and 395 the time allowed for each i.e. The ourt shall determine the amount of time for oral arguments. 396 A motion to postpone the argument or po extend the argument time frame shall be filed at least 397 five (5) business days before the hearing date 398 154.12-3. Citation of Authoritic an Oral Argument. Unless primited by the Court, parties may 399 not cite or discuss a case at an oral gue ent unless the case has been cited in one (1) of the 400 briefs. 401 402 **154.13.** Entry and Form of Judgment 403 154.13-1. Entry. A judgment is entered when it is not d on the docket. Levelerk shall prepare, 404 sign, and enter the judgment after receiving the Cover s opinion 405 (a) The decision and opinion of the Court shall be by majority vote. 406 (b) The Court may: 407 (1) Reverse, affirm, or modify the judgmen or order as any or all parties; 408 (2) Remand the matter to the Trial Court or orginal caring ody and order a new 409 trial on any or all issues presented; the order manding a case shall contain 410 specific instructions for the Trial Court or original haring ody; 411 (3) If the appeal is from a part of a judgment or order, the Court may reverse, 412 affirm or modify as to the part which is appealed; 413 (4) Direct the entry of an appropriate judgment or order; or 414 (5) Require such other action or further proceeding as may be appropriate to each 415 individual action. 416 (c) On the date when judgment is entered, the Clerk shall serve all parties with a copy of 417 the decision and opinion as entered. 418 154.13-2. Form. All decisions of the Court shall be in writing and accompanied by an opinion 419 stating the legal issues and the basis for the decision. Decisions of the Court shall be issued no 420 154-9

(a) Amicus curiae briefs shall conform to the requirements of 154.11-2 (Appellant's

(1) Amicus curiae shall file his or her brief no later than seven (7) days after the

brief of the party being supported is filed. Amicus curiae that do not support

Brief), except as provided in the following:

374

375

376 377

For OBC consideration (clean) 03/11/15

- 421 later than sixty (60) days after the conclusion of oral argument or after the expiration of time to
- 422 file a *Reply Brief* or *Response Brief* if no oral argument is held.
- (a) The time for issuing a decision and opinion may be extended provided all parties are 423 notified of the extension in writing. The notice of extension shall include the cause for 424 and length of such extension. 425

154.14. Interest of Judgments 427

154.14-1. Unless the law provides otherwise, if a money judgment in a civil case is affirmed, 428 whatever interest is allowed by law is payable from the date when the Trial Court's or original 429 hearing body's judgment was entered. If the Court modifies or reverses a judgment with a 430 direction that a money judgment be entered in the Trial Court or by the original hearing body, the 431 mandate shall contain instruction bout the allowance of interest. 432

433 154.15. Penalties 434

426

455

464

- 154.15 Friv lous Appeals. If a peabor cross-appeal is found by the Court to be frivolous, 435 the Court may avard to the successful party costs and attorney's or advocate's fees. (a) Costs may be assessed against the Appellant or cross-Appellant, the (cross)-436
- 437 Appellant' at prney or advocate, o both he (cross)-Appellant and his/her attorney or 438 advo até vir ly. 439
- 440
- (1) Court courts shall be based on a tual cost or defined by the Court.(b) A finding court rolors appeal or cross-appeal shall be made if one (1) or more of the 441 following elements are bound by the Court: 442
- (1) The appendix appeal was file, used, continued in bad faith, solely 443 for purposes of relay, hardssment or injuring the opposing party; or 444
- (2) The party or party's atterney or advocate known should have known, that 445 the appeal or cross espeal was without any regionable be is in law or equity and 446 could not be supported y a rood faith argument for an extension, modification or 447 reversal of existing law 448
- 154.15-2. Delay. If the Court finds that an appeal or cross-appeal way taken for the purpose of 449 delay, it may award one (1) or more of the ferrowing to the opposing party: 450
- (a) Double costs: 451
- (b) A penalty of additional interest not excerding ten percent (10%) on the award amount 452 453 affirmed;
- (c) Damages caused by the delay; and/or 454
 - (d) Attorney's or advocate's fees.
- 154.15-3. Non-Compliance with Rules. Failure of a party to complexity with requirement of these 456 Rules or an order of the Court, does not affect the jurisdiction the Court over the appeal but 457 may be grounds for one (1) or more of the following: 458
- (a) Dismissal of the appeal; 459
- (b) Summary reversal of the Trial Court or original hearing body; 460
- (c) Striking of a paper, document or memorandum submitted by a party; 461
- (d) Imposition of a penalty or costs on a party or party's attorney or advocate; and/or 462
- (e) Other action as the Court considers appropriate. 463

465 **154.16.** Substitution of Parties

154.16-1. Death of a Party. Death of a party does not automatically end a party's right to 466 467 appeal.

For OBC consideration (clean) 03/11/15

- (a) *After Notice of Appeal Is Filed.* If a party dies after a notice of appeal has been filed
 or while a proceeding is pending in the Court, the decedent's personal representative may
 be substituted as a party on motion filed with the Clerk by the representative or by any
 party. A party's motion shall be served on the representative. If the Decedent has no
 representative, any party may suggest the death on the record, and the Court may then
 direct appropriate proceedings.
- (b) *Before Notice of Appeal Is Filed-Potential Appellant*. If a party entitled to appeal dies before filing a notice of appeal, the decedent's personal representative, or if there is no personal representative, the decedent's attorney or advocate of record, may file a notice of appeal within the time prescribed by these Rules. After the notice of appeal is filed, substitution shall be in accordance with 154.16-1(a).
- (c) *Before Notice of Appel Is Filed-Potential Respondent.* If a party against whom an appeal may be taken diss after entry of a judgment or order in the Trial Court or original hearing body, but before a notice of appeal is filed, an Appellant may proceed as if the eath had not occurred. After the notice of appeal is filed, substitution shall be in accordance with 154.10-1(2).
- 484 154.162. Substitution for a Reason ConcerThan Death. If a party needs to be substituted for any
 485 reason other than it at , the procedure set in 154.101(a) applies.

486 487 **154.17. Costs**

- 488 154.17-1. *Costs*. Lost in an appeal shall be a follows unly s otherwise ordered by the Court:
- (a) Against the appellant when the appeal is dismissed or the judgment or order affirmed;
 (b) Against the respondent when the judgment or order is reversed.
- 491 154.17-2. Allowable Cosis. A towable posts shall include:
- 492 (a) Cost of printing and assembling the number of correl and briefs and appendices
 493 required by the Rules;
- (b) Fees charged by the Court adverters;
- 495 (c) Cost of the preparation of the transcript of testimony of the reliard on ppeal; and
- 496 (d) Other costs as ordered by the Court

497 154.17-3. *Recovery of Costs.* A party seeking to recover costs in the Court shall file a statement 498 of the costs within fourteen (14) days of the filing on the decision of the Court. An opposing 499 party may file, within eleven (11) days after service of the statement, a motion objecting to the 500 statement of costs.

- 501
- 502 *End*.
- 503 504 Adopted BC-04-25-14-B
- 505 Emergency Amendments BC-12-19-14-A

Oneida Business Committee Meeting Agenda Request Form Page 61 of 340

1. Meeting Date Requested: 03 / 11 / 15 2. Nature of request Session: 🛛 Open Executive - justification required. See instructions for the applicable laws that define what is considered "executive" information, then choose from the list: Agenda Header (choose one): Follow Up Agenda item title (see instructions): Accept Oneida Gaming Commission continuing resolution closeout report Action requested (choose one) Information only ⊠ Action please escribe: Accept Oneida Camil g Commission ontinuing Jution closeout report 3. Justification Why BC action is required instructions): sе Per BC directive. 4. Supporting Materials **Instructions** Memo of explanation with required information (see instructions) Contract (check he box below if streather required) ⊠ Report □ Resolution Other - please list (Note: multi-media presentation due to Trible Clerk 2 days prior to meeting) 1. 3 2. 4 Business Committee signature required 5. Submission Authorization Brandon Stevens, Council Member Authorized sponsor (choose one): Requestor (if different from above): Mark A. Powless, Sr., OGC Chairperson Name, Title / Dept. or Tribal Member Additional signature (as needed): Name, Title / Dept. Additional signature (as needed): Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Page 62 of 340



CLOSE OUT REPORT CONTINUING (FISCAL) RESOLUTION

To: Oneida Business Committee

From: Mark A. Powless, Sr., Oneida Gaming Commission Chairperson **Date:** 02/09/14

The OGC and staff worked diligenting to restrict spending during the period of continuing resolution for cost containment. Operational moacts were felt and a few large expenses could not be avoided.

Operation 1 Imp. cts

While restrictions on donation specifically effect the OGC's departments, the remaining aspects of the resolution and impact operations.

- **Hiring** The Surveillance department was not able to hire two vital positions: Surveillance Technician and Administrative Assistant.
- Wage Increases Restricting wages continues to affect employee morale and initiative.
- **Overtime** Overtime was lept to a minimum as much as possible. However, in order to provide necessary training or all Surveillance staff, or ertimely as incurred November.
- **Travel** Travel was halted, which as acted the OGC and its departments' FY15 operational goals of training and development. This indirectly inducted projects and created opportunity costs for maintaining status que instead of improving processes.
- **Contracting** A contract request for a new vendor had to be routed brough the OBC for approval, delaying the anticipated time line of implementation.

Exceptions

Exceptions to the adjusted budget (1/12 of 75% of the ellocation) occurred due to items not falling within a standard monthly payment schedule. Main exceptions to the adjusted allocations include: a Surveillance capex purchase from FY14 (\$31, 90) that was not received timely and had to be paid from FY15's supply line, an emergency equipment repair (\$3,600), Legal Services for travel to represent OGC unable to travel, and monthly rent expenses that could not be reduced.

Process Improvement Recommendations

Although difficult, across the board restrictions are not practical. Some departments already conduct annual assessments and budget based on minimal needs. To require additional reductions can detrimentally and unnecessarily impact operations. Furthermore, it is recommended that gaming operations be exempted from cost restriction expectations.

Oneida Business Committee Meeting Agenda Request Form Page 63 of 340

1. Meeting Date Requested: 03 / 11 / 15 2. Nature of request Session: 🛛 Open Executive - justification required. See instructions for the applicable laws that define what is considered "executive" information, then choose from the list: Agenda Header (choose one): Agenda item title (see instructions): Accept Oneida Gaming Division continuing resolution closeout report Action requested (choose one) Information only ☑ Action please escribe: Accept the close pat r port 3. Justification instructions): Why BC action is required зe BC follow-up directive 4. Supporting Materials **Instructions** ☐ Memo of explanation with required information (see instructions) Contract (check he box below if streather required) ⊠ Report □ Resolution Other - please list (Note: multi-media presentation due to Trible Clerk 2 days prior to meeting) 1. 3 2. 4 Business Committee signature required 5. Submission Authorization Authorized sponsor (choose one): Requestor (if different from above): Louise Cornelius, General Manager/Gaming Name, Title / Dept. or Tribal Member Additional signature (as needed): Name, Title / Dept. Additional signature (as needed): Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org



Memo

To: Business Committee Direct Reports, Oneida Boards, Committees and Commissions

From: Lisa Summers, Tribal Secretary

Date: February 3, 2015

Re: Close Out of Continuing Resolution Reporting

This correspondence is to inform you that the Oneida Business Committee has requested submission of final close out reports to include the operation impacts of the continuing resolution, any information pertaining to the exceptions that were made and any recommendations moving for the don how to improve the continuing resolution process

Background

Resolution C-09-24-14-Accequire a exceptions to be set forth in a report which specifically identified the exception, the next for the exception, cost for non-compliance and compliance, and alternatives the were considered but spected and the reasons for rejection.

The following violing was made as the January 22 2015, Oneida Business Committee meeting: Motion by A elinda J. Danfrach to direct the Tribal Secretary to issue a memorandum b car Direct Poports, Non-Divisional area sincluding the Judiciary, and all Boards, Committees as a Commissions to submit final close out reports to the February 11, 2015, regular Busines committee meeting agenda to include the operation impacts of the continuing resolution, any a formation pertaining to the exceptions that were made and any recommendations moving forward on here to improve the court ing resolution process, seconded by Lisa Summers. Motion can educated upminously

Next Steps

The motion indicates that your close one report is to be submitted to the February 11, 2015 regular Business Committee meeting. The submission acadline for the meeting is today at 4:30 p.m., but we are extending this deadline to 10 and, on Friday, February 6, 2015.

Thank you for your attention. Questions can be directed to the arext. 4434 or via email at <u>lsummer2@oneidanation.org</u>, or to Lisa Ligging Executive Assistant at <u>lliggins@oneidanation.org</u>.





Gaming Division Overtime Report OCTOBER, NOVEMBER & DECEMBER 2014 AND JANUARY 2015

October 2014 – January 2015 OVERTIME COMPARISON FY10 vs. FY15



Gaming Division Overtime Exceptions:

- New Instal ation of Bingo System (Cot/Nov)
- Training of New Bingo System (act/New)
- Minimum Starfing Levels Short Stared
- o Call-ins
- Medical Leave Of Assence(s)
- Approved Vacation/Personal Time Requests
- Regulatory Impressed count bank
- Regulatory Over/Short: SOP Employee required to stay until issue resolved

IN

A

- Coverage for Funeral Time
- Regulatory Paper Room coverage
- Regulatory testing
 - Holiday Coverage

Overtime Summary of Alternatives:

- Revised Departmental Shifts to Accommodate Customers
 - 8 hour shifts changed to 10 hour shifts
 - Reduced coverage on floor
- Employees assumed additional duties and responsibilities
- Managers & Supervisors assumed frontline duties and responsibilities



Gaming Division Staffing Activity 10/01/2014 to 01/31/2015

October 2014	November 2014				
4 Full-Time Regular	2 Full-Time Regular				
9 Full-Time Emergency Temp	30 Full-Time Emergency Temp				
4 Part-Time Emergency Temp	3 Part-Time Emergency Temp				
4 Full-Time Reassignments	1 Half-Time Emergency Temp				
	2 Full-Time Reassignments				
December 2014	January 2015				
18 Full-Time Emergency Temp	7 Full-Time Emergency Temp				
2 Part-Time Emergency Temp	1 Part-time Regular				
6 Full-Time Emergency Temp	2 Part-Time Emergency Temp				
	1 Half-Time Emergency Temp				

Gaming Division Accumulative Staffing Activity Totals for 10.01/2014 to 01/31/2015

	FT REG	TET	1050	рт нт		FT Inter REASSIGNN		Staffing Activity
ост	4	9	MLG	4		4	ALINIS	21
NOV	2			3	1	2		38
DEC	0	18		2		6		26
	0	7			1			11
Total	6	64			2	.2		96
	g Levels 2014	2015	YTD Variance	Ż		ng Lever 1° 2014	2015	1 ST QTR Variance
Oct	1347	1321	-26		ct	1321		
Nov	1342	1321	-21	-	Nov	1.71		
Dec	1340	1318	-22	-	Dec	1318		-3
lan	1338	1303	-35		Jan		1303	-15
lan *The Ga	1338 ming D	1303 ivision s	-35		Jan	ecliner in 1		te

Oneida Tribe of Indians of Wisconsin | Staffing Activity – Cost Containment Resolution Close Out Summary



Gaming Division Staffing Activity October 2009 – January 2015

								Accounting	273
IONTH	2009	2010	2011	2012		2014	2 15	Administration	18
								Pingo/OTB	44
ост	1,478	1,439	1,439	1,402	1,382	1,347	1,32	Compliance	5
							1.001	Sust Vial	118
NOV	1,466	1,441	1,441	1,401	1,39	1 42	1,321	Crut. Se. vice	130
					1.000	77	1.010	∠SD	28
DEC	1,456	1,441	1,440	1,385	1,398	1,? .0	1,318	S curity	179
							T	MIS	15
JAN	1,556	1,447	1,397	1,374	1,397	1,338	1,7 .5	Maintennce	32
								Marketing	51
FEB	1,545	1,445	1,401	1,379	1,392	1,344		Sints	173
								Surveillance	37
MAR	1,545	1,447	1,425	1,379	1,395	1,340		Vable Games	200
							100000		
APR	1,528	1,450	1,451	1,378	1,395	1,346		ótal	1,. 93
MAY	1,525	1,438	1,445	1,379	1,390	1,334		J. NUARY AN	PLOYEE COUNT
JUN	1,510	1,441	1,391	1,391	1,386	1,337	122	1,556	
								1,447	
JUL	1,522	1,427	1,392	1,382	1,375	1,347		1,30	1,374 1,397
				Constant.					1,338
AUG	1,512	1,424	1,388	1,383	1,371	1,334			
SEP	1,495	1,424	1,402	1,386	1,368	1,329	12000		

Gaming Division Staffing Transactions 10/1/14 to 1/31/15

Position Requested for Hiring	Job Code	Grade	Salary	Duties	Positions Filled				Difference
					Oct-14	Nov-14	Dec-14	Jan-15	
Administration	and the second s			6 FTE's Minimum staffing					
Admin Asst I – Reception – POOL POSTED	1829	NE06	\$10.02 to \$14.03	Perform and/or oversee a variety of associated administrative, fiscal, personnel support, and blanning activities. Coordinate and facilitate meetings, program functions, and/or special events, as appropriate.			1 FT ET		0
Drop Count				45.5 FTE's Minimum Staffing					
Drop Count - POOL POSTED	249	NE05	\$9.17 to \$12.76	erform collect on/transporting cash boxes, emptoing orch boxes, counting entire contents, and intering oformation in database using a personal computer. In the Count Area, will count, when, yearly and prepare all monies to be transferred to the value rea.		1 FT REG 4 HT ET		3 HT ET	-0.88
Cage Vault			V	Cage Cashier 100 Fr. Min Staff/ ault 5, 2c 41 TE N. Staff/ Vault Spec II.2 FTE Min Str / Cage/ ault Supervisor 2.7 FTE's Budm.ed					
Cage Cashier - POOL	81	NE06	\$10.02 to \$14.03	Provide even exchanges for currency, con- inity TITC coup and check cashing for oneide Catho internal and external customers. Process the e-collments for Check Service. Welcome III guests and assist in explanation of benefits for monoership programs. Obtain proper identification from eastomers for required coorting surposes.	1 FT F FT REG 2 PT ET	1 FT REG	3 FT ET	1 FT ET 1 PT ET	-1.42
Specialist - Vault - POOL	482	NE08	\$12.13 to \$16.98	Maintain accurate records of all transmoor and expedite the transmoof the concercy received from Gaming to the consignent depository banking institution. Maintan confidentiality of all privileged internation.	3 FT ET	1 FT ET	2 FT ET	2 FT ET	-4.92
Specialist - Vault II	1594	NE08	\$12.13 to \$16.98	Maintain accurate records of all transaction and expedite the transfer of the currency received from Gaming to the designated depository banking institutions. A valid driver's license, reliable transportation and insurance are required.			1 FT ET		-1.19
Position Requested for Hiring	Job Code	Grade	Salary	Duties		Position	s Filled		Differen
---	--------------------	----------	------------------------	--	----------	--------------------	--------------------------	---------------------	-----------
					Oct-14	Nov-14	Dec-14	Jan-15	
Super - Cage Vault	06006/00053	NE10	\$14.68 to \$20.55	Supervise Cage/Vault activities of the department by performing, training, supervising the Cashiers.			4 FT INTERIM EXTENDED		-0.75
Finance				Acctg Asst 14 FTE Min Staff/ Admin Asst III 3 FTE Budgeted					
Assistant - Accounting	4	NEOZ	\$11.03 to \$55.44	Review and verify all receipts in the Gaming Division against daily computerized printouts of Gamua activity to ensure proper recording of sales, prize payout, and cash our rages/shortages.	1 FT REG	1 FT ET			0
Admin Asst III - Accounting Operations Director / Drop Count	1991A	NE08	⊂ 113 to \$2098	Peinem and/o overseco variety of associated administrative, fiscoly performed support, and planning activities Coordinate and facilitate meetings, provide the ctions, and/or special events, as appropriate	1 FT REG				-0.75
Valet	Contraction of the	a factor	1200 M	10 FTE's Min S if					
Valet - POOL	479	NE06	\$10.05 to \$14.03	Park and retrieve customer's cars in usafe and timely reanner for the Oneider sasing Ensure the services are provide, effectively and efficiently to much the needs of the Oneid Casino Customer.	1. PT ET	1 PT ET	1 PT ET		0
Food & Beverage				7.5 FTE'r nin Staff					
Server - POOL POSTED	1830/6034	NE03	\$7.53 to \$10.54	Greet cus emers, answer customer questions, provide directions, obtain change for customers at intrchines, and transfer lost and found items whe approvriate areas.	2 FT 1	4 FT ET 2 PT ET		2 FT ET 1 PT REG	0
Employee Services				T&D Spec 3 FTE Budget / Comm Spec 1 FTE Budge d					1 Charles
Specialist - Training & Development	918	E03	\$15.91 to \$25.46	Develop and implement specified ain that and development programs and initiatives to have assessed needs for the Gaming Division.		1 FT ET			-7.94

Position Requested for Hiring	Job Code	Grade	Salary	Duties	and the second second	Positions	Filled		Difference
					Oct-14	Nov-14	Dec-14	Jan-15	
Specialist - Communications	2161	E03	\$15.91 to \$25.46	Develop an employee communication strategy and implement methods and processes to effectively distribute and communicate information to all levels of Gaming employees. Oversee various Gaming communication itiatives and continuous improvement efforts that align with Gaming's Vision, Mission, and Values This position is responsible for the continuation and facilitation of employee event within the Gaming Division.	1 FT REASSIGNMENT				1.68
Maintenance			1	Carper C. 2 FTE B. Ngeted / Groundskeeper I 6 C. E's Min. taff / Groundskeeper II 7 FTE's M. S. F					
Carpenter	163	NE11	\$16.14 to 227.0	Responsible on the overall carpentry and maintenance wall Gaming buildings. Perform skilled tasks and complete work where in accordance with bandard practices with induction and carpentry tracks and codes. Complete inspections of phous areas and perform reventive maintenance as preded Ensure wat services are provided mective and efficiency to the Oneida pasing		1 FT ET			4.30
Groundskeeper I POOL	629	NE05	\$9.11 to \$12.76	Proofms younds keeping and recycling duties for the exkeep on the Gaming Division as well as assist which building repairs, pick- up/delivery of equipment/supplies, and moving office furnities. Ensure that provides are provided effectively and efficiently for the Gaming Division.	0	1 FT ET			-0.44
Groundskeeper II POOL	119	NE06	\$10.02 to \$14.03	Performs grounds ke ming and recycled duties for the upkeep of the Gaming advision as well as assist with building recurs, tick up/delivery of equipment/supplies and motiog office furniture. Ensure that services are provided effectively and efficiently is the Gaming Division. *Tribal Vehicle Clearant required*			1 FT ET		-0.44

Position Requested for Hiring	Job Code	Grade	Salary	Duties	DA SHOW ST	Positions	Filled		Difference
					Oct-14	Nov-14	Dec-14	Jan-15	
Custodian - POOL POSTED	156	NE04	\$8.28 to \$11.60	Maintain all Gaming Division facilities in an exceptional manner. Perform basic and routine custodial duties for the upkeep of the Gaming Division.			1 FT ET		0
Marketing				1 FTE Budgeted					
Coordinator - Direct Marketing	2063	E03	\$15.91 to \$25.46	A complish the Casino Promotions act atment objectives by coordinating, tracking ar maintaining all functions related to Direct many ting/Casino Management Structure database operations including act sting increating direct marketing efforts, innovementing and that ging projects and controlling very ors. Players Club, P. J. 41.5. TE's Min Staff / Players	1 FT REASSIGNMENT				1.17
Casino Promotions				Club a nervisor 5 Budget.					
Rep - Players Club POOL	2387	NE07	\$11.03 to \$15.4	Provide guest services for Oranda Caono custo eers to include: Player Club Membro, Tour but special event ticlet, gift certificates sales, all Oneida Casino guest. Provide membership ervices including a swering customer crestices and explaining benuits o Naver Creatmembership.	1 FT REASSIGNMENT				0
Players Club Supervisor	371	NE09	\$13.34 to \$18.68	Ac emplish the Marketing Department objectives by a custing in the planning, organizing, promoting guest satisfaction and directing of all unctions required to operate and maintain or partment activities and services. Ensure all services are provided effectively and efficiently to meet the necessory the Oneida Casino custor ars in activities with federal standards	1 FT REASSIIGNMENT				0
Bingo				Bingo Caller 5 FTE's Min Staff / Bingo Cashler 16 FTE's Min Staff / Bingo Worker 10 n E's Min Staff / Paper Room Clerk 5 FTE's Min Staf OTB Supervisor 2 FTE's Budgeted	0				

Position Requested for Hiring	Job Code	Grade	Salary	Duties		Positio	ns Filled	il in the second	Difference
					Oct-14	Nov-14	Dec-14	Jan-15	
Caller - Bingo - POOL POSTED	70	NE04	\$8.28 to \$11.60	Provide excellent guest services and will work as part of a team to ensure that the guests have a pleasant experience; calls the bingo game, records numbers and verifies the winning cards.		3 FT ET			0
Cashier - Bingo – POOL	71	NFC	\$8.28 to \$14.60	Operate and perform cash transactions and ensure accurate sales to customers. Sell parts and electronics. Issue bingo cards to astroners and maintains a log of bingo sales. Issue produts for winning cards.	1 PT ET 1 FT ET	2 FT ET			0
Worker - Bingo POOL POSTED	72	'EO3	\$7.52 to \$10.54	Sell'an o game on the floor and ensure chanliness of Bingo area.	1 FT ET	2 FT ET	1 PT ET 1 FT ET	1 PT ET	
Clerk - Paper Room - POOL	1836	NE04	\$8.8 to \$1060	Assumin the control of a hand inventory in the Bings Paper from and cout of inventory into the bric System	1 FT ET	1 FT ET	2 FT ET		-0.88
Off Track Betting Supervisor	1887	NE10	\$14.68 to 2018	Oversee the operational and to opical maintenance of the Off Track Pround operation of the Off Track Pround operation of Track Betting equipment.			1 FT INTERIM REASSIGNMENT		0
Table Games				1' 5 FTE's Min Staffing					
Dealer- POOL POSTED	4130	NE01	\$6.07 to \$8.50	Conclust as sened game according to Oneid proc. tures. Exchange tash for gaming chip deal multiple card groues, and provide qualit, custorier senice within the policies, procedures and expectations of the Table Games Department.	6	7 FT ET	5 FT ET		0
Slots				Slot Rep 97 FTE's Min St. 'f / Slot Interse, St. y Spec 3 FTE's Budgeted / St. PM Co., 6 FTE's Budgeted					
Representative - Slot POOL POSTED	129	NE06	\$10.02 to \$14.03	Serve the patrons of the Oneida Cosino by resolving guests' problems such as tilu, tick jams, jackpots, and fills in a friendly, professional, and courteous manner; addres questions or concerns about how slot machines are played and the types of pay outs; provide accurate change as requested.		6 FT ET	1 FT ET	2 FT ET	- 2.66

Position Requested for Hiring	Job Code	Grade	Salary	Duties		Position	s Filled		Difference
					Oct-14	Nov-14	Dec-14	Jan-15	
Slot Internal Services Specialist	3420	NE11	\$16.14 to \$22.60	Ensure that the Slot Department meets regulatory compliance with NIGC MICS, OGMICR, and the State Compact. Position will work in cooperation with Gaming Management, Gaming Compliance, and Internal Audit to reach a goal of 100% compliance with all regulations mentioned above. Regulate the Slot Department by encloing and implementing regulatory cophuls option on-going basis and develop Statulate Operating Procedures.		1 FT INTERIM REASSIGNMENT			-4.39
PM Crew - Slots	2169	NE07	•\$11.58 to \$15.44	Insport, it bintain, and record data on all slot not fines it ensure regulatory compliance. Perfore preventative nanteenance checks as well as essist in rechine to ging verification process.		1 FT INTERIM REASSIGNMENT EXTENSION	1 FT INTERIM REASSIGNMENT EXTENSION		0
					G				
					$\mathbf{)}$				



То:	Oneida Business Committee
From:	Chad Fuss, Assistant Gaming General Manager- Finance
CC:	Louise Cornelius- Gaming General Manager
Date:	February 10, 2015
Re:	BC Resolution 09-24-14-A Continuing Resolution for Fiscal Year 2015

BC Resolution 09-24-14-A Continuin R solution for Fiscal Year 2015 stated that:

"Operational Expenses. Operational expenses are not to exceed one-twelfth (1/12) of 75% of the allocation in the CY 2014 budget for October through January. Provided that, this restriction shall exclude personnel and personnel related lines, which shall continue at 100% funding and be subject to the further limited and personnel in this resolution"

The Gaming Division created a set dsheet for all 3 plus busines. Units to utilize in order to by compliant with the resolution. The spreadsheet condisted of the hapported Fisca Year 2014 budget for each Business Unit along with the allocated amount for each month, the actual for each amount and a variance for each month. Removed from the spreadsheet were allow venue lines, cost of stres lines personnel lines, depreciation/amortization lines, compart fee associated lines and indirect condities.

The remaining lines exceeded one-twelfth (2/12) of the 35% allocation on O tober by \$188,845, November by \$247,537 and December by \$264,019. Most of variances have to do with expresses related directly to revenues, expenses related to revenue generation, expenses related to safety, security clear mess and **contractual obligations**.

Even though the Gaming Division did exceed the one-twelfth (712) of the 75% allocation outlined in BC Resolution 09-24-14-A Continuing Resolution for Fiscal Year 2015, revenues exceeded prior year revenues and Gaming Net Profit exceeded both Fiscal Year 2015 Budget and Fiscal Year 2014 Actual.

Page 79 of 340

		Oct-14	Oct-14	Oct-14		Nov-14	Nov-14	Nov-14		Dec-14	Dec-14	Dec-14	
		75%	Actual			75%	Actual			75%	Actual		and a state of the
DESCRIPTION	ACCOUNT NUMBER	BUDGET	Spend	Variance	EXPLANATION	BUDGET	Spend	Variance	EXPLANATION	BUDGET	Spend	Variance	EXPLANATION
													and the second

CUSTOMER SERVICE - AD													
SUPPLIES & MATERIALS	1205009-000-700001-000	31	0	31		31	50	-18	is required to have \$50	31	58	-27	to have \$50 or more of supplies in
NEWSP/SUBSCRIPTIONS	1205009-000-700070-000	13	31	-19	The memberships offer online :	a 13	31	-18	2 Employees have	13	31	-18	2 Employees have Annual SHRM
TRANSP. & PER DIEM	1205009-000-701000-000	178	0	178		178		178		178		178	
PROFESSIONAL FEES	1205009-000-702030-000	9,425	1,215	8,21		1	13,889	-4,464	Language Line Services payments are \$50.00	9,425	1,265	8,160	
TRAINING & EDUCATION	1205009-000-705010-000	20	27	-7	to intinue her education	20	62	-42	Director attends these	20	25	-5	attends these meetings to continue
PROMOTIONAL COUPONS	1205009-000-705010-000	0	27		to continue of reducation	0	62	-62	Director attends these	0	475	-475	attends these meetings to continue
CUSTOMER INCENTIVE	1205009-000-705111-000	250	0	250		2.		50		250		250	
I/T - PRINTING	1205009-000-750009-000	750	0	750		750		750		750		750	
TOTAL		10,667	1,301	9,366	0	10,667	14,0	-3,427		10,667	1,854	8,814	
CUSTOMER SERVICE - CA													
SUPPLIES & MATERIALS	1205009-100-700001-000	63	0	63		2		63		63		63	
TRANSP. & PER DIEM	1205009-100-701000-000	169	0	169		I.		169		- 19		169	
TRAINING & EDUCATION	1205009-100-705010-000	194	100	94		194	100	94		194	100	94	
CUSTOMER INC - RESTAUR	1205009-100-705111-400	0	0	0		0	4	-48	Comps for customer issu	0		0	
I/T - PRINTING	1205009-100-750009-000	75	0	75		75		75	\wedge	75		75	
I/T REPAIRS/MAINT - VEHIC	1205009-100-753100-300	63	0	63		63		63		63		63	
VEHICLE EXP - GAS & OIL - TOTAL	1205009-100-753101-000	125 688	42 142	83 546	0	125 688	45 192	80 495		125 688	35 134	91 554	
CUSTOMER SERVICE - VA													
SUPPLIES & MATERIALS	1205009-200-700001-000	125	124	1		125		125		125		125	
REPAIRS/MAINT - VEHICLE	1205009-200-703100-300	94	0	94		94		94		94		94	-
TIPS OVER/SHORT	1205009-200-709004-300	0	- 0	0		0		0		0		0	

Page 80 of 340 Dec-14 Dec-14 Oct-14 Oct-14 Oct-14 Nov-14 Nov-14 Nov-14 Dec-14 75% 75% 75% Actual Actual Actual DESCRIPTION **EXPLANATION** EXPLANATION BUDGET Variance **EXPLANATION** ACCOUNT NUMBER BUDGET Spend Variance BUDGET Spend Variance Spend I/T - PRINTING 205009-200-750009-000 281 0 281 281 281 281 281 TOTAL 500 124 376 0 500 0 500 500 0 500 **CUSTOMER SERVICE - FO** 94 94 94 SUPPLIES & MATERIALS 1205009-300-700001-000 94 89 5 94 1205009-300-701000-100 0 0 0 0 **TRAVEL - AIRFARE** 0 0 0 0 1205009-300-701000-200 0 0 TRAVEL - LODGING 0 **TRAVEL - MEALS** 1205009-300-701000-300 0 0 0 0 0 0 0 TRAVEL - TAXIS/SHUTTLES 1205009-300-701000-500 0 0 0 0 0 0 97 60 3 63 63 63 63 to make weekly checks, MILEAGE EXPENSE 1205009-300-701001-000 REPAIRS/MAINT - EQUIPME 1205009-300-703100-200 750 244 506 750 426 750 94 656 1205009-300-705001-000 UNIFORMS 6 6 6 6 Continually new employ 43 75 43 43 43 43 1205009-300-705010-000 Alcohol license TRAINING & EDUCATION Product is on demand, working Order all beverage on floor, 52 6,383 53,914 78,677 74,850 47,53 53,914 14 October had 5 weeks of 1205009-300-705100-000 with vendor to adjust par levels PROMOTION/ITEMS TIPS OVER/SHORT 1205009-300-709004-300 0 0 0 0 1205009-300-750009-000 31 31 0 31 31 31 31 I/T - PRINTING 144 144 144 SUPPLIES & MATERIALS 1205009-303-700001-000 144 0 144 144 uip. nt Repair REPAIRS/MAINT - EQUIPME 1205009-303-703100-200 609 188 188 0 188 188 188 Prod 1205009-303-705100-000 19,257 30,242 Order all beverage on floor, 19,257 21,474 t is on demand, 19,257 26,495 Product is on demand, working PROMOTION/ITEMS 1205009-304-105400-000 0 0 0 0 0 0 0 EQUIPMENT REPAIRS/MAINT - EQUIPME 1205009-304-703100-200 78 78 78 78 0 78 78 1205009-304-705100-000 986 1,067 Order all beverage on floor, 986 770 216 986 1,318 Product is on demand, working PROMOTION/ITEMS **PROMOTION/ITEMS** 1205009-305-705100-000 1,199 1,341 -142 Order all beverage on floor, 1,199 1,416 Product is on demand, 1,199 1,924 Product is on demand, working 4,533 75,840 108,414 -32,574 76,752 107,968 -31,216 76,752 72,219 TOTAL 0

DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
ADMINISTRATION													
SUPPLIES & MATERIALS	1205010-000-700001-000	3,938	6,072	-2,135	Copy Paper that is ordered for a	3,938	76	3,861		3,938		3,938	
DUES	1205010-000-700071-000	0		0		0	170	-170	This expense will be mov	0		0	
DISCOUNTS EARNED/LOS	Г 1205010-000-700999-000	0		0		- 0		0		0		0	
TRANSP. & PER DIEM	1205010-000-701000-000	1,563	0	1,563		1,563		1,563		1,563		1,563	
RAVEL - AIRFARE	1205010-000-701000-100	0	0	0				0		0		0	
RAVEL - LODGING	1205010-000-701000-200	0	50	-50	The is budgeted in TRANSP.	0		0		0		0	
RAVEL - MEALS	1205010-000-701000-300	0	0	0		0		0		0		0	
TRAVEL - RENTAL CAR	1205010-000-701000-400	0		. 0				0		0		0	
RAVEL - PERSONAL CAR	1205010-000-701000-410	0	_	0		0	\wedge	0		0		0	
RAVEL - TOLLS/PARKING	i 1205010-000-701000-430	0		0		0		0		0		0	
RAVEL - TAXIS/SHUTTLE	s 1205010-000-701000-500	0	0	0						0		0	
RAVEL - REGISTRATION	1205010-000-701000-600	0		0		0	700	-700	T s experience was from	0		0	
MILEAGE EXPENSE	1205010-000-701001-000	36		36		36		36		36	52	-16	Reimbursed mileage per policy
													Do Not have access to verify
DUTSIDE SERVICES	1205010-000-702010-000	45,813	34,923	10,889		813	55,151	-9,338	Do Not hacces to ve	15,813	128,737	-82,924	charges
OTHER CONTRACTED SEF	1205010-000-702010-000	0	0	0		0		0				0	
PROFESSIONAL FEES	1205010-000-702030-000	625		625		625		0		625		625	
REPAIRS/MAINT - EQUIPM	F 1205010-000-703100-200	0	254	-254	This should be expensed to Out	0	179	-179		0	757	-757	
MAINT-AGREEMENTS	1205010-000-703110-000	3,125		3,125		3,125		3,12		3,125	13,411	-10,286	Quarterly payment, do not have access to this account currently
EMPLOYEE INCENTIVE	1205010-000-705000-000	0		0		0		0		0		0	
EMPLOYEE INCENTIVE-PI		0		0		0		0		0		0	
JNIFORMS	1205010-000-705001-000	31	32	0		31		31		31		31	
RAINING & EDUCATION	1205010-000-705010-000	0		0		0		0		0		0	
EMPLOYEE DISCOUNTS	1205010-000-705020-000	15,000	14,894	106		15,000	19,190		Budget if based on	15,000	20,002		Budget if based on historical

Page 82 of 340

DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
PROMOTION/ITEMS	1205010-000-705100-000	1,950	185	1,765		1,950		1,950		1,950	4,264	-2,314	Expense for adjustement in inventory for Community Relations Items
PROMOTIONAL COUPONS	1205010-000-705101-100	0	0	0		0		0		0		0	
SPECIAL EVENTS	1205010-000-705105-000	1,750		1,750		1,750		1,750		1,750		1,750	
SPONSORSHIPS	1205010-000-705110-100	313		313		313		313		313		313	
ADVERTISING/BC REQUEST	1205010-000-705110-200	0		0				0		0		0	
CUSTOMER INCENTIVE	1205010-000-705111-000	0	635	-635	Ad inistration portion of Radi	0		0		0		0	
CUSTOMER INCENTIVE-RA	1205010-000-705111-100	1,250		250		,250	234	1,016		1,250	786	464	
CUSTOMER INC - RESTAUR	1205010-000-705111-400	0		0				0		0		0	
RENT EXPENSE	1205010-000-705201-000	29,723	21,943	7,780		29,723	<u> 33,615</u>	-53,89	Two months of Rent at	29,723	52,779	-23,056	Mainly Rent at Radisson. Oneida
RENTAL USAGE	1205010-000-705202-000	6,589	4,829	1,760	1/	6,589		6,589		6,589	8,836	-2,248	Do not have access to this accoun currently
HEAT & LIGHTS	1205010-000-705211-000	87,500	89,527	-2,027	Based on Usage	87,5 0	97,207	2,707	Base on the facility beir	87,500	26,769	60,731	
WATER & SEWER	1205010-000-705212-000	8,125	-9,799	17,924		8,125	1,398	6,727		8,125	26,825	-18,700	Typically a quarterly charge
TELEPHONE	1205010-000-705213-000	8,125	-4,417	12,542		810	7,8.	267		8,125	5,008	3,117	
BUSINESS EXPENSE	1205010-000-705300-000	250	325	-75	Send Flowers for Deaths of em	2.		250		10	348	-98	Flowers for deaths of employee and family members
CREDIT CARD CHARGES	1205010-000-705402-000	3,438	5,569	-2,131	Based on the ability of custome	3,438	4,184	-73	Based on custom, usage	3,438	4,920	-1,482	Based on customer usage
NSURANCE	1205010-000-705500-000	21,548	19,700	1,849		21,548	19,7.0	1,849		21,548	19,700	1,849	
MISCELLANEOUS	1205010-000-705900-000	0		0		0		0	\wedge	0		0	
CASH/OVER SHORT	1205010-000-709004-000	0		0		0	665	-665	This is the Over/Under	0	-327	327	
CLAIM SETTLEMENT	1205010-000-709800-000	1,563	649	914		1,563		1,563		1,563	1,984	-422	Backpay
/T SUPPLIES & MATERIALS	1205010-000-750001-000	0		0		0		0		0		0	
/T - PRINTING	1205010-000-750009-000	250		250		250		250		250		250	
/T REPAIRS/MAINT - VEHIC	1205010-000-753100-300	125	14	111		125		125		125		125	
VEHICLE EXP - GAS & OIL -	1205010-000-753101-000	81	102	-21	Mail has been making more de	81	62	19		81	49	32	
I/T SPECIAL EVENTS	1205010-000-755105-000	0		0		0		0		0		0	

	÷.												Page 83
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
I/T RENT EXPENSE	1205010-000-755201-000	7,031	9,375	-2,344	We are bound by contract for r	7,031	9,375	-2,344	Airview, IMAC, Main C	7,031	9,375	-2,344	Airview, IMAC, Main Casino lease contracts with Land Management
I/T RENT EXPENSE	1205010-303-755201-000	7,206		7,206		7,206	9,608	-2,402	West Mason lease contra	7,206	9,608	-2,402	West Mason lease contracts with Land Management
TOTAL		256,946	194,861	62,085	0	256,946	309,372	-52,426		16,256	21,016	-4,760	
ACCOUNTING FINANCE			-	6		1		1					Needed to purchase 1099 Pro in
SUPPLIES & MATERIALS	1205011-000-700001-000	438	240	198				438		438	2,883	-2,445	order to submit tax information
TRANSP. & PER DIEM	1205011-000-701000-000	184	i.	184				184		184		184	
MILEAGE EXPENSE	1205011-000-701001-000	147	130	17		147	76	71		147	209	-62	Reimbursed mileage per policy
PROFESSIONAL FEES	1205011-000-702030-000	313		313		313		313		313		313	
UNIFORMS	1205011-000-705001-000	0		0						0		0	
CONTRA - MIS	1205011-000-709509-000	145,191	210,774	-65,583	This is made up of prim y sal	a 145,191	176,219	-31,028	a his is stand in the rep	145,191	185,881	-40,690	This is detailed in the report that MIS submits
I/T SUPPLIES & MATERIALS	1205011-000-750001-000	0		0				0		0		-0	
I/T - PRINTING	1205011-000-750009-000	188		188		18		188		88	1,042	-855	Yearly ordering of forms.
TOTAL		146,459	211,144	-64,685	0	146,459	1 ,294	-29,835		14,,459	190,014	-43,555	
ACCOUNTING - HARD/SO		1,188	382	806		1,188		1,18		1,188	1,299	-104	Labels for drop boxes to ensure the are attributed to the correct
SUPPLIES & MATERIALS	1205011-200-700001-000												machine when dropped.
MILEAGE EXPENSE	1205011-200-701001-000		30	126		156	. 11	145		156	55	101	
OUTSIDE SERVICES	1205011-200-702010-000		0	34		34		34		34		34	
REPAIRS/MAINT - EQUIPMI	1205011-200-703100-200	313	0	313		313		313		313		313	
MAINT-AGREEMENTS	1205011-200-703110-000		13,377	-2,795	25 mo. contract signed prior to		13,377	-2,795	25 mo. contract signed p	10,582	13,377	-2,795	25 mo. contract signed prior to Resolution 09-24-14-A
UNIFORMS	1205011-200-705001-000	63	25	38		63	75	-13	ET's must purchase	63		63	
TRAINING & EDUCATION	1205011-200-705010-000	38	0	38		38		38		38		38	

Page 84 of 340

						-				1			
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
I/T SUPPLIES & MATERIALS	1205011-200-750001-000	75	0	75		75		75		75		75	
I/T - PRINTING	1205011-200-750009-000	44	0	44		44	41	3		44		44	
TOTAL		12,491	13,814	-1,323	0	12,491	13,504	-1,013		12,491	14,731	-2,240	
ACCOUNTING - CAGE/VAI													
CASH/OVER SHORT	1205011-300-709004-000	0					41	-41	There are procedures in		-741	741	
CASH/OVER SHORT	1205011-301-709004-000	0		0				-41	There are procedures in		-5,181	5,181	
CASH/OVER SHORT	1205011-303-709004-000	0		0		0	41	-41	There are procedures in		668	-668	There are procedures in place as to
CASH/OVER SHORT	1205011-304-709004-000	0		0		0		<u> 182.</u> 785	There are procedures in		1	-1	There are procedures in place as to
CASH/OVER SHORT	1205011-305-709004-000	0		0		0	1		Shere are procedures in		149	-149	There are procedures in place as to
SUPPLIES & MATERIALS	1205011-310-700001-000	3,750	3,439	311		3,750	6,721	-2,97	\$2,4.9 due to invoice	3,750	2,453	1,297	
MILEAGE EXPENSE	1205011-310-701001-000	188	184	4		18.	157	30		188	370	-183	Reimbursed mileage per policy
REPAIRS/MAINT - EQUIPME	1205011-310-703100-200	773	0	773		773		773		773		773	
MAINT-AGREEMENTS	1205011-310-703110-000	16,089	21,916	-5,827	25 mo. contract signed prior to	b 1/ 89	21,915	-5,827	25 mo. co pre signed	16,089	21,916	-5,827	25 mo. contract signed prior to Resolution 09-24-14-A
TRAINING & EDUCATION	1205011-310-705010-000	94	0	94		94		94				94	
BUSINESS EXPENSE	1205011-310-705300-000									0	6	-6	Reimbursement
MISCELLANEOUS	1205011-310-705900-000	94	11	83		94	12.	-28	Mutilated/Counterfeit cu	94	52	41	
CASH/OVER SHORT	1205011-310-709004-000	0		0		0		0		0		0	
CLAIM SETTLEMENT	1205011-310-709800-000	0		0		0		0		0		0	
I/T SUPPLIES & MATERIALS	1205011-310-750001-000	0		0		0		0		0		0	
I/T - PRINTING	1205011-310-750009-000	1,563	1,331	232		1,563	2,165	-602	Didn't the order to	1,563	1,500	63	
I/T REPAIRS/MAINT - VEHIC			35	· 121		156	72	84		156	22	134	
VEHICLE EXP - GAS & OIL -			662	57		719	604	115		719	593	126	
I/T TRAINING & EDUCATIO				0		0		0		0		0	
CASH/OVER SHORT	1205011-321-709004-000			0		0	10,801	-10,801	There are procedures in		-5,399	5,399	
TOTAL		23,424	27,578	-4,154	0	23,424	42,694	-19,270		18,714	18,690	24	

										Accession and the			Page 8
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
ACCOUNTING - MIS DEP	R												
REPAIRS/MAINT - EQUIPM	IE 1205011-400-703100-200	938	19,819	-18,881	This expense will be moved to	938	-19,819	20,756		938		938	
FOTAL	§	938	19,819	-18,881		26,799	36,517	-9,718		22,089	15,405	6,684	
ACCOUNTING - SHIPPIN	G		e'	1		1							
SUPPLIES & MATERIALS	1205011-500-700001-000	306	0	306		2		306		306		306	
TRANSP. & PER DIEM	1205011-500-701000-000	19	0	19		19		19		19		19	
OUTSIDE SERVICES	1205011-500-702010-000	813	0	813	YA	813	A	03		813		813	
REPAIRS/MAINT - EQUIPM	IE 1205011-500-703100-200	31	0	31		31	119	-88-	Forklift battery repair - w	31	96	-64	Needed service on Forklift
NSURANCE	1205011-500-705500-000	75		75		75	182	-11	Submit of by Risk Mana	75	182	-107	Submitted by Risk Management t Oneida Casino
/T REPAIRS/MAINT - VEH	IC 1205011-500-753100-300	31	0	31		31	153	-122		31		31	
EHICLE EXP - GAS & OIL	- 1205011-500-753101-000	156	158	-1	Filled up delivery van twic du	156	-3	3		156	63	93	
TOTAL	1	1,431	158	1,274	0	481	608	824		1,431	341	1,090	
COMPLIANCE							<		· (
SUPPLIES & MATERIALS	1205012-000-700001-000	9	0	9		9		9		9		9	
DUES	1205012-000-700071-000	9	13	-4	IIA Membership, resource for	9	170	-161	IV	9		9	
RANSP. & PER DIEM	1205012-000-701000-000	116	0	116		116		116		116		116	
AILEAGE EXPENSE	1205012-000-701001-000	6	0	6		6		6		6		6	
RAINING & EDUCATION	1205012-000-705010-000	0	0	0		0		0		0		0	
ONTRA - INTERNAL AUD	01 1205012-000-709509-000	15,025	18,714	-3,689	We do not control this account	15,025	29,268	-14,243	This is detailed in the rep	15,025	9,154	5,871	
CONTRA - BACKGROUND	n 1205012-000-709509-100	32,709	40,846	-8,137	We do not control this account	32,709	50,814	-18,105	This is detailed in the rep	32,709	39,753	-7,043	This is detailed in the report that Backgrounds submits
CONTRA - SURVEILLANCI	E 1205012-000-709509-200	158,742	173,196	-14,454	We do not control this account	158,742	267,612	-108,870	This is detailed in the rep	158,742	180,913	-22,171	This is detailed in the report that Surveillance submits
TOTAL		206,617	232,769	-26,152	0	206,617	347,864	-141,247		206,617	229,819	-23,202	

Page 86 of 340

8

		Oct-14	Oct-14	Oct-14		Nov-14	Nov-14	Nov-14		Dec-14	Dec-14	Dec-14	
		75%	Actual			75%	Actual			75%	Actual		Stand Address and
DESCRIPTION	ACCOUNT NUMBER	BUDGET	Spend	Variance	EXPLANATION	BUDGET	Spend	Variance	EXPLANATION	BUDGET	Spend	Variance	EXPLANATION

CUSTOMER RELATIONS -

SUPPLIES & MATERIALS	1205013-200-700001-000	94	0	94		94		94		94		94	
TRANSP. & PER DIEM	1205013-200-701000-000	156	0	156		156		156		156		156	
REPAIRS/MAINT - EQUIPME	1205013-200-703100-200	281	164	117				281		281	720	-439	Refill foam for Hydroclean equipment
REPAIRS/MAINT - VEHICLE	1205013-200-703100-300	2,500	500	2,000		2,500		2,500		2,500	602	1,898	
VEH. EXP-GAS & OIL	1205013-200-703101-000	5,478	3,348	130		,478	443	5,035		5,478	6,005	-527	
UNIFORMS	1205013-200-705001-000	19	0	19			5	-31	Due to the garage area	19		19	
TRAINING & EDUCATION	1205013-200-705010-000	68	0	68		68		68		68		68	
INSURANCE	1205013-200-705500-000	890	1,103	-213	Based on Risk Magazient	890	1,1 3	-213	Subilitted by Risk Mana	890	1,103	-213	Submitted by Risk Management to Oneida Casino
I/T - PRINTING	1205013-200-750009-000	6	0	6		6		6		6		6	
I/T REPAIRS/MAINT - VEHIC	1205013-200-753100-300	2,500	3,765	-1,265	Repairs are necessary to eep u	2,500	1,607	893		2,500	2,291	209	
VEHICLE EXP - GAS & OIL -	1205013-200-753101-000	1,688	3,415	-1,728	Using regular gas units more du	1/.	2,8.	-1,149	Leto do a tim	1,688	2,274	-587	Due to down time (repairs) of
I/T TRAINING/EDUCATION	1205013-200-755010-000	6	0	6				6		6		6	
TOTAL		13,685	12,295	1,390	0	13,685	.40	7,645		12, /85	12,995	690	
MARKETING					a.			(
SUPPLIES & MATERIALS	1205014-000-700001-000	563	-1,178	1,741		563	87	75		563	125	438	
NEWSP/SUBSCRIPTIONS	1205014-000-700070-000	456	0	456		456		456		456		456	
DUES	1205014-000-700071-000	140	0	140		140		140		140		140	
TRANSP. & PER DIEM	1205014-000-701000-000	387	0	387		387		387		387		387	
MILEAGE EXPENSE	1205014-000-701001-000	50	30	20		50	27	23		50	11	39	
OUTSIDE SERVICES	1205014-000-702010-000	7,094	495	6,599		7,094	160	6,934		7,094	533	6,561	
PROFESSIONAL FEES	1205014-000-702030-000	16,125	0	16,125		16,125		16,125		16,125		16,125	
REPAIRS/MAINT - EQUIPME	1205014-000-703100-200	263	0	263		263	31	231		263		263	

					-				-				Page 87
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
TRAINING & EDUCATION	1205014-000-705010-000	56	0	56		56		56		56		56	
PROMOTION/ITEMS	1205014-000-705100-000	0	8,225	-8,225		0	-	0		0		0	
ENTERTAINMENT-PROMO	1 1205014-000-705100-003	31,250	0	31,250		31,250	4,400	26,850		31,250	16,550	14,700	
PROMOTIONAL COUPONS	1205014-000-705101-100	0	37,500	-37,500	Auventisting Actual Spend – \$212.452	0		0		0		0	
ADVERTISING	1205014-000-705110-000	130,206	0	130,206		130,206		130,206		130,206		130,206	
BROADCAST MEDIA	1205014-000-705110-102	0	0				54,485	-54,485	Was budgeted under "Ad	0		0	
BROADCAST PRODUCTION	1205014-000-705110-103	0	50	-50		0	5,814	-25,814	Was budgeted under "Ad	0		0	
COPYWRITING	1205014-000-705110-104	0	3,305	305	Budgeted Advertising, does	0	590	-590	Was budgeted under "Ad	0	100	-100	Was budgeted under "Advertising
NTERACTIVE MARKETING	3 1205014-000-705110-105	0	1,330	-1,330	out sted (Advertising, does		37 5	1288	Was budgeted under "Ad	0	165	-165	Was budgeted under "Advertising
MEDIA PLANNING	1205014-000-705110-106	0	15,699	-15,699	Budget I in After ing, does	0	1,938	-1,93(Was budgeted under "Ad	0	2,660	-2,660	Was budgeted under "Advertising
OUTDOOR/STATIC & DIGIT	1205014-000-705110-107	0	128,625	-128,625	Budgetee A Advensing uses	0	7, 25	-7,685	w budgeted under "Ad	0	12,375	-12,375	Was budgeted under "Advertising
DUTDOOR/OFF-PROPERTY	1205014-000-705110-108	0	0	0	1	0	41,125	+1,125	Vus bude under "Ad	0	41,125	-41,125	Was budgeted under "Advertising revenue generation
PHOTOGRAPHY/VIDEO	1205014-000-705110-109	0	25,128	-25,128	Budgeted in Advertising,	0		0		0		0	
PRINT	1205014-000-705110-110	0	1,816	-1,816	Budgeted in Advertising, does	0		0		0	4,800	-4,800	Was budgeted under "Advertising
SIGNAGE-SUPPLIES	1205014-000-705110-111	0	0	0		0	-6	684				0	
AGENCY ACCOUNT SERVI	c 1205014-000-705110-112	0	0	0		0	169	-1/	Was budgeted unor "Ad	0		0	
CUSTOMER INCENTIVE-RA	1205014-000-705111-100	5,375	689	4,686		5,375	540	4,829		5,375	710	4,665	
CUSTOMER INCENTIVE-RE	1205014-000-705111-400									0	5,117	-5,117	Was budgeted under "Customer Incentive Radisson" revenue generation
LICENSE, CERT, FEES	1205014-000-705401-000	575	2,628	-2,053	IT Printing was not trended	575	678	-103	h, sic licer, ag fee fo,	575	4,464	-3,889	Music licensing fee for live
/T - PRINTING	1205014-000-750009-000	11,191	0	11,191		11,191	2,625	8,566		11,191	17,195	-6,004	Increased Direct Mail outs per strategy
/T ADVERTISING EXPENSI	E 1205014-000-755110-000	13	0	13		13		13		13		13	
TOTAL		203,743	224,342	-20,599	0	203,743	142,964	60,779		147,360	88,712	58,648	

SECURITY

DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
SUPPLIES & MATERIALS	1205015-000-700001-000	938	315	623		938	238	700		938	288	649	
NEWSP/SUBSCRIPTIONS	1205015-000-700070-000	56	0	56		56		56		56		56	
TRANSP. & PER DIEM	1205015-000-701000-000	144	0	144		144		144		144		144	
IILEAGE EXPENSE	1205015-000-701001-000	19	0	19		19		19		19		19	
OUTSIDE SERVICES	1205015-000-702010-000	116	0	116		116		116		116		116	
EPAIRS/MAINT - EQUIPMI	1205015-000-703100-200	500	40	46				500		500	166	334	
EPAIRS/MAINT - VEHICLE	1205015-000-703100-300	63	0	63		63		63		63		63	
INIFORMS	1205015-000-705001-000	13	0	13		13	50	-38	Reimbursed employee \$5	13		13	
RAINING & EDUCATION	1205015-000-705010-000	763	0	763				763		763		763	
NSURANCE	1205015-000-705500-000	259	521	-262	Based in Riv Management	259	521	-262	ubmitted by Risk Mana	259	521	-262	Submitted by Risk Management to Oneida Casino
T - PRINTING	1205015-000-750009-000	250	0	250		250		250		250		250	
T REPAIRS/MAINT - VEHI	1205015-000-753100-300	469	408	61		46	120	349		469	26	443	
EHICLE EXP - GAS & OIL ·	1205015-000-753101-000	1,000	791	209		1,000	674	326		1,000	543	457	
TOTAL		4,587	2,075	2,512	0	4-	1,60	2,985		4,587	1,544	3,043	
E MPLOYEE SERVICES - A SUPPLIES & MATERIALS	1205016-000-700001-000	375	149	226		375	2		(375		375	
	1205016-000-700070-000	50	72	22	Des Deid, Deine Franzend	50	72	22	Deine	50	242	-192	Pre-Paid; Being Expensed -
EWSP/SUBSCRIPTIONS				-22	Pre-Paid; Being Expensed -			-22	Present Being		242		rie-raid, Being Expensed -
RANSP. & PER DIEM	1205016-000-701000-000	234	0	234		234		204		234		234	
IILEAGE EXPENSE	1205016-000-701001-000	22	0	22		22		22		22		22	
UTSIDE SERVICES	1205016-000-702010-000	26	0	26		26	100	26		26	100	26	
MPLOYEE INCENTIVE	1205016-000-705000-000	0	0	0		0	100		For safety L some	0	100	-100	For safety reasons employees are
RAINING & EDUCATION	1205016-000-705010-000	69	0	69		69	93		2 Oct certified	69	55	14	
T SUPPLIES & MATERIALS	S 1205016-000-750001-000	6	0	6		6		6		6		6	
/T - PRINTING	1205016-000-750009-000	25	0	25		25		25		25	38	-13	Printing for forms utilized in department

		1											Page 8
DESCRIPTION A	CCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
TOTAL		807	221	586	I	807	265	543		807	434	373	
EMPLOYEE SERVICES - EI													
MILEAGE EXPENSE 12	205016-100-701001-000	19	0	19		19		19		19		19	
PROFESSIONAL FEES 12	205016-100-702030-000	938	0	938		938		938		938		938	
I/T - PRINTING 12	205016-100-750009-000	63	0					63	1	63		63	
TOTAL		1,019	0	1,019	0	1,01	0	1,019		1,826	434	1,392	
EMPLOYEE SERVICES - W							• /						
SUPPLIES & MATERIALS 12	205016-200-700001-000	375	300	75		375	6	369		375	8	367	
MILEAGE EXPENSE 12	205016-200-701001-000	6	0	6		6		6		6		6	
DUTSIDE SERVICES 12	205016-200-702010-000	250	350	-100	This is not an action ont	- 2	350	10	This is of an actual	250	350	-100	This is not an actual monthly
CONSULTANT EXPENSE 12	205016-200-702100-001	156	0	156		15		156		156	- 0	156	
REPAIRS/MAINT - EQUIPME 12	205016-200-703100-200	16	0	16		I		16	11	16	85	-69	Scissor sharpening that happens occasionally
	205016-200-705001-000		5,271	3,479		2 50	3,318	5,432		750	8,685	65	
	205016-200-705001-100		0	156		156		156		6		156	
	205016-200-705111-000	6	0	6		6				6		6	
	205016-200-709004-000	0	0	0		0		0		0		0	
	205016-200-750009-000	45	0	45		45		45		45	359	-314	Printing for forms and tickets utilized in department
FOTAL	203010 200 730003 000	9,761	5,921	3,840	0	9,761	3,674	6,087		9,761	9,487	273	
IUIAL		9,701	5,921	3,040	0	9,701	3,074	0,087		9,701	9,407	213	
													÷
EMPLOYEE SERVICES - TI													
	205016-300-700044-000	344	0	344		344		344		344		344	
NEWSP/SUBSCRIPTIONS 12	205016-300-700070-000	44	75	-31	Our Survey Monkey	44	75	-31	Our Survey Monkey	44	75	-31	Our Survey Monkey subscription
DUES 12	205016-300-700071-000	200	335	-135	The SHRM membership	200	-35	235		200	120	80	
BOOK PURCHASES 12	205016-300-700072-000	31	0	31		31		31		31		31	

													Page 9
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
VIDEOS/COLLECTIONS	1205016-300-700074-000	31	0	31		31		31		31		31	
MILEAGE EXPENSE	1205016-300-701001-000	13	0	13		13		13		13		13	
CONSULTANT EXPENSE	1205016-300-702100-001	938	0	938		938		938		938		938	
TRAINING & EDUCATION	1205016-300-705010-000	0	0	0		0		0		0		0	
I/T - PRINTING	1205016-300-750009-000	831	539	292		831	927	-96	Mandated Key Watcher	831	1,223	-392	Mandated Key Watcher Training
/T TRAINING/EDUCATION	1205016-300-755010-000	6	0	6		6		6		6		6	
TOTAL		2,437	949	1,488	0	2	,,	1,470		2,437	1,418	1,019	
EMPLOYEE SERVICES - T		-	^	-	YA		6					1	
TRAINING MATERIALS	1205016-400-700044-000	6	0	6		6		6		6		6	
MILEAGE EXPENSE	1205016-400-701001-000	9	0	9		9		9		9		9	
I/T - PRINTING	1205016-400-750009-000	50	0	50		50	•	50		50		50	
TOTAL		66	0	66	0	66	0	66	///	2,503	1,418	1,085	
EMPLOYEE SERVICES - S	5.4					C			· Vr				
NEWSP/SUBSCRIPTIONS	1205016-500-700070-000	103	129	-27	Pre-Paid; Expense -	103	29	-2.7	Pre-Paid; Expen -	1.3	129	-27	Pre-Paid; Expense - Membership
MILEAGE EXPENSE	1205016-500-701001-000	56	0	56		56		-0	Mileage expense	56	32	25	
/T - PRINTING	1205016-500-750009-000	13	0	13		13		13	\wedge	13		13	
TOTAL		171	129	42	0	171	205	- 4		2,674	1,579	1,095	
EMPLOYEE SERVICES - I	21								V)				
NEWSP/SUBSCRIPTIONS	1205016-700-700070-000	47	46	1		47	46	1		47	211	-164	Yearly SHRM renewal, in order have access to most current HR policies and procedures
MILEAGE EXPENSE	1205016-700-701001-000	63	0	63		63		63		63	47	15	
TOTAL		109	46	63	0	109	46	63		2,796	1,837	958	

													Page 91
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
CUSTODIAL													
SUPPLIES & MATERIALS	1205017-000-700001-000	17,750	20,187	-2,437	Custodial supplies based on	17,750	10,067	7,683		17,750	26,487	-8,737	Restocking of custodial cleaners,
MILEAGE EXPENSE	1205017-000-701001-000	13	0	13		13		13		13		13	
OUTSIDE SERVICES	1205017-000-702010-000	7,813	9,749	-1,937	Towels/Mats/Mops; this	7,813	4,340	3,473		7,813	16,000	-8,188	Novembers dumpster expense,
REPAIRS/MAINT - EQUIPMI	1205017-000-703100-200	250	821	-571	invoices that were never sai	250	53	197		250	581	-331	season, repair hose and clips for
UNIFORMS	1205017-000-705001-000	13	0					13		13		13	
RAINING & EDUCATION	1205017-000-705010-000	63	0	63		6		63		63	83	-21	Annual license for Electrical Inspector
MEETING EXPENSE	1205017-000-705301-000	31	0	31		31		31		31		31	
SUPPLIES & MATERIALS	1205017-303-700001-000	5,000	5,594	-594	Custo, supplies sed on	5,00	<u>A</u>			5,000	8,635	-3,635	Restocking of custodial cleaners,
OFFICE SUPPLIES	1205017-303-700001-000	0	5,594	-5,594	Custo, al sy utes ased on	0	24	-24	his will be reclassified	to Supplies and	-24		
OUTSIDE SERVICES	1205017-303-702010-000	1,438	2,856	-1,419	Towels/Mats/More, th	1,438	1,530	-9*	Started winter mat	1,438	5,744	-4,307	Started winter mat schedule for
REPAIRS/MAINT - EQUIPMI	1205017-303-703100-200	63	0	63		6.		63		63		63	
SUPPLIES & MATERIALS	1205017-304-700001-000	500	654	-154	Custodial supplies based on	500	543	-43	Cleani st plies based	500	942	-442	Cleaning supplies based on
OUTSIDE SERVICES	1205017-304-702010-000	625	546	79		F	21.	407		625	2,796	-2,171	Novembers dumpster expense,
SUPPLIES & MATERIALS	1205017-305-700001-000	844	1,322	-478	Custodial supplies based on		518	326		44	993	-150	Restocking of custodial cleaners,
OUTSIDE SERVICES	1205017-305-702010-000	938	706	232		938	32	545			1,547	-609	ARAMARK past due invoices
SUPPLIES & MATERIALS	1205017-321-700001-000	531	559	-28	Custodial supplies based on	531	- 13	.8		531	961	-430	Restocking of custodial cleaners,
OUTSIDE SERVICES	1205017-321-702010-000	531	744	-213	Towels/Mats/Mops; this	531	344	187		531	853	-322	ARAMARK past due invoices
FOTAL		36,400	49,332	-12,932	0	36,400	22,757	15,643		18,650	39,112	-20,486	
MAINTENANCE									、 へ				
SUPPLIES & MATERIALS	1205018-000-700001-000	13,125	8,885	4,240		13,125	8,321	4,804		13,125	22,143	-9,018	Salt stock up for winter, needed for saftety of all customers and employees.

TRANSP. & PER DIEM

TRAVEL - MEALS

MILEAGE EXPENSE

1205018-000-701000-000

1205018-000-701000-300

1205018-000-701001-000

REPAIRS/MAINT-BLDGS & (REPAIRS/MAINT - EQUIPME12UNIFORMS12TRAINING & EDUCATION12RENTAL USAGE12MEETING EXPENSE12INSURANCEMISCELLANEOUS12BUILDING REPAIR/MAINT-I12		2,125 9,375 5,000 44 194 94 50	2,897 13,210 2,702 76 0 156	-772 -3.835 2,298 -32 194	Quarterly Sprinkler Main Casino Surveillance area Safety shoe reimbursement for	2,125 9,375 5,000 44	2,897 652 913	8,723	Otis elevator &	2,125 9,375	4,022 4,784	-1.897 4,591	Otis elevator & escalator monthly
REPAIRS/MAINT - EQUIPME12UNIFORMS12TRAINING & EDUCATION12RENTAL USAGE12MEETING EXPENSE12LICENSE, CERT, FEES12INSURANCE12MISCELLANEOUS12J/T - PRINTING12BUILDING REPAIR/MAINT-I12	205018-000-703100-200 205018-000-705001-000 205018-000-705010-000 205018-000-705202-000 205018-000-705301-000	5,000 44 194 94	2,702 76 0	2,298 -32		5,000		1.		9,375	4,784	4,591	
UNIFORMS12TRAINING & EDUCATION12RENTAL USAGE12MEETING EXPENSE12LICENSE, CERT, FEES12INSURANCE12MISCELLANEOUS12J/T - PRINTING12BUILDING REPAIR/MAINT-I12	205018-000-705001-000 205018-000-705010-000 205018-000-705202-000 205018-000-705301-000	44 194 94	76 0	-32	Safety shoe reimbursement for		913	1.007					
TRAINING & EDUCATION12RENTAL USAGE12MEETING EXPENSE12LICENSE, CERT, FEES12INSURANCE12MISCELLANEOUS12I/T - PRINTING12BUILDING REPAIR/MAINT-I12	205018-000-705010-000 205018-000-705202-000 205018-000-705301-000	194 94	0		Safety shoe reimbursement for	44		4,087		5,000	2,841	2,159	
RENTAL USAGE 12 MEETING EXPENSE 12 LICENSE, CERT, FEES 12 INSURANCE 12 MISCELLANEOUS 12 I/T - PRINTING 12 BUILDING REPAIR/MAINT-I 12	205018-000-705202-000 205018-000-705301-000	94		194				44		44		44	
MEETING EXPENSE 12 LICENSE, CERT, FEES 12 INSURANCE 12 MISCELLANEOUS 12 I/T - PRINTING 12 BUILDING REPAIR/MAINT-I 12	205018-000-705301-000		156	2508/02		194		194		194	85	109	
LICENSE, CERT, FEES 12 INSURANCE 12 MISCELLANEOUS 12 I/T - PRINTING 12 BUILDING REPAIR/MAINT-I 12		50		2		4	162	-68	Cylinder rental of oxyger	94	156	-62	Cylinder rental of oxygen & acetylene for torches based on usage
INSURANCE 12 MISCELLANEOUS 12 I/T - PRINTING 12 BUILDING REPAIR/MAINT-I 12	205018-000-705401-000	30	0	50				50		50		50	
MISCELLANEOUS 12 I/T - PRINTING 12 BUILDING REPAIR/MAINT-I 12		69	16	53				53		69	16	53	
I/T - PRINTING 12 BUILDING REPAIR/MAINT-I 12	205018-000-705500-000	180	261	-81	*Out a pur entrol	180	261	-80	Submitted by Risk Mana	180	261	-80	Submitted by Risk Management to Oneida Casino
BUILDING REPAIR/MAINT-I	205018-000-705900-000	3	0	3		3		3		3		3	
	205018-000-750009-000	94	0	94		94	147	-53	Need a to restock work	94		94	
	205018-000-753100-100	2,188	0	2,188		2,188		2,188		2,188		2,188	
EQUIP/REPAIR MAINT-INTR 12	205018-000-753100-200	688	0	688		6.		688		688		688	
I/T REPAIRS/MAINT - VEHIC 12	205018-000-753100-300	938	362	576		18		938		38	16	921	
VEHICLE EXP - GAS & OIL - 12	205018-000-753101-000	1,875	2,662	-787	Oil changes and gas purchased	1,875	19	-104	Extra gas purcha es due	1, 75	2,001	-126	Extra gas puchases due to snow
SUPPLIES & MATERIALS 12	205018-303-700001-000	506	1,552	-1,046	HVAC filters for this location	506	26	- A.	HVAC filter annual	506	693	-187	Filters for HVAC units, needed for
OUTSIDE SERVICES 12	205018-303-702010-000	250	0	250		250		250		250	375	-125	Back-up generator inspection
REPAIRS/MAINT-BLDGS & (205018-303-703100-100	1,875	290	1,585		1,875	361	1 14		1,875		1,875	
REPAIRS/MAINT - EQUIPME 12	205018-303-703100-200	1,875	546	1,329		1,875	811	1,064		1,875	145	1,730	
BUILDING REPAIR/MAINT-I	205018-303-753100-100	0	0	0		0		0		0		0	
	205018-304-700001-000	175	224	-49	HVAC filters for this location	175	601	-426 -	HVAC fn. annual	175	-269	444	
	205018-304-702010-000	344	0	344		344		344		344	375	31	billing for Total Energies, typical
REPAIRS/MAINT-BLDGS & (12		1,875	0	1,875		1,875	145	1,730		1,875	-599	2,474	
REPAIRS/MAINT - EQUIPME 12		625	0	625		625	- 10	625		625	145	480	
BUILDING REPAIR/MAINT-I		025	0	023		025		025		0	. 10	0	

													Page 93
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
SUPPLIES & MATERIALS	1205018-305-700001-000	219	202	17		219	27	191		219	217	1	
OUTSIDE SERVICES	1205018-305-702010-000	406	0	406		406		406		406	375	31	
REPAIRS/MAINT-BLDGS &	1205018-305-703100-100	125	290	-165	HVAC filters/electrical for	125	145	-20	HVAC filter annual	125		125	
REPAIRS/MAINT - EQUIPMI	1205018-305-703100-200	188	0	188		188		188		188	145	43	
BUILDING REPAIR/MAINT-	1 1205018-305-753100-100	0	0	0		0		0		0	-	0	
SUPPLIES & MATERIALS	1205018-321-700001-000	63	0	62				63		63		63	
OUTSIDE SERVICES	1205018-321-702010-000									0		0	
TOTAL		44,640	34,332	308	0	.,640	18,343	26,296		8,525	1,602	6,923	
							1						
PLAYER DEVELOPMENT	-				YN.		\frown						
SUPPLIES & MATERIALS	1205020-200-700001-000	606	0	606		606		401		606	3,005	-2,399	Additional Players Club cards
DUES	1205020-200-700071-000	0	0	0			•			0		0	
MILEAGE EXPENSE	1205020-200-701001-000	19	0	19		19		19		19		19	
OUTSIDE SERVICES	1205020-200-702010-000	11,250	0	11,250		11,25	1. 211	-1,561	Feel or gin card service	11,250	16,136	-4,886	Fee's for gift card services
PROMOTION/ITEMS	1205020-200-705100-000	8,156	25,030	-16,874	Purchase of SVM Mobile Gas	156	10,112	-1,955	Retail gas ar (point	8,156	22,002	-13,846	Retail gas cards (point purchase) is
PLAYERS CLUB REDEMPT	1205020-200-705100-210	0	5,845	-5,845	Patrons can redeem points on	0		0				0	
TOUR BUS COUPONS	1205020-200-705101-200	0	0	0		0		0		0		0	
CUSTOMER INCENTIVE	1205020-200-705111-000	5,625	0	5,625		5,625		3,625		5,625	0	5,625	
CUSTOMER INCENTIVE-RA	1205020-200-705111-100	11,363	0	11,363		11,363	6,824	1,538		11,363	7,639	3,724	
CUSTOMER INC-RAD PTS I	1205020-200-705111-101	0	0	0		· 0	7,692	-7,69	Point	0	5,518	-5,518	Point redemption/discount for
CUSTOMER INCENTIVE-LC). 1205020-200-705111-300	0	3,523	-3,523	Budgeted in Customer Incentiv	v 0	14,305	-14,305	I nt	0	14,123	-14,123	Point redemption/discount for
CUSTOMER INC - LODGE P	1205020-200-705111-301	0	603	-603	Budgeted in Customer Incentiv	v 0	1,922	-1,922	Point	0	2,148	-2,148	Point redemption/discount for
CUSTOMER INC - REST PTS	5 1205020-200-705111-401	0	0	0		0	7,565	-7,565	Point	0	6,697	-6,697	Point redemption/discount for
MISCELLANEOUS	1205020-200-705900-000	0	0	0		0		0		0		0	
CASH/OVER SHORT	1205020-200-709004-000	. 0	0	0		0		0		0	-320	320	
CREDIT CARD VARIANCE	1205020-200-709004-400		0	0		0	2,725	-2,725	Customer dispute on cha	0	-2,142	2,142	
I/T SUPPLIES & MATERIAL			0	0		0		0		0		0	
								1					

.

Page 94 of 340

		and the second se											Page 94,
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
I/T - PRINTING	1205020-200-750009-000	3,125	0	3,125		3,125		3,125		3,125	1,408	1,717	
TOTAL		40,144	35,001	5,143	0	40,144	64,161	-24,017		39,519	73,209	-33,690	
PLAYER DEVELOPMENT													
SUPPLIES & MATERIALS	1205020-400-700001-000	156	20	136		156		156		156		156	
MILEAGE EXPENSE	1205020-400-701001-000	16		0		-0		16		16		16	
OUTSIDE SERVICES	1205020-400-702010-000	1,165	599	566			1, 98	-243	Services include Plants	1,165	258	908	
TRAINING & EDUCATION	1205020-400-705010-000	0				0		0		0		0	
PROMOTION/ITEMS	1205020-400-705100-000	9,000	41,233	-32,233	YA	9,000	1,71	-18,/10	Comps for gift cards, Pla	9,000	30,521	-21,521	Comps for gift cards, Player preference revenue generation
PROMOTIONAL COUPONS	1205020-400-705101-100	0		0	-///	0		0		0		0	
SPONSORSHIPS	1205020-400-705110-100	625		625		5				625		625	
CUSTOMER INCENTIVE	1205020-400-705111-000	8,208	9,982	-1,774		8,20.8	11,598	-3,390	Lomps , r e ents, player	8,208	14,427	-6,220	Comps for events, player preference revenue generation
CUSTOMER INCENTIVE-RA	1205020-400-705111-100	42,025	41,566	459		47 25	47,61	-5,586	Comps for or player	42,025	49,569	-7,544	Comps for events, player preference revenue generation
CUSTOMER INCENTIVE-FR	1205020-400-705111-200	0	1,085	-1,085		0	1 0	-1,100	Comps for food, layer g		1,100	-1,100	Comps for events, player preference revenue generation
CUSTOMER INC - RESTAUF	1205020-400-705111-400	0	17,643	-17,643		0	24, 99	.609	Comps for food, player p	0	26,832	-26,832	Comps for events, player preference revenue generation
LICENSE, CERT, FEES	1205020-400-705401-000	7,132	12,232	-5,100		7,132	14,132	7,000	Preatid accounts for	7,132	12,232	-5,100	Prepaid accounts for Packer and
TOTAL		68,327	124,359	-56,033	0	68,327	128,167	-59,84		68,327	134,938	-66,612	
CASINO PROMOTIONS									$\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{\mathbf{$				
SUPPLIES & MATERIALS	1205021-000-700001-000	469	234	234		469	179	290		469	132	337	
NEWSP/SUBSCRIPTIONS	1205021-000-700070-000	55		55		55		55		55		55	
TRANSP. & PER DIEM	1205021-000-701000-000	241		241		241		241		241		241	
TRAVEL - AIRFARE	1205021-000-701000-100	0		0		0	2	0		0		0	
TRAVEL - LODGING	1205021-000-701000-200	0		0		0		0		0		0	

Page 95 of 340

													Page
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	
TRAVEL - MEALS	1205021-000-701000-300	0		0		0		0		0		0	
MILEAGE EXPENSE	1205021-000-701001-000	9		9		9		9		9		9	
OUTSIDE SERVICES	1205021-000-702010-000	2,313		2,313		2,313		2,313		2,313		2,313	
MAINT-AGREEMENTS	1205021-000-703110-000	0		0		0		0		0		0	
PROMOTION/ITEMS	1205021-000-705100-000	9,656	500	9,156		9,656	3,510	6,146		9,656	7,094	2,563	
CUSTOMER INCENTIVE	1205021-000-705111-000	5,375	-135	.10		<u>075</u>	135	5,240		5,375		5,375	
CUSTOMER INCENTIVE-RA	1205021-000-705111-100	17,563	351	7,211		17	22 78	-4,516	Point	17,563	72,419	-54,856	Point redemption/discount for
CUSTOMER INCENTIVE-LO	1205021-000-705111-300	0	-			0	79	-79	Point	0		0	
CUSTOMER INC - RESTAUR	1205021-000-705111-400	0	989	-989	Budge n Custor Incentiv	v O		~ 3	Point	0		0	
/T - PRINTING	1205021-000-750009-000	31,031	16,290	14,741		31,031	1,829	29,202		31,031		31,031	
/T PROMOS/PROMO ITEMS	1205021-000-755100-000	469		469		469		469		469		469	
FOTAL		67,180	18,230	48,951	0	67,1 0	27,985	2,195		66,657	79,513	-12,856	
DIRECT MAIL	1205021-000-705111-000	0	-135	135			-135	135	1/	0	383	-383	Coupon offer for food at restaurants (loyalty), Increase revenue
CUSTOMER INCENTIVE-RA	1205021-000-705111-100	0	351	-351	Coupon offer for food at restau	0	644	-64	Coupon offer for foc	0	394	-394	Coupon offer for food at restaurants (loyalty), Increase revenue
CUSTOMER INCENTIVE-LO	1205021-000-705111-300	0		0		0		0		0		0	
CUSTOMER INC - RESTAUR	1205021-000-705111-400	0	989	-989	Budgeted in Customer Incentiv	0	619	-619	Coupon offer from that	0	980	-980	Coupon offer for food at restaurants (loyalty), Increase revenue
/T - PRINTING	1205021-000-750009-000	0	16,290	-16,290	Coupon offer for food at restau	0	20,131	-20,131	Coupon of the for format	0	70,791	-70,791	Coupon offer for food at restaurants (loyalty), Increase revenue
/T PROMOS/PROMO ITEMS	1205021-000-755100-000	0		0		0		0		0		0	
TOTAL		0	17,495	-17,495	0	0	21,260	-21,260		0	72,547	-72,547	
LODGE													
SUPPLIES & MATERIALS	1205030-303-700001-000	2,625	1,009	1,616		2,625	3,243	-618	Purchased fryer for \$122	2,625	2,202	423	
						1							

Page 96 of 340

													Page 96
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
TRANSP. & PER DIEM	1205030-303-701000-000	5	0	5		5		5		5		5	
OUTSIDE SERVICES	1205030-303-702010-000	625	0	625		625		625		625		625	
REPAIRS/MAINT - EQUIPMI	1205030-303-703100-200	325	612	-287	Micro maintenance, and ice m	a 325	244	81		325	389	-64	Van's Fire and Safety needed to clean hood for safety reasons
INIFORMS	1205030-303-705001-000	34	0	34		34		34		34		34	
RAINING & EDUCATION	1205030-303-705010-000	0	0	0		0		0		0		0	
ICENSE, CERT, FEES	1205030-303-705401-000	125	0	_5			100	25		125		125	
ASH/OVER SHORT	1205030-303-709004-000	0	28	-28	Over /Shorts happen		- 1	19		0	23	-23	There are procedures in place as to
REDIT CARD VARIANCE	1205030-303-709004-500	0	0			0		0		0		0	
LAIM SETTLEMENT	1205030-303-709800-000	0	0	0		0				0		0	
T LICENSES & FEES	1205030-303-755401-000	6	100	-94	Food 5 fety / cens paid	6	\leftarrow	6		6		6	
OTAL		3,746	1,749	1,997	Food Striety Lens paid	3,746	3, 18	178		3,746	2,614	1,132	
BINGO SUPPLIES & MATERIALS	1205040-201-700001-000	531		531		531		531	Δ.	531	138	393	
								1.			138	1	
RANSP. & PER DIEM	1205040-201-701000-000		296	-97 .		59		199		199		199	
RAVEL - MEALS	1205040-201-701000-300			0		0		0				0	
IILEAGE EXPENSE	1205040-201-701001-000	1.0		0		0		0		0		0	
EPAIRS/MAINT - EQUIPMI	1205040-201-703100-200	63		63		63	-1,0	1,125		63	-1,063	1,125	
RAINING & EDUCATION	1205040-201-705010-000	56		56		56		56	\leftarrow	56		56	
ROMOTION/ITEMS	1205040-201-705100-000	5,672	1,309	4,363	-	5,672	701	4,97		5,672	861	4,811	
USTOMER INCENTIVE	1205040-201-705111-000	47	0	47		47		47		47		47	
ASH/OVER SHORT	1205040-201-709004-000	0		0		0	7	-7	On 11/1, there was a	0	5,015	-5,015	There are procedures in place as to
INGO VARIANCE	1205040-201-709004-400	0		0		0	713	-713	The Bingo variance	0	252	-252	There are procedures in place as to
REDIT CARD VARIANCE	1205040-201-709004-500	0		0		0	414	-414	On 11/25 (pm session)	0		0	*
T - PRINTING	1205040-201-750009-000	2,813	2,908	-96	Updated the game schedules	2,813	2,748	65		2,813	1,945	868	
T PROFESSIONAL FEES	1205040-201-752030-000	0		0		0		0		0		0	
TOTAL	-	9,381	4,513	4,868	0	9,381	3,520	5,860		9,381	7,148	2,233	

Page 97 of 340

	Oct-14	Oct-14	Oct-14		Nov-14	Nov-14	Nov-14		Dec-14	Dec-14	Dec-14	
	75%	Actual			75%	Actual			75%	Actual		
DESCRIPTION ACCOUNT NUMBER	BUDGET	Spend	Variance	EXPLANATION	BUDGET	Spend	Variance	EXPLANATION	BUDGET	Spend	Variance	EXPLANATION

TABLE GAMES

SUPPLIES & MATERIALS	1205050-000-700001-000	406	675	-269	Ordered four layouts as	406		406		406	135	271	
CARD EXPENSE	1205050-000-700011-000	4,094	5,722	-1,628	Utilize 144 decks a day and	4,094	5,351	-1,256	Unable to reduce expense	4,094	5,170	-1,076	Unable to reduce expense as cards
DICE EXPENSE	1205050-000-700014-000	464	618	-154	Utilize 17 sleeves of ce ? er	464	577	-113	Unable to reduce expense	464	569	-105	Unable to reduce expense as cards
SHIPPING & FREIGHT	1205050-000-700019-000	81	0	1		1		81		81	110	-29	
NEWSP/SUBSCRIPTIONS	1205050-000-700070-000	63	127	-65	Utmzed to help employees		7	-64	Periodicals are ordered	63	127	-64	Periodicals are ordered on a annual
TRANSP. & PER DIEM	1205050-000-701000-000	463	-620	1,005	G2E .ers lg journal entry	463		463		463		463	
PROFESSIONAL FEES	1205050-000-702030-000	469	0	469		465		9		469		469	
REPAIRS/MAINT - EQUIPMI	1205050-000-703100-200	563	166	397		563	166	397		563	166	397	
TRAINING & EDUCATION	1205050-000-705010-000	0	0	0		0		0		0		0	
CUSTOMER INCENTIVE-RA	1205050-000-705111-100	0	0	0		0		0		0		0	
RENTAL USAGE	1205050-000-705202-000	1,250	13,254	-12,004	Shuffler are utilized for time	1,250	13,254	-12,004	Jnable , re-uce as	1,250	10,875	-9,625	Unable to reduce as annual
GAMING EQUIPMENT LEAS	1205050-000-705203-000	20,506	11,569	8,937		20	11,5 2	8,937		20,506	12,841	7,665	
I/T SUPPLIES & MATERIALS	1205050-000-750001-000	94	0	94		4		94		94		94	
I/T - PRINTING	1205050-000-750009-000	63	0	63		63		63				63	
TOTAL		28,514	31,511	-2,997	0	28,514	3. 943	-7 28		28,514	29,993	-1,478	
POKER		÷.							$\boldsymbol{\wedge}$				
SUPPLIES & MATERIALS	1205053-000-700001-000	394	0	394		394		394		394		394	
CARD EXPENSE	1205053-000-700011-000	719	667	52		719	791	-72	Implemented an extra r	719	558	161	
SHIPPING & FREIGHT	1205053-000-700019-000	0	0	0		0		0		0		0	
TRANSP. & PER DIEM	1205053-000-701000-000	84	0	84		84		84		84		84	
CONSULTANT EXPENSE	1205053-000-702100-001	0	0	0		0		0		0		0	
REPAIRS/MAINT - EQUIPME		0	0	0		0		0		0		0	
RENTAL USAGE	1205053-000-705202-000	2,790	3,720	-930	Game protection and increase	2,790		2,790		2,790	3,720	-930	Contractual obligation for the
		4.55								-,	-,.=v		when a son button for the

Page 98 of 340

1													Page 90
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
CASH/OVER SHORT	1205053-000-709004-000	0	-16	16		0	-15	15		0	-15	15	
I/T - PRINTING	1205053-000-750009-000	31	0	31		31		31		31		31	
TOTAL		4,018	4,371	-353	0	4,018	776	3,243		4,018	4,263	-245	
отв													
SUPPLIES & MATERIALS	1205055-000-700001-000	438	51	30				438		438	395	43	
OUTSIDE SERVICES	1205055-000-702010-000	300		300		X		300		300	418	-118	Interface fees based on amount of wagers on OTB from Sportstech
REPAIRS/MAINT - EQUIPM	E 1205055-000-703100-200	344		344				344		344		344	
PROMOTION/ITEMS	1205055-000-705100-000	63		63		63		63		63		63	
PROMOTIONAL COUPONS	1205055-000-705101-100	0		0		0		0		0		0	
CUSTOMER INCENTIVE	1205055-000-705111-000	86		86				0		86		86	
RENTAL USAGE	1205055-000-705202-000	875		875		87	290	585		875		875	
I/T - PRINTING	1205055-000-750009-000	31		31		31		31		31		31	
OTB OPERATING FEE	1205055-301-702121-000	11,331		11,331		,331	15,277	-3,946	This is band of bets the	11,331	9,653	1,678	
CASH/OVER SHORT	1205055-301-709004-000	0	-16	16		0		3			-4	4	
FOTAL		13,468	35	13,433	0	13,468	5,565	-2,0	0	13,468		3,006	
SLOTS													
SUPPLIES & MATERIALS			2,779	1,409		4,188	1,068	3,11		4,188	107	4,081	
TICKET PAPER	1205060-000-700002-000		0	11,266		11,266	12,888	-1.622	N ket paper asage is	11,266	25,776		Ticket paper usage is based on
SHIPPING & FREIGHT	1205060-000-700019-000		2,856	-1,419	No control over this cost.	1,438	395	1,043		1,438	1,360	77	
NEWSP/SUBSCRIPTIONS			0	6		6		6		6		6	
TRANSP. & PER DIEM	1205060-000-701000-000		0	982		982		982		982		982	
TRAVEL - AIRFARE	1205060-000-701000-100	0		0		0		0		0		0	
TRAVEL - MEALS	1205060-000-701000-300	0		0		0		0		0		0	
TRAVEL - LODGING	1205060-000-701000-300	0	170	-170	Budgeted in TRANSP. & PER	0		0		0		0	

Page 99 of 340

							and the second se	All Company of the local division of the loc		and the second se	and the second se		Pade 99
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Oct-14 Actual Spend	Oct-14 Variance	EXPLANATION	Nov-14 75% BUDGET	Nov-14 Actual Spend	Nov-14 Variance	EXPLANATION	Dec-14 75% BUDGET	Dec-14 Actual Spend	Dec-14 Variance	EXPLANATION
TRAVEL - TOLLS/PARKIN	1205060-000-701000-430	0		0		0		0		0		0	
FRAVEL - TAXIS/SHUTTLI	1205060-000-701000-500	0		0		0		0		0		0	
FRAVEL - REGISTRATION	1205060-000-701000-600	0	713	-713	Budgeted in TRANSP. & PER	0		0		0		0	
MILEAGE EXPENSE	1205060-000-701001-000	938	878	60		938	1,001	-63	Slot Supervisors and	938	838	99	
PROFESSIONAL FEES	1205060-000-702030-000	0	0	0		0		0		0		0	
REPAIRS/MAINT - EQUIPM		34,375	17,216	,159		7,375	44,372	-9,997	The ordering of parts is	34,375	43,742	-9,367	The ordering of parts is necessary
REPAIRS/MAINT-SLOT MA		37,500	78,584	41,084	IG1 Conversion for APO-	37		37,500		37,500		37,500	
INIFORMS	1205060-000-705001-000	50	50			50		50		50	32	18	
CUSTOMER INCENTIVE-R		0		0		0	\mathbf{X}			0		0	
RENTAL USAGE	1205060-000-705202-000	625	0	625		625	395	230		625	395	230	
NSURANCE	1205060-000-705500-000	86	162	-76	Based on Risk Magnent	86	162	.75	Sub, itted by Risk Mana	86	162	-75	Submitted by Risk Management to
/T - PRINTING	1205060-000-750009-000	1,313	0	1,313		1,3		,313		1,313		1,313	
/T REPAIRS/MAINT - VEH		188	467	-280	Higher than anticipated	188		188		188	596	-409	
VEHICLE EXP - GAS & OII		563	675	-113	Gas, oil changes and etc. are		62	-59	S. a Super sors ad	563	478	85	
GAMING EQUIPMENT LEA		96,875	132,157	-35 282	Based on Slot Play	\$75	30,607	-33.728	This cost ased	875	127,243	-30,368	This cost is based in part on
GAMING EQUIPMENT LEA		43,750	36,062	7,688		43,750	1 .023	3,727		47 50	37,745	6,005	
GAMING EQUIPMENT LEA			83,076	-26.826	Based on Slot Play	56,250	2 131		A small percentage	56,250	73,596		A small percentage of our gaming
RENT PAID TO SEVEN GEI		31,300	41,733	-10,433	Due to contractual terms for O	31,300	41,733	-10,433	Luse parto Seven Gen	31,300	41,733	-10,433	Lease paid to Seven Generation for the Travel Center
GAMING EQUIPMENT LEA		10,238	7,842	2,396		10,238	12,096	-1,85	A sm. percentage of	10,238	8,746	1,491	
GAMING EQUIPMENT LEA			13,950	-3.713	Based on Slot Play	10,238	13,500	-3,263	is cost is fused in	10,238	13,500	-3,263	This cost is based in part on
CASH/OVER SHORT	1205060-305-709004-000		0	0		0		0		0		0	
/T RENT EXPENSE	1205060-305-755201-000	6 134	8,178	-2,045	Due to contractual terms for H	6,134	8,178	-2,045	Packerland One Stop leas	6,134	8,178	-2,045	Packerland One Stop lease agreement
GAMING EQUIPMENT LEA	1205060-321-705203-000	0	13,950	-13,950	Based on Slot Play	0	6,300	-6,300	A small percentage of	0	6,300	-6,300	
I/T RENT EXPENSE TOTAL	1205060-321-755201-000	0 348,299	8,178 449,675	-8,178 -101,376	Due to contractual terms for H	0 348,299	7,081 404,547	-7,081 -56,248	Highway 54 lease agreen 0	0 348,299	7,081 397,608	-7,081 -49,309	0

													Page 100	of 340
DESCRIPTION	ACCOUNT NUMBER	Oct-14 75% BUDGET	Actual	Oct-14 Variance	EXPLANATION	75%	Actual		EXPLANATION	Dec-14 75% BUDGET	Actual	Dec-14 Variance	EXPLANATION	
TOTAL		1,637,413	1,826,258	-188,845		1,663,274	1,910,812	-247,537		1,309,289	1,562,822	-264,019		



Page 101 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: 🗵 Open 🛛 Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one): Report
	Agenda item title (see instructions):
	DPW continuing resolution closeout port
	Action requested (choose one)
	Action please escribe:
3.	Justification
	Why BC action is required seconstructions):
	required reporting
4	Supporting Materials
4.	Supporting Materials Instructions Instructions
	☑ Report □ Resolution □ Contract (check ine box below if signature required)
	☐ Other - please list (Note: multi-media presentation, due to Trib) Clerk 2 days prior to meeting)
	1 3
	2. 4.
	Business Committee signature required
5.	Submission Authorization
	Authorized sponsor (choose one):
	Requestor (if different from above): Bruce Danforth, Asst. Division Director/Development Operations
	Name, Title / Dept. or Tribal Member
	Additional signature (as needed): Name, Title / Dept.
	Additional signature (as needed):
	Name, Title / Dept.

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Interoffice

MEMORANDUM

To:Oneida Business CommitteeFrom:Bruce Danforth, and istant Division Director Development OperationsDate:February 6, 2014Subject:Close Out Reporting for continuing Resolution 09-24-14-A

Oversitht Reporting

DPW has maintrared, nonthly spread heets for exclusioness unit in order to monitor and assure compliance with the Continuing Resolution 092.7-14-2

For the month of December, the verall DPW operating expresses were \$924,598 compared to the FY14 December budge of \$821,284. This resulted in a variance of \$103,314.

The following Business Units were able to achieve compliance with the 25% reduction in expenses as required in the Continuing Restation 09-27-12-A for Lecember.

Plumbing Custodial Automotive

The following Business Units were unable to achieve compliance with the 25% reduction in expenses as required in the Continuing Resolution 2-27-14-Ac. December.

BIA Roads Maintenance was \$66,542 over the targeted 21% reduction Administration was \$16,057 over the targeted 25% reduction Facilities was \$29,947 over the targeted 25% reduction Groundskeeping was \$8,664 over the targeted 25% reduction

The business units were over due to the following main reasons:

Bulk Rock Road salt (1,244.81 tons) was purchased in December Custodial Armark invoices were applied to Admin's Outside Services line in error. On 1/22/15 expenses were transferred to Custodial Fleet vehicle repairs for v#9451 C.H.R. vehicle was involved in an accident with a deer. 1/15/15 received reimbursement Annual alarm monitoring invoices paid in month of December Emergency Plumbing Repairs (Internal expense) Overtime for snow removal Building demolition expenses Equipment repairs on loader and payloader Vehicle repairs to Groundskeeping Truck# 9400 and Truck# 9439

Operational Impacts of Resolution

Expenses:

Operational expenses such as supplies & materials, repairs & maintenance, and training were kept at a minimum. This resulted in some work requests being put on hold because they did not meet the health, safety, or regreative requirements. These work requests will be reprioritized among all outstanding work requests for future completion. The priority during this period was to address preventative maintenance and emergency repairs keeping expenses at a minimum.

Hiring Freeze and Overtime:

Vacant positions even not filled. Personnel were temporarily reassigned to priority areas and created a backlog of work a some areas. Overtine, was utilized only if absolutely necessary and typically was due of cover employees on vacation, show removal, and emergency calls. In some cases, instead of replacing quipment there were temporar repairs made and now there will be a need to go back and readders there to aporary conditions.

Travel, Contracting, Donation CIP:

There was no impact to DPW due to the constraints.

Recommendations

As a priority, address the factors in the budget process that are barriers to a beving an approved budget before the beginning of the fiscal year. This will eliminate the need for a continuing resolution, the communication necessary to implement it, and the operational impacts that result from it.

If reports are required, there should be a defined purpose for aporting, a process for evaluating and consolidating the information from all of the various reports and a communication mechanism put in place to inform the business units how effective the restruction has been on a monthly basis. The reports were time-consuming and it was unclear what purpose they served.

Department of Public Works December Oversight Report

		2014	Monthly	Dec-14 75%	Dec-14 Actual	Dec-14
DESCRIPTION 5262*02	ACCOUNT NUMBER	BUDGET	BUDGET	BUDGET	Spend	Variance
BIA Roads Maintenance						
TOTAL		70,000	5,833	4,375	70,917	-66,542
		2014	Monthly	Dec-14	Dec-14	Dec-14
DESCRIPTION 1207020	ACCOUNT NUMBER	BUDGET	BUDGET	75% BUDGET	Actual Spend	Variance
Plumbing TOTAL		763,661	63,638	58,787	54,989	3,798
		2014	Monthly	Dec-14	Dec-14	Dec-14
DESCRIPTION 4201010	ACCOUNTIONPLA	BUDGET	BUDGET	75% BUDGET	Actual Spend	Variance
DPW Administration	. 🔨					
TOTAL		721 1	60,129	59,271	75,328	-16,057
		26-4	Monthly	Dec-14	Dec-14	Dec-14
DESCRIPTION	ACC UNT JMBER	UDGET	BUDC.	75% BUDGET	Actual Spend	Variance
4201017 DPW Custodial				·		
TOTAL		3,434,147	286,179	273 36	260,678	13,258
TOTAL		3,434,147	200,17	2/ 10	200,078	13,238
		2014	Monthly	De 14 7 %	Dec-14 Actual	Dec-14
DESCRIPTION 4201040	ACCOUNT NUMB	BUDG	BUDGET	ВЦ ∠Т	эрен	Variance
DPW Facilities						
TOTAL		520,517	3,376 <mark>ک</mark>	276,893	306 11	-29,947
		201	Monthly	Dec-14	Dec-14	Dec-14
DESCRIPTION	ACCOUNT NUMBER	BUDGET	BUD LT	BUDG_T	Actual Spend	Variance
4201050 DPW Groundskeeping						
TOTAL		1,231,001	102,583	886, 9	1 6,652	-8,664
		2014	Monthly	De. 14 75%	Dec-14 Actual	Dec-14
DESCRIPTION 4201060	ACCOUNT NUMBER	BUDGET	BUDGET	BUDGET	Spend	Variance
DPW Automotive		•				
TOTAL		675,146	56,262	50,034	49,194	840
					ACTUAL	VARIANCE
	TOTAL DPW			821,284	924,598	-103,314

Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: 🛛 Open 🛛 Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one): Report
	Agenda item title (see instructions):
	Accept Final Audit Memo and Delete the continuing resolution closeout reports
	Action requested (choose one)
	 ☐ Information only ⊠ Action - please describe:
	Action - please escribe:
	1. Acceptions memory adum and fine (report as information.
	2. Delete the continuing Resolution Reports" section from the agenda.
3.	Justification
	Why BC action is required seconstructions):
	BC Resolution 09-24-14-B required continuing resolution excersion reports and BC requested final
	closeout reports.
4.	Supporting Materials Instructions
	☑ Report □ Resolution □ Contract (check the box below if signature required)
	□ Other - please list (Note: multi-media presentation, due to Trib) Clerk 2 days prior to meeting)
	13
	2. 4.
	Business Committee signature required
5.	Submission Authorization
	Authorized sponsor (choose one): Lisa Summers, Tribal Secretary
	Requestor (if different from above):
	Name, Title / Dept. or Tribal Member
	Additional signature (as needed): Name, Title / Dept.
	Additional signature (as needed):
	Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

INTEROFFICE MEMORANDUM

TO: **ONEIDA BUSINESS COMMITTEE**

FROM: LISA SUMMERS, TRIBAL SECRETARY

SUBJECT: FINAL CONTINUING RESOLUTION CLOSEOUT REPORTS

DATE: MARCH 3, 2015

Background

BC Resolution 09-24-14-B required exceptions to the resolution be reported to the Business Committee. With the approver of the FY-2015 Budget on January 19, 2015, final continuing resolution reports needed to submitted.

On the January 28, 2015, the Pasin as Committee directed all Direct Reports and Non-Di isional ar as, including the Junician and all Boards, Committees and Commissions, to sub ait final closeout reports. They final reports were to include the following: 1. The granation impacts of the continuing solution,

- 2. Any information pertaining to the exposition, that were made, and
- 3. Any recommendations moving for ward in how to improve the continuing resolution process.

Attached is a final report nation analysis the reports which were submitted.

Requested Action

- 1. Accept this memorandum approval report as information
- 2. Delete the "Continuing Resolution Reports" section from the genda.

Page 107 of 340 <u>Continuing resolution closeout reports - 2015</u>

Area	Name	Date report on BC agenda				
	AJRCCC	no report submitted				
	Arts Board	no report submitted				
	Child Protective Board	2/11/2014				
	ERB Board	2/11/2014				
Appointed BCC	Library Board	not notified				
Appointed BCC	ONVAC	2/11/2014				
	Personnel Commission	2/25/2014				
	Police Commission	no report submitted				
	Pow Wow Committee	no report submitted				
	SEOTS	no report submitted				
	Compressive Health	2/11/2014				
	De ele ment	no report submitted				
	development rations	3/11/2014				
	Enconmercal Hearth & Safety	2/11/2014				
<u>Divisio</u>	Gamin	3/11/2014				
	Governmer al Services	no report submitted				
	Internal Servic	2/11/2014				
	and Management	no report submitted				
	Retail Paterprise	2/25/2014				
	Election Board	no report submitted				
	Saming Commission	3/11/2014				
	Lau Claims Cor mission	no report submitted				
Elected BCC	Lans Commission	no report submitted				
	ONCOA	2/11/2014				
	School Board	no to prt submitted				
	Trust/Enrollment	2/ 5/2014				
	Appeals	ns eport submitted				
Judicial System	Family Court	2/11/2014				
	Judiciary	no report submitted				
	Chief Counsel	not notified				
	Emergency Management	no report submitted				
	Finance	no report submitted				
	Housing Authority	o report submitted				
	HRD	2/25/2014				
Other Direct Reports	Intergovernmental Affairs and Communications	no report submitted				
	Internal Audit	2/11/2014				
	Ombudsman	no report submitted				
	Organiztional Development	2/25/2014				
	Records	no report submitted				
	Self-Governance	no report submitted				
	Utilities	not notified				
Tota		17				

Page 108 of 340 **Oneida Business Committee Meeting** Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: Open Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one): Legislative Operating Committee
	Agenda item title (see instructions): Accept February 18, 2015 LOC Meeting Minutes
	Action requested (choose one)
	Action - please, escribe:
	Accept LOC merangeninutes
3.	Justification
	Why BC action is required seconstructions):
4.	Supporting Materials
	Memo of explanation with required information (see instructions)
	 Report Resolution Contract (check the box below if signature required) Other - please list (Note: multi-media presentation, due to Trick) Clerk 2 days prior to meeting)
	1. Minutes 3.
	2. 4.
	Business Committee signature required
5.	Submission Authorization
	Authorized sponsor (choose one): Brandon Stevens, Council Member
	Requestor (if different from above):
	Name, Title / Dept. or Tribal Member Additional signature (as needed):
	Name, Title / Dept.
	Additional signature (as needed): Name, Title / Dept.

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org
Oneida Tribe of Indians of Wisconsin

Legislative Reference Office P.O. Box 365 Oneida, WI 54155 (920) 869-4376 (800) 236-2214 http://oneida-nsn.gov/LOC



Committee Members

Brandon Stevens, Chairperson Tehassi Hill, Vice Chairperson Fawn Billie, Councilmember Jennifer Webster, Councilmember

LEGISLATIVE OPERATING COMMITTEE MEETING MINUTES

Business Committee Conference Room-2nd Floor Norbert Hill Center February 18, 2015 9:00 a.m.

PRESENT: Fawn Billie, Tehe st J.Ill, Brendon Stevens

EXCUSED Jennier Webster

OTHERS PRESELT: Candice Skenandor, Denelle Wilson, Taniquelle Thurner, Fawn Cottrell, Rae Skenandor, Lynn Franzmeier, Phil Wisners, Michelle Mays

I. Call To crock and Approval of the Agenca

Brandon Stevers crite the February 8, 2015 Legislative Operating Committee meeting to order at 9:00 a.m.

Motion by Tehassi and to approve the agenda, secondar by Fawn Billie. Motion carried unanimously.

II. Minutes to be approved

1. February 4, 2015 LOC Meeting Muutes

Motion by Tehassi Hill to approve the Foruary 4, 2015 LOC Meeting Minutes; seconded by Fawn Billie. Motion carried unanimously.

III. Current Business

1. Investigative Leave Policy Amendments (256-04-28)

Motion by Tehassi Hill to defer the Investigative Leav Policy Amendments for a legislative analysis and a fiscal impact statement; seconded by Fact Billie. Motion carried unanimously.

2. Removal Law Amendments (04:39–10:56)

Motion by Fawn Billie defer the Removal Law Amendments for a legislative analysis and a fiscal impact statement; seconded by Tehassi Hill. Motion carried unanimously.

3. Environmental, Health and Safety Law (11:00 –13:56)

Motion by Tehassi Hill to defer the Environmental, Health and Safety Law to the sponsor's office for further review and re-drafting; seconded by Fawn Billie. Motion carried unanimously.

IV. New Submissions

1. Administrative Procedures Act Emergency Amendments (14:00–19:46)

Motion by Tehassi Hill to approve the resolution and forward to the Oneida Business Committee for consideration of adoption; seconded by Fawn Billie. Motion carried unanimously.

2. Appeals Commission References Removal (19:53 – 22:34)

Motion by Tehassi Hill to approve the resolution and forward to the Oneida Business Committee for consideration of adoption; seconded by Fawn Billie. Motion carried unanimously.

V. Additions

VI. Administrative Updates

 LOC Sponsor List 22: 0 –29:00) Notion by Tehassi Hill to acc pt the LOC Sponsor List as FYI; seconded by Fawn Billie. Motion cirried unanimously

VII. Executive Sec

VIII. Recess/Adjour Motion by Favn Julii to adjourn the February 18, 2015 Legislative Operating Committee Meeting at 9:29 c.m.; seconded by Tehassi Fill. Motion Corried unanimously.

Page 111 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15		
2.	Nature of request Session: ⊠ Open □ Executive - justification required. See instructions for the applicable laws that		
	define what is considered "executive" information, then choose from the list:		
	Agenda Header (choose one): Finance Committee		
	Agenda item title (see instructions):		
	FC Meeting Minutes of 03/02/15 & FC E-Poll Approving Minutes		
	Action requested (choose one)		
	Action - please escribe:		
	OBC Approval committee		
3.	Justification		
	Why BC action is required seconstructions):		
	The Finance Committee is Standby Committee (the ODC		
	The Finance Committee is Standing Committee of the OBC		
Л	4. Supporting Materials		
4.	■ Memo of explanation with required information (see instructions)		
	□ Report □ Resolution □ Contract (check the box below if signature required)		
	☑ Other - please list (Note: multi-media presentation, due to Trik Clerk 2 days prior to meeting)		
	1. FC Meeting Minutes of 03/02/15 3.		
	2. FC E-Poll Approving 03/02/15 Minutes 4.		
	Business Committee signature required		
5.	Submission Authorization		
	Authorized sponsor (choose one): Trish King, Tribal Treasurer		
	Requestor (if different from above):		
	Name, Title / Dept. or Tribal Member Additional signature (as needed):		
	Name, Title / Dept.		
	Additional signature (as needed): Name, Title / Dept.		

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org



Oneida Tribe of Indians of Wisconsin

Finance Administration Office Phone: 920- 869-4325 FAO@oneidanation.org

MEMORANDUM

- **TO:** Finance Committee
- CC: Business Committee
- FR: Denise Vigue, Executive Assistant
- **DT:** March 3, 2015
- RE: <u>*d*-Poll Results of *Pc* Veeting Minutes of March 2, 2015</u>

An E-Poll vote of the Jonance Committee was conducted to approve the March 2, 2015 Juna ce Committee meeting minutes. The results of the completed E-Pollius as follows:

<u>E-POLL RESULTS</u>:

There was a Majority <u>5 YES</u> vers from Larry Barto, Jonnifer Webster, Chad Fuss, Fawn Billie and Wesley Jarth, Jroto approve the Marco 2, 2015 Finance Committee Meeting Minutes.

The minutes will be placed on the next of agende of March 11, 2015 for approval and the next Finance Committee agenda of March 16, 2015 to accept this E-Poll action.

YuwAko



ONEIDA FINANCE COMMITTEE

March 2, 2015 Business Committee Executive Conference Room

REGULAR MEETING MINUTES

Members Present:

Trish King, Treasurer/FC Chair Jenny Webster, BC Council Member Chad Fuss, Gaming AGGM Larry Barton, CFO/FC Vice-Chair Fawn Billie, BC Council Member Wesley Martin, Jr., Community Elder Member

Members Excused: Patrick Stensloff, Purchasing Director

Others Present: Jeff Carlson, Elizabeth Somers, Mike Finn, Lloyd E. Powless, Jr., David Emerson, Sam VanDen Heuvel, Paul Witek and Denise Vigue, FC Recording Secretary

- I. Call to Order: The Finance of mmittee meeting was called to order by the FC Chair at 10:00 a.m.
- II. Apre-vec of the Agenua: Notion by ennifer Webster to approve the February 16, 2015 Finance Committee genda with the colowing charges: move New Business item #8 and Community Fund items #8, #5, & #10 right after Tabled susine s. Seconded by Wesley Martin, Jr. Motion carried unpaimously.

III. Approxim on the dinutes:

1. February 16 2015 (pproved via E-P (on CV17/15

Motion by Fawn Billie to raify the E-poll action or Feb. 17, 2010 approving the Finance Committee meeting minutes of Feb. 13, 2015. Seconded by Jennifer Viebster. Motion carried unanimously.

IV. Tabled Business:

1. Jezeski Accounting & Txx Service / Incont: \$433,000.00 David Larson, Community Charles Program, OCHC

Motion by Fawn Billie to remove from the table for discussion. Second select W Martin, Jr. Motion carried unanimously.

Jeff Carlson was present and provided handour of memo from Mr. Larson as we as copy of contract to FC member that asked for it at the previous mating. There was discussion of services provided by vendor; committee discussion of needing to assess all carrent vendors to get as much value added for services and bidding out of services.

Motion by Wesley Martin Jr. to approve noting this to be id or the September and the copy of contact provided is being placed in the file for reference. Seconded by Fawi Billie Hotion carried unanimously.

V. Capital Expenditures:

1. Ainsworth (9) Slot Machines

Amount: \$171,165.

David Emerson, Gaming Slots Director

David Emerson was present to provide details; noting items #1, #2, & #4 under New Business are also capital expenditure-it will be corrected in the minutes. Speaking to all five requests David explained these are slot machines for this year under their allocated budget; some discussion of timing and discounts; inventory sales and difference between these purchases and lease games.

Motion by Wesley Martin Jr. to approve the Gaming Capital Expenditures for items 1-5 for fifty-five slot machines from the vendors identified and in the amounts specified. Seconded by Jennifer Webster. Motion carried unanimously.

		Fage 114 01 540
2. Se	Komani (16) Slot Machines David Emerson, Gaming Slots Director ee Capital Expenditures motion #1.	Amount: \$277,891.80.
3.	Aristocrat (16) Slot Games David Emerson, Gaming Slots Director	Amount: \$285,936.00
Se	e Capital Expenditures motion #1.	
4.	WMS (6) Slot Games David Emerson, Gaming Slots Director	Amount: \$498,520.00.
Se	e Capital Expenditures motion #1.	
5.	Bally (8) Slot Games David Emerson, Gaming Slots Director	Amount: \$181,160.00
Se	e Capital Expenditures motion #1.	
/I.	New Business:	
Α.	Internal Requests:	
1.	Slot Conversions- Arig or at (25)	Amount: \$45,000.00.
	Slownversions- CTEC I (20)	Amount: \$57,600.00
	David Entersions- WM (200) David Enterson, Gaming Sloth Director	Amount: \$90,195.00
	vid Ellerson capie hed these comprisions are rent merculars; the funding comes from their	e for tames being swapped out and new ones placed in cuerating budget.

Motion by Jennife Web te to approve the order for highly flot Conversions from the vendors identified in the amounts identified. Leconded by Larry Barton. Motion carried unanimously.

2. Supply Advance – Jr War - Sommittee Lloyd Powless, Jr., W Committee Chairman

V

Lloyd Powless, Jr. was present to discuss with the Committee me details of the supply advance request for the 4th of July Pow Wow including: bre kould for costs, recruitment of volunteers, payouts, and the suggestion to offer cash cards rather than octual cash to some paid wilker. Lloyd noted that costs for admission and booths will be going up and last year was their best year or review received.

Motion by Wesley Martin, Jr. to approve the supply advance for the Pow Wow Commilee in the amount of \$96,950.000. Seconded by Larry Barton. Maion carried manimously.

3. Howard Immel Contract-Park Upgrade Physe

Sam VanDen Heuvel, Project Manager, Engine ting

Paul Witek and Sam VanDen Heuvel were present to discuss an obased of the park(s) upgrade; this is the design work; next phase will be to bid out for actual upgrade work; included in the submitted packet was the bidder rating sheet as well as the threline with veral costs of entire project; to avoid change orders some areas specified may have a wai. After further explaining the bidding of this phase it was recommended all bid information is placed on cover request sheet.

Motion by Larry Barton to approve the Howard Immel contract in the and of \$69,800.00 for the design work for the Park(s) Upgrades. Seconded by Fawn Billie. Motion carried unanimously.

4. Northeast Asphalt – Change Order #1

Michael Finn, TTP

Michael Finn was present to provide additional information on the Hillside project which has been completed. The change order is for the soft spots that needed additional fill, rock and labor; entire project including change order is covered under the Roads Program and is under budget

Motion by Jennifer Webster to approve Change Order #1 with Northeast Asphalt in the amount of \$3,018.01. Seconded by Chad Fuss. Motion carried unanimously.

5. Home Instead Senior Care David Larson, COPS - OCHC Amount: \$105,736.00

Item reviewed directly after Tabled Business. Jeff Carlson provided details of request: their current vendor has discontinued their services; this vendor will replace those services. The Committee members had some concerns regarding comments on legal review as well as if a request for a waiver of sovereign immunity to the BC is needed; Committee suggests backup on updated legal review and an explanation to address the waiver issue be forwarded to FC recording secretary so an E-Poll of the FC can be conducted.

Motion by Larry Barton to defer until additional information as discussed is received and if time sensitive an E-Poll of the Committee be conducted for approval. Seconded by Wesley Martin, Jr. Motion carried unanimously.

- B. External Donation Reques
- 1. Oneida Sobriety Group String Conference Reconstor: Marilyn King

Westey Martin Jr. excused his cell from this discussion. The Committee discussed the merits of the domation request.

Motion 22 arry Practic to approve from the bipance Committee donation line \$5,000.00 to be used for the Spring Converse of the Oneida Schriety Group. Seconded by Jennifer Webster. Motion carried unanimously.

2. Oneida Nation Conchouse - Cookhouse Repairs Requestor: Apache Lapforth

The requestor was unable to at and; Committee had questions of the costs of labor and if being done by tribal employees the tax implications as yell as funds possibly received from another tribal program(s).

Motion by Larry Barton to approve from the finance Committee denation line \$5,000.00 and stipulate in the check letter that none of the funds of the euse for labor costs. Second d by Fawn Billie. – After further discussion motioner and second without without motion.

Motion by Jennifer Webster to defer for two weres for clarification of request and to determine if money for this project has been received by any other of the Triber Program. Seconded b Wesley Martin, Jr. Motion carried unanimously.

3. Visions-Independent Film Project Requestor: Audrey Geyer

There was discussion about the merits of the proposed project and there other more appropriate areas of the Tribe that can provide input to gain a definitive response possible consideration.

Motion by Larry Barton to defer for two weeks and forward to Tourism and Joint Marketing for consideration and/or input. Seconded by Fawn Billie. Motion carried a pair ously.

VII. Executive Session: No requests submitted

VIII. Community Fund:

1. Oneida Nation High School – Graduation

Motion by Fawn Billie to approve from the Community Fund 15 Cases of Coca-Cola Product for refreshments to be used at the Oneida Nation High School's Graduation Reception. Seconded by Larry Barton. Motion carried unanimously.

2. Oneida Nation High School – Prom

Motion by Larry Barton to approve from the Community Fund 15 Cases of Coca-Cola Product for refreshments to be used at the Oneida Nation High School's Prom. Seconded by Fawn Billie. Motion carried unanimously.

3. Oneida Relay-for-Life

Motion by Wesley Martin, Jr. to approve from the Community Fund 20 Cases of Coca-Cola Product to be used at the Oneida Relay-for-Life Community Event in July 2015. Seconded by Larry Barton. Motion carried unanimously.

4. Oneida Fourth of July Powwow

Motion by Jennifer Webster to approve from the Community Fund 100 Cases of Coca-Cola Product (water) to be used as part of the fundraising activities of the Powwow Committee at the Oneida Fourth of July Powwow. Seconded by Fawn Billie. Motion carried unanimously.

5. UWGB Rock Academy / a mer camp) for son

Motion by Jennifer Webster to a prove from the Community Fund the UWGB Summer Camp Rock Academy feet for son of requisiter and a amount of \$469.00. Seconded by Larry Barton. Motion carried unanimously.

6. Nine nsin Pitz ard participation feet for so

Motion by Jawn Bill's to approve from the Community Lond the Wisconsin Blizzard Basketball participation fees of some requestor in the aboun of \$450.00. Seconded by Jennifer Webster. Motion carried unanimously.

7. YMCA Camp U-Nat Lina for daughter

Motion by Larry Barton to approve from the Community, Fund the MCA Camp U-Hah-Li-Ya fees for daughter of requestor in the a nount of \$-81,50. Seconded by Fresler Frankrish, Jr. Motion carried unanimously.

8. Oneida Cultural Enrichment Tear

Item reviewed directly after Tabled Busicess. Motion by Larry Barton to approve from the Community Fund the Oneida Cultural Enrichment Tour registration fees for this and the following two requests. Seconded by Jennifer Webster. Motion carried manimourly.

9. Oneida Cultural Enrichment Tour (son of recessor)

See item #8 for motion

10. Oneida Cultural Enrichment Tour (daughter of required) See item #8 for motion

11. Freedom High School Band Trip to NY

Motion by Jennifer Webster to approve from the Community Fund the Freedor. High School Band Trip to NY fees for daughter of requestor in the amount of \$500.00. Seconded by Fawn Billie. Motion carried unanimously.

IX. Follow Up: No follow up

X. For Your Information:

1. American Gaming Systems (AGS LLC) Lease Game David Emerson, Gaming Slots Director

2. GTECH USA LLC (3) Lease Games @WMSC David Emerson, Gaming Slots Director

Motion by Larry Barton to accept as FYI items 1 & 2. Seconded by Jennifer Webster. Motion carried unanimously.

XI. Adjourn: Motion by Jennifer Webster to adjourn. Seconded by Fawn Billie. Motion carried unanimously. Meeting ended at 11:25 a.m. The next Finance Committee meeting is scheduled for Monday, March 16, 2015 at 10:00 a.m. in the BC-Executive Conference Room.

Minutes taken and transcribed by: Denise Vigue, Executive Assistant in Finance & Finance Committee Recording Secretary

Finance Committee E-Poll Minutes Approval Date: <u>March 3, 2015</u>



Page 118 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	. Meeting Date Requested: 03 / 1	1 / 15	
2 .	Nature of request		
	Session: 🗵 Open 🔲 Executive - justification required. See instructions for the applicable laws that		
		s considered "executive" information, then choose from the list:	
	Anthem Blue Cross Blue Shield Contract 2014-1170		
	Agenda Header (choose one): New Business/Request		
	Agenda item title (see instructions):		
	Agenda item ittle (see instructions). Approve Anthem Blue Cross Blue Skield Contract 2014-1170		
	Action requested (choose one)		
	☑ Action - please escribe:		
	Approval contra wit Anthem Blu	e cross and Poor Shield	
3.	Justification		
	Why BC action is required user instr	uctions):	
	Law Office indicates the OBC needs to	approve the contract	
4.	4. Supporting Materials		
	☐ Memo of explanation with require	ed information (see instructions)	
	Report Resolution	Contract (check we box below if signature required)	
	Other - please list (Note: multi-me	dia presentation due to Tribel Clerk 2 days prior to meeting)	
	1.	3.	
	2.	4.	
_	Business Committee signature required		
5.			
	Authorized sponsor (choose one): De	obie Danforth, Division Director/Operations	
		frey R Carlson	
		me, Title / Dept. or Tribal Member	
	Additional signature (as needed): Na	me, Title / Dept.	
	Additional signature (as needed):		
	Na	me, Title / Dept.	

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

ONEIDA LAW OFFICE

CONFIDENTIAL: ATTORNEY/CLIENT WORK PRODUCT

TO: Maria J. Doxtator-Alfaro Oneida Health Center Use this number on future correspondence:

2014-1170

Purchasing Department Use

Contract Approved Contract Not Approved (see attached explanation)

FROM: Kelly M. McAndrews, Staff Attorney KMM HUW

DATE: March 3, 2015

RE: Anthem Blue Cross and Blue Shield-Provider Agreement - Revised

If you have any questions or comments regarding this review, please call 869-4327.

The attached agreement, contract policy and/or guaranty has been reviewed by the Oneida Law Office or legal content only. Please note the Allowing:

- ✓ The docurrent in appropriate legal form *excrecution is a management decision.*)
- Requires Business committee approved price to execution, pursuant to the Tribe's Sovereign Lamonity Law

Comments: I continue to moment to HIS Model OuP Adder turn be signed with this agreement, as it resolves some assues (see previous review). Begardings, Business Committee approval is necessary as the Agreement contrars a mandaton arbit a on clause (subject to JAMS- Judicial Arbitration and Mailation Services), requires compliance with the managed care organization contract and incorporates the disuments between the F ar and the State of Wisconsin (in the Wisconsin Medicaid Participation Attachment).

ONEIDA LAW OFFICE

CONFIDENTIAL: ATTORNEY/CLIENT WORK PRODUCT

 TO:
 Maria J. Doxtator-Alfaro Oneida Health Center
 Use this number on future correspondence:

 FROM: Kelly M. McAndrews, Staff Attorney
 2014-1170

 DATE: October 17, 2014
 Purchasing Department Use Contract Approved Contract Approved (see attached exploanation)

If you have any designs or comments regarding this review, please call 869-4327.

The stached greement, contact, policy and/or guaranty has been reviewed by the Oneida Law Of ice for legil content only. Please not the following:

This compact may of impacted by K solution #BC 22-24-2014-A entitled Continuing Resolution for FY 2014. Shis contract should be a viewed for compliance with this resolution prior to signing.

 \checkmark Not in appropriate legation.

First, I recommend to Model QHP Addendum for adian Heran Care Providers be incorporated into the Agreement are signed by the Parties to clariful issues unique to IHS. The IHS addendum is required to be incorporated by reference a a cord with Sec. 9.5 and can be incorporated into Sec. 1.6 (sample: This Agreement facture ng it has incorporated herein by reference) and the Model QHP Addendum for Indian Healt, Care Providers constitutes the entire understanding...).

The Model QHP clarifies the following is also, minimally, in the agreement A ND INCORPORATED PROVIDER MANUAL (incorporated at Sec. 2.10 of the ement, subject to change on a yearly basis without approval hop provider, there is a 186 page manual posted which is effective December 15, 2014) :

- o Persons eligible for service (agreement- Arter II, Sc. 2.1 and 2.16)
- Dispute Resolution and Arbitration (p. 9 provider manual article VII Agreement)
- o Insurance Requirements (p. 11, provider manual
 - o Claims Requirements (p. 27, 43 provider manual, agreement- Acicle II, Sec 2.5)
 - o Governing law (agreement- Article IX, Sec. 9.8)

The current Agreement and Provider Manuel do not recognize the uniqueness of IHS related services and the Model QHP clarifies issues unique to Indian Health Care Providers. I have previously provided the Model QHP.

Second, please note this Agreement incorporates the Wisconsin Medicaid Participation Attachment to the Blue Cross Blue Shield Provider Agreement which states, "Provider agrees not to bill Medicaid Covered Individuals for missed appointments while enrolled in the Medicaid programs." P. 25. I mention this because there has been conversations about billing for missed appointments, which is not prohibited by many agreements, but is prohibited by this document.

Last, after it is determined whether or not the above changes will be made and the Model QHP will be signed by the Parties, Business Committee approval is necessary as the attachments incorporate Wisconsin law and incorporate applicable state contracts, which appear to waive of sovereign immunity.

ANTHEM BLUE CROSS AND BLUE SHIELD PROVIDER AGREEMENT

WITH



ANTHEM BLUE CROSS AND BLUE SHIELD PROVIDER AGREEMENT

This Provider Agreement (hereinafter "Agreement") is made and entered into by and between Blue Cross Blue Shield of Wisconsin doing business as Anthem Blue Cross and Blue Shield (hereinafter "Anthem") and **Oneida Community Health Center and Ka Ni Kuhli Yo Family Center** (hereinafter "Provider"). In consideration of the mutual promises and covenants herein contained, the sufficiency of which is acknowledged by the parties, the parties agree as follows:

ARTICLE I DEFINITIONS

"Affiliate" means any entity owned or controlled, either directly or through a parent or subsidiary entity, by Anthem, or any entity which injunder common control with Anthem and that accesses the rates, terms or conditions of this Agreement Anthem will have a current listing of such Affiliates available through a commonly available website upon request.

amount pat Provider and anther nave preed upon as set forth in the Plan Compensation Schedule ("PCS"), he Anthem Rate shall epres at payment in full to Provider for Covered Services.

"Contitation" reads the amount expre-partment made by Anthem to a provider or management services organization on a per member per hanth basis for the respective services or the total cost of care.

"Case Rate means and all inclusive Antheor Rap for an entire admission or one outpatient encounter. "Global Case Rate means the all inclusive in them hate with the includes facility, professional and physician services for specific Caded Solvice Identifier(s).

"Claim" means either the uniferential claim form or electronic claim form in the format prescribed by Plan submitted by a promer for payment by a Plan for thatth Service's rendered to a Covered Individual. "Complete Claim" means unless state aw otherwise requires, an occurate Claim submitted pursuant to this Agreement, for which all information necessary to process such claim at make a benefit determination is included.

"Coded Service Identifier(s)" means culsticul of descriptive terms and identifying codes, updated from time to time by the Centers for Medicare and Medicaid Solvices ("CMS") or other indicary curce, for reporting Health Services on the CMS 1500 cum form or a successor. The codes of clude but are not limited to, American Medical Association Current Procedural Terminology ("CPT®-44, CMS Heathcare Common Procedure Coding System ("HCPCS"), International Classification of Dispases, 5th Revision, Clinical Modification ("ICD-9-CM"), and National Drug Pode ("NDC" or their successors.

"Cost Share" means, with respect to Covered Service, an amount which a Covered Individual is required to pay under the terms of the applicable Health Benefit Plan. Such hyment may be referred to as an allowance, coinsurance, copayment, deductible, penalty or other covered individual payment responsibility, and may be a fixed amount or a percentage of applicable response for Covered Services rendered to the Covered Individual.

"Covered Individual" means any individual who is eligible, as determined by Pinn, to receive Covered Services under a Health Benefit Plan. For all purposes related to the Agreement including all schedules, attachments, exhibits, manual(s), notices and communications related to this Agreement, the term "Covered Individual" may be used interchangeably with the terms Insured, Cover Person, Member, Enrollee, Subscriber, Dependent Spouse/Domestic Partner, Child or Contract Holder, and the meaning of each is synonymous with any such other.

"Covered Services" means Medically Necessary Health Services, as determined by Plan and described in the applicable Health Benefit Plan, for which a Covered Individual is eligible for coverage. Covered Services do not include the preventable adverse events as set forth in the provider manual(s).

"Emergency Condition" means a medical condition manifesting itself by acute symptoms of sufficient severity (including severe pain) such that a prudent layperson, with an average knowledge of health and medicine, could reasonably expect the absence of immediate medical attention to result in serious jeopardy to the health of the individual, or in the case of a pregnant woman, the health of the woman or her unborn

child; serious impairment to bodily functions; or serious dysfunction of any bodily organ or part. "Emergency Services" means those Covered Services provided in connection with an Emergency Condition.

"Encounter Data" means Claims information submitted by a Provider under capitated or risk-sharing arrangements, for Health Services rendered to Covered Individuals.

"Health Benefit Plan" means the document(s) describing the partially or wholly insured, underwritten, and/or administered, marketed health care benefits, or services program between the Plan and an employer, governmental entity, or other entity or individual.

"Health Service" means those services or supplies that a health care provider is licensed, equipped and staffed to provide and which he/she/it customarily provides to or arranges for individuals.

"Medically Necessary" or "Medical Necessity" means the definition as set forth in the Health Benefit Plan, unless a different definition is the unreaded by statute or regulation.

"Network" means a group of providers that support, through a direct or indirect contractual relationship, and or an and an angle and angle angle and angle angle angle and angle ang

"Network Participating Provider" near a povider designated by Plan to participate in one or more Network(s).

"Other Proors" means persons or entities, utilizing the Network(s)/Plan Program(s) pursuant to an agree net with Anthem or an Affilian, including with at limitation, other Blue Cross and/or Blue Shield Plans that a phot Affilines, and employers or in views providing Health Benefit Plans pursuant to insured, self-administrated or en insured programs.

"Participation and a part of this Agreement which identifies the additional druges and/or obligations related to Nework(s) and/or Plan Program(s).

"Percentage Rate" means the Anthem hate that is expressed as a procentage of allowed Provider Charges.

"Per Diem Rate" means the Anthem Rate of t is expressed at the attributive fixed payment for Covered Services rendered on a single latit of service.

"Per Hour Rate" means the Anthem Rate that is applicable when paymen, in deriver the sed on an increment of time multiplied by the Anthem Rate in the applicable fee schedule.

"Per Unit Rate" means the Anthem Rate that is applicable when payment derived based on a unit of service multiplied by the Anthem Rate in the applicable feer medule(s).

"Per Visit Rate" means the Anthem Rate that is excessed as the all inclusive fixed payment for one outpatient encounter.

"Plan" means Anthem, an Affiliate as designated by Anthem and ar an Other Payor. For purposes of this Agreement, when the term "Plan" applies to an entity other can Anthem, "Plan" shall be construed to only mean such entity.

"Plan Compensation Schedule" ("PCS") means the document(s) and here to, made a part of this Agreement which sets forth the Anthem Rate(s) and compensation relief d terms for the Network(s) in which Provider participates. The PCS may include additional Provider blight is and specific Anthem compensation related terms and requirements.

"Plan Fee Schedule(s)" means the schedule of the maximum amounts that Plan will pay for Covered Services, less Cost Shares if applicable. The Plan Fee Schedule(s) applicable for the Network(s) in which Provider participates is further described in the PCS.

"Plan Program" means any program now or hereafter established, marketed, administered, sold, or sponsored by Plan, or Blue Cross Blue Shield Association ("BCBSA") (and includes the Health Benefit Plans that access, or are issued, or entered into in connection with such program). Plan Program shall include but is not limited to, a health maintenance organization(s), a preferred provider organization(s), a point of service product(s) or program(s), an exclusive provider organization(s), an indemnity product(s) or

program(s), and a quality program(s). The term Plan Program shall not include any program excluded by Plan or BCBSA.

"Provider Charges" means the regular, uniform rate or price Provider determines and submits to Anthem as charges for Health Services provided to Covered Individuals. Such Provider Charges shall be no greater than the rate or price Provider submits to any person or other health care benefit payor for the same Health Services provided, regardless of whether Provider agrees with such person or other payor to accept a different rate or price as payment in full for such services.

ARTICLE II SERVICES/OBLIGATIONS

- 2.1 <u>Covered Individual Identification</u>. Anthem shall ensure that Plan provides a means of identifying Covered Individual either by issuing a poler, plastic, or other identification document to the Covered Individual or by a telephonic, paper or electronic communication to Provider. This identification need not include all information necessary to be the mine Covered Individual's eligibility at the time a Health Service is rendered, but shall include information necessary to contact Plan to determine Covered Individual's participation and ne applicable Health Belefit and Provider acknowledges and agrees that possession of such identification document of country to access bigibility information telephonically or electronically, in and of itself, does not qualify the holder there as a Covered Individual, nor does the lack thereof mean that the person is not a Covered Individual.
- 2.2 <u>Provides on-discrimination</u>. Provider shall provide realth Services to Covered Individuals in a manner similar to and vithin the same time a allabilitient whice Provider provides Health Services to any other individual. Provider variable differentiate, or discriminate against any Covered Individual as a result of his/her enrollment in a Plot, or because of race, cour, creationational origin, ancestry, religion, sex, marital status, age, disability, mymear source, state of health need for health services, status as a litigant, status as a Medicare or Medicaid lengt dary, sexual orientation, or any other basis prohibited by law. Provider shall not be required to provide any two counters of Health Service to Covered Individuals that he/she/it does not customarily provide to there.
- 2.3 <u>Publication and Use of Provider Information</u> Provider agrees that Anthon Plans or its designees may use, publish, disclose, and display, either direct, for through a unit dearty information related to Provider, including but not limited to demographic information, information regaring usedentialing and affiliations, performance data, Anthem Rates, and any other information related to Provider for transparency initiatives, for commercially reasonable general susiness purpores.
- 2.4 <u>Use of Symbols and Marks</u>. Neither pairs to this Agreement shall publish, dopy, reproduce, or use in any way the other party's symbols, service marks) or trademarks) without the vior water consent of such other party. Notwithstanding the foregoing, the parties agree that they may iden by Provider as a participant in the Network(s) in which he/she/it participates.
- 2.5 <u>Submission and Payment of Claims</u>. Unless otherwise instructed, our equired by state or federal law, Provider shall submit Claims to Plan, using appropriate and current Code Service Identifier(s), within one hundred eighty (180) days from the date the Health Service are to dered or Plan will refuse payment. If Plan is the secondary payor, the one hundred eighty (180) day period will not begin until Provider receives notification of primary payor's responsibility.
 - 2.5.1 Provider agrees to provide to Anthem, unless otherwise instructed, at no post to Anthem, Plan or the Covered Individual, all information necessary for Plan to different payment liability. Such information includes, without limitation, accurate and Complete Clause or Covered Services. Once Anthem determines Plan has any payment liability, all Complete Claims will be paid in accordance with the terms and conditions of a Covered Individual's Health Benefit Plan and the PCS.
 - 2.5.2 Provider agrees to submit Claims in a format consistent with industry standards and acceptable to Plan either (a) electronically or (b) if electronic submission is not available, utilizing paper forms.
 - 2.5.3 If Anthem or Plan asks for additional information so that Plan may process the Claim, Provider must provide that information within sixty (60) days, or before the expiration of the one hundred eighty (180) day period referenced above, whichever is longer.

- 2.5.4 In no event, shall Provider bill, collect, or attempt to collect payment from the Covered Individual for Claims Plan receives after the applicable period(s) as set forth above, regardless of whether Plan pays such Claims.
- 2.5.5 In all events, however, Provider shall only look for payment (except for applicable Cost Share or other obligations of Covered Individuals) from the Plan that provides the Health Benefit Plan for the Covered Individual for Covered Services rendered.
- 2.6 <u>Plan Payment Time Frames</u>. To the extent a Complete Claim is subject to Wis. Stat. § 628.46, such Complete Claim shall be paid within thirty (30) days as provided for under the terms of Sec. 628.46.

2.7 Payment in Full and Hold Harmless.

- 2.7.1 Provider agrees to accept as payment in full, in all circumstances, the applicable Anthem Rate whether such payments in the form of a Cost Share, a payment by Plan, or payment by another source, such as the gh coordination of benefits or subrogation. Provider shall bill, collect, and accept compensation for Cost Shares. Provider agrees to make reasonable efforts to verify Cost Shares prior or billing for success Shares. In no event shall Plan be obligated to pay Provider or ny person acting on tenal of Provider for services that are not Covered Services, or any amounts excess of the Anthem tate loss post Shares or payment by another source, as set forth above. Consistent with the for going Provide agrees to accept the Anthem Rate as payment in full if the Covered Individual has not set subject subject eductible.
- 2.7.2 Soviety agrees that in no event, including belond limited to, nonpayment by applicable Plan, or breach of this greement, or comm provent denois or adjustment requests or recoupments based on a scoding or other billing errors and the wider, built, charge collect a deposit from, seek compensation from or have any oner recourse against a Covered Nevideul, or a person legally acting on the Covered Incident's behalf for Covered Services provided pursuant to this Agreement. This section does not provider from collecting eimbursement for the following from the Covered Individual:

2.7.2.1 Cost Shares, if applicable

- 2.7.2.2 Health Services that an not Covered Services (other nanoveventable adverse events). However, Provider may seek payment for a Hearn Service that is not Medically Necessary or is experimental/involtigational only if Provider Statistics advritten waiver that meets the following o teria:
 - a) The waiver notifies the Covered In vidual that the Rhalth Shrvice is likely to be deemed not Medically Noressary an experimental/investigation at
 - b) The waiver notifies the Cover of Individual of the Health Service being provided and the date(s) of service;
 - c) The waiver notifies the Covered Incidual the approximate cost of the Health Service;
 - d) The waiver is signed by the Covered Individue, or a person legally acting on the Covered Individual's behalf, prior to receipt of the Health Sector.
- 2.7.2.3 Any reduction in or denial of payment as a result of the covered Individual's failure to comply with his/her utilization management program pursuant to his/her Health Benefit Plan, except when Provider has been designated by Anthem to comply with utilization management for the Health Service provided by Provider to the Covered Individual;
- 2.7.2.4 Health Services which are not payable in the Covered Individual's Health Benefit Plan because Provider does not participate in the applicable Plan Program.
- 2.7.3 If subject to §609.92, Wis. Stats., Provider agrees not to opt-out of its obligations thereunder as referenced in Attachment A.

- 2.8 <u>Adjustments for Incorrect Payments</u>. Provider shall refund all duplicate or erroneous Claim payments including but not limited to credit balances, regardless of the cause, including, but not limited to, payments for Claims where the Claim was miscoded or otherwise billed in error, whether or not the billing error was fraudulent or abusive, provided such erroneous payment has been identified by either party and notice of the error has been provided to Provider within one (1) year from the date of payment or explanation of payment, unless otherwise set forth in the provider manual. In lieu of a refund, Plan may offset future Claim payments.
- 2.9 <u>Provider Subcontractors</u>. Provider may fulfill some of his/her/its duties under this Agreement through subcontractors or delegates. Hereinafter, subcontractors and delegates are referred to as "subcontractors". Provider shall assure the compliance of his/her/its subcontractors with the terms and conditions of this Agreement as applicable, including, but not limited to, the Payment in Full and Hold Harmless provisions of section 2.7 hereof. Provider shall be solely responsible to pay subcontractor for any Health Services. If Anthem has a direct contract with the subcontractor ("direct contract"), the direct contract shall prevail over this Agreement.
- 2.10 Compliance with Province Manual(s) and Policies, Programs and Procedures. Provider agrees to abide by, and comply with, Anthem's provider canual(s) and all other policies, programs and procedures (collectively "Policies") established an amplementer of Plan. Anthem or its designees may modify the provider manual(c) and Policies by making a good faith effort to provide notice to Provider at least forty-five (45) days in advance of the effective dats of material module atom thereto.
- 2.11 In Network Referrals and Transfers. Provider shall us best efforts when medically appropriate, to refer and transfer covered Individuals to Netterk/Participating Providers. Additionally, Provider represents and warrants that provider opes not give, provider containe or receive any incentives or kickbacks, monetary or otherwise, in exchange for the referral of a povered addividual and if a Claim for payment is attributable to an instance in youch provider provided or received an incentive or kickback in exchange for the referral, such Claim shall not to pay ole and, if paid in ends, shall be reduced to Anthem.
- 2.12 <u>Programs and Provide Parks</u>. Provider acknowledge and as of the Effective Date, it participates only in those Networks designated on the signifure page. Provider acknowledge that Plan may have, develop, or contract to develop, various networks or programs that have a variety of provider panels, program components and other requirements, and varie Plan may discontinue or modify such networks or programs. In addition to those Network's accignated on the signature page of the Agreement's, Anthem may also identify Provider as a Network/Participating row or in additional Networks and or products designated in writing from time to time by Anthem. The terms and onditions of Provider's participations as set forth in this Agreement unless otherwise agreed to in writing y Provider and Anthem.
 - 2.12.1 Provider further acknowledges and understands that Anthem participate in the Federal Employees Health Benefit Program ("FEHBP") the head r insurance plan for federal employees. Provider further understands and acknowledges that the FEHBP is offederal government program and the requirements of the program are subject to change at the cale crection and discretion of the United States Office of Personnel Management. Provider actives to able by the rules, regulations and other requirements of the FEHBP as they exist and on they may be amended or changed from time to time. Provider further agrees that in the event on conflic between this Agreement and/or the provider manual, and the rules/regulations/other requirements of the FEHBP shall control.
- 2.13 <u>Provider's Inability to Carry Out Duties</u>. Provider shall promptly send witten notice, in accordance with the Notice section of this Agreement, to Anthem of:
 - 2.13.1 Any change in Provider's business address;
 - 2.13.2 Any legal, governmental, or other action involving Provider which could materially impair the ability of Provider to carry out his/her/its duties and obligations under this Agreement, except for temporary emergency diversion situations; or
 - 2.13.3 Any change in accreditation, provider affiliation, insurance, licensure, certification or eligibility status, or other relevant information regarding Provider's practice or status in the medical community.

- 2.14 <u>Provider Credentialing</u>. Where applicable, Provider agrees that he/she/it meets Anthem's credentialing standards or other applicable standards of participation for Networks in which Provider participates. A description of the credentialing program or applicable standards of participation, including any applicable accreditation requirements, is set forth in the provider manual(s).
- 2.15 <u>Adjustment Requests</u>. If Provider believes a Claim has been improperly adjudicated for a Covered Service for which Provider timely submitted a Claim to Plan, Provider must submit a request for an adjustment to Plan within one (1) year from the date of Plan's payment or explanation of payment, unless otherwise set forth in the provider manual. The request must be submitted in accordance with Plan's payment inquiry process. Requests for adjustments submitted after this date may be denied for payment, and Provider will not be permitted to bill Anthem, Plan, or the Covered Individual for those services for which payment was denied.
- 2.16 <u>Blue Cross Blue Shield Out of Area Program</u>. Provider agrees to provide Covered Services to any person who is covered under another CBSA out of area or reciprocal programs and to submit Claims for payment in accordance with current of SA Claims filing guidelines. Provider agrees to accept payment by Plan at the Anthem Rate for the equivalent Network as payment in full except Provider may bill, collect and accept provider Services under the order and a creciprocal programs. Provider further agrees to comply with other similar pagrams of the BeBSA or Crecipical Individuals who are enrolled under BCBSA out of area or reciprocal programs, Provider shall county with the applicable Plan's utilization management policies.
- 2.17 <u>Supervision of Survices</u>. Providen grees to t all Holth Services provided to Covered Individuals under this Agree to t shall be provided by Provider or by a dailord person under Provider's direction. Provider shall warrant that by nurse for other health, tofesser als employed by or providing services for Provider shall be duly licensed or certice, under applicable lat
- 2.18 <u>Pass-Through marge</u>. Provider agrees not to pass through tractan or the Covered Individual any charges which Provider incur as a result of providing supplies or making referrals to another provider or entity. Examples include, are not instead to, pass-through charges associated with laboratory services, pathology services, indicided services and durable method equipment. If Anthem has a direct contract with the subcontractor, the direct contract shall prevail over this Agreement.
- 2.19 <u>Coordination of Benefits/St. regation</u>. Forvious agrees to cooperate with Plan regarding subrogation and coordination of benefits, as set forther, the provider manual, aga to nulify than promptly after receipt of information regarding any Covered adivided who may have a Claim inverse subregation or coordination of benefits.
- 2.20 <u>Preventable Adverse Events</u>. Notwithstanding any provision in this Agreement to the contrary, when any preventable adverse event as set forth in the provider manual (s) occurs with respect to a Covered Individual, the Provider shall neither bill, nor seek to a fect from nor accept any pays and for Plan or Covered Individual for such event. If Provider receives any pays and from Plan or Covered Individual for such event, it shall refund such payment within ten (10) business days of beilding aware of such receipt. Further, Provider shall cooperate with Anthem, to the extent masonable of any Anthem initiative designed to help analyze or reduce such preventable adverse events.
- 2.21 <u>Cost Effective Care</u>. Provider shall provide Covered Services in the unit cost effective, clinically appropriate setting and manner.
- 2.22 <u>Covered Individual Grievance Procedure</u>. Provider agrees to fully coperate with any applicable Covered Individual grievance procedure. Provider shall, in accordance with applicable law (including without limitation, Wis. Admin. Code § Ins 18.03(2)(c), as amended from time to time, provide Plan with necessary records and responses to questions regarding quality issues, Covered Individual grievances or complaints. The information shall be provided within ten (10) business days of Anthem's request. Claims of Provider's medical malpractice shall not be subject to said grievance procedure. This provision shall survive termination of this Agreement with respect to Covered Services rendered to Covered Individuals prior to termination.

7

ARTICLE III CONFIDENTIALITY/RECORDS

- 3.1 <u>Proprietary Information</u>. Except as otherwise provided herein, all information and material provided by either party in contemplation of or in connection with this Agreement remains proprietary to the disclosing party. This Agreement, including but not limited to the Anthem Rates, is Anthem's proprietary information. Neither party shall disclose any information proprietary to the other, or use such information or material except: (1) as otherwise set forth in this Agreement; (2) as may be required to perform obligations hereunder; (3) as required to deliver Health Services or administer a Health Benefit Plan; (4) to Plan or its designees; (5) upon the express written consent of the parties; or (6) as required by law or regulation, except that either party may disclose such information to its legal advisors, lenders and business advisors, provided that such legal advisors, lenders and business advisors, lenders and business advisors.
- 3.2 <u>Confidentiality of Personally Identifiable Information</u>. Both parties agree to abide by state and federal laws and regulations regarding conventiality of the Covered Individual's personally identifiable information.
- 3.3 <u>Network Provider/Patient Decussion</u>. Notwithstanding any other provision in this Agreement and garon as of any benefit or consistence exclusions or limitations associated with a Health Benefit Plan, Provider shall not be protoned from discusing fully with a Covered Individual any issues related to the Covered Individual's health including recommended treatments, treatment alternatives, treatment risks and the consequences of any benefit or or are or payment decisions made by Plan or any other party. Nothing a this Agreement shall prohibit Provider from disclosing to the Covered Individual the general methodology by which avoid it is compensated under this Agreement. Plan shall not refuse to allow or to continue the participation of only otherwise eligible provider, configure compensate Provider in connection with services rendered, solvy because Provider has in got face communicated with one or more of his/her/its current, former or projective totients regarding the provisions, terms or requirements of a Health Benefit Plan as they relate to the mediationed of such patient.
- 3.4 <u>Plan Access to and peneests in Provider Records</u>. Provider shall comply with all applicable state and federal record keepin nequitaments, and, as set for that the proviner manual(s), shall permit Plan or its designees to have, with an inopriate weaking space and without datage, ex-site access to and the right to examine, audit, photocopy excerpt and transcribe any books, occurrent papers, and records related to Covered Individual's medica and billing in any tion within the possestion of Provider and inspect Provider's operations, which involve transcribes notion to Covered Individual's medica and billing in any tion within the possestion of Provider and inspect Provider's operations, which involve transcribes of care, including but not limited to, assessing quality of care, Medical Necessity, appropriateness of care, includand provider shall submit incords to Pran, the Covered Individual or their respective designees to photocopy or electronic transmittant no charge. Provider shall make such records available to the state and federal authorities involved in passessing quality of care or investigating Covered Individual grievances on the state.
- 3.5 <u>Transfer of Medical Records</u>. Provider shall share a Covered Indicatual's medical records, and forward medical records and clinical information in a timely manner to other reactions are providers treating a Covered Individual, at no cost to Anthem, Plan, a Covered Individual, or other treating healthcare providers.

ARTICLE IV

- 4.1 <u>Anthem Insurance</u>. Anthem shall self-insure or maintain insurance as nall be necessary to insure Anthem and its employees, acting within the scope of their duties.
- 4.2 <u>Provider Insurance</u>. Provider shall self-insure or maintain insurance in types and amounts acceptable to Anthem as set forth in the provider manual(s).

ARTICLE V RELATIONSHIP OF THE PARTIES

5.1 <u>Relationship of the Parties</u>. For purposes of this Agreement, Anthem and Provider are and will act at all times as independent contractors. Nothing in this Agreement shall be construed, or be deemed to create, a relationship of employer or employee or principal and agent, or any relationship other than that of independent entities contracting with each other for the purposes of effectuating this Agreement. In no way shall Anthem or Plan be construed to be providers of Health Services or responsible for the provision of

such Health Services. Provider shall be solely responsible to the Covered Individual for treatment and medical care with respect to the provision of Health Services.

- 5.2 <u>Blue Cross Blue Shield Association (BCBSA)</u>. Provider hereby expressly acknowledges his/her/its understanding that this Agreement constitutes a contract between Provider and Anthem, that Anthem is an independent corporation operating under a license from the Blue Cross and Blue Shield Association, an association of independent Blue Cross and/or Blue Shield Plans ("Association"), permitting Anthem to use the Blue Cross and/or Blue Shield Service Marks in the state (or portion of the state) where Anthem is located, and that Anthem is not contracting as the agent of the Association. Provider further acknowledges and agrees that he/she/it has not entered into this Agreement based upon representations by any person other than Anthem, and that no person, entity or organization other than Anthem shall be held accountable or liable to Provider for any of Anthem's obligations to Provider created under this Agreement. Provider has no license to use the Blue Cross and/or Blue Shield names, symbols, or derivative marks (the "Brands") and nothing in the Agreement shall be deemed to grant a license to Provider to use the Brands. Any references to the Brands made by Provide in his/her/its own materials are subject to review and approval by Anthem. This section shall not created in provisions of this Agreement.
 - <u>Contracting Party</u>. If Provider is a numership, corporation, or any other entity other than an individual, all references herein to "Provider" shor also usen and refer to each individual within such entity who Provider certifies is owned or employed by Provider, and who has applied for and been accepted by Plan as a Network/Participating Provider.

AF 10 E VI INDEMNIFICATION OF LIABILITY

- 6.1 Indemnification. Asthem and Provider shall each indemnify, original and hold harmless the other party, and his/her/its directors, orders comployees, age is and subsidiaries, from and against any and all losses, claims, damages, liabilities costs and expenses (acluding winout mitation, reasonable attorneys' fees and costs) arising from the operform his/her/its obligations under this Agreement, and or the indemnifying party's solution of any law, statute, ordinance, order, standard of care, rue or regulation. The obligation to prome indomnifying party with prompt written notice of any claim in which indemnified newer that the indemnifying arty agrees not to enter into any settlement of such claim indemnified newer that the indemnifying arty agrees not to enter into any settlement or compromise of any claim or action in a manner net admits fault or imposes any restrictions or obligations on an informified party without that indemneed prov's pair written consent which will not be unreasonably withher and coordinating fully with the indemnifying party is connection with such defense and settlement.
- 6.2 <u>Limitation of Liability</u>. Regardless of whether here is a total and fundamental chack of this Agreement or whether any remedy provided in this Agreement alls cause essential purpose, in no event shall either of the parties hereto be liable for any amounts representing loss of revenue loss of profits, loss of business, the multiple portion of any multiplied damage award, or neidental, indirect, consequential, special or punitive damages, whether arising in contract, tort (including ne ligenta), or othe wise regardless of whether the parties have been advised of the possibility of such damages, arising in any way out of or relating to this Agreement. Further, in no event shall Plan be liable to Provide for any extra an actual damages relating to any claim or cause of action assigned to Provider by any person continue.
- 6.3 <u>Period of Limitations</u>. Unless otherwise provided for in this Agreement, the providemanual(s), or Policies, neither party shall commence any action at law or equity, including but in Ulimiter to, an arbitration demand, against the other to recover on any legal or equitable claim arising out of the agreement more than two (2) years after the events which gave rise to such claim, unless compliance with this section would compel a party to violate the terms of the Health Benefit Plan. The deadline for initiating an action shall not be tolled by the appeal process, provider dispute resolution process or any other administrative process. To the extent a dispute is timely commenced, it will be administered in accordance with Article VII of this Agreement.

Wisconsin Provider Agreement © 2013.1.1 Blue Cross Blue Shield of Wisconsin Onelda Community Health Center and Ka Ni Kuhli Yo Family Center 09/29/2014

5.3

ARTICLE VII DISPUTE RESOLUTION AND ARBITRATION

- 7.1 <u>Dispute Resolution</u>. All disputes between Anthem and Provider arising out of or related in any manner to this Agreement shall be resolved using the dispute resolution and arbitration procedures as set forth below. Provider shall exhaust any other applicable provider appeal/provider dispute resolution procedures and any applicable state law exhaustion requirements as a condition precedent to Provider's right to pursue the dispute resolution and arbitration procedures as set forth below.
 - 7.1.1 In order to invoke the dispute resolution procedures in this Agreement, a party first shall send to the other party a written demand letter that contains a detailed description of the dispute and all relevant underlying facts, a detailed description of the amount(s) in dispute and how they have been calculated and any other information that the Anthem provider manual(s) may require Provider to submit with respect to such dispute. If the total amount in dispute as set forth in the demand letter is learn an two hundred thousand dollars (\$200,000), exclusive of interest, costs, and attorneys' fact then within twenty (20) calendar days following the date on which the receiving party receives the dimand letter, representatives of each party's choosing shall meet to discuss the dispute in person of telephonic cally in an effort to resolve the dispute. If the total amount in dispute is set forth in the daman effect to resolve the dispute. If the total amount in dispute is set forth in the daman effect to resolve the dispute. If the total amount in dispute is set forth in the daman effect to resolve the dispute. If the total amount in dispute is set forth in the daman effect to resolve the dispute. If the total amount in dispute is set forth in the daman effect on the hundred thousand dollars (\$200,000) or more, exclusive of interest, costs, and attendeys' sets, even within ninety (90) calendar days following the date of the demand letter, the parties shall engage in non-binding mediation in an effort to resolve the dispute unlear of the parties agree in writhing to waive the mediation requirement. The parties shall mutually unlear of the parties agree in writhing to do to pludicial Arbitration and Mediation Services ("JAMS") s all be authorized to appoint a mediator.
- 7.2 <u>Arbitration</u>, any dispute within the scope of subsection 7.1. that remains unresolved at the conclusion of the applicate process putlined in subsection 1.1 shall be rescribed by binding arbitration in the manner as set forth below. Except to be extent as set for below, the colluction shall be conducted pursuant to the JAMS Comprehensive Arbitration Dules and Procedures, solvided, however, that the parties may agree in writing to further more the caMS Comprehensive Arbitration Rules and Procedures. The parties agree to be bound by the findings of the arbitration(s) with respect to such dispute, object to the right of the parties to appeal such findings as set forth herein. No arbitration demand shall be finded until after the parties have completed the dispute resolution efforts deviced in section 7.4 abov.
 - 7.2.1 Selection and Replacement of arbitrator(s). If the total amount or dispute as set forth in the demand letter is less that two notion dollars (\$2,000,000) and using on otherest, costs, and attorneys' fees, the dispute shall be decided by a single arbitrator calected, and replaced when required, in the manner described in the JAMS Comprehensive Arbitration Rules and Procedures. If the total amount in dispute as set orth in the demond letter is two tillion callars (\$2,000,000) or more, exclusive of interest, costs, an attorneys' fees, the dispute shall be defined by a single arbitrator calected and Procedures. If the total amount in dispute as set orth in the demond letter is two tillion callars (\$2,000,000) or more, exclusive of interest, costs, an attorneys' fees, the dispute shall be decided by a single arbitrator.
 - 7.2.2 Appeal. If the total amount of the arbitration ward in ive mixing dollars (\$5,000,000) or more, inclusive of interest, costs, and attorneys' fees, the part is shall have the right to appeal the decision of the arbitrator(s) pursuant to the JAM. Option of Arbitration Appeal Procedure. In reviewing a decision of the arbitrator(s), the appeal panel shall arriv the same standard of review that a United States Court of Appeals would apply in aviewing a similar decision issued by a United States District Court in the jurisdiction in which the arbitration hearing was held.
 - 7.2.3 Waiver of Certain Claims. The parties, on behalf of themselves more that they may now or hereafter represent, each agree to and do hereby waive any right to join or consolidate claims in arbitration by or against other individuals or entities to pursue, on a class basis, any dispute; provided however, that if an arbitrator or court of competent jurisdiction determines that such waiver is unenforceable for any reason with respect to a particular dispute, then the parties agree that section 7.2 shall not apply to such dispute and that such dispute shall be decided instead in a court of competent jurisdiction.

ARTICLE VIII TERM AND TERMINATION

- 8.1 <u>Term of Agreement</u>. The term of this Agreement shall commence at 12:01 AM on the Effective Date and shall continue in effect until such time it is terminated as provided herein.
- 8.2 <u>Termination Without Cause</u>. Either party may terminate this Agreement without cause at any time by giving at least one hundred eighty (180) days prior written notice of termination to the other party.
- 8.3 <u>Breach of Agreement</u>. Except for circumstances giving rise to the Termination With Cause section, if either party fails to comply with or perform when due any material term or condition of this Agreement, the other party shall notify the breaching party of its breach in writing stating the specific nature of the material breach, and the breaching party shall have thirty (30) days to cure the breach. If the breach is not cured to the reasonable satisfaction of the non-breaching party within said thirty (30) day period, the non-breaching party may terminate this Agreement by providing written notice of such termination to the other party. The effective date of such termination of some than sixty (60) days after such notice of termination.

8.4 Terrination With Cause

8.4.1 his Agreement way be to minate a mediately by Anthem if:

- 8.4.1.1 Provider commitse my act or conduct for which his/her/its license(s), permit(s), or any governmental of poard a thorization(s) or approval(s) necessary for business operations or to provide Healt. Services are not business or voluntarily surrendered in whole or in part; or
- 8.4. 2 Prover commits a fraud or mores any material misstatements or omissions on any occupients related to this preemen which exubmits to Anthem or to a third party; or
- 8.4.1.3 Product ales for bankruptcy, makes a assument for the benefit of its creditors with a Anthone bitten consent, or if a ceiver is appointed; or
- 8.4.1.4 Provider insurance overage as required by the Agreement lapses for any reason; or
- 8.4.1.5 Provider this to main ampliance with Anthem's predentialing standards or other applicable standards a participation; or
- 8.4.1.6 Anthem reasonably believes based on Provider's connect or maction or allegations of such conduct or inaction, that the well-being of patients may be jeopardia d; or
- 8.4.1.7 Provider has been abuve to a Concred Individual, an A them employee or representative; or
- 8.4.1.8 Provider and/or his/her/its employers, contractors, obcontractors, or agents are identified as ineligible persons on the General Services A ministration list of Parties Excluded from Federal Programs and/or HHS/OIG List of Excluded Individuals/Entities, and in the case of an employee, contractor, subcontractor or agent fails to remove such individual from responsibility for, or involvement with, the provider's busines, operations related to this Agreement; or
- 8.4.1.9 Provider is convicted of a felony or misdemeanor.
- 8.4.2 This Agreement may be terminated immediately by Provider if:
 - 8.4.2.1 Anthem commits any act or conduct for which its license(s), permit(s), or any governmental or board authorization(s) or approval(s) necessary for business operations are lost or voluntarily surrendered in whole or in part; or
 - 8.4.2.2 Anthem commits a fraud or makes any material misstatements or omissions on any documents related to this Agreement which it submits to Provider or to a third party; or
 - 8.4.2.3 Anthem files for bankruptcy, or if a receiver is appointed; or

- 8.4.2.4 Anthem's insurance coverage as required by this Agreement lapses for any reason.
- 8.4.3 If applicable, Anthem reserves the right to terminate individual providers under the terms hereof while continuing the Agreement for one or more providers in a group.
- 8.4.4 Anthem shall have the right to terminate this Agreement upon thirty (30) days prior written notice to Provider as set forth in subsection 9.3.2.
- 8.5 <u>Transactions Prior to Termination</u>. Termination shall have no effect on the rights and obligations of the parties arising out of any transaction occurring prior to the date of such termination.
- 8.6 <u>Continuance of Care-Termination</u>. Unless otherwise set forth in this Agreement, the Health Benefit Plan, or required by statute or regulation, Continuance of Care-Termination shall apply as follows: Provider shall, upon termination of this Agreement for reasons other than the grounds set forth in the Termination With Cause section of this Agreement, pursuant to 609.24, Wis. Stats., continue to provide and be compensated for Covered Services reported to Covered Individuals under the terms and conditions of this Agreement until the earlier of such medically a propriate arrangements have been made for a Network/Participating Provider to render mealth Services and network overed Individual. Provider shall comply with the notice posting requirements of INS 9.35(1m) (if opplicate).
- 8.7 Sontinuance of pare-Insolvency adhless otherwise set forth in the Health Benefit Plan, or required by statute of regulation, in the even of a Han's of an Other Payor's insolvency or other cessation of operations. Provider shall continue to provide overed Services to Covered Individuals of such Plan or Other Payor's accordince with this Agreement until the varier of: (i) the expiration of the period through which the premium has been paid for or erage order the applicable Health Benefit Plan; or (ii) the expiration of such other period as may be required by laws or regulations applicable to the Provider, the Plan or the Other Payor. Provider agrees not a bill or otherwise pursue Covered Individual for Covered Services for which secto covered Person is not liable under §609.91, Wis. Stats. This section will be construed in favor of a covered Individual and supersects any oral dividual if the contrary agreement between Provider and a Covered Individual or the representative of a Covered Individual if the contrary agreement is inconsistent with this provision or the provisions of this Agreement acting continuance of care after termination of this Agreement.
- 8.8 <u>Survival</u>. In the event of termination the green ent, the following provider shall survive:
 - 8.8.1 Publication and Use of Proving Information (Section 2.3) excluding transparency of formation;
 - 8.8.2 Payment in Full and Hold Harmless / ection 2.7);
 - 8.8.3 Adjustments for Incorrect Payments (Section 2
 - 8.8.4 Confidentiality/Records (Article III);
 - 8.8.5 Indemnification and Limitation of Liability (Article VI)
 - 8.8.6 Dispute Resolution and Arbitration (Article VII);
 - 8.8.7 Continuance of Care-Termination (Section 8.6); and
 - 8.8.8 Continuance of Care-Insolvency (Section 8.7).

ARTICLE IX GENERAL PROVISIONS

9.1 <u>Amendment</u>. Except as otherwise provided for in this Agreement, Anthem retains the right to amend this Agreement, the Anthem Rate, any attachments or addenda by making a good faith effort to provide notice to Provider at least forty five (45) days in advance of the effective date of the amendment. If Provider decides not to accept the amendment, Provider has the right to terminate this Agreement without the amendment taking effect by providing written notice within thirty (30) days from receipt of such notice from Anthem. Provider's termination shall take effect on the later of the amendment effective date identified by Anthem or ninety (90) days from the date Provider has provided notice of his/her/its intention to terminate pursuant to

this section. Failure of Provider to provide such notice to Anthem within the time frames described herein will constitute acceptance of the amendment by Provider.

9.2 <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the respective legal successors and assignees of the parties. However, neither this Agreement, nor any rights or obligations hereunder may be assigned, either by operation of law or otherwise, transferred in whole or in part, without the prior written consent of the other party, except that Anthem retains the right to assign, either by operation of law or otherwise, transferred any rights or obligations of law or otherwise, transfer in whole or in part, this Agreement to an Affiliate or to delegate any rights or obligations under this Agreement to a designee.

9.3 Scope/Change in Status.

- 9.3.1 Anthem and Provider agree that this Agreement applies to Health Services rendered at the Provider's location(s) on file with Anthem. Anthem may, if in Anthem's judgment the circumstances require such, limit this greement to Provider's locations, operations or business or corporate form, status or structure of distance on the Effective Date of this Agreement and prior to the occurrence of any of the following events:
 - 3.3.1.1 Provider self an or substantially all of his/her/its assets; or
 - .3.1.2 Provider transfers or trol or his/her/its management or operations to any third party, including Provider energing into a management contract with a physician practice management company which doe not manage Provider as of the Effective Date of this Agreement, or there is a subsequent, thange in control of Provider's current management company; or
 - 9.3 .3 provider acquires or controls any other medical practice or entity or is in any manner oth wise acquired or controlled by any other party, whether by purchase, merger, con blightion, alliance, joint venture, partnership, association or expansion; or
 - 9.3.1.4 Provider merwise changes his/her/it flocations, disiness or operations, or business or corporate form or stalls; or
 - 9.3.1.5 Provider or paters or otherwise, operates a licensial health montenance organization or commercial health prior whether such creation or operation indirect or through a Provider affiliate).
- 9.3.2 Without limiting any of Anthen orights a set forth elsewhere in this A freement, A them shall have the right to terminate this Agreement upon thirty (30) days writter notice to Provider if Anthem determines, that as a result of any of the transactions listed in subsection 5.8.1, Provider cannot satisfactorily perform the obligations of Provider Jereunder, or cannot complete with one or more of the terms and conditions of this Agreement, including but not limited to the confidentiality provisions herein; or Anthem elects in its reasonable to siness discretion bot to do business with Provider, the successor entity or new management company, as a roult of one or more of the events as set forth in subsection 9.3.1.
- 9.3.3 Provider shall provide Anthem with thirty (30) days pre-written notice
 - 9.3.3.1 A change in providers who are part of the group, if policable. In y new providers must meet Anthem's credentialing standards or other oplicable randards prior to being designated as a Network/Participating Provider; or
 - 9.3.3.2 Any new physical location, tax identification number, mailing address or similar demographic information; or
 - 9.3.3.3 A change in operations, business or corporate form as set forth in subsections 9.3.1.1 through 9.3.1.5 above.
- 9.4 <u>Definitions</u>. Unless otherwise specifically noted, the definitions as set forth in Article I of this Agreement will have the same meaning when used in any attachment, the provider manual(s) and Policies.

Wisconsin Provider Agreement © 2013.1.1 Blue Cross Blue Shield of Wisconsin

Oneida Community Health Center and Ka Ni Kuhli Yo Family Center 09/29/2014

- 9.5 <u>Entire Agreement</u>. This Agreement (including items incorporated herein by reference) constitutes the entire understanding between the parties and supersedes all prior oral or written agreements between them with respect to the matters provided for herein. If there are any conflicts between any of the provisions of this Agreement and the provider manual, this Agreement will take precedence.
- 9.6 <u>Force Majeure</u>. Neither party shall be deemed to be in violation of this Agreement if such party is prevented from performing any of his/her/its obligations hereunder for any reason beyond his/her/its reasonable control, including without limitation, acts of God, acts of any public enemy, floods, statutory or other laws, regulations, rules, or orders of the federal, state, or local government or any agency thereof.
- 9.7 Compliance with Federal and State Laws. Anthem and Provider agree to comply with all requirements of the law relating to their obligations under this Agreement, and maintain in effect all permits, licenses and governmental and board authorizations and approvals as necessary for business operations. Provider agrees that he/she/it shall be and remain licensed and certified (including Medicare certification in unqualified, unrestricted states) in accordance with all state and federal laws and regulations (including those applicable to utilization eview and Claims payment) relating to the provision of provider services to Covered Individuals. Provider shall upply evidence of such licensure, compliance and certifications to unnew upon request. Provider fault, agrees to immediately notify Anthem if he/she/it loses or voluntarily surrende such licensure percedition, a units, authorizations or approvals, or when applicable no longer meets A them's credentialing mandance. From time to time legislative bodies, boards, departments or agencies may enact, issue or ameromany, rules, or regulations to the extent applicable, and to cooperate with the ameromany ely abide by all aid laws rules, or regulations to the extent applicable, and to cooperate with the ameromany ely abide by all aid laws rules, or regulations to the event of a conflict between this section and any going provision in this Agreement. In the event of a conflict between this section and any going provision in this Agreement, this section shall control.
 - 9.7.1 In addition to be for going, Providen variants and receivents that at the time of entering into this Agreement, weither he/she/it nor any of his/her/it em, loyees, contractors, subcontractors or agents are releable means identified on the Coneral Services Administrations' List of Parties Excluded from Fernal Programs (available through the internet at http://www.epls.gov/ or its successor) and the HHS/OIG ist of Excluded individuals antities available through the internet at http://www.oig.hh gov/fraud/exclusions.asp or its successor) of as otherwise designated by the Federal government. If Provider and we employees, obcordact is or agents thereof becomes an ineligible person after entering to this agreement or otherwise usits to asclose his/her/its ineligible person status, Provider sharman obligation to (1) immediarily usify Anthem of such ineligible person status and (2) within ten (10, lays of uch notice, removided in wides from responsibility for, or involvement with, the newider's burness operations related to as Agreement.
- 9.8 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the state where Anthem is located, as identified by the legal entry name in the preactile, these such state laws are otherwise preempted by federal law. However, concreage issues specific to a Health Benefit Plan are governed by the state laws where the Health Benefit Plan is issued unless such state laws are otherwise preempted by federal law.
- 9.9 <u>Intent of the Parties</u>. It is the intent of the parties that this threemost is to be effective only in regards to their rights and obligations with respect to each other; it is encressly not the ment of the parties to create any independent rights in any third party or to make any used party or third party beneficiary of this Agreement, except to the extent specified in the Payment in wall and Hold Formless section of this Agreement.
- 9.10 <u>Non-Exclusive Participation</u>. None of the provisions of this Agreement shows event Provider or Plan from participating in or contracting with any provider, preferred provider organization, health maintenance organization/health insuring corporation, or any other health delivery or insurance program. Provider acknowledges that Plan does not warrant or guarantee that Provider will be utilized by any particular number of Covered Individuals.
- 9.11 <u>Notice</u>. Any notice required to be given pursuant to the terms and provisions of this Agreement shall be in writing and shall be delivered by electronic mail, by facsimile, by hand, or by mail. Unless specified otherwise in writing by a party, Anthem shall send Provider notice to an address that Anthem has on file for Provider, and notice initiated by Provider shall be sent to Anthem's address as set forth on the signature page. Notice shall be effective upon the marked date associated with the corresponding delivery method noted above. Notwithstanding the foregoing, Anthem may post updates to its provider manual(s) and

Policies on its web site.

- 9.12 <u>Severability</u>. In case any one or more of the provisions of this Agreement shall be invalid, illegal, or unenforceable in any respect, the remaining provisions shall be construed liberally in order to effectuate the purposes hereof, and the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby. If one or more provisions of the Agreement are invalid, illegal or unenforceable and an amendment to the Agreement is necessary to maintain its integrity, the parties shall make commercially reasonable efforts to negotiate an amendment to this Agreement and any attachments or addenda to this Agreement which could reasonably be construed not to contravene such statute, regulation, or interpretation. In addition, if such invalid, unenforceable or materially affected provision(s) may be severed from this Agreement and/or attachments or addenda to this Agreement without materially affecting the parties' intent when this Agreement was executed, then such provision(s) shall be severed rather than terminating the Agreement or any attachments or addenda to this Agreement.
- 9.13 <u>Waiver</u>. Neither the waiver be either of the parties of a breach of any of the provisions of this Agreement, nor the failure of either of the parties, on one or more occasion, to enforce any of the provisions of this Agreement, shall thereaf or be construed as a waiver of any subsequent breach of any of the provisions of the pro
- 9.14 <u>Abandor bent</u>. Nothing herein shall be construed as authorizing or permitting Provider to abandon any patient.
- 9.15 <u>Exertangent onlysis specifically set forth on the Net bork listing on the signature page and/or in the PCS or as offering set signated by Anthem, the Anthere Raw shall not apply to any products or Plan Programs which Anthere may offer on state-basid, received or federal health insurance exchanges ("Exchanges") established to the Port of Protection and Affeddable Care Act.</u>

Each party warrants that it has full power and authority to enter into this Agreement and the person signing this Agreement on behalf of either party warrants that he/she has been duly authorized and empowered to enter into this Agreement.

THIS AGREEMENT CONTAINS A BINDING ARBITRATION PROVISION WHICH MAY BE ENFORCED BY THE PARTIES



As of the Effective Date of this Agreement, Provider will be designated as a Network/Participating Provider in the following:

Commercial lines of business:

Health Benefit Plans in which Covered Individuals have access to a network of providers and receive an enhanced level of benefits when they obtain Covered Services from Network/Participating Providers regardless of product licensure status or funding source. Such Health Benefit Plans include but are not limited to:

- HMO (includes group HMO and POS products such as: Blue Preferred and Blue Preferred Plus and CDHP)
- PPO (includes PPO products such as: Blue Access and CDHP)
- Indemnity/Traditional/Standard (includes indemnity/traditional/standard products such as: Traditional)
- Other State Specific (such as: Blue Priority WI, Blue Priority X WI)

Governmental lines of business:

Health Benefit Plans issued pursion to an agreement between Plan and the federal or state government and in which Covered Individuals navy access to a network of providers and receive an enhanced level of benefits much they obtain cover d Sense of from Network/Participating Providers regardless of product licens a status. Such Health Brance Plans include but are not limited to:

- den S products such as: Medicare Advantage HMO) uch a Medicare Advantage PPO) dicare HMO (includes group HM and Vicare PPO (includes PPO products ad
- Me
- Medicaid

ATTACHMENT A - WISCONSIN STATUTES §609.91 AND §609.94 ACKNOWLEDGEMENT APPLICABLE TO SERVICES RENDERED BY PROVIDER IN HMO PROGRAM

Ins 9 APPENDIX C

NOTICE

THIS NOTICE DESCRIBES HOLD-HARMLESS PROVISIONS WHICH AFFECT YOUR ABILITY TO SEEK RECOURSE AGAINST HMO ENROLLEES FOR PAYMENT FOR SERVICES

Section 609.94, Wis. Stat., requires each health maintenance organization insurer ("HMO insurer"), to provide a summary notice to all of its participating providers of the statutory limitations and requirements in Sections 609.91 to 609.935, and 609.97 (I), Wis. Stats.

SUMMARY

Under Wiscensin law a health one provider monot hold HMO enrollees or policyholders ("enrollees") liable for costs covered inder a HMO policy if the provider is subject to statutory provisions, which "hold harmless" the enrollees. For most health are providers as a cation of the entutory hold-harmless is "mandatory" or it applies unless the provider elects to opt-out." A provider monittee to "op out" must file timely notice with the Wisconsin Office of the Commissioner of Insurance ("OCI").

Some types of a wider care are subject to the statutes only if the provider voluntarily, "opts-in." An HMO may partially satisfy its insulatory capital and surplus requirements if walth care providers elect to remain subject to the statutory hold-harmles providers.

This notice is only a summary of the low. Every effort has been made to occurately describe the law. However, if this summary is inconsistent with a provision of the law or incomplete, the low was control.

Filings for exemption with OC sust in on the pascribed form in the der to be elective

HOLD HARMLESS

A health care provider who is subject to the star bory hot harmless providens is prohibited from seeking to recover health care costs from an enrollee. The provider hav not oill, charge, collect a lept at from seek remuneration or compensation from, file or threaten to file with a creat reporting agency or have the recourse as just an enrollee or any person acting on the enrollee's behalf, for health care unsts for which the enrollee is not liable, the prohibition on recovery does not affect the liability of an enrollee for any Cost Shares, or for premiting owed under the policy or certificate issued by the HMO.

- A. MANDATORY FOR HOLD HARMLESS. An expoller of an HMO insurer is not hable to a health care provider for health care costs that are covered under policy by that MO if:
 - 1. Care is provided by a provider who is an Affin te of the HMO in urer, owns at least five percent (5%) of the voting securities of the HMO, is direct or interectly involved with the HMO through direct or indirect selection of or representation by on or more board, embers, or is an Individual Practice Association ("IPA") and is represented, or an Affiliate is represented, by one of at least three HMO insurer board members who directly or insurer's represented one or more IPAs or affiliates of IPAs; or
 - 2. Care is provided by a provider under a contract with or through the mbership in an organization identified in 1; or
 - 3. To the extent the charge exceeds the amount the HMO has contractually agreed to pay the provider for that health care service; or
 - 4. The care is provided to an enrolled medical assistance recipient under a Department of Health and Retail Services prepaid health care policy.
 - 5. The care is required to be provided under the requirements of Ins 9.35 Wis. Adm. Code.

B. "OPT-OUT" HOLD HARMLESS.

If the conditions described in A do not apply, the provider will be subject to the statutory hold-harmless unless the provider files timely election with OCI to be exempt if the health care is:

- 1. Provided by a hospital or an IPA; or
- 2. A physician service, or other provider services, equipment, supplies or drugs that are ancillary or incidental to such services and are provided under a contract with the HMO or are provided by a provider selected by the HMO insurer; or
- 3. Provided by a provider, other than a hospital, under a contract with or through membership in an IPA that has not elected to be exempt. Note that only the IPA may file election to exempt care provided by its member providers from the statutory hold harmless. (See Exemptions and Elections, No. 4.)

C. "OPT-IN" HOLD HARML

a provider of health care a prestill act to the conditions described in A or B, the provider may elect to be subject to the statutory hold-harm assign whoms by filing a notification with OCI stating that the provider elects to be subject with respect to an specific HMO insurer. A provider may terminate such a notice of election by stating the termination doe in pat notice or in a separate notification.

CONDITIONS AFF CTING IMMUNITY

An enrollee's immunity under the statutory hold harm to s is not affected by any of the following:

- 1. Any agreement interest into the a provider, an N4O, or any other version, whether oral or written, purporting to hold the enrollee light of costs (except a notice of election or ten ination permitted under the statute);
- 2. A breach of or default on my agreement by the HMC an IPA, or any other person to compensate the provider for health care cores for which he enrollee is not liable;
- 3. The insolvency of the HMO is any person connacting with the HMC, or the compencement of insolvency, delinquency or bankruptcy proceeding, involving the HMO or other prisons which would affect compensation for health care costs or which an enrollee is not liable under the statutes, hold-harmless;
- 4. The inability of the provider or other person where owed compensation to obtain compensation for health care costs for which the enrollee is not liable;

5. Failure by the HMO to provide notice to provide of the strutory hold-harmless providens; or

6. Any other conditions or agreement existing at any tim

EXEMPTIONS AND ELECTIONS

Facilities, IPAs, and providers of physician services who may "opt-out may elect to be exempt from the statutory hold-harmless and prohibition on recovery of health care costs under the allowing nonditions and with the following notifications:

- 1. If the facility, IPA, or other provider has a written contract with the HMO, he preder must within thirty (30) days after entering into that contract provide a notice to OCI of the provider selection to be exempt from the statutory hold-harmless and recovery limitations for care under the contract.
- 2. If the facility, IPA, or other provider does not have a contract with an HMO, the provider must notify OCI that it intends to be exempt with respect to a specific HMO and must provide that notice for health care costs incurred at least ninety (90) days in advance.
- 3. A provider who submits a notice of election to be exempt may terminate that election by stating a termination date in the notice or by submitting a separate termination notice to OCI.

- 4. The election by an IPA to be exempt from the statutory provisions, or the failure of an IPA to so elect, applies to costs of health care provided by any provider, other than a hospital, under contract with or through membership in the IPA. Such a provider, other than a hospital, may not exercise an election separately from the IPA. Similarly, an election by a clinic to be exempt from the statutory limitations and restrictions or the failure of the clinic to elect to be exempt applies to costs of health care provider through the clinic. An individual provider may not exercise an election to be exempt separate from the clinic.
- 5. The statutory hold-harmless "opt-out" provision applies to physician services only if the services are provided under a contract with the HMO or if the physician is a selected provider for the HMO, unless the services are provided by a physician for a hospital, IPA or clinic which is subject to the statutory hold-harmless "opt-out" provision.

NOTICES

All notices of election and terminated must be in writing and in accordance with rules promulgated by the Commissioner of Insurance. All proceed of election or termination filed with OCI are not affected by the renaming, reorganization, merger, consolid ago or change in control of the provider, HMO, or other person. However, OCI may promulated rules requiring an inform donation of these events occur.

Notices to the Office of the Commission of Insurance must be written, on the prescribed form, and received at the Office current address:

J. Box 873, Madison, WI 53707-7873

HMO CAPITAL AN SI CURITY SURPLUS

Each HMO is require to meet minimum capital and surplus tandard ("compulsory surplus requirements"). These standards are highen if the HMU has fewer than ninety percent (90%) of its liabilities covered by the statutory hold-harmless. The compulsory surplus equirement shall be bleast the greater of \$750,000 or six percent (6%) of the premiums earned by the HMC in the last 10 months if its covered abilities are less than ninety percent (90%), or three percent (3%) of the premiums earned by the premiums earned by the HMC in the last 10 months if its covered abilities are less than ninety percent (90%), or three percent (3%) of the premiums earned by the HMO in the east twelve (12) months if its covered liabilities are ninety percent (90%) or more. In addition to capital and surplus, an HMO must also maintain a security surplus in the amount set by the Commissioner of surplus earned.

FINANCIAL INFORMATION

An HMO is required to file financial statements with CCI. You may request finance statements from the HMO. OCI also maintains files of HMO financial statements that can be aspected by the public.

20

PLAN COMPENSATION SCHEDULE ("PCS")

I. PROVIDER TYPE

a. "Multispecialty Group Practice" means a group of licensed practitioners with varying specialties who provide Health Services to Covered Individuals.

To the extent required by law or an accrediting body, upon termination without cause, Provider will provide timely, sixty (60) day, notice to affected Covered Individual(s) of termination of this Agreement or termination of an individual Network participation.

b. "Behavioral Health Practitioner" means a licensed or certified mental health and/or substance abuse practitioner, or a group of licensed or supervised practitioners with varying specialties, who work either in an independent private practice, a group setting in one or more locations, or at an appropriately licensed clinic/facility or agency provide previous health and/or substance abuse Health Services.

C LUSIC REIMBURG ME T TEP

For Covered Services provided by or condenalf of Provider to a Covered Individual who is enrolled in a product and/or program that is uppend by an Network designated on the signature page other than the Medicare Advantage Network, Provider a reset to accept the lesser of Provider's Charges or the applicable fram Fee Guided e. With respect with Medicare Advantage Network, Provider agrees to accept the lesser of the unit of the Charges or the applicable Plan fee Schedule which is based on the current CMS Medicare Fee Schedule of ffective anuary 1st of the year with the exception of drugs that are updated quarterly. Plan will automatically united its allowance for inectable lrugs in accordance with CMS quarterly updates to the Drug Pricine File Replaced adjustments wade by CMS to the Drug Pricing File or the Medicare Fee Schedule shall not be made of Anthem's Drug Preing File or the applicable Plan Fee Schedule.

III. GENERAL PROVISI

П.

Billing Form and Claims reporting Requirements. Provider shall submit all Claims on a CMS 1500 claim form or its successor. Provider shall reported II Health Services in accordance with the reporting guidelines and instructions contained in an MA Court, CP Assistant, and HCCCS miblications. Plan audits that result in identification of Health Services character no reported in accordance with the AMA CPT, and CPT Assistant publications, will be subject to recovery through remittance adjustment or other recovery action. In addition, updates to Anthem's Claims processing fillers and edits, as a moult or anages in AMA CPT, and CPT Assistant reporting guidelines and instructions, shall take place automatically and do not require any notice, disclosure or amendment to Provider.

<u>Claim Submissions for Pharmaceuticals</u>. Proviner agree that the NDC must be tested on each Claim that includes Federal Legend Drugs.

<u>Coding Updates</u>. Coded Service Identifier(s) used to a fine spenic rate are updated from time to time to reflect new, deleted or replacement codes. Anthem shall used on nercially reasonable efforts to update all applicable Coded Service Identifiers within sixty (60) days of clease in CMS on other applicable authority. If an update is delayed beyond the sixty (60) days, Anthem shall notify Provider. Claims processed prior to the implementation of the revised codes shall not be reprocessed. In audition, claims with codes which have been deleted will be rejected.

Not Otherwise Classified Codes (NOC) and/or Individual Consideration Soder (C). Anthem reserves the right to price NOC and/or IC codes individually, and may require the submission of medical records prior to the adjudication of such Claims.

<u>Out-of-Network Compensation</u>. Except for state and federal health programs, if Provider renders services to a Covered Individual who accesses a Network in which Provider does not participate, Provider will receive compensation as follows:

Plan shall compensate Provider for Emergency Services rendered to a Covered Individual based on the applicable Indemnity/Traditional/Standard Anthem Rate. Provider agrees to accept the Indemnity/Traditional/Standard Anthem Rate as payment in full and shall only bill for the applicable Cost Share.

Except for Emergency Services, if the Covered Individual's Health Benefit Plan requires authorization by the Plan or a Provider for out of Network Covered Services in order for the Covered Individual to have the highest level of benefits, and such authorization has been given, then Plan shall compensate Provider for such authorized Covered Services based on the applicable Network/Participating Provider Anthem ("Indemnity/Traditional/Standard") Rate. Provider agrees to accept the Indemnity/Traditional/Standard Anthem Rate as payment in full and shall only bill for the applicable Cost Share. Except for Emergency Services, if the Covered Individual's Health Benefit Plan does not have out-ofnetwork benefits unless authorized by the Plan or Provider, Plan shall have no liability for services rendered without such authorization. In that event, Provider shall bill the Covered Individual for Health Services rendered.

Except for Emergency Services, if the Covered Individual's Health Benefit Plan has out-of-network benefits without authorization being realized by the Plan or Provider, and no authorization has been given, then Plan will compensate Provider for overed Services based on the Anthem Rate established for the Network and/or Plan Program the supports the Covered Individual's Health Benefit Plan. For example, if the Covered Individual's does is supported by PPO Network, compensation is based on the applicable anthem Rate for the PPO Network provider shall only bill for the applicable Cost Share as well as any amount resignated as the covered Individual's responsibility on the provider payment voucher (or other written notice of explanation of ayment). In the event shall payment from Plan and the Covered Individual exceed Provider's Charge for such 6 wend Services.

<u>System odate</u>. Unless otherwise required by the supplicable regulation, any updates to the Anthem Rates tied by governmental agency vendors of other untity shall be effective no later than sixty (60) days after Anthem received such rate changes. Examples include, but are not limited to, CMS state specific fee schedular and one maceutical rates.

The parties recognize and agree that Anthem parment systems cannot be modified immediately when the above changes are near, but nucleinstead be modified using a timeframe of sufficient length to allow for the loading of payment system changes and updates including oblinges and updates that allow for the recognition of new fees and codes. In them shall use reasonable effect to load such payment system changes as quickly as pravicable following the release date of ruch charges. Claims processed prior to the implementation of the new a them Rates inductor be reprocessed.

<u>Workers' Compensation</u>. Provider an ees that in the event a Covered Individual, who is covered for workers' compensation benefits by an An ate or Other dayor under a workers' compensation arrangement administered by an Affiliate, seeks services for an ork-related illness or injury Provider shall provide such Provider services as are Medically Nece same compensable Medical Care. Compensate Medical Care" means medical care determined by the carrier or administrator to be covered, under a Workers' Compensation Plan. As payment for such Provider services rendered, Provider service care to accept the lesser of (i) the Anthem Rate as set forth in this PCS or (ii) the applicable State's Workers' Compensation fee schedule or the reasonable allowable defined by the state for workers' compensation in effect on the date the services or expenses were incurred.

Wisconsin Provider Agreement - PCS © 2013.1,2 Blue Cross Blue Shield of Wisconsin Oneida Community Health Center and Ka Ni Kuhli Yo Family Center 09/29/2014

22

WISCONSIN MEDICAID PARTICIPATION ATTACHMENT TO THE ANTHEM BLUE CROSS AND BLUE SHIELD PROVIDER AGREEMENT

This is a Participation Attachment to the Anthem Blue Cross and Blue Shield Provider Agreement ("Agreement"), entered into by and between Anthem and Provider and is incorporated into the Agreement.

- 1. This Medicaid Participation Attachment (the "Attachment") is limited to the terms and conditions governing the provision of and payment for Health Services provided to Medicaid enrollees who are also enrolled in Plan's managed care Medicaid program (hereinafter referred to as "Medicaid Covered Individuals"). Provider agrees to participate as a Network Provider in Plan's managed care Medicaid network (hereinafter "Medicaid Network") and to provide Health Services to Medicaid Covered Individuals.
- 2. For purposes of this Attachment, "Medicaid" means medical assistance provided under a plan approved under Title XVI, Title XIX, ad or Title XXI of the Social Security Act. (e.g. Medicaid, Badgercare and any other Medicaid Manager, Cr. e Programs). For purposes of this Attachment, "Covered Services" means user indically necessary lealth for the for which a Medicaid Covered Individual is eligible.
- 3. All of Provider's duties and obligations to Covered Individuals set forth in the Agreement shall also apply to Medicaid Covered Individuals. In reaction, Provider agrees to the following with respect to Medicaid Covered Individuals:
 - a. In length and hold harmless the State of Wisconsin, its agencies, officers, and employees from all claim, and subs, including court cores, itorney fees, and other expenses, brought because of injulies or languages received or se tained by any person, persons, or property that is caused by any action of Provider;
 - b. Comply with neverms explicable to providers set with in the managed care organization contract (Contract for Medicaid and BadgerCare HMC Services) including incorporated documents, between Plan and the State of Wisconsin, which applie to the same incorporated herein by reference. Plan arrees to provide Provider with a descrution of the applicable terms upon request (Contract Article X paragraph A.4
 - c. Comply with all State and Fide, Llaws, jules, regulations and ordinances applicable to Medicaid Covered Individuals;
 - d. Cooperate and comply with Provider Arricals Process for purposes of claims displace resolution;
 - e. Maintain Medicaid certification pursu at to HFS 12, Wis. Adm. Code, and obtain a unique provider identification number as specified in Section 117 (b) of the Social Security 12,
 - f. Be duly licensed in accordance with the approable state litencing board of the State of Wisconsin. Provider further agrees to remain in good stancing with still board (42 CFR 438.602);
 - g. Obtain and maintain all required permits, licenses and approvals and comply with all applicable health, safety and environmental statutes, rules, re-ulations or order ances necessary for the performance of Health Services (42 CFR 438.606);
 - h. Submit all encounter data and claims for Health Services rendered to Mericaid Covered Individuals that do not involve a third party payor within one hundred eighty 1800 alendar days from the date of service;
 - Cooperate with any program designed to monitor Medicaid program compliance by providers who participate in Plan's Medicaid Network and comply with any corrective actions related thereto;
 - j. Submit all encounter data or claims for Health Services rendered to Medicaid Covered Individuals in accordance with Plan's specifications for the submission of such encounter data (Contract Article X, paragraph A.5);
 - k. Provide a copy of a Medicaid Covered Individual's medical record at no charge upon reasonable request by the Medicaid Covered Individual (Contract Article X, paragraph A.9);

i.
- Facilitate the transfer of the Medicaid Covered Individual's medical record to another provider at said Medicaid Covered Individual's request (Contract Article X, paragraph A.9);
- m. Cooperate with and permit evaluations, through on-site inspection or other means, of the quality, appropriateness, and timeliness of Health Service rendered to Medicaid Covered Individuals. Such evaluations may be conducted by Plan, the Wisconsin Department of Health and Family Services, the Federal Department of Health and Human Services, or other duly authorized State or Federal agency (Contract Article X, paragraph A.8);
- n. Cooperate with and permit inspections of any records pertinent to Provider's delivery of Health Services to Medicaid Covered Individuals. Such inspections may be conducted by Plan, the Wisconsin Department of Health and Family Services, the Federal Department of Health and Human Services or the reduction of the State or Federal agency (Contract, Article X, paragraph A.8);
 - Maintain an edectrate receive keeping system as required in Article IV F of the Contract for Medicaid and Bodge part AMO Services for recording services, charges, dates and other pommonly accepted information elements for Health Services rendered to Medicaid Covered individuals including out proclimited to all special compliance requirements on abortions, sterilizations, hysterectomic an Health Check reporting requirements;
 - participate in any internal and external quality assurance, utilization review, peer review, and go whice procedures estably bed by than for Medicaid Covered Individuals including but not limit d to the rovision of medical accrection Plan pursuant to grievances received by Plan within fifte in (11, but iness days (42 CFR 38.402),
- q. Comply with the equirements of 42 CR 489, Stepart related to maintaining and distributing written police stand processes respecting advance directives.
- r. Prepare and subjit required data for quality assessment performance improvement programs as requested by Pla. (Contract Afticle X, paragraph A.3):
- s. In the event of Plan's insolvery, continue to provide Heard Services to Medicaid Covered Individuals until the end of the month in which insolvency has or direct and to provide inpatient Health Services until the core of ouchard for any Medicaid Dovered individual institutionalized when insolvency occurs.
- t. Abide by the terms of the Contract or Medicaid and BadgerCare HMM Services (Article III, E 9) for the timely provision of emergency and urgent and the mergency care cases stipulated in any required hospital/emergency room MOUs signed of Plan in accelerance with Article III, E, 9, c and Addendum I of the Contract for Medicaid and Badger are TMO Services (Contract Article X, paragraph A.4);
- u. Ensure confidentiality of family planning services (Contract Anticle X angraph A.10);
- v. Not to create barriers to access to care by imposing requirements on Medicaid Covered Individuals that are inconsistent with the provision of Medically Necessary Covered Dervices (Contract Article X, paragraph A.11).
- w. All laboratory testing sites providing Health Services to Medicaid Covered Individuals pursuant to this Attachment must have a current Clinical Laboratory Improvement Amendments (CLIA), certificate of waiver, certificate of accreditation or a certificate of registration along with a CLIA identification number and comply with CLIA regulations at 42 CFR Park 493.
- 4. Termination of Medicaid Network Attachment.
 - a. This Attachment shall automatically terminate upon the occurrence of any one of the following:
 - i. Termination of Provider's license;

1.

n.

- ii. Failure to comply with section 3.g above;
- iii. Termination/expiration of Plan's Contract for Medicaid and BadgerCare HMO Services with the State of Wisconsin; or
- iv. Failure to meet Plan's credentialing standards or other applicable standards of participation for the Medicaid Network.
- b. Except as otherwise provided herein, either party hereto may terminate this Attachment without cause upon one hundred eighty (180) calendar days prior written notice to the other party.
- c. Except as otherwise provided herein, if either party fails to comply with or perform any term or condition of this Attachment the other party shall notify the defaulting party of its default in writing, and the defaulting party shall have thirty (30) days to cure the default. If the default is not cured within said thirty (30) have period, this Attachment is automatically terminated, unless otherwise specified by the party faulting party (42 CFR 438.708).
- The Hachment shall be an omation of mended to conform to applicable changes to State or Federal laws, rules, revulations or ordinate or related to Medicaid Covered Individuals or the Wisconsin Medicaid program without the necessity of executing writter an edments.
- Provider agrees that Plan's payment constitutes payment in full for any Covered Services rendered to interacial covered Individuals. Fourier agrees if shall not seek payment from the Medicaid Covered Individual bis/lour representative or the State of Mescon in for any Health Services rendered pursuant to this Attachment, with the exception of Cos. Share of any, or payment for non-covered services otherwise requested by and tool ded to, the Covered individuals for missed a pointments while enrolled in the Medicaid Covered Individuals for missed a pointments while enrolled in the Medicaid programs. This provider shall remain in effect wen in the event Plan becomes insolvent (Contract, Article X, paragraph A.11).
 - Fred Services if the Medicaid Covered Medicaid overed Individue for non-C 6.1. Provider may bill ce being rendered. The form of Individual agrees writing to ay for the service prior e admissions, s vices ocedures that are non-Covered pecifically state agreement must moul of out of pocket pen to beincurred by the Medicaid Services and the a mate Covered Individual.
- 7. Nothing herein shall be construed to prohibit Provider from contracting out offer Medicaid managed care organizations.
- 8. Compensation:

5.

6.

- a. For covered services provided to Covered Incuiduals, Provider shall be compensated at the lesser of the State of Wisconsin Medicaid Fee Shedule ("Feel Shedule") on file with Plan or billed charges.
- b. Upon written notification from the State of Wisconstato Place of a charge in the State of Wisconsin Medicaid Fee Schedule, Plan will have up to ninety 90) days to use the changes in rates and notify Provider of any fee schedule updates. Plan hall us best e forts to update the fee schedules within thirty (30) days of receipt of the new Medicaid fee schedule files from the State of Wisconsin. In no event shall the effective date of the rate changes extend past the effective date announced by the State or thirty (30) days after the date the State provided the new Medicaid Fee Schedule to Plan, whichever is later.
- 9. Notice: Provider shall provide thirty (30) days prior notice of any change in information or status that would affect Provider's participation in Plan's Medicaid Network or claims payment status (e.g. change of address, physician status change, etc.) to the following address:

Anthem Blue Cross and Blue Shield Attn: VP Network Management N17 W24340 Riverwood Drive Waukesha, WI 53188

- 10. Order of Precedence: All other provisions of the Agreement shall remain in full force and effect. In the event of a) a conflict between the provisions of this Amendment and the provisions of the Agreement or b) any inconsistency or ambiguity in this Amendment, such conflict, inconsistency or ambiguity shall be resolved by giving precedence in the following order: i) State or Federal law, rule, regulation or ordinance; ii) this Amendment for purposes of administering Plan's managed Medicaid plan or rendering services to Medicaid enrollees and iii) the Agreement.
- 11. Nothing herein will be construed to prohibit or restrict Provider from advising a Medicaid Covered Individual about his/her/its health status, medical care, or treatment, regardless of whether benefits for such care are available for the Medicaid Covered Individual, if Provider is acting within the lawful scope of practice. However, this provision does not require Provider to provide Health Services if Provider objects to such service on moral or religious grounds (42 CFR 438.102).
- 12. Provider certifies that neither it nor its principals nor any of its subcontractors are presently debarred, suspended, proposed for encomment, declared ineligible, or voluntarily excluded from entering into this Attachment by any Federa alency or by any department, agency or political subdivision of the State. For purposes of this Attachment "principal" means an officer, director, owner, partner, key employee, or other ensonwith primary management or upervisory responsibilities, or a person who has a critical influence or substant e control over covider oper ones (42 CFR 438.610). The Provider agrees to comply with requirements set forth in 42 CFF 455 not through 455.106 (regarding disclosure by providers of ownership and control information and doclosure control on a provider's owners' and other persons' conviction of the state required to control owners against Meditare, Molicaid, or Title XX services program) and will agree to provide required to control entities against to notify the HMC within curteen (14) days of any changes to the required disclosures.
- 13. Provider agrees to able by clan's marketing aformation requirements. Provider shall forward to Plan for prior approval all flyer, browners, letters and paraphlets Provider stends to distribute to Medicaid Covered Individuals concerning in payar filiations, or changes in affiliation or relating directly to Medicaid and BadgerCare population. Provider without distribute and marketing or recipient informing materials without the consent of Plan or the Wiscons) Department of Health and Farry Services (Contract Article X, paragraph A.16).

ng:

- 14. Provider agrees to comply with the following of the second se
 - a. Title VI of the Civil Rights set of 1960 (Public 88-352), as amened, are an requirements imposed by or pursuant to the regulation of the Popartment of Health and Honan Services (45 CFR Part 80), to the end that, in accordance with Title VI of that Act and the Regulation, ho person in the United States shall on the ground curace, color, or national origin, as excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under any program or activity for which Provider receives Feorgal associate.
 - b. Section 504 of the Rehabilitation Act of 1970 (Pub. L. 920.12) has amended, and all requirements imposed by or pursuant to the Regulation of the Deprement of Health and Human Services (45 CFR Part 84), to the end that, in accordance with faction 504 of that Act and the Regulation, no otherwise qualified handicapped individual in the Dited States show, holely by reason of his/her handicap, be excluded from participation in, be defied the Monefits of, or be subjected to discrimination under any program or activity for which Provider a ceives Federal assistance.
 - c. The Age Discrimination Act of 1975 (Pub. L 94-135), as amended, and a requirements imposed by or pursuant to the Regulation of the Department of Health and Human Services (45 CFR Part 91), to the end that, in accordance with the Act and the Regulation, no person in the United States shall, on the basis of age, be denied the benefits of, be excluded from participation in, or be subjected to discrimination under any program or activity for which Provider receives Federal assistance.
 - d. The Americans with Disabilities Act of 1990 (Pub. L. 101-336), as amended, and all requirements imposed by or pursuant to the Regulation of the Department of Justice (28 CFR 35.101 et seq.), to the end that in accordance with the Act and Regulation, no person in the United States with a disability, be excluded from participation in, be denied the benefits of, or otherwise be subjected to discrimination under any program or activity for which Provider receives Federal financial assistance.

- e. Title IX of the Educational Amendments of 1972, as amended (30 U.S.C. sections 1681, 1783, and 1685-1686), and all requirements imposed by or pursuant to regulation, to the end that, in accordance with the Amendments, no person in the United States shall, on the basis of sex, be excluded from participation in, be denied the benefits of, or otherwise be subjected to discrimination under any program or activity for which Provider receives Federal financial assistance.
 - §16.765 Wis. Stats., and the Civil Rights Act of 1964, as amended, and any other applicable State or Federal law, regulations and executive orders prohibiting discrimination, in that Provider shall not discriminate against any employee or applicant for employment in the performance of this Attachment. Provider shall not discriminate with respect to the hire, tenure, terms, conditions or privileges of employment or any matter directly or indirectly related to employment, because of race, color, religion, sex, disability, national origin, ancestry or status as a veteran. Breach of this provision shall be considered default; and
- g. All requirements appropriate to Provider under the Health Insurance Portability and Accountability Act of 1996.

f.

MEDICARE ADVANTAGE PARTICIPATION ATTACHMENT TO THE ANTHEM BLUE CROSS AND BLUE SHIELD PROVIDER AGREEMENT

This is a Participation Attachment to the Anthem Blue Cross and Blue Shield Provider Agreement ("Agreement"), entered into by and between Anthem and Provider and is incorporated into the Agreement.

ARTICLE I DEFINITIONS

The following definitions shall apply to this Medicare Advantage Participation Attachment:

- 1.1 "Clean Claim" means a Claim that has no defect or impropriety, including a lack of required substantiating documentation, or particulated cumstances requiring special treatment that prevents timely payment from being made on the Claim. A claim is clean even though Plan refers it to a medical specialist within Plan for examination. If additional documentation (e.g., a medical record) involves a source outside Plan, then the praim is not considered clean.
- 1.2 "Covered Individual" means, for purposes of this Attachment, a Medicare beneficiary covered under a Medicare agreement between CM and Plan under Part C of Title XVIII of the Social Security Act "Medicare Arise tage Program").
- 1.3 "Energence of Emergency Medical condition means a medical condition manifesting itself by acute symptoms of sufficiences verify (including a very pain) such that a prudent layperson with an average knowledge of heat of a dimedicine, could reisonably expect the absence of immediate medical attention to result in: (1) services je pardono the health of the individual or of the case of a pregnant woman, the health of the woman or her onbot or child; (2) serious in pairment to pools, functions; or (3) serious dysfunction of any bodily organ or pote
- 1.4 "Emergency Services" means covered inpatient and outpatient H with Services that are: (1) furnished by a provider qualified to furning emergency services; and (2) neared to a uate or stabilize an Emergency Medical Condition.
- 1.5 "CMS" means the Centers for Medice e an Medicaid Services.
- 1.6 "Downstream Entity(ies)" means any party that evers into a written arrangement, acceptible to CMS, with persons or entities involved with the readicate Advantage benefit, below the level of the arrangement between Anthem and a First Tier Entity. These written arrangements continue down to the level of the ultimate provider of both health and administrative service.
- 1.7 "First Tier Entity(ies)" means any party that enter into a written greement, acceptable to CMS, with Anthem or applicant to provide administrative services or healing can service for a Medicare eligible individual under the Medicare Advantage Program.
- 1.8 "Medicare" means the Health Insurance for the Aged Act, the XVIN of the social Security Act, as then constituted or later amended.
- 1.9 "Related Entity(ies)" means any entity that is related to Anthem by summon own rship or control and (1) performs some of Anthem's management functions under contract or subgation (2) furnishes services to Medicare enrollees under an oral or written agreement; or (3) leases results property or sells materials to Anthem at a cost of more than twenty-five hundred dollars (\$2,500) during a contract period.
- 1.10 "Urgently Needed Care" means Covered Services provided when a Covered Individual is either:
 - 1.10.1 Temporarily absent from Plan's Medicare Advantage service area and such Covered Services are Medically Necessary and immediately required: (1) as a result of an unforeseen illness, injury, or condition; and (2) it was not reasonable, given the circumstances, to obtain the services through Plan's Medicare Advantage Network; or
 - 1.10.2 Under unusual and extraordinary circumstances, the Covered Individual is in the service area but Plan's provider Network is temporarily unavailable or inaccessible and such Covered Services are

Medically Necessary and immediately required: (1) as a result of an unforeseen illness, injury, or condition; and (2) it was not reasonable, given the circumstances, to obtain the services through Plan's Medicare Advantage Network.

ARTICLE II SERVICES/OBLIGATIONS

- 2.1 <u>Participation-Medicare Advantage</u>. As a participant in Plan's Medicare Advantage Program, Provider will render Covered Services to Covered Individuals enrolled in Plan's Medicare Advantage Program in accordance with the terms and conditions of the Agreement and this Attachment. Except as set forth in this Attachment, or the Plan Compensation Schedule ("PCS") attached to the Agreement, all terms and conditions of the Agreement will apply to Provider's participation in Plan's Medicare Advantage Program(s). This Agreement does not apply to any of the Plan's Medicare Advantage Private Fee for Service or Medical Savings Account Programs.
- 2.2 <u>Participation-Out of Area Partams</u>. Pursuant to the Blue Cross and Blue Shield Out of Area Program section of the Agreement Privider hereby acknowledges and agrees that Provider shall provide Covered Services to any person which is covered under another Blue Cross and Blue Shield Plan under the Blue pross and Blue Shield As ociented of Area Program, including, but not limited to, a network sharing PPO developed to support under a Adventure Programs.
- 2.3 <u>Participation-Medicare Advantage Fogu</u>m. By virtue of the fact that Provider is a Medicare Advantage lock/Participating ring Provider, Fovider is reby atknowledges and agrees that Provider shall provide service any viedicare Advantage Covered Include Lenrolled in a Plan insured product that utilizes the Medicare Alventage Network.
- 2.4 <u>Covered Invidual Covered Service-Define</u>. The parties as as that all references in the Agreement to Covered Individual(s) clude covered Individuals of Plan's Mediane Advantage Program and all references to Covered Services include services offered pursuant to Plane. Mediane Advantage Program.
- 2.5 <u>Medical Necessity</u> Indicat recessity tecisions regarding Covered adividuals will be made in compliance with CMS guidelines.
- 2.6 <u>Accountability/Oversight</u>. Pixe delegates a Provider its responsibility and a its Madicare Advantage contract with CMS to provide the services as a solution in this Attachment of Covered Matviduals. Plan may revoke this delegation, including, if applicable, the delegated responsibility to use CMS reporting requirements, and thereby terminate the Attachment if CMS of Plan determinement. Provider has not performed satisfactorily. Such revocation shall be consistent with the termination provisions of this Attachment. Performance of the Provider shall be consistent with the termination provisions of this Attachment. Performance of the Provider shall be consistent with the termination provisions of this Attachment. Performance of all services. Provide acknowledges that Plan is accountable to CMS no the functions and responsibilities described in the Medicare Actionate acknowledges that Plan shall oversee and is accountable to CMS for the performance of all services. Provide acknowledges that Plan shall oversee and is accountable to CMS for the functions and responsibilities described in the Medicare Actionate acknowledges that Plan shall oversee and is accountable to CMS for the functions and responsibilities described in the Medicare Actionate and the Medicare Advantage regulatory standards as cultificately responsible to CMS for the functions and responsibilities described in the Medicare Advantage regulatory standards. Further, Provider acknowledges that Plan may only de gate such anchores and responsibilities in a manner consistent with the standards as set forth in 42 CFR § 420-504/144.
- 2.7 <u>Accountability/Credentialing</u>. Both parties acknowledge that a countability share in a manner consistent with the requirements as set forth in 42 CFR § 422.504(i)(4). Therefore the forwing are acceptable for purposes of meeting these requirements:
 - 2.7.1 The credentials of medical professionals affiliated with the tran or the Provider will be either reviewed by the Plan if applicable; or
 - 2.7.2 The credentialing process will be reviewed and approved by the Plan and the Plan must audit the Provider's credentialing process and/or delegate's credentialing process on an ongoing basis.
- 2.8 <u>Medicare Provider</u>. Provider must have a provider and/or supplier agreement, whichever is applicable, with CMS that permits them to provide services under original Medicare.

ARTICLE III ACCESS: RECORDS/FACILITIES

- 3.1 <u>Inspection of Books/Records</u>. Provider acknowledges that Plan, Health and Human Services department (HHS), the Comptroller General, or their designees have the right to timely access to inspect, evaluate and audit any books, contracts, medical records, patient care documentation, and other records of Provider, or his/her/its First Tier, Downstream and Related Entities, including but not limited to subcontractors or transferees involving transactions related to Plan's Medicare Advantage contract through ten (10) years from the final date of the contract period or from the date of the completion of any audit, or for such longer period provided for in 42 CFR § 422.504(e)(4) or other applicable law, whichever is later. For the purposes specified in this section, Provider agrees to make available Provider's premises, physical facilities and equipment, records relating to Plan's Covered Individuals, including access to Provider's computer and electronic systems and any additional relevant information that CMS may require. Provider acknowledges that failure to allow HHS, the Comptroller General or their designees the right to timely access under this section can subject Provider period fifteen thousand dollar (\$15,000) penalty for each day of failure to comply.
- 3.2 <u>Confidentiality</u>. Each party agrees to abide by all federal and state laws applicable to that party regarding connectitiality and disclosure for agrees to baintain records, medical records, other health information, and enrollee nformation. Provider agrees to baintain records and other information with respect to Covered Individures in an accurate and onely danner, to ensure timely access by enrollees to the records and information that pertain to their, and essafegued the privacy of any information that identifies a particular enrollee. Information from, or condex of, becords may be released only to authorized individuals. Provider must enrolle the unauthorized individuals cannot us access to or alter patient records. Original medical records must be released only in accordance to an federal or state laws, court orders or subpoenas. Both parties acknowledge out Plan, HHS, the Compteller General or its designee have the right, pursuant to section 3.1 power of udit and/or inspector ovider opremises to monitor and ensure compliance with the CMS requirements for maintaining the privace and security of potected health information (PHI) and other personally identifiable information of Covered Interviduals.

ARTICLE IV ACCESS BENEFITS AND COVERAGE

- 4.1 <u>Non-Discrimination</u>. Provider shall not determinit, or condition the fremsting of Health Services to Covered Individuals of Plan on the basis of any actor that is related to the alther attraction, but not limited to medical condition; claims experience net sign of health care; medical his ory genetic information; evidence of insurability, including conditions using out of acts of domestic violence or disability.
- 4.2 This provision intentionally left blank.
- 4.3 <u>Direct Access</u>. Provider acknowledges the Covered baividuals may obtain covered mammography screening services and influenza vaccinations from a participating provider where a referral and that Covered Individuals who are women may obtain we ten's routine and preventive Health Services from a participating women's health specialist without a referral.
- 4.4 <u>No Cost Sharing</u>. Provider acknowledges that covered influence vectors and pneumococcal vaccines are not subject to Covered Individual Cost Share obligations.
- 4.5 <u>Timely Access to Care</u>. Provider agrees to provide Covered Services or sistent with Plan's: (1) standards for timely access to care and member services; (2) policies and procedures that all w for individual Medical Necessity determinations; and (3) policies and procedures for the provider's consideration of Covered Individual input in the establishment of treatment plans.
- 4.6 <u>Continuity of Care</u>. A Provider who is a Primary Care Provider, or a gynecologist or obstetrician, shall provide Health Services or make arrangements for the provision of Health Services to Covered Individuals on a twenty-four (24) hour per day, seven (7) day a week basis to assure availability, adequacy and continuity of care to Covered Individuals. In the event a Provider is not one of the foregoing described providers, then Provider shall provide Health Services to Covered individuals on a twenty-four (24) hour per day, seven (7) day a week basis or at such times as Health Services are typically provided by similar providers to assure availability, adequacy, and continuity of care to Covered Individuals. If Provider is unable to provide Health Services as described in the previous sentence, Provider will arrange for another Network/Participating Provider to cover Provider's patients in Provider's absence.

ARTICLE V BENEFICIARY PROTECTIONS

- 5.1 <u>Cultural Competency</u>. Provider shall ensure that Covered Services rendered to Covered Individuals, both clinical and non-clinical, are accessible to all Covered Individuals, including those with limited English proficiency or reading skills, with diverse cultural and ethnic backgrounds, the homeless, and individuals with physical and mental disabilities. Provider must provide information regarding treatment options in a cultural-competent manner, including the option of no treatment. Provider must ensure that individuals with disabilities have effective communications with participants throughout the health system in making decisions regarding treatment options.
- 5.2 <u>Health Assessment</u>. Provider acknowledges that Plan has procedures approved by CMS to conduct a health assessment of all new Covered Individuals within ninety (90) days of the effective date of their enrollment. Provider agrees to cooperate with Plan as necessary in performing this initial health assessment.
- 5.3 <u>Identifying Complex and Star as Medical Condition</u>. Provider acknowledges that Plan has procedures to identify Covered Individuals with complex or serious medical conditions for chronic care improvement increases; and to assess these connects, including medical procedures to diagnose and monitor them on an ongoing basis; and establish and implement a treatment plan appropriate to those conditions, with an adequate number of direct access visits of becialists to accommodate the treatment plan. To the extent applicable, Provider agrees to assist in the development and implementation of the treatment plans and/or chronic care improvement initiative.
- <u>Ves.</u> Provider shall betablish any mantain written policies and procedures to implement uals' rights to make decisions of perning heir health care, including the provision of written all and Covered Individuals regarding their rights under state and federal law to make roling their right to accept or of use medical treatment and the right to execute an advance le. Provider different agrees to bocument or of use the documentation in the Covered idical records whether or not the Covered Individual has an advance directive, that Provider and records for advance directives and that Provider will provide for education Nired 5.4 Adv 20 Covered Inc. uals' ri all a information ardi decisions r medical direct Individuals' medical r ral requirements for advance directives and that Provider will provide for education e community chadvance directives. will follow state and of his/her/its staff and the c
- 5.5 <u>Standards of Care</u>. Provider agrees to provide Covered Services in a mamer consistent with professionally recognized standards of heat page.
- 5.6 <u>Hold Harmless</u>. Provider agrees that in the event including but not linked to con-payment by Plan, insolvency of Plan or breach of the ingreement, sho the Provider bill, using explicit a Veposit from, seek compensation, remuneration or reimbulgement from, or have any recourse a alinst a Covered Individual or persons other than Plan acting on their behal for Covered Services provide pursuant to the Agreement. This section does not prohibit the collection supplements charges or Cost phares of Plan's behalf made in accordance with the terms of the Covered adjuidual's fleath Benefit Plan or promounts due for services that have been correctly identified in advance as an on-Covered Service, subject to medical coverage criteria, with appropriate disclosure to the Covered andividual of the financial obligation. This advance notice does not apply to services not covered due to a statutor exclusion from the Medicare Advantage Program.
 - 5.6.1 Provider further agrees that for Covered Individuals who are dual using ble enrollees for Medicare and Medicaid, that Provider will ensure they will not bill be Covered Individuals for Cost Sharing that is not the Covered Individual's responsibility and such Covered Individuals will not be held liable for Medicare Parts A and B Cost Sharing when the State is liable for the Cost Sharing. In addition, Provider agrees to accept the Plan payment as payment in funer by using the appropriate state source.
- 5.7 <u>Continuation of Care-Insolvency</u>. Provider agrees that in the event of Plan's insolvency, termination of the CMS contract or other cessation of operations, Covered Services to Covered Individuals will continue through the period for which the premium has been paid to Plan, and services to Covered Individuals confined in an inpatient hospital on the date of termination of the CMS contract or on the date of insolvency or other cessation of operations will continue until their discharge.
- 5.8 <u>Survival of Attachment</u>. Provider further agrees that: (1) the hold harmless and continuation of care sections shall survive the termination of the Covered Individual; (2) these provisions supersede any oral or written contrary agreement now existing or hereafter entered into between Provider and a Covered Individual or

persons acting on their behalf that relates to liability for payment for, or continuation of, Covered Services provided under the terms and conditions of these clauses; and (3) any modifications, addition or deletion to these provisions shall become effective on a date no earlier than fifteen (15) days after the Administrator of CMS has received written notice of such proposed changes.

5.8.1 Survival after Termination. To the extent the Agreement terminates before this Attachment, the parties agree that all necessary terms of the Agreement will survive to allow continuation of this Attachment until the effective date of the termination of the Attachment.

ARTICLE VI COMPENSATION AND FEDERAL FUNDS

- 6.1 <u>Compensation-Medicare Advantage</u>. For Covered Services provided to Covered Individuals, Provider shall be compensated in accordance with the Medicare Advantage rate in effect at the time the Covered Service is rendered, and as set forth of the PCS attached to the Agreement. Such PCS may be amended from time to time as provided for in the greement.
- 6.2 <u>nompeRayment</u>. Plan agrees to use a best efforts to pay a majority of Clean Claims for Covered Services submittee by or on behalf a covered Individuals, within forty-five (45) days of receipt by Plan. Plan agrees to make best efforts to pay all a maining Clean Claims for Covered Services submitted by or on behalf of Covered Individuals, within study (60 days of receipt by Plan. Plan agrees to make best efforts to pay all a maining Clean Claims for Covered Services submitted by or on behalf of Covered Individuals, within study (60 days of receipt by Plan. Plan agrees to make best efforts to pay all a clean Clean Clean Clean Clean Clean for Covered Services significant of the necessary document days to adjudicate the Claim.
- 6.3 <u>Federal Funct</u>. Provent acknowledges the payments Provider receives from Plan to provide Covered Services to lover a I dividuals are, in while or pay from oderal funds. Therefore, Provider and any of his/her/its subcouract is are subject to certai, laws that are a Nicable to individuals and entities receiving federal funds, which if ay include but is not limite to, Title VI of the Civil Rights Act of 1964 as implemented by 45 CFR Part 84; the use Provider and Act of 1975 are implemented by 45 CFR Part 91; the Americans with Disabilities Act to be Ruabilitatio. Act of 1973 are any other egulations applicable to recipients of federal funds.

REPORTING ALL DISCLOSURE REQUIREMENTS

- 7.1 <u>Risk Adjustment Data Validation Vidits</u>. Pue and provider are required in secondarie with 42 CFR § 422.310(e) to submit a sample of melical record for Covered Individuals for the purpose of validation of risk adjustment data. Accordingly, Plan, wither designee, shall have the right as set forth in section 3.1 to obtain copies of such documentation on at hist an annual tesis. Provider agrees to provide the requested medical records to Plan, or their designee, within for seen (14) calendar laws om Plan's, or their designee's, written request. Such records shall be provided to Plan, or their designee, at no additional cost.
- 7.2 Data Reporting Submissions. Provider agrees to provide to Plan a rink mation necessary for Plan to meet its data reporting and submission obligations to CMs sincluster, but not limited to, data necessary to characterize the context and purpose of each encounter between a Covered Individual and the Provider ("Risk Adjustment Data"), and data necessary for Plan to must its reporting a ligations under 42 CFR §§ 422.516 and 422.310. In accordance with the CMS requirements, the Plan reserves the right to assess Provider for any penalties resulting from Provider's submission of face circa.
- 7.3 <u>Risk Adjustment Data</u>. Provider's Risk Adjustment Data shall include a informer on necessary for Plan to submit such data to CMS as set forth in 42 CFR § 422.310 or any submated or additional regulatory provisions. If Provider fails to submit his/her/its Risk Adjustment Data accurately, completely and truthfully, in the format described in the 42 CFR § 422.310 or any subsequent or additional regulatory provisions, then this will result in denials and/or delays in payment of Provider's Claims.
- 7.4 <u>Accuracy of Risk Adjustment Data</u>. Provider further agrees to certify the accuracy, completeness, and truthfulness of Provider generated Risk Adjustment Data that Plan is obligated to submit to CMS. Within thirty (30) days after the beginning of every Fiscal Year or as required by CMS while the Medicare Advantage Participation Attachment is in effect, Provider agrees to give Plan a certification in writing, in a format that Plan specifies, that certifies to the accuracy, completeness, and truthfulness of Provider's Risk Adjustment Data submitted to Plan during the specified period.

32

ARTICLE VIII QUALITY ASSURANCE/QUALITY IMPROVEMENT REQUIREMENTS

- 8.1 <u>Independent Quality Review Organization</u>. Provider agrees to comply and cooperate with an independent quality review and improvement organization's activities pertaining to the provision of Covered Services for Covered Individuals.
- 8.2 <u>Compliance with Plan Medical Management Programs</u>. Provider agrees to comply with Plan's medical policies, quality improvement and performance improvement programs, and medical management programs to the extent provided to or otherwise made available to Provider in advance.
- 8.3 <u>Consulting with Network/Participating Providers</u>. Plan agrees to consult with Network/Participating Providers regarding its medical policies, quality improvement program and medical management programs and ensure that practice guidelines and utilization management guidelines: (1) are based on reasonable medical evidence or a consensus of botth care professionals in the particular field; (2) consider the needs of the enrolled population; (3) are acceleded in consultation with participating physicians; (4) are reviewed and updated periodically; and (5) are communicated to providers and, as appropriate, to Covered Individuals. Planelso agrees to consult that do bions with respect to utilization management, Covered Individual aducation coverage of Heath Carvinus, and other areas in which the guidelines apply are consistent with the guidelines.

- 9.1 <u>Compliance Ledicare haws/Regulations</u>. Prover agrees to comply, and to require any of his/her/its subcontractor to comply, with all applicable when are laws, regulations, and CMS instructions. Further, Provider agrees that any Covered Services rovided by the nevider or his/her/its subcontractors to or on behalf of the Fran's covered Individuals with the consistent of the and will comply with Plan's Medicare Advantage contractual obligations.
- 9.2 <u>Compliance: Excluse a from Federal realth Care Program</u>. Provide may not employ, or subcontract with an individual, or have periods with othership or control interest, when are been convicted of criminal offenses related to their in plycement in Medicaid, Medicare, or occile et ices programs under Title XX of the Social Security Act, and has have been et gluded from participation it any federal health care program under §§ 1128 or 1128A of the Act (or oth an entity that employ, or contract with such an individual) for the provision of any of the following:
 - 9.2.1 healthcare;
 - 9.2.2 utilization review;
 - 9.2.3 medical social work; or
 - 9.2.4 administrative services.
- 9.3 <u>Compliance: Appeals/Grievances</u>. Provider agrees to couply we Plan's policies and procedures in performing his/her/its responsibilities under the Agreement. Provider specially agrees to comply with Medicare requirements regarding Covered Individual appeals an grievance and expedited appeals, including the gathering and forwarding of information in a timely manner and compliance with uppeals decisions.
- 9.4 <u>Compliance: Policy and Procedures</u>. Provider agrees to comply with Pours policy and procedures in performing his/her/its responsibilities under the Agreement and this Attachment including any supplementary documents that pertain to Plan's Medicare Advantage Program such as the Product Guide.
- 9.5 <u>Illegal Remunerations</u>. Both parties specifically represents and warrants that activities to be performed under this Agreement are not considered illegal remunerations (including kickbacks, bribes or rebates) as defined in 42 USCA § 1320(a)-7b.
- 9.6 <u>Compliance: Training, Education and Communications</u>. In accordance with, but not limited to 42 CFR §§ 422.503(b)(4)(vi)(C)&(D) and 423.504(b)(4)(vi)(C)&(D), Provider agrees and certifies that it, as well as its employees, subcontractors, Downstream Entities, Related Entities and agents who provide services, to or

for Plan's Medicare Advantage and/or Part D Covered Individuals or to or for the Plan itself, shall participate in applicable compliance training, education and/or communications as reasonably requested by the Plan or its designee annually or as otherwise required by applicable law, and must be made a part of the orientation for a new employee, new First Tier, Downstream or Related Entity and for all new appointments of a chief executive, manager, or governing body member. Both parties agree that the Plan or its designee may make such compliance training, education and lines of communication available to Provider in either electronic, paper or other reasonable medium. Provider shall be responsible for documenting applicable employee's, subcontractor's, Downstream Entity's, Related Entity's and/or agent's attendance and completion of such training. Upon notice, Provider shall provide such documentation to Plan, unless otherwise not required by CMS regulation. In addition, the training requirement set forth herein is not required for providers or suppliers who have met the fraud, waste and abuse certification requirements through enrollment into the Medicare program, as those providers and/or suppliers are deemed to have met that portion of the fraud waste and abuse training required by CMS.

ARTICLE X MARKETING

10.1

<u>coroul of Materials</u>. Both particle gree to comply, and to require any of his/her/its subcontractors to comply, with all applicable in aerals and stronglaws, regulations, CMS instructions, and marketing activities under this Agreement, including but not united to the Medicare Marketing Guidelines for Medicare Managed Care Plans and any requirements for all spring pproval of materials. Any printed materials, including but obtained to contract to Plan Govered addividuals, brochures, advertisements, telemarketing scripts, packaging upper ed or produced by Provider or an context bis/her/its subcontractors pursuant to this Agreement muscles is milled to Plan for review and appropriate activities planning stage (i.e., creative, copy, mechanicals, blue lines, et al. to asser compliance with fed tail, tate, and Blue Cross/Blue Shield Association guidelines. Plan agrees is approved will not be unrease ably witcheld or relayed.

ARTALE XI TERM AND TERMINATION

- 11.1 <u>Notice Upon Termination</u>. Plan decrees to terminate this Attachment Plan shall give Provider written notice, to the extent required under CMS regulations, of the rear as for a faction, including, if relevant, the standards and the profiling state the organization used to evaluate a row for and the numbers and mix of Network/Participating Provide. The numbers are decreased written notice shall also be forth Provider's right to appeal the action and the process an arm of for requesting a hearing.
- 11.2 <u>Termination for Medicare Exclusion</u>. Provider act nowledges that this attactment shall be terminated if Provider, or a person or entity with owner bip or control interest in Provider, is excluded from participation in Medicare under § 1128A of the Social Security Act or from participation in any other federal health care program.
- 11.3 <u>Termination Without Cause</u>. Either party may terminate this Medicar. Advantage Participation Attachment without cause by giving at least one hundred eighty (1.0) days prior with an notice of termination to the other party. Upon your notice of Termination Without Cause, Providence required to notify Covered Individual(s) sixty (60) days prior to your effective date of termination with Attachment
- 11.4 <u>Term/Termination</u>. This Attachment shall continue in effect unit is otherwise terminated as provided for in this Attachment or in the Agreement.

ARTICLE XII GENERAL PROVISIONS

- 12.1 <u>Inconsistencies</u>. In the event of an inconsistency between terms of this Attachment and the terms and conditions as set forth in the Agreement, the terms and conditions of this Attachment shall govern. Except as set forth herein, all other terms and conditions of the Agreement remain in full force and effect.
- 12.2 <u>Interpret According to Medicare Laws</u>. Provider and Plan intend that the terms of the Agreement and this Attachment as they relate to the provision of Covered Services under the Medicare Advantage Program shall be interpreted in a manner consistent with applicable requirements under Medicare law.
- 12.3 <u>Subcontractors</u>. Provider agrees that if Provider enters into subcontracts to perform services under the terms of this Attachment, Provider's subcontracts shall include: (1) an agreement by the subcontractor to

comply with all of the Provider's obligations in the Agreement and this Attachment; (2) a prompt payment provision as negotiated by the Provider and the subcontractor; (3) a provision setting forth the term of the subcontract (preferably one (1) year or longer); and (4) dated signatures of all the parties to the subcontract.

- 12.4 <u>Delegated Activities</u>. If Plan has delegated activities to Provider, then the Plan will provide the following information to Provider and Provider shall provide such information to any of its subcontracted entities:
 - 12.4.1 A list of delegated activities and reporting responsibilities;
 - 12.4.2 Arrangements for the revocation of delegated activities;
 - 12.4.3 Notification that the performance of the contracted and subcontracted entities will be monitored by the Plan;
 - 12.4.4 Notification that the pentialing process must be approved and monitored by the Plan; and
 - 12.4.5 Notification the full contracted and subcontracted entities must comply with all applicable Medicare laws, regulate ns and CMP productions.
- 12.5 Delegation of Provider Selection on addition the responsibilities as set forth in section 12.4 above, to the extent that Plan has delegated selection of the providers, contractors, or subcontractor to Provider, the Plan retains the right approve, suspend, or terminate any such arrangement.

Plan Compensation Schedule (PCS) Attachment

Set forth below are the financial terms for the Agreement between Anthem and Provider. Anthem may, in its sole discretion, review and amend the reimbursement described below from time to time. Coded Service Identifiers referenced in the Plan Fee Schedule shall be updated from time to time.

Commercial Lines of Business

- 1. Primary Care Provider will be compensated in accordance with the following:
 - (a) Vaccinations: Will be reimbursed at the Anthem Vaccine Fee Schedule.
 - (b) Drugs: Will be reimbursed based on one hundred percent (100%) of CMS and adjusted quarterly.
 - (c) HCPCS/DME: Will or reimbursed at one hundred percent (100%) of the Wisconsin CMS DMEPOS Fee Schedule are are disted annually.
 - (c. Plan Fee Schedule (PEStave 12/WI113), which is based upon a Wisconsin Adjusted Resource Based Relative Value State (RFAVS), corporating CMS site of service differentials and gap-filled with ngenix relative value state, cunican ab services will be based on Wisconsin Adjusted CLIA with Medicare and Optum gap-fillen a Plan is e Schedule is outlined on Exhibit A.
- 2. ed a overed Individual that is not listed by code on the I be based on an amount equal to the billed charge less Con tion or a Covered Service provided applicable Fee S edule references abr a twenty-five ercer %) discount. The mount the discount will be calculated using the charge prior to the application of C Notwiths inding the foreging, Anthem reserves the right to price Not st Shares. Otherwise Clar fied (COC) des in accordance with its NOC in ursement policies.
- 3. With respect to certain reality enefit clans, psychiatric or a and cherneal dependency rehabilitation services are Covered Services only of the extent that these services are indentify to Medically Necessary Health Services provided to a Covered Individual and such services are not or envise reimbursable by a party subcontracting with Anthema provide such a rage.

Government Lines of Business

Medicare HMO / PPO

1. For Covered Services provided to Covered Individuals, Provider shall be companisated at the Anthem Rate which is one hundred percent (101.6) of the current Medicare Advantage Plan Fee Schedule for the year in which services were provided.

Medicaid

2. For Covered Services provided to Covered Individuals, Provider chall be compensated at the lesser of the State of Wisconsin Medicaid Fee Schedule ("Medicaid Fee Schedule" on file with Anthem of billed charges.

Exhibit A Plan Fee Schedule

	NEW FEE SCHEDULE SOURCE	WI112/WI113
CPT CATEGORY	BASED ON	
Allergy 95165	MEDICARE	110.0%
Allergy 95105	BASED ON	
Aller y Testing	MEDICARE	110.0%
Alle yy restilig	BASED ON	
Autor omionervous eastern function text	MEDICARE	100.0%
Autor and her vous a territarioteria	BASED ON	
Cardiology Holte, Nonitor	MEDICARE	110.0%
	BASED ON	
Chemotherapy as ministration	MEDICARE	185.0%
	BA FED ON	
Chiropractic	MEDICARE	83.0%
	BA ED ON	
Clinical Lab	AEDIC A	107.0%
	BALLD (N	
Clinical Laboratory	MEDIC/ RE	107.0%
	BASED FIXE	
Delivery 59400	RATE	\$3,832.91
	BASED ON FKED	
Delivery 59409	RATE	\$1,499.40
	BASED ON FIXED	
Delivery 59410 ,	RA _	\$1,910.70
	SED C V FIXED	
Delivery 59510		\$4,232.31
	BASED ON ALT	
Delivery 59514	IN TE	\$1,684.10
	BASL ON FIXER	
Delivery 59515	RATE	\$2,311.18
	BASED ON	
Dermatology - Destruction of premalignant or benign lesions	MEDICARE	150.0%
DME	BASED ON CURRENT	100.0%
	BASED ON MEDICARE	100.0%
DME	BASED ON	
Endovenous ablation therapy	MEDICARE	130.0%
Endovenous ablation therapy		
	1	

Evaluation and Management - ER	BASED ON	200.0%
	MEDICARE	
	BASED ON	
Evaluation and Management - Neonatal	MEDICARE	150.0%
	BASED ON	
Evaluation and Management - Preventive	MEDICARE	165.0%
· · · · · ·	BASED ON	1001070
Evaluation and Management	MEDICARE	150.0%
	BASED ON	100.070
Immunization Admin	MEDICARE	100.0%
	BASED ON	100.070
Injectible Drug - Non-chemotherary	MEDICARE	100.0%
	BASED ON	100.070
In-Office Drugs Yey Generics	MEDICARE	100.0%
	BASED ON	100.070
Intra perative neurophysiology monitoring	MEDICARE	100.0%
man et autor insure d'aloregy mentioning	BASED ON	100.078
Lab - Pathology	MEDICARE	110.0%
	BASED ON FIXED	\$
Lab tests - other - 005	ATE	43.70
	JA SD ON	43.70
Lab tests - other - Drug scree	MEDICARE	107.00/
Law tests other brug serve	BAS JON FIXED	107.0%
Major procedure, orthopedic 2, 533	P ATE	2794.05
	BASJO	2794.05
Medicine - Allergy & Clinical Immunology		135.0%
meaner / mergy of entited minimizations	BASED	133.0%
Medicine - Cardiovascular	MEDICAKE	130.0%
Medicine - Immun./Vaccines	BASED ON CURREN	100.0%
	BASED ON	
Medicine - Neurology	MEDICARE	145.0%
<u> </u>	BASCON	1010/0
Medicine - Ophthalmology	DICAN	110.0%
Medicine - Other	BASE ON CUPRENT	100.0%
	ASED ON	100.078
Medicine - Other	MEDICAL	100.0%
	BASEL ON	100.0%
Medicine Health and Behavior Assessment	MEDICAN	100.0%
Medicine Health and Benavior Assessment	BASED ON	100.0%
Medicine Neuro Sleep Study	MEDICARE	120.00/
	BASED ON	130.0%
Medicine Nutrition	MEDICARE	100.00/
	BASED ON FIXED	100.0%
Medicine PT OT 97139		\$
Medicine Fr OF 57 155		5.00
Medicine PT OT	BASED ON	00.004
	MEDICARE	86.0%

.

#1061709754

,

,

Medicine Vascular Diagnostic Studies	BASED ON MEDICARE	130.0%
	BASED ON	
Minor procedures - musculoskeletal 20550-20553	MEDICARE	175.0%
Miscellaneous Service	BASED ON CURRENT	100.0%
	BASED ON	
Miscellaneous Service	MEDICARE	100.0%
· ·	BASED ON	
Neurology Nerve Conduction Tests	MEDICARE	120.0%
	BASED ON	
Post operative sinus debridement	MEDICARE	90.0%
	BASED ON	
Radiology High Tech	MEDICARE	155.0%
	BASED ON	100.000
Radiology Imaging CAD	MEDICARE	100.0%
	BASED ON	100.00/
Radio gy WRT	MEDICARE	100.0%
	BASED ON MEDICARE	145.0%
Radiology Non-O. Magnoric Ultrasound	BASED ON	145.0%
Dadialamy Ultrace in midance		145.0%
Radiology Ultrasovic aidaace	BALED ON	1-13:070
Radiology	MED ARE	155.0%
Tradiology	B' _ ED O FIXED	\$
Radiopharmaceutical agent A9 00	ATE	80.00
	BALLO CALFIXE RATE	\$ 24.00
Specialist - ophthalmology - 92015	BASED	24.00
Currow Auditory	MEDICARE	135.0%
Surgery - Auditory	BASED ON	100.070
Surgery - Cardiovascular	MEDICARE	140.0%
	BASED ON	
Surgery - Digestive	MEJINRE	130.0%
	DI SED ON	
Surgery - Endocrine	MED SARE	135.0%
	PASED ON	
Surgery - Eye & Ocular Adnexa	MLDIC KE	130.0%
	BASE. ON	
Surgery - Female Genital	MEDICAL	165.0%
	BASED ON	
Surgery - Hemic & Lymphatic	MEDICARE	135.0%
	BASED ON	450.00
Surgery - Integumentary	MEDICARE	150.0%
	BASED ON	160.0%
Surgery - Male Genital	MEDICARE	160.0%
Surgery - Maternity & Delivery	BASED ON	
	MEDICARE	150.0%

.

	BASED ON	
Surgery - Mediastinum & Diaphragm	MEDICARE	130.0%
		150.0%
	BASED ON	
Surgery - Musculoskeletal	MEDICARE	175.0%
	BASED ON	
Surgery - Nervous	MEDICARE	150.0%
	BASED ON	
Surgery - Respiratory	MEDICARE	160.0%
	BASED ON	
Surgery - Urinary	MEDICARE	175.0%
	BASED ON	
Venipuncture	MEDICARE	133.3%



Plan Compensation Schedule (PCS) Attachment

Set forth below are the financial terms for the Agreement between Anthem and Provider. Anthem may, in its sole discretion, review and amend the reimbursement described below from time to time. Coded Service Identifiers referenced in the Plan Fee Schedule shall be updated from time to time.

Commercial Lines of Business

1. Behavior Health Provider will be compensated in accordance with the following:

Plan Fee Scheduler PFS - WI378411556/WI379/WI557/WI380/WI558/WI381/WI559/WI382/WI560), which is based upon a Wisco sin fajor of Resource Based Relative Value Scale (RBRVS) incorporating CMS site of service differences and pap-filler with Ingenix relative value units. A representative sample of the Plan see Schedule in effect on the time the representative secured is attached.

- 4. Compared tion or a Covered Service provided to a covered Individual that is not listed by code on the applicable. We Fee Sciedule referenced above will be used on an amount equal to the billed charge less a twenty-five reception %) discount. The provided the discount will be calculated using the charge prior to the application of C at Shares. Notwith transing the fore sing. Anthem reserves the right to price Not Otherwise Class and (OC) rules in accordance with its NOC runbursement policies.
- 5. With respect to certal. Heat r Benefit Hans, psychiatric care and chemical dependency rehabilitation services are Covered Services only to the extent that these services are incidently to Medically Necessary Health Services provided to a Covered Individual and such services are no cherwise reimbursable by a party subcontracting with Anthem to provide such coverage.

Government Lines of Business

Medicare HMO / PPO

1. For Covered Services provided to Covered Inductuals, Provider shall be compendent the Anthem Rate which is 100% of the current Medicare Advantage Plan ree Schedulerfor the year in which services were provided.

Medicaid

1. For Covered Services provided to Covered Individuals, Provider hall be compensed at the lesser of the State of Wisconsin Medicaid Plan Fee Schedule on file with Plan or biller charges.

Effective 10/1/2013											1	
Type of Service	СРТ	Description	WI378 MD	WI556 MD	WI379 CADC	WI567 CADC	Wi380 Master's Level	WI658 Master's Level	WI381 PHD/PS YD	W1559 PHD/PS YD	WI382 RN/NP	WI560 RN/NP
			Non Facility	Facility	Non Facility	Facility	Non Facility	Facility	Non Facility	Facility	Non Facility	Facility
		Psychiatric diagnostic					1				, , , , , , , , , , , , , , , , , , ,	
		evaluation (no						ļ				
Medicine - Other	90791	medical services)	\$183,92	\$143.45	\$109.11	\$85.10	\$155.84	\$121.55	\$165.22	\$128.86	\$155.84	\$121.55
		Psychiatric diagnostic										
		evaluation with										
Medicine - Other	90792	medical servicer	152.31	\$148.22	\$90.35	\$87.93	\$129.06	\$125.59	\$136.82	\$133.15	\$129.06	\$125.59
		complexity d										
Medicine ver	785	on code	-13	\$5.73	\$3.40	\$3.40	\$4.86	\$4.86	\$5.15	\$5.15	\$4.86	\$4.86
Medici - Other	9 32	Psychotherapy,3	46	50.11	\$45.36	\$35.66	\$64.79	\$50.94	\$68.69	\$54,00	\$64.79	\$50.94
		30 minute			\$ 15.55	\$55.00	201.75	40.000	208.05		304.79	\$50.94
Medicine -	90833	psychotherapy	\$56	\$50.	\$30	¢20.01	¢42.00	¢40.70	A45 63			4
Weddenie -	9000	F chotherapy.	200	\$50	\$30	\$29.91	\$43.08	\$42.73	\$45.67	\$45.30	\$43.08	\$42.73
Medicine - Other ┥	90. 1	min	\$102.00	\$102.00	.60.51	\$60.51	\$86.44	\$86.44	\$91.63	\$91.63	\$86.44	\$86.44
		5 minute psychot ^u										
Medicine - Other	90836	add o ode	\$82.61	\$82.6	\$49.0	\$49.6	\$70.00	\$70.00	\$74.21	\$74.21	\$70.00	\$70.00
Medicine - Other	90837	Predother y,	A144 70	¢120.44	tor or	40.0						
Medicine - Other	50037	60 minutes	\$144.73	\$136.14	\$85.86	\$80.7	22.63	\$115.36	\$130.01	\$122.30	\$122.63	\$115.36
Medicine - Other	90838	psychoth add on c	\$133.41	\$132.59	\$79.14	\$78.66	\$1 4	\$112.35	\$119.84	\$119.11	\$113.04	\$112.35
Medicine - Other	90845	PSYCHOANALY SIS	\$93.25	\$91.12	\$55.32	ÉEA DC	¢70.00	677.04	600 TT	404.00	A	4
medicine - other	50045	FAMILY PSYTX	\$95.25	\$91.12	\$55.32	\$54.06	\$79.07	\$77.21	\$83.77	\$81.86	\$79.02	\$77.21
Medicine - Other	90846	W/O PATIENT	\$98.26		\$58.35	\$56.84	.34	\$81.18	\$88.36	\$86.06	\$83.34	\$81.18
Medicine - Other	90847	FAMILY PSYTX W/PATIENT	\$124.00	\$1. 10	\$.3.56	\$73.56	\$105.08	\$1 .08	\$111.39	\$111.39	\$105.08	\$105.08
	· · ·	MULTIPLE							, JIII.05	, fill, 55	\$105.00	\$105.00
Medicine - Other	90849	FAMILY GROUP PSYTX	\$37.04	\$33.21	.98	\$19.70	\$31.39		¢22.20	¢20.04	¢21.20	***
		GROUP		J.J.J.L.1		\$13.70	\$51,35	\$2 4	\$33.28	\$29.84	\$31.39	\$28.14
Medicine - Other	90853	PSYCHOTHERA PY	¢125.00	cape of	400.00							
Wedicine - Other	90855	NARCOSYNTHE	\$135.00	\$135.06	\$80.08	¢ ,8	\$114.41	\$114	\$1, 7	\$121.27	\$114.41	\$114.41
Medicine - Other	90865	SIS	\$175.43	\$148.60	14.07	\$88.16	\$148.65	\$125.92	ə157,59	\$133.49	\$148.65	\$125.92
		ELECTROCONV ULSIVE										
Medicine - Other	90870	THERAPY	\$154.57	\$97.08	\$91.69	\$57.59	130.97	\$82.26	\$138.85	\$87.21	\$130.97	\$82.26
		PSYCHOPHYSI										
Medicine - Other	90875	OLOGICAL THERAPY	\$80.48	\$68.55	\$47.74	\$40.	\$68.	ST	\$72.29	\$61.58	\$68.19	\$58.09
		PSYCHOPHYSI	,						<i>VILLU</i>		\$08,15	\$30.05
Medicine - Other	90876	OLOGICAL THERAPY	\$119.65	\$108.58	\$70.98	\$64.41	01.00	top op	107.10	407.54		
Wieuleine - Other	30070	HYPNOTHERAP	\$119.05	2108.28	\$70.98	\$64.41	101.38	\$92.00	107.48	\$97.54	\$101.38	\$92.00
Medicine - Other	90880	Y	\$117.95	\$110.28	\$69.97	\$65.42	\$99.9-	\$93.45	\$105.95	\$99.07	\$99.94	\$93.45
		PSY EVALUATION							,			
Medicine - Other	90885	OF RECORDS	\$55,35	\$55.35	\$32.84	\$32.84	\$46.90	\$46.90	\$49.72	\$49.72	\$46.90	\$46.90
Medicine - Other	90887	CONSULTATION	¢05.00	¢04.21	ATC DA	650 04	404.40					
	30007	WITH FAMILY PSYCHO	\$95.80	\$84.31	\$56.84	\$50.01	\$81.18	\$71.44	\$86.06	\$75.74	\$81.18	\$71.44
Medicine - Other	96101	TESTING BY PSYCH/PHYS	\$98.00	\$98.00	\$58.14	\$58.14	\$83.05	\$83.05	\$88.03	\$88.03	\$83.05	\$83.05
		PSYCHO TESTING BY TECHNICIAN										
Medicine - Other	96102	· · · · ·	\$58.33	\$25.97	\$34.61	\$15.41	\$49.43	\$22.01	\$52.40	\$23.33	\$49.43	\$22.01
		PSYCHO		+_0.57	1011-11	47.54	64,04,0	722,01		+20,00	ə49.43	\$22.01
Madicina Other	06100	TESTING ADMIN	¢10.00	A	An	40						
Medicine - Other	96103	BY COMP	\$43.00	\$43.00	\$25.51	\$25.51	\$36.44	\$36.44	\$38.63	\$38.63	\$36.44	\$36.44

PLAN COMPENSATION SCHEDULE/ATTACHMENT

Wisconsin Provider Agreement Attachment © 2013.1 Blue Cross Blue Shield of Wisconsin Oneida Community Health Center and Ka Ni Kuhli Yo Femily Center 09/29/2014

		NEUROPSYCH TST BY										
Medicine - Other	96118	PSYCH/PHYS	\$127.00	\$127.00	\$75.34	\$75.34	\$107.63	\$107.63	\$114.08	\$114.08	\$107.63	\$107.63
		NEUROPSYCH										
Madiaina Other	06110	TESTING BY	\$74.94	\$28.10	\$44.46	\$16.67	\$63,50	\$23.81	\$67.32	\$25.24	\$63.50	\$23.81
Medicine - Other	96119	TEC ASSESS	\$74,94	\$28.10		\$10.07		323.01	307.32	Ş2J.24	90.206	
		HLTH/BEHAVE,										
Medicine - Other	96150	INIT	\$21.29	\$20.93	\$14.90	\$14.65	\$21.29	\$20.93	\$22.57	\$22.18	\$21.29	\$20.93
		ÎNTERVENE HLTH/BEHAVE,										
Medicine - Other	96152	INDIV	\$19.48	\$19.12	\$13.64	\$13.39	\$19.48	\$19.12	\$19.48	\$19.12	\$19.48	\$19.12
		INTERV										
Maliature Others	00454	HLTH/BEHAV,	¢10.12	¢10.70	ć12 20	¢12.14	\$19.12	\$18,76	\$19.12	\$18.76	\$19,12	\$18.76
Medicine - Other	96154	FAM W/PT MED	\$19.12	\$18.76	\$13.39	\$13.14	\$15.1Z	\$18,70		<i></i>		
		NUTRITION,										
Medicine - Other	97803	INDIV, SUBSEQ	\$28.53	\$26.40	\$16.92	\$15.66	\$24.17	\$22.37	\$25.63	\$23.72	\$24.17	\$22.37
		DEVELOP OF										
		COGNITIVE										
_		SKILLS, PATIENT										
		CONTACT										
Mar distant and the sec	0751	PROVIDER,EAC	\$10		¢10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
Medicin Other	975	H 15 MINUTES	\$10		\$10.00	\$10.00	\$10.00	\$10,00	\$10.00	\$10.00	\$10.00	\$10.00
Evaluation and	99205	OFFICE/OUTIEN	\$211	\$177.	\$125 5	\$105.59	\$179.32	\$150.81	\$190.10	\$159.88	\$179.32	\$150.81
Managemen	99200	C ICE/OUTPA		2111		\$101,15	Ş173.32	<i></i>	\$150.10	<i></i>	<i>\$115.52</i>	
Evaluation and 🔫		INT VISIT										
Management	99212	ST	\$42,58	27.25	5.26	16.17	\$36.08	\$23,09	\$38.25	\$24.48	\$36.08	\$23.09
Evaluation and		OFFICE A										
Management	99213	E ²	\$73.24	\$54.08	\$43.45	\$32.08	\$62.06	\$45.82	\$65.79	\$48.58	\$62.06	\$45.82
		FICE/O PA										
Evaluation and	00014	EST		\$83.46	\$64.92	.51	\$92.7	\$70.72	\$98,30	\$74.97	\$92.73	\$70.72
Management	99214	INITIAL		205.40			<i>QJ2.</i>	\$70.72	\$50.50		<i>452.115</i>	<i></i>
Evaluation and		HOSPITAL							ъ.			
Management	99221	CARE	\$106,02	106.02	\$62.90	\$62.90	\$89.84	\$89.84	\$95.24	\$95.24	\$89.84	\$89.84
Evaluation and		INITIAL HOSPÍTAL										
Management	99222	CARE		4.77	\$85.88	\$85.88	.22.67	\$122,¢	\$130.05	\$130.05	\$122.67	\$122,67
		INITIAL										
Evaluation and	99223	HOSPITAL	\$212.90	\$212.90	\$126	\$126.30	\$180.40	\$180.4	\$191.	\$191.25	\$180.40	\$180.40
Management	99223	SUBSEQUENT	\$212,50	\$212,50	\$120	\$120,50	\$180.40	\$180.		\$151.25		
Evaluation and		HOSPITAL										
Management	99231	CARE	\$43.01	\$43.01	\$25.51	\$25,51	\$36.44	\$ 44	38.63	\$38.63	\$36.44	\$36.44
Evaluation and		SUBSEQUENT										
Management	99232	CARE	\$77.07	\$77.07	45.72	\$45.72	\$65,30	\$65.30	69.23	\$69.23	\$65.30	\$65.30
		SUBSEQUENT										
Evaluation and Management	99233	HOSPITAL	\$111.13	\$111.13	\$65,93	\$65.93	,94.17	\$94.17	\$99.83	\$99.83	\$94.17	\$94.17
wanagement	00200	HOSPITAL			, , , , , , , , , , , , , , , , , , , ,	405.35		45			+0.027	
Evaluation and		DISCHARGE										
Management	99238	DAY	\$75.79	\$75.79	\$44.96	\$4	\$64	\$67.2	\$68.09	\$68.09	\$64.22	\$64.22
Evaluation and		HOSPITAL DISCHARGE					4					
Management	99239	DAY	\$111.13	\$111.13	\$65.93	\$65.93	94.17	\$94.17	\$99.83	\$99.83	\$94,17	\$94.17
Evaluation and		OFFICE										
Management	99244	CONSULTATION	\$206.94	\$174.58	\$122.76	\$103.57	\$175.5	\$147.93	\$185.90	\$156,82	\$175.35	\$147.93
Evaluation and	00251		\$55.35	ČEE DE	\$32.84	\$32.84	\$46.90		\$49.72	\$49.72	\$46.90	\$46.90
Management Evaluation and	99251	INPATIENT	400. 0 0	\$55.35	432.84	40,24	,40.90			<i></i>	9-10.3U	-+U,JU
Management	99252	CONSULTATION	\$85.16	\$85.16	\$50.52	\$50.52	\$72.16	\$72,16	\$76.50	\$76.50	\$72.16	\$72.16
Evaluation and		INPATIENT										
Management	99253	CONSULTATION	\$129.87	\$129.87	\$77.04	\$77.04	\$110.04	\$110.04	\$116.66	\$116.66	\$110.04	\$110.04
Evaluation and	00054		6100 C2	¢100 C3	¢111.00	¢111.00	¢150.02	¢150.02	\$160 AF	\$169.45	\$159.83	\$159,83
Management	99254	CONSULTATION NURSING FAC	\$188.63	\$188,63	\$111.90	\$111.90	\$159.83	\$159.83	\$169.45	\$103.43	\$139.03	\$132,02
Evaluation and		CARE, SUBSEQ										
Management	99308		\$70.68	\$70.68	\$41.93	\$41.93	\$59.89	\$59.89	\$63.50	\$63.50	\$59.89	\$59.89
Rehabilitative		MENTAL HLTH ASSESMENT/N										
Services	H0031	ON PHYS	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
	1	1	,		,		L					

	T	MENTAL HLTH					Γ		I			T
		SERVICE PLAN					1					
Rehabilitative		DEVELOPMENT/									410.00	
Services	H0032	NON PHYS	\$10,00	\$10.00	\$10.00	\$10.00	\$10,00	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00
		BEHVIORALHLT H DAY										1
Rehabilitative		TRATMENT,										
Services	H2012	PER HOUR	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00
		THERAPEUTIC										
		BEHAVIORALSE										
Rehabilitative	110040	RVICE, PER 15										
Services	H2019	MIN	\$10.00	\$10.00	\$10.00	\$10.00	\$10.00 WI380	\$10.00 WI558	\$10.00 WI381	\$10.00 WI559	\$10.00	\$10.00
Type of			WI378	WI556	WI379	WI557	Master's	Master's	PHD/PS	PHD/PS	WI382	WI560
Service	CPT	Description	MD	MD	CADC	CADC	Level	Level	YD	YD	RN/NP	RN/NP
Adaptive	1											
Behavior												
Assessments and												
Treatment –		BEHAVIOR										
Autism Spectrum		IDENTIFICATIO										
Disorders	0359T	N ASSESSM	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00	\$40.00
Adaptive Behavior		OBSERVATION										
Assessore its and		AL										
Treatn t –		BEHAVIORAL	ſ									1
Autism		ASSESS. FIRST										
Disorde	0360T	30 MINUTES	\$41,1	14	\$28.80	\$28.80	\$41.14	\$41.14	\$41,14	\$41.14	\$41,14	\$41.14
Adaptive		<u> </u>										
Behavior								1				1
Assessments and												1
Treatment -		EACH										1
Autism Spectrum		ADDIT NA 30							1			
Disorders	0361T	MIN ÉS	\$41.14	\$41.14	\$28.80	\$28.8	\$41.14	\$41.14	\$41.14	\$41.14	\$41.14	\$41.14
Adaptive Behavior	· ·											
Assessments and		ÉXPOSU										l
Treatment		BEHAVIC AL										
Autism Spectrum		ASSESS. IRST			•							l i
Disorders	0362T	30 MINUTES	\$61.71	\$61.71	\$43.20	\$43.20	\$61.71	\$61.71	\$61.71	\$61.71	\$61.71	\$61,71
Adaptive								1				
Behavior												1
Assessments and												1
Treatment ~		EACH										1
Autism Spectrum		ADDITIONAL 30								4		
Disorders	0363T	MINUTES	\$61.71	\$61.71	\$47	\$43.20	\$61.71	\$61	\$61.	\$61.71	\$61.71	\$61.71
Adaptive Behavlor	1	ADAPTIVE										
Assessments and		BEHAVIOR										1
Treatment –		TREATMENT BY PROTOCOL.										
Autism Spectrum		FIRST 30										ł
Disorders	0364T	MINUTES	\$20.00	\$20.00	20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00	\$20.00
Adaptive												
Behavior	1											1
Assessments and												
Treatment		EACH										
Autism Spectrum	ODOFT	ADDITIONAL 30	620.00	£20.00	600 00	C D	600		600.00	622.02		
Disorders	0365T	MINUTES GROUP	\$20.00	\$20.00	\$20.00	\$20.	\$20.55		\$20.00	\$20.00	\$20.00	\$20.00
Adaptive	1	ADAPTIVE										1 · · · ·
Behavior		BEHAVIOR		·								
Assessments and		TREATMENT BY										
Treatment –		PROTOCOL.										
Autism Spectrum	ODEET	FIRST 30	64.50	64.50	62.29		64 CD		64 CD	<u> </u>	64.60	4.00
Disorders Adaptive	0366T	MINUTES	\$4.69	\$4.69	\$3,28	\$3.28	\$4,69	\$4.69	\$4.69	\$4.69	\$4.69	\$4.69
Behavior												
Assessments and	1											
Treatment -		EACH					1					
Autism Spectrum		ADDITIONAL 30					1					
Disorders	0367T	MINUTES	\$4.69	\$4.69	\$3.28	\$3.28	\$4.69	\$4.69	\$4.69	\$4.69	\$4.69	\$4.69
Adaptivo	1	ADAPTIVE										
Adaptive Behavior	1	BEHAVIOR										İ
Assessments and	1	TREATMENT WITH										
Treatment -		PROTOCOL					1					
ileatilient -				1]					1
Autism Spectrum		MODIFICATION										

Wisconsin Provider Agreement Attachment © 2013.1 Blue Cross Blue Shield of Wisconsin Oneida Community Health Center and Ka Ni Kuhli Yo Family Center 09/29/2014

		MINUTES					1		1			
Adaptive												
Behavior												
Assessments and												
Treatment –		EACH				1			1			
Autism Spectrum	0369T	ADDITIONAL 30 MINUTES	\$38,96	\$38.96	\$23.38	\$23.38	\$38.96	\$38,96	\$38,96	\$38.96	\$38.96	\$38.9
Disorders	03691	FAMILY	\$56.90	329:20	\$23,36							
Adaptive		ADAPTIVE										
Behavior		BEHAVIOR									1	
Assessments and		TREATMENT									1	
Treatment –		GUIDANCE W/O	l· 1									
Autism Spectrum	ł	PATIENT				4				605 1C	\$85.16	\$85.1
Disorders	0370T	PRESENT	\$85.16	\$85,16	\$\$9.60	\$59.60	\$85.16	\$85,16	\$85.16	\$85.16	282.10	285.I
		MULTIPLE- FAMILY GROUP										
Adaptive		ADAPTIVE					-				ł	
Behavior		BEHAVIOR										
Assessments and		TREATMENT										
Treatment –		GUIDANCE W/										
Autism Spectrum		PATIENT	1 📕 🛛									
Disorders	TT7	PRESENT	\$18,76	18.76	\$13,12	\$13.12	\$18,76	\$18.76	\$18.76	\$18.76	\$18.76	\$18.7
Adaptive												
Behavio												
Assessionts and		BEHAVIOR										
Treatm -		TREATMENT										
Autism Sp. trum		AL SKILLS										
Disorders	0372T	GF UP	\$18.	\$18	\$13	\$13.12	\$18.76	\$18.76	\$18.76	\$18.76	\$18.76	\$18.7
		F POSURE										
		DAPTIVE										
		EHAVIO										
		TREAT										
		WIT'										
Adaptive		DIFICA ON										
Behavior		REQUIRI										
Assessments and		TWO OR										
Treatment		TECHNIC IS.										
Autism Spectrum	1	FIRST 60						· ·				
Disorders	0373T	MINUTES	\$77.92	\$77.92	\$68.20	\$68.20	\$77.92	\$77.92	\$77.92	\$77.92	\$77.92	\$77.9
Adaptive	1											
•												
Adaptive Behavior Assessments and												
Behavior Assessments and		EACH										
Behavior		EACH ADDITIONAL 30		\frown					\$38.9	\$38,96	\$38.96	\$38.9

Wisconsin Provider Agreement Attachment © 2013.1 Blue Cross Blue Shield of Wisconsin Oneida Community Health Center and Ka Ni Kuhli Yo Family Center 09/29/2014

#1061709754





Oneida Business Committee Meeting Agenda Request Form

- 1. Meeting Date Requested: 03 / 11 / 15 2. Nature of request Session: 🛛 Open Executive - justification required. See instructions for the applicable laws that define what is considered "executive" information, then choose from the list: United Healthcare Insurance Contract 2014-0648 Agenda Header (choose one): New Business/Request Agenda item title (see instructions): United Healthcare Insurance Contract 2014-0648 Action requested (choose one) Information only ⊠ Action please escribe: Approve United Lealt care Insurance Contract 20 648 3. Justification Why BC action is required instructions): ьe Law Office indicates the OBC needs to approve the contract 4. Supporting Materials **Instructions** Memo of explanation with required information (see instructions) Contract (check he box below if streather required) Report □ Resolution Other - please list (Note: multi-media presentation due to Trible Clerk 2 days prior to meeting) 1. 3 2. 4. Business Committee signature required 5. Submission Authorization Debbie Danforth, Division Director/Operations Authorized sponsor (choose one): Requestor (if different from above): Jeffrey R Carlson Name, Title / Dept. or Tribal Member Additional signature (as needed): Name, Title / Dept. Additional signature (as needed): Name, Title / Dept.
- 1) Save a copy of this form in a pdf format.
- 2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Page 169 of 340

ONEIDA LAW OFFICE

CONFIDENTIAL: ATTORNEY/CLIENT WORK PRODUCT

TO: Maria J. Doxtator-Alfaro Oneida Health Center Use this number on future correspondence:

FROM: Kelly M. McAndrews, Staff Attorney $\sqrt{\lambda^1}$

DATE: February 2, 2015

RE: United HealthCare Insurance Company-Medical Group Participation Agreement (revised) 2014-0648

 Purchasing Department Use

 Contract Approved

 Contract Not Approved

 (see attached explanation)

If you have any creations or comments regarding this review, please call 869-4327.

The attached *revired* agreement, contract, acticy and/or guaranty has been re-reviewed by the Oneida Law Office for legal content only. Please note the following:

 Required pusicless Committee approval profile execution, because the Agreement contains post ole wayyers of soversign an punity.

It is my understanding merchasticen in-depth discussion with United Health Care (UHC) regarding this agreement. The has agreed to sign the agreement and the IHS Model QHP, but will not agree to any other many s.

Even with the Model QHP, the Agreement in is the division to the Visconsin Regulatory Requirements Appendix (with Exmott A). Additionally, the Agreement cabjects the division to the terms of the State Medicaid Contract through the Wisconsin Bage a carrier and Medicaid SSI Regulatory Requirements Appendix. The requirements are likely weivers of overeign immunity.



UnitedHealthcare

...

September 26, 2014

Oneida Tribe of Indians of Wisconsin Cindy Mooren PO Box 365 Oneida, WI 54155

Dear Cindy,

Please find enclosed revised copies of the Medical Group Participation Agreement proposal between UnitedHealthcore of Wisconsing Included Optida Tribe of Indians of Wisconsin.

If you are in agreement with this propose, please **sign, date and return the three (3) original copies to my attention.** In order to meet a 12/2/2010 fective date, I will need the contracts returned to me no later than Friday, October 24, 2014.

We look forward to purparticipation with UnitedHeat ocare and in maintaining a positive working relationship with you. Please rel free to contact the with any questions or concerns at 414-443-4189.

Sincerely,

Khanh Nguyen Network Account Manager UnitedHealthcare of Wisconsin, Inc. 10701 West Research Drive Wauwatosa, WI 53226

Medical Group Participation Agreement

This Agreement is entered into by and between UnitedHealthcare Insurance Company, contracting on behalf of itself, UnitedHealthcare of Wisconsin, Inc. and the other entities that are United's Affiliates (collectively referred to as "United"), and Oneida Tribe of Indians of Wisconsin ("Medical Group").

This Agreement is effective on the later of the following dates (the "Effective Date"):

- i) 12/01/2014 or
- ii) the first day of the first calendar month that begins at least 30 days after the date when this Agreement has been executed by all parties.

Through contracts with physicians a provider providers of health care services, United maintains one or more networks of providers that the vailable to Customers. Medical Group is a provider of health care services.

United vishes to a range to make Medical Group eservices available to Customers. Medical Group wishes a provide such services, under the cases and conditions set forth in this Agreement.

The parties the 210, 2 er er into this Agreen, nt.

Article I. Definitions

The following terms when used in this Agreement have the meaning set for the below:

- 1.1 Benefit Plan means a certificate of coverage, summary plan esception or other document or agreement, whether delivered in proceeds electronic, or other format under which a Payer is obligated to provide coverage of covered Services for a Customer
- **1.2** Covered Service is a health care service a product for which a Customer identified to receive coverage from a Payer, pursuant to the tends of the Costomer's Benefit Man y th that Payer.
- **1.3** Customary Charge is the fee for health care so vices charge by Medical Group that does not exceed the fee Medical Group would ordinarily charge and er person regardless of whether the person is a Customer.
- **1.4 Customer** is a person eligible and enrolled to receive cover ge from a Payer for Covered Services.
- 1.5 Medical Group Physician is a Doctor of Medicine ("M.D."), or a Doctor of Osteopathy ("D.O."), duly licensed and qualified under the laws of the jurisdiction in which Covered Services are provided, who practices as a shareholder, partner, or employee of Medical Group, or who practices as a subcontractor of Medical Group. However, a subcontractor of Medical Group is a Medical Group Physician only with regard to services rendered to patients of Medical Group and billed under Medical Group's tax identification number. Additionally, a subcontractor is not a Medical Group Physician with regard to any services rendered in a physician's office other than those locations listed in Appendix 1.

- 1.6 Medical Group Non-Physician Provider is a surgical assistant, physician assistant, nurse practitioner, physical therapist, occupational therapist, speech therapist, mental health provider, or licensed social worker, who is duly authorized under the laws of the jurisdiction in which Covered Services are provided, and who renders Covered Services as an employee or subcontractor of Medical Group. However, a subcontractor of Medical Group is a Medical Group Non-Physician Provider only with regard to services rendered to patients of Medical Group and billed under Medical Group's tax identification number. Additionally, a subcontractor is not a Medical Group Non-Physician Provider with regard to any services rendered in a physician's office other than those locations listed in Appendix 1.
- **1.7** Medical Group Professional is a Medical Group Physician or a Medical Group Non-Physician Provider.
- **1.8 Payment Policies** are the gradelines adopted by United for calculating payment of claims under this Agreement. The Payre . Policies may change from time to time as discussed in section 7.4 of this Agreement.
- **1.9 Cayer** is in entity obligated to a Customer to provide reimbursement for Covered Services under the Customer's Benefit Plan, and a provide orized by United to access Medical Group's services under the Aureement
- 1.10 Protocols aroune programs, protocol and to ministrative procedures adopted by United or a Payer to be followed to Medical Groups (providing services and doing business with United and Payers under the Agreement. These Protocols may include among other things, credentialing and recredentialing processes, utilization management and cake management processes, quality improvement, peer notewordstomer grievance, concurrent review, or other similar United or Payer programs. The Proceeds may change from time to time as docussed in section 5.4 of this Agreement.
- **1.11 United's Affiliates** are those entities controlling, controlled by, courser common control with UnitedHealthcare Insurance Company.

A acle II. <u>Representation</u> and <u>Varranties</u>

- 2.1 **Representations and Warranties of Medical Group.** Merical croup, by virtue of its execution and delivery of this Agreement, represents and warrants a to lows:
 - i) Medical Group is a duly organized and validly existing legerentity in good standing under the laws of its jurisdiction of organization.
 - Medical Group has all requisite corporate power and author, the conduct its business as presently conducted, and to execute, deliver and perform its obligations under this Agreement. The execution, delivery and performance of this Agreement by Medical Group have been duly and validly authorized by all action necessary under its organizational documents and applicable corporate law.
 - iii) The execution, delivery and performance of this Agreement by Medical Group do not and will not violate or conflict with (a) the organizational documents of Medical Group, (b) any material agreement or instrument to which Medical Group is a party or by which Medical Group or any material part of its property is bound, or (c) applicable law.

Medical Group has the unqualified authority to bind, and does bind, itself and Medical Group Professionals to all of the terms and conditions of this Agreement, including any Appendices, Attachments and Exhibits, as applicable.

- iv) Medical Group has obtained and holds all registrations, permits, licenses, and other approvals and consents, and has made all filings, that it is required to obtain from or make with all governmental entities under applicable law in order to conduct its business as presently conducted and to enter into and perform its obligations under this Agreement.
- Medical Group has been given an opportunity to review the Protocols and Payment Policies. See the Additional Manuals Appendix for additional information regarding the Protocols and Payment Policies applicable to Customers enrolled in certain Benefit Plans.
- vi) Each submission to celaim by Medical Group pursuant to this Agreement shall be deemed to constant, the representation and warranty by it to United that (a) the appresentations an awarant es of it set forth in this section 2.1 and elsewhere in this Agreement are the and correct a of the date the claim is submitted, (b) it has complied with the requirement of this Agreement with respect to the Covered Services involved and the abmission of sum claim, (c) the charge amount set forth on the claim is the Contemport of the claim is available claim.
- 2.2 **Representations and Warranties of Unced.** United, by virtue of its execution and delivery of this Agreement, apprents and warrants as follows:
 - i) United is a colle organized and validly existing legal entity in good standing under the laws of its jurisdiction of organization.
 - ii) United has all requisite corporate power and authority to conducte the business as presently conducted, and to execute a cover an operform its colligations rader this Agreement. The execution, delivery and performance of this Agreement be denited two been duly and validly authorized by all acconnecessary under its organizational documents and applicable corporate law.
 - iii) The execution, delivery and performance of this Agreement by United do not and will not violate or conflict with (a) the organizational document of United, (b) any material agreement or instrument to which United to a party or by which United or any material part of its property is bound, or (c) applicable late.
 - iv) United has obtained and holds all registrations, per aits, lice uses, and other approvals and consents, and has made all filings, that it is required to chain from a make with all governmental entities under applicable law in order to conduct its ausiness as presently conducted and to enter into and perform its obligations under us Agreement.

Article III. Applicability of this Agreement

3.1 Medical Group's Services. This Agreement applies to Medical Group's practice locations set forth in Appendix 1. In the event Medical Group begins providing services at other locations (either by opening such locations itself, or by acquiring, merging or coming under common ownership and control with an existing provider of services that was not already under contract with United or one of United's Affiliates to participate in a network of health care providers), such additional locations will become subject to this Agreement 30 days after United receives the notice required under section 5.7(v) of this Agreement.

In the event Medical Group acquires or is acquired by, merges with, or otherwise becomes affiliated with another provider of health care services that is already under contract with United or one of United's Affiliates to participate in a network of health care providers, this Agreement and the other agreement will each remain in effect and will continue to apply as they did prior to the acquisition, merger or affiliation, unless otherwise agreed to in writing by all parties to such agreements.

Medical Group may transfer all or some of its assets to another entity, if the result of such transfer would be that all or some of the Covered Services subject to this Agreement will be rendered by the other entity rather than by Medical Group, but only if Medical Group requests that United approve the assignment of an Agreement as it relates to those Covered Services and only if the other entity agrees to as any this Agreement. This paragraph does not limit United's right under service 10.4 of this Agreement the other to approve the assignment of this Agreement.

- **3.2** Payers and Benefit Plan types. Useed may allow Payers to access Medical Group's services a der this Agreement for the Berguit Parn types described in Appendix 2. Appendix 2 may be modified by United upon 30 days written or electronic notice.
- **3.3** Services not over a nder a Benefit Fan. This Agreement does not apply to services not covered under the applicable Benefit Plan. Medical Group may seek and collect payment from a Customer for such service, provided that the Medical Coupeerst obtain the Customer's written consent.

This section does not authorize Medical Group to bill or cenect for Customers for Covered Services for which claims are denied or oth rwise not paid. That is us is oddressed in sections 7.5 and 7.8 of this Agreement.

- **3.4** Patients who are not Customers. This Agreement does not apply to privices realered to patients who are not Customers at the time me services were rendered. Section 7.5 of this Agreement addresses circumstances in which claims for services rendered to set persons are inadvertently paid by a Payer.
- **3.5** Health Care. Medical Group acknowledges that his Agree nent and Customer Benefit Plans do not dictate the health care provided by Medical Group or Medical Group Professionals, or govern Medical Group's or Medical Group Professional's determination of what sare to provide its patients, even if those patients are Customers. The decision regarding what care is to be provided remains with Medical Group Professionals and with Customers and not will United or any Payer.
- 3.6 Communication with Customers. Nothing in this Agreement is intended to limit Medical Group's or Medical Group Professional's right or ability to communicate fully with a Customer regarding the Customer's health condition and treatment options. Medical Group and Medical Group Professionals are free to discuss all treatment options without regard to whether or not a given option is a Covered Service. Medical Group and Medical Group Professionals are free to discuss Medical Group and Medical Group Professionals are free to including describing at a general level the payment methodologies contained in this Agreement.

Article IV. Participation of Medical Group Professionals in United's Network

- **4.1** Medical Group Professionals as Participating Providers. Except as described under section 4.2, all Medical Group Professionals will participate in United's network. Medical Group has the authority to bind, and will bind, all new Medical Group Professionals to the obligations of this Agreement.
- **4.2** Medical Group Professionals who are not Participating Providers. The following Medical Group Professionals are not participating providers in United's network:
 - i) A Medical Group Physician (or a Medical Group Non-Physician Provider, in the event such provider is of a rovider type that United credentials) who has been denied participation in united's credentialing program, whose credentialing application has not been submitted, or whose condentialing application remains pending; or

A Medical Group Procession 1 who has been terminated from participation in United's network pursuant to section 4. of this Agreement.

- **4.3** Credentian of Medical Group and Jedica Group Physicians will participate in and cooperate with United's credent ing program. Medical Group Non-Physician Providers will participate in and cooperate with U1 ted's oredentialing program to the extent such Medical Group Non-Physician Providers are surject to credentialing by United.
- 4.4 New Medical Group Processionals Medical Group will not by United at least 30 days before a physician becomes a Medical Group Physician. In the event that an Medical Group's agreement with the new Medical Group Physician provides for a starting rate nat would make it impossible for Medical Group to provide 30 days dvance notice to United, then Medical Group will give notice to United within five business days after reaching agreement will cabmin and complete a credentialing application to United within 7 days of the new Medical Group Physician's agreement to join Medical Group, unless the new Medical Group Physician already has been credentialed by United and is already a participant in United's network.

The requirements of this section 4.4 also apply to sew Medical Group Non-Physician Providers who are subject to credentialing by United.

- **4.5** Termination of a Medical Group Professional from United's Network. United may terminate a Medical Group Professional's participation in United's network, without arminating this Agreement, immediately upon becoming aware of any of the following:
 - i) material breach of this Agreement that is not cured by Medical Group Professional within 30 days after United provided notice to Medical Group of the breach;
 - ii) the suspension, revocation, condition, limitation, qualification or other material restriction on a Medical Group Professional's licenses, certifications and permits by any government agency under which the Medical Group Professional is authorized to provide health care services;

i)

- the suspension, revocation, condition, limitation, qualification or other material restriction of a Medical Group Physician's staff privileges at any licensed hospital, nursing home or other facility at which the Medical Group Physician has staff privileges during the term of this Agreement;
- iv) an indictment, arrest or conviction for a felony, or for any criminal charge related to the practice of Medical Group Professional's profession;
- v) a sanction imposed by any governmental agency or authority, including Medicare or Medicaid; or
- vi) pursuant to United's Credentialing Plan.

United will notify Medical Group of the Medical Group Professional's termination according to the notice provision set for a section 10.8 of this Agreement.

4.6 Covered Services by Medical Croup Professionals who are not Participating Providers. Indical Croup will staffe a service for upons so that Covered Services can appropriately be rendered to Customers by Medical Croup R of essionals who participate in United's network. A Redical Group Professional who noes not participate in United's network, pursuant to section 4.2 of this Accement, will not render Covered Services to a Customer.

In the event Covered's revices are rendered by a Medical Group Professional who does not participate in United's network, Medical Croup and the Nadical Group Professional will not submit a claim or other request for payment to United on ayor, and will not seek or accept payment from the Customer

> Auticle V. Intigraf Madical Group

- 5.1 Provide Covered Services. Medical Group with provide Covered Services to Contomers.
- 5.2 Nondiscrimination. Medical Group will not discriminate against any patient with regard to quality of service or accessibility of service on the basis that the patient production medical Group will not require a Customer to pay a "memorship fee" or other fee in order to access Medical Group for Covered Services (except for to payment, consurance and/or deductibles provided for under Customer's Benefit Plan) and we not us riminate against any Customer based on the failure to pay such a fee.
- **5.3** Accessibility. Medical Group will provide or arrange for the provision of a vice and assistance to Customers in emergency situations 24 hours a day, seven day, a week.
- **5.4 Cooperation with Protocols.** Medical Group will cooperate with and be bound by United's and Payers' Protocols. The Protocols include but are not limited to all of the following:
 - i) Medical Group will use reasonable commercial efforts to direct Customers only to other providers that participate in United's network, except as otherwise authorized by United or Payer.

- ii) If the Customer's Benefit Plan requires the Customer to receive certain Covered Services from or upon referral by a primary care physician, all referral physicians must adhere to the following additional protocols when those Covered Services are provided:
 - a) Notify Customer's primary care physician of referrals to other participating or non-participating providers.
 - b) Covered Services must be provided pursuant to the terms and limitations of the referral notification issued by or on behalf of the Customer's primary care physician.
 - c) If the Medical Group Physician providing the Covered Services is a referral physician, the Medical Group Physician must also notify the Customer's primary care physician of all admissions in accordance with the required time frames.

iii) Medical Group will provide notification for certain Covered Services, accept and return lephone cans from United staff, and respond to United requests for clinical information, as required by United or cayer a described in the Protocols.

The Protocols will be made available to Medical Group online or upon request. Some or all Protocols as a nay be disseminated in the form of an administrative manual or guide or in other communications.

United may charge the Protocols from time to time. United will use reasonable commercial efforts to inform Medical boroup at least 30 do vs in advance of any material changes to the Protocols. United meanplement shanges in the Protocols without Medical Group's consent if such change is applicable to all or substantially all of the medical groups in United's network located in the same state of Medical Group and that practice the one speciality as Medical Group. Otherwise, change to the Protocol proposed by United to be applicable to Medical Group are subject to the terms of science 10.2, of this Agreement that are applicable to amendments.

- **5.5** Licensure. Medical Group and Medical Group Professionals will manitain, with jut material restriction, such licensure, registration, and permits as the necessary to mable Medical Group and Medical Group Professionals to lawfully perform the Agreement.
- 5.6 Liability Insurance. Medical Group will assure that Medical Group and all Medical Group Professionals are covered by liability insurance. Except of the extent coverage is a state mandated placement, the liability coverage must be placed with espondble, financially sound insurance carriers authorized or approved to write coverage in the state in which the Covered Services are provided. The liability insurance shall be, at a minimum, of the types and in the amounts set forth below. Medical malpractice insurance shall be either or surrence or claims made with an extended period reporting option. Prior to the Effective state of this Agreement and within 10 days of each policy renewal thereafter, Medical Group shall submit to United in writing evidence of insurance coverage.

TYPE OF INSURANCE	MINIMUM LIMITS
Medical malpractice and/or	Three Million Dollars (\$3,000,000.00) per occurrence
professional liability insurance	and Five Million Dollars (\$5,000,000.00) aggregate, if
	Medical Group insures all Medical Group Professionals
	in a single policy

	This insurance requirement will also be satisfied if the
	Medical Group insures each Medical Group
	Professional separately, and the coverage for each
	Medical Group Professional is at least One Million
	Dollars (\$1,000,000.00) per occurrence and Three
	Million Dollars (\$3,000,000.00) aggregate
Commercial general and/or	One Million Dollars (\$1,000,000.00) per occurrence and
umbrella liability insurance	aggregate

In lieu of purchasing the insurance coverage required in this section, Medical Group may, with the prior written approval of United, self-insure its medical malpractice and/or professional liability, as well as its commercial general liability. Medical Group shall maintain a separate reserve for its self-insurance. If Medical Group will use the self-insurance option described in this paragraph, Medical Group will provide to United, prior to the Effective Date, a statement verified by an independent at litor or actuary that its reserve funding levels and process of funding appears adequate to meet the requirements of this section and fairly represents the financial condition of the find care acal Group will provide a similar statement during the term of this Agreement upon baned? require which will be made no more frequently than annually. Medical Group will assure the its sections and will comply with applicable laws and regulations.

- 5.7 Notice. Medical Group will give not se to be ited within 10 days after any event that causes Medical Group to be out of compliance with section 5.5 or 5.6 of this Agreement, or of any change in Medical Group's name, ownership, control, or a expayer Identification Number. This section does not apply to manages of ownership or control the result in Medical Group being owned or controlled as an entry with which it was already affiliated prior to the change. In addition, Medical Group will give written notice to Onited within 10 days after it learns of any of the following:
 - i) any suspension, revocation are dition, limitation, qualification or other material restriction on a Medical croup Professional's licenses, certifications and permits by any government agency under which a Merical Group Professional is authorized to provide health care services;
 - ii) any suspension, revocation, conditional limit alon, qualification or other material restriction of a Medical Group Physician staff privilences at any licensed hospital, nursing home or other facility at which a Medical Group Physician has staff privileges during the term of this Agreement;
 - iii) indictment, arrest or conviction of a Medical Group Professional for a felony, or for any criminal charge related to the practice of the Medical Group Professional's profession;
 - iv) the departure of any Medical Group Professional from Medical Group; or
 - v) any changes to the information contained in Appendix 1.
- **5.8 Customer consent to release of Medical Record Information.** Medical Group will obtain any Customer consent required in order to authorize Medical Group to provide access to requested information or records as contemplated in section 5.9 of this Agreement, including copies of the Medical Group's medical records relating to the care provided to Customer.

5.9 Maintenance of and Access to Records. Medical Group will maintain adequate medical, financial and administrative records related to Covered Services rendered by Medical Group under this Agreement, including claims records, for at least 6 years following the end of the calendar year during which the Covered Services are provided, unless a longer retention period is required by applicable law.

Medical Group will provide access to these records as follows:

i) to United or its designees, in connection with United's utilization management/ care management, quality assurance and improvement and for claims payment and other administrative obligations, including reviewing Medical Group's compliance with the terms and provisions of this Agreement and appropriate billing practice. Medical Group will provide access during ordinary business hours within fourteen days after a request is made, except in case of a United audit involving a fraud investigation or the health and safety of a Custor of an excedited Customer appeal or grievance (in which case, access will be given so as to enally the theory to reasonably meet the timelines for determining the appeal or grievance); an

to agencies of the government, in accordance with applicable law, to the extent such encess innecessary to concely with recent ory requirements applicable to Medical Group, United or Payers.

Medical Group year cooperate with United on a timely basis in connection with any such audit including, among other theory, in the schedun of and carticipation in an audit exit interview within 30 days of United's request

If such information and records are requested by United, Medical Coup shall provide copies of such records free of charge

5.10 Access to Data. Medical Group spresers that is conducting its protations, mollects and reviews certain quality data relating to care required by Medical Group that is reported in a manner which has been validated by a third party as having a clear, evidence based link to quality or safety (e.g., AHRQ standards) or which has been created by employer coalcions as proxies for quality (e.g., Leapfrog standards).

United recognizes that Medical Group has the sole discretize to select the metrics which it will track from time to time and that Medical Group's primare goal in so tracking is to advance the quality of patient care. If the information that Medical croup chooser dependence report on is available in the public domain in a format that includes all data element required by United, United will obtain quality information directly from the source to whom Menceal Group reported. If the Medical Group does not report metrics in the public domain, on equarter basis, Medical Group will share these metrics with United as tracked against a database of the commercial patients (including patients who are not United customers). United may publish this data to entities to which United renders services or seeks to render services, and to Customers.

- **5.11 Compliance with law.** Medical Group will comply with applicable regulatory requirements, including but not limited to those relating to confidentiality of Customer medical information.
- **5.12** Electronic connectivity. When made available by United, Medical Group will do business with United electronically. Medical Group will use <u>www.unitedhealthcareonline.com</u> to check eligibility status, claims status, and submit requests for claims adjustments for Customers enrolled

in products supported by <u>www.unitedhealthcareonline.com</u>. Medical Group agrees to use <u>www.unitedhealthcareonline.com</u> for additional functionalities (for instance, notification of admission) after United informs Medical Group that such functionalities have become available for the applicable Customer.

5.13 Employees and subcontractors. Medical Group will assure that its employees, affiliates and any individuals or entities subcontracted by Medical Group to render services in connection with this Agreement adhere to the requirements of this Agreement. The use of employees, affiliates or subcontractors to render services in connection with this Agreement will not limit Medical Group's obligations and accountability under this Agreement with regard to such services.

For laboratory services, Medical Group will only be reimbursed for services that Medical Group is certified through the Clinical Laboratory Improvement Amendments (CLIA) to perform, and Medical Group must not bill fustomers for any laboratory services for which Medical Group lacks CLIA certification.

Article VI. Denes <u>a United and Payers</u>

- 6.1 **Payment** a claims. As described in further deal in Article VII of this Agreement, Payers will pay Medica Goup for rendering Covered Services to Customers.
- 6.2 Liability Incarate. United will procure and maintain professional and general liability insurance and other insurance, as United reas nably determines may be necessary, to protect United and United's protocol against claims, liabilities, damages or judgments that arise out of services provided by United or United's employees ander this agreement.
- **6.3** Licensure. United will m intain, with at caterial restriction and licensure, registration, and permits as are necessary to enable United to h wfully perform this agreement.
- 6.4 Notice. United will give written notice to Mencal Group within 0 days after an event that causes United to be out of compliance with section 6.2 or 6.3 of this Agreement, or of any change in United's name, ownership, control, or expayer Ide affication Number. The section does not apply to changes of ownership or control the result in United being owned a controlled by an entity with which it was already affiliated prior to the change.
- 6.5 Compliance with law. United will comply with applicable regulatory requirements, including but not limited to those relating to confidentiality of Cus omer medical, formation and those relating to prompt payment of claims, to the extent those requirements are opplicable.
- **6.6** Electronic connectivity. United will do business with Medical woup electronically by providing eligibility status, claims status, and accepting requests for cann adjustments, for those products supported by <u>www.unitedhealthcareonline.com</u>. United will communicate enhancements in <u>www.unitedhealthcareonline.com</u> functionality as they become available, as described in Section 5.12, and will make information available as to which products are supported by <u>www.unitedhealthcareonline.com</u>.
- 6.7 Employees and subcontractors. United will assure that its employees, affiliates and any individuals or entities subcontracted by United to render services in connection with this Agreement adhere to the requirements of this Agreement. The use of employees, affiliates or
subcontractors to render services in connection with this Agreement will not limit United's obligations and accountability under this Agreement with regard to such services.

Article VII. Submission, Processing, and Payment of Claims

7.1 Form and content of claims. Medical Group must submit claims for Covered Services in a manner and format prescribed by United, as further described in the Protocols. Unless otherwise directed by United, Medical Group shall submit claims using current CMS 1500 form or its successor for paper claims and HIPAA standard professional or institutional claim formats for electronic claims, as applied e, with applicable coding including, but not limited to, ICD, CPT, Revenue and HCPCS count.

dedical Group will submit realize on both r services performed by Medical Group or Medical Group state. Pass through billing is not pay able under this Agreement.

- 7.2 Electronic and of claims. Within six months after the Effective Date of this Agreement, Medical Group will us pelectronic schmissing for an of its claims under this Agreement that United is able to acress electronically.
- 7.3 Time to file claims. All diagramation necessary to proceed a claim must be received by United no more than 90 days first the claim to Covered Services are rendered. In the event United requests additional information in order to process the claim, Medical brown will provide such additional information within 90 days of United's request. If Payer is not the rimary payer, and Medical Group is pursuing payment from the primary payer, the >0 day rilling limit will begin on the date Medical Group receives the claim rop use from the primary paye.
- 7.4 **Payment of claims.** Payer will pay claims for Covered Services according to the lesser of Medical Group's Customary Charge of the opplicable fee schedule (as further described in the Payment Appendix(ices) to this Agreement), and in an ordance with Payment Policies.

Claims for Covered Services subject to coordination of benefits will be puid in accordance with the Customer's Benefit Plan and applicable law.

The obligation for payment under this Agreement is solely that on Payer and not that of United unless United is the Payer.

Ordinarily, fee amounts listed in the Payment Appendix(ices) are based upon primary fee sources. United reserves the right to use gap-fill fee sources where primary fee sources are not available.

United routinely updates its fee schedule in response to additions, deletions, and changes to CPT codes by the American Medical Association, price changes for immunizations and injectable medications, and in response to similar changes (additions and revisions) to other service coding and reporting conventions that are widely used in the health care industry, such as those maintained by the Centers for Medicare and Medicaid Services (for example HCPCS, etc.).

Ordinarily, United's fee schedule is updated using similar methodologies for similar services. United will not generally attempt to communicate routine maintenance of this nature and will generally implement updates within 90 days from the date of publication.

United will give Medical Group 90 days written or electronic notice of non-routine fee schedule changes which will substantially alter the overall methodology or reimbursement level of the fee schedule. In the event such changes will reduce Medical Group's overall reimbursement under this Agreement, Medical Group may terminate this Agreement by giving 60 days written notice to United, provided that the notice is given by Medical Group within 30 days after the notice of the fee schedule change.

United will make its Payment Policies available to Medical Group online or upon request. United may change its Payment Policies from time to time.

7.5 Denial of Claims for Mot Vollowing Protocols, Not Filing Timely or Lack of Medical

Notessity Payment may be detected whole or in part if Medical Group does not comply with a rotocol of does not file clamely claim a der section 7.3 of this Agreement. Payment may also be denied for services provided that we determined by United to be medically unnecessary, and Medical Group may not bill the flaston of a lack at medical necessity, agreed in writing to be responsible for payment of those charges.

In the event nate ayount of a claim is denied for lack of a tification or for untimely filing, the denial will be reversed if Leedical Group appeals within 2 honths after the date of denial and can show all of the following

- i) that, at the time the Protocol required notification or at the time the claim was due, Medical Group do not know and has unable to reasonable determine that the patient was a Customer,
- ii) that Medical Group took re-conable stops to learn that the patient was a sustomer, and
- iii) that Medical Group promptly provided notification, or filed the claim, after learning that the patient was a Customer.
- 7.6 Retroactive Correction of Information Regarding Witcher Patient Is a Customer. Prior to rendering services, Medical Group will ask the patient to presench is concerned Customer identification card. In addition, Medical Group may contact United to obtain the most current information on the patient as a Customer.

However, Medical Group acknowledges that such information provided by United is subject to change retroactively, under the following circumstances: (i) if United has not yet received information that an individual is no longer a Customer; (ii) if the individual's Benefit Plan is terminated retroactively for any reason including, but not limited to, non-payment of premium; (iii) as a result of the Customer's final decision regarding continuation of coverage pursuant to state and federal laws; or (iv) if eligibility information United receives is later proven to be false.

If Medical Group provides health care services to an individual, and it is determined that the individual was not a Customer at the time the health care services were provided, those services shall not be eligible for payment under this Agreement and any claims payments made with regard to such services may be recovered as overpayments under the process described in section 7.10 of this Agreement. Medical Group may then directly bill the individual, or other responsible party, for such services.

- 7.7 **Payment under this Agreement is Payment in Full.** Payment as provided under section 7.4 of this Agreement, together with any co-payment, deductible or coinsurance for which the Customer is responsible under the Benefit Plan, is payment in full for a Covered Service. Medical Group will not seek to recover, and will not accept any payment from Customer, United, Payer or anyone acting in their behalf in excess of payment in full as provided in this section 7.7, regardless of whether such a bound is less than Medical Group's billed charge or Customary Charge.
 - 8 **Custome: Hold Harmles:** Mencal Coup will not bill or collect payment from the Customer, or seek to impose a lien, for the difference between the amount paid under this Agreement and Medical Group' billed charge or customary Charge, or for any amounts denied or not paid under this Agreement us to:
 - i) Medial Great s failure to comre, with the Protocols,
 - ii) Medical Grop's alure to file a timely claim,
 - iii) Payer's Payment Slicies,
 - iv) inaccurate or incorrect claim rock sing,
 - v) insolvency or other failur (by Pager to maintain its obligation to force laims payments, if Payer is United, or is an entity required by applicable lawwo as are that in Customers not be billed in such circumstances, or
 - vi) a denial based on medical necessity r price authorization, except as permitted under section 7.5.

This obligation to refrain from billing Customers applied even in those cases in which Medical Group believes that United or Payer has made an incorrect determination. In such cases, Medical Group may pursue remedies under this Agreement against United of Payer as applicable, but must still hold the Customer harmless.

In the event of a default by a Payer other than those Payers covered by the above clause v), Medical Group may seek payment directly from the Payer or from Customers covered by that Payer. However, Medical Group may do so only if it first inquires in writing to United as to whether the Payer has defaulted and, in the event that United confirms that Payer has defaulted (which confirmation will not be unreasonably withheld), Medical Group then gives United 15 days prior written notice of Medical Group's intent to seek payment from Payer or Customers. For purposes of this paragraph, a default is a systematic failure by a Payer to fund claims payments related to Customers covered through that Payer; a default does not occur in the case of a dispute as to whether certain claims should be paid or the amounts that should be paid for certain claims. This section 7.8 and section 7.7 will survive the termination of this Agreement, with regard to Covered Services rendered prior to when the termination takes effect.

7.9 Consequences for Failure to Adhere to Customer Protection Requirements. If Medical Group collects payment from, brings a collection action against, or asserts a lien against a Customer for Covered Services rendered (other than for the applicable co-payment, deductible or coinsurance), contrary to section 7.7 or 7.8 of this Agreement, Medical Group shall be in breach of this Agreement. This section 7.9 will apply regardless of whether Customer or anyone purporting to act on Customer's behalf has executed a waiver or other document of any kind purporting to allow Medical Group to collect such payment from Customer.

In the event of such a breach, Payer may deduct, from any amounts otherwise due Medical Group, the amount wrongfulls collected from Customers, and may also deduct an amount equal to any costs or expenses in unced by the Customer, United or Payer in defending the Customer from such action and otherwise enforcing sections 7.7 through 7.9 of this Agreement. Any arcums deducted by Payer in a containce with this provision shall be used to reimburse the customer and to satisfy a cost ancured. The remedy contained in this paragraph does not preclude builted from invoking any oner remedy for breach that may be available under this cureement.

7.10 Correction of verpagements or uncerpagements of claims. In the event that either Party believes that claiments is not been paid correctly or that funds were paid beyond or outside of what is provided for under this Agreement, either party musseek correction of the payment, except that Medical Crown may not seek correction of an aynometic more than 12 months after it was made.

Medical Group will repay overpayments within 30 days of notice of the overpayment. Medical Group will promptly report any tredit value that it maintains with regard to any claim overpayment under this Agreement and will obturn such overpayment to United within 30 days after posting it as a credit balance.

Medical Group agrees that recovery or overpayments may be accomplished by of sets against future payments.

7.11 Claims Payment Issues Arising from Departure of Medical Group Professionals from Medical Group. In the event a Medical Group Profession of departs from Medical Group and uncertainty arises as to whether Medical Group or some there only is entitled to receive payment for certain services rendered by such former Medical Group Profession as the parties will cooperate with each other in good faith in an attempt to rescave the retuation appropriately.

In the event that Medical Group's failure to give timely notice under section 5.7 (iv) of this Agreement resulted in claims payments being made incorrectly to Medical Group, Medical Group shall promptly call the situation to United's attention and return such payments to United. In the event Medical Group fails to do so, United may hold Medical Group liable for any attorneys' fees, costs, or administrative expenses incurred by United as a result.

In the event that both Medical Group and some other entity assert a right to payment for the same service rendered by the former Medical Group Professional, United may refrain from paying either entity until the payment obligation is clarified. Provided that United acts in good faith, Medical Group will waive any right to receive interest or penalties under any applicable law relating to the prompt payment of claims.

Article VIII. Dispute Resolution

The parties will work together in good faith to resolve any and all disputes between them (hereinafter referred to as "Disputes") including but not limited to all questions of arbitrability, the existence, validity, scope or termination of the Agreement or any term thereof.

If the parties are unable to resolve any such Dispute within 60 days following the date one party sent written notice of the Dispute to the other party, and if either party wishes to pursue the Dispute, it shall thereafter be submitted to binding arbitration in accordance with the Commercial Dispute Procedures of the American Arbitration Association as they may be amended from time to time (see <u>http://www.adr.org</u>). Unless otherw or agreed to in writing by the parties, the party wishing to pursue the Dispute must initiate the arbitration one year after the date on which notice of the Dispute was given or shell to have wrived its right to pursue the dispute in any forum.

Any arl tration proceeding under this agreement shall be conducted in Milwaukee County, WI. The arbitrate (s) may construe or interpret but nation to vary or ignore the terms of this Agreement and shall be bound by controlling inw. The arbitrater(s) shall have no authority to award punitive, exemplary, indirect or special controlling set, except in connection with a station vary claim that explicitly provides for such relief.

The parties expressly incend that are dispute relating to the business relationship between them be resolved on an individual basis of that mother dispute with are third party(ies) may be consolidated or joined with our dispute. The parties agree that any arbitration ruling beam arbitrator allowing class action arbitration or requiring consolidated dispute involving any third party(its) would be contrary to their intent and would require immediate judicial review of such ruling

If the Dispute pertains to a matter which ingeneally administered by cert priorited procedures, such as a credentialing or quality improvement plan, he policies and procedures set orthog that plan must be fully exhausted by Medical Group before Medical Group ray invoke any right to an itration under this Article VIII.

The decision of the arbitrator(s) on the points in dispute year be binding, and judgment on the award may be entered in any court having jurisdiction thereof. The parties acknowledge that because this Agreement affects interstate commerce the Federal Arbitration Act approx.

In the event that any portion of this Article or any part of this Agreement is chemed to be unlawful, invalid or unenforceable, such unlawfulness, invalidity or unenforceability shall not serve to invalidate any other part of this Article or Agreement. In the event any court determines that his arbitration procedure is not binding or otherwise allows litigation involving a Dispute to proceed, the parties hereby waive any and all right to trial by jury in, or with respect to, such litigation. Such litigation would instead proceed with the judge as the finder of fact.

In the event a party wishes to terminate this Agreement based on an assertion of uncured material breach, and the other party disputes whether grounds for such a termination exist, the matter will be resolved through arbitration under this Article VIII. While such arbitration remains pending, the termination for breach will not take effect.

This Article VIII governs any dispute between the parties arising before or after execution of this Agreement and shall survive any termination of the Agreement.

Article IX. <u>Term and Termination</u>

- **9.1** Term. This Agreement shall take effect on the Effective Date. This Agreement shall have an initial term of three years and renew automatically for renewal terms of one year, until terminated pursuant to section 9.2.
- 9.2 Termination. This Agreement may be terminated under any of the following circumstances:
 - i) by mutual written agreement of the parties;
 - ii) by either party, pop at least 90 days prior written notice, effective at the end of the initial term or effective a the error any renewal term;
 - ii) by either party upon 60 days writte photice in the event of a material breach of this Agreement by the other ready, except that such a termination will not take effect if the breach to cured within 60 days after notice of the termination; moreover, such termination may be deferred as further a scribed to Art. le VIII of this Agreement;
 - iv) by emery arts upon 10 days written notice in the event the other party loses licensure or other governments authorization necessary to proto in this Agreement, or fails to have insurance as generated ther section 5.6 or section 6.2 of this Agreement; or
 - v) by Medical Group, as described in section 7.4 of the Agree ment in the event of a nonroutine fee schedule change.
- **9.3** Ongoing Services to Certain Creations: After Termination Takes Effect. In the event a Customer is receiving any of the Covered Services listed below, an of the date that ermination takes effect, Medical Group will contraine to ender those Covered Services to tha Customer and this Agreement will continue to apply to those Covered Services, after the tendination takes effect, for the length of time indicated below.

Inpatient Covered Services	30 days or until ascherge, whichever comes first
Pregnancy, Third Trimester – Low Risk	Through post at m follow up visit
Pregnancy, First, Second or Third	
Trimester – Moderate Risk and High	Through postportum follow up visit
Risk	
Non-Surgical Cancer Treatment	30 days or a conclete cyce of radiation or
	chemotherapy, whichever greater
End Stage Kidney Disease and Dialysis	30 days
Symptomatic AIDS undergoing active	30 days
treatment	
Circumstances where Payer is required	As applicable
by applicable law to provide transition	
coverage of services rendered by	
Medical Group after Medical Group	
leaves the provider network accessed by	
Payer.	

Article X. Miscellaneous Provisions

- **10.1** Entire Agreement. This Agreement is the entire agreement between the parties with regard to the subject matter herein, and supersedes any prior written or unwritten agreements between the parties or their affiliates with regard to the same subject matter.
- **10.2 Amendment.** United can amend this Agreement or any of the appendices on 90 days written or electronic notice by sending Medical Group a copy of the amendment. Medical Group's signature is not required to make the amendment effective. However, if the amendment is not required by law or regulation and would impose a material adverse impact on Medical Group, then Medical Group may to a mate this Agreement on 60 days written notice to United by sending a termination process within 30 days after receipt of the amendment.
- **10.3 Nonwaive.** The waiver of either part of any breach of any provision of this Agreement shall not operate as a waiver of any subscription breach of the same or any other provision.
- **10.4** Assignment. This Agreement may not be assigned by either party without the written consent of the other party except that this Agreement of the bear signed by United to any of United's Affiliates.
- **10.5** Relationship of the leart is. The sole relationship between the parties to this Agreement is that of independent contractors, this Agreement does not create a joint venture, partnership, agency, employment or other relationship between the parties.
- **10.6** No Third-Party Beneficiation Unity, and Medical Group are the only initial with rights and remedies under the Agreement.
- **10.7** Delegation. United may delegate (but not as egn) certain of its administrative dules under this Agreement to one or more other entities. No such delegation will relieve United of its obligations under this Agreement.
- **10.8** Notice. Any notice required to be given under this Agreement shall be in writing, except in cases in which this Agreement specifically permits electronic notice, or a otherwise permitted or required in the Protocols. All written or electronic notices shall be deemed to have been given when delivered in person, by electronic communication, by facsimilear, of delivered by first-class United States mail, on the date mailed, proper postage prepard and properly addressed to the appropriate party at the address set forth on the signature portion of this Agreement or to another more recent address of which the sending party has received written notice. Notwithstanding the previous sentence, all notices of termination of this Agreement by enter party must be sent by certified mail, return receipt requested.

Each party shall provide the other with proper addresses, facsimile numbers and electronic mail addresses of all designees that should receive certain notices or communication instead of that party.

10.9 Confidentiality. Neither party will disclose to a Customer, other health care providers, or other third parties any of the following information (except as required by an agency of the government):

- i) any proprietary business information, not available to the general public, obtained by the party from the other party;
- ii) the specific reimbursement amounts provided for under this Agreement, except for purposes of administration of benefits.

At least 48 hours before either party issues a press release, advertisement, or other media statement about the business relationship between the parties, that party will give the other party a copy of the material the party intends to issue.

- **10.10** Governing Law. This Agreement will be governed by and construed in accordance with the laws of the state in which Medical Group renders Covered Services, and any other applicable law.
- **10.11** Regulatory Appendices, and or more regulatory appendix may be attached to this Agreement, setting forth additional provisions included in this Agreement in order to satisfy regulatory requirements under applicable base these regulatory appendices, and any attachments to them, are expressly incorporate into this Agreement and are binding on the parties to this Agreement. In the event of any inconsister or contrary inguage between a regulatory appendix and any other part of this Agreement, including but not limited to appendices, amendments and exhibits, the provisions of the regulatory appendix will on rol, to the extent it is applicable.
- **10.12** Severability. Any reclision of this Agreement that is unlawful, invalid or unenforceable in any situation in any in as a tion shall not affect the vandity or inforceability of the remaining provisions of this Agreement or the lawfulnes, validity or enforceability of the offending provision in any other quaties, or jurisdiction.
- **10.13** Survival. Sections 5.9, 77, 7.8, Article VIII and sections 5.3 and 2.9 (except for the last paragraph) of this Agreement will survive the termination of this Agreement.

THIS AGREEMENT CONTAINS A BINDING ARBITRATION PROVISION THAT MAY BE ENFORCED BY THE PARTIES.

Oneida Tribe of Indians of Wisconsin	
Signature:	Street: PO Box 365
Print Name:	City: Oneida
Title:	State: WI Zip Code: 54155
D/B/A:	Phone: 920-869-2711
Date:	E-mail:

United Lealthcar Insurance Company on boalf of itself, United Healthcare of Wisconsin, Inc. and its other affiliates, as signed by its a morified representative:

Signature:	
Print Name: Cathe, ne burn	
Title: Vice President, Network Mr. agen. nt	
Date:	
Address to be used for giving notice to U accurder the Agreement:	
Street: 10701 W. Research Dr.	
City: Milwaukee	
City. Willwalkee	
State: WI Zip Code: 53226	
State: w1 Zip Code. 55220	
For office use only: 961880	
Month and year in which Agreement is first effective: 12/1/2014	

Attachments as of the Effective Date:

- [X] Appendix 1: Medical Group Practice Locations
- [X] Appendix 2: Benefit Plan Descriptions
- [X] Additional Manuals Appendix

Payment Appendices:

- [X] All Payer Appendix(ices)
- [_____] Options PPO Payment Appendix
- [X] Medicare Advantage Payment Appendix
- [X] Medicaid and/or CHIP Payment Appendix(ices)

[_____]

Regulatory Appendices:

[X] State Regulatory Requirement ppendix (list all states as applicable)

[WI]

- [X] Midicare Advantage Regulatory Recomments Appendix
- [X] Med. and And IP Regulatory requirements opendix(ices)

Other Attachments:

Appendix 1 Medical Group Practice Locations

Medical Group attests that this Appendix identifies all services and locations covered under this Agreement.

IMPORTANT NOTE: Medical Group acknowledges its obligation under Section 5.7 to promptly report any change in Medical Group's name or Taxpayer Identification Number. Failure to do so may result in denial of claims or incorrect payment.

BILLING ADDRESS

Identify only if a common name and ordress appears on all Medical Group practice location bills that utilize the Medical Group's Tax 2011 der the Agreement.

Practice Name Orbida Tribe of Indian of Wilcowin Street Address PO Box 365 City One-Codetate Wilcow 54155 Tax ID Number (AN) 296081138 National Provide ID (NPI) 1689908527

PRAC' ICF A	CATIONS (com, tete on, for e.	h service location)
Clinic Name	Chic Name	Vinic Name
Oneida Tribe of Indians of	Oneide Tribe of Indians of	
Wisconsin	Visconsi	
Street Address	street Adaress	Seet / dress
2640 W Point Rd	25 Airport V	
City	Chy	
Green Bay	Oneid	· · · · · · · · · · · · · · · · · · ·
State and Zip Code	State and Zip Code	State a Zj Code
WI 54304	WI 54155	
Phone Number	Phone Number	Phone Number
920-490-3700	920-869-2711	
TIN (If different from above)	TIN (If different from sove)	TIN (If different from above)
National Provider ID (NPI)	National Provider ID (NPI)	x. ional x ovider ID (NPI)
1689908527	1689908527	

Clinic Name	Clinic Name	Clinic N m
Street Address	Street Address	Street Address
City	City	City
State and Zip Code	State and Zip Code	State and Zip Code
Phone Number	Phone Number	Phone Number

TIN (If different from above)	TIN (If different from above)	TIN (If different from above)
National Provider ID (NPI)	National Provider ID (NPI)	National Provider ID (NPI)



Appendix 2 Benefit Plan Descriptions

Section 1. United may allow Payers to access Medical Group's services under this Agreement for the Benefit Plan types described in each line item below, unless otherwise specified in section 2 of this Appendix 2:

- Benefit Plans where Customers are offered a network of participating providers and must select a primary physician, who in some cases must approve any care provided by other health care providers. Such Benefit Plans may or may not include an out-of-network benefit.
- Benefit Plans where Customers are offered a network of participating providers but are not required to select primary physician. Such Benefit Plans may or may not include an out-of-network but at.
- _____ Vrcar SecureHorizons Medicare A wan ge Benefit Plans.
- Unite Heat ac re Community Ph. 1 Med. are Act antage Benefit Plans.
- Wisconsin M di and CHIP Benefit Plans.

Section 2. Notwithstanding the above section 1 of this Appendix, this tagreement will not apply to the Benefit Plan types described in the following time items:

- Benefit Plans where Customers are not offered a network of participating providers from which they may receive Covered Services.
- Deere Premier Benefit Plans sponsored by Deere & Company on behalf of its United Auto Workers Customers and other collectively bargained benefit plans as indicated by a reference to "Deere Premier" on the face of the valid identification and of any Customer eligible for and enrolled in that Benefit Plan.
- Medicaid or CHIP Benefit Plans other than those separately addressed in this Appendix 2.
- Benefit Plans for Medicare Select.
- Medicare Advantage Private Fee-For-Service Benefit Plans and Medicare Advantage Medical Savings Account Benefit Plans.
- Other Governmental Benefit Plans.

- TRICARE Benefit Plans.
- Benefit contracts for workers' compensation benefit programs

Note: Excluding certain Benefit Plans or programs from this Agreement does not preclude the parties or their affiliates from having or entering into a separate agreement providing for Medical Group's participation in a network for such Benefit Plans or programs.

Section 3. Definitions:

Note: United may adopt a different name for a particular Benefit Plan, and/or may modify information referenced in the definitions below regarding Customer identification cards. If that happens, section 1 or section 2 of this Appendix will continue to apply to those Benefit Plans as it did previously, and United will provide Medical Group with the pdated information. Additionally, United may revise the definitions in this section 3 to reace changes in the names or roles United's business units, provided that doing so thes ne change Medical from so articipation status in Benefit Plans impacted by that change, and further provided that United provide Medical Group with the updated information.

MEDIC RE-

- Medic re Advantage Benefic Plans chans Benefit Plans sponsored, issued or administered to a Medicare Advantage organization as part of:
 - i) the Mudicare Advantage program under Little XVIII, Part C of the Social Security Let, or
 - ii) the Interference on the program to other with the Prescription Drug program under Titl XVIII, hart C and Part 2, respectively, of the Social Security Act, as those program names may change from time to one.
- UnitedHealthcare Commune, Plan Nedicare Advantage Lenetz Plans means Medicare Advantage Benefit Plans subject to the UnitedHealthcare Commune, Plan Protocols. Those Benefit Plans can be dentified arough a reference to "Cor" on the tack of the valid identification card of any Current eligible for and encoded in thos. Benefit Plans.
- UnitedHealthcare Medicare Solution, Medicare Advantage Beneficie and means Medicare Advantage Benefit Plans subject to the Protocols of the UnitedHealthcare Medicare Solutions business unit. Those Denefit Plans can be identified through a reference to "UHC" (or in certain parts of the country othrough a reference to "OXH" or "West") on the back of the valid identification ord of a y Curlemer eligible for and enrolled in those Benefit Plans.
- Wisconsin Medicare and Medicaid Enrollees (MME) Be efit Plan means an integrated Benefit Plan issued in Wisconsin that provides benefits to the function of
MEDICAID, CHIP AND OTHER STATE PROGRAMS:

- Medicaid Benefit Plans means Benefit Plans that offer coverage to beneficiaries of a program that is authorized by Title XIX of the federal Social Security Act, and jointly financed by the federal and state governments and administered by the state.

- Wisconsin Medicaid Benefit Plans means Medicaid Benefit Plans issued in Wisconsin that include a reference to "UnitedHealthcare Community Plan" on the identification card of any Customer eligible for and enrolled in that Benefit Plan.
- Children's Health Insurance Program ("CHIP") Benefit Plans are Benefit Plans under the program authorized by Title XXI of the federal Social Security Act that is jointly financed by the federal and state governments and administered by the state.
- Wisconsin CHIP Benefit Plans are CHIP Benefit Plans issued in Wisconsin that include a reference to "UnitedHealthcare Community Plan" on the valid identification card of any Customer eligible for and enrolled in that Benefit Plan.
- Other Governmental Benefit Plans are Benefit Plans that are funded wholly or substantially by a state or district government or a subdivision of a state (such as a city or county), but do not at lude Benefit Plans for:
 i) employ es a state government or a subdivision of a state and their dependents;

 - students as a prime aniversity, college or school; employer based over 5 of private sector employees, even if the employer receives a government subsety to help fund the coverage;

i)

iii

iv)

tedicaid benefic aries, and, hildren's Health Insurance Program (CHIP) beneficiaries.

Additional Manuals Appendix

For services rendered to Customers enrolled in certain Benefit Plans that may be included under this Agreement, Medical Group will be subject to additional requirements described in or made available to Medical Group through one or more additional provider manuals ("Additional Manuals"). When this Agreement refers to Protocols or Payment Policies it is also referring to the Additional Manuals. The Additional Manuals will be made available to Medical Group on a designated website or upon request.

In the event of any conflict between this Agreement or the "UnitedHealthcare Physician, Health Care Professional, Facility and Ancillary Provider Administrative Guide" or other UnitedHealthcare Protocols and Payment Policies, and any Additional Manual, in connection with any matter pertaining to Customers enrolled in the Benefit Plans to which the Additional Manual applies, that Additional Manual will govern, unless statutes and regulations dictate otherwise. United may make changes to the Protocols and Payment Policies subject to this Appendix in Fordance with the provisions of the Agreement relating to Protocol and Payment Policy changes.

The Benear Plans names of the Additions Maruals, and name of the website to view and download the manual, when applicable, are set forther the able below. United will notify Medical Group of any changes in the location of the Additional Maruals. Nedical Group may request a copy of the Additional Manual.

Type of Benefit Plan	Description of Applicable Adortional Margar	Website
Wisconsin Medicaid and CHIP tenefit Plans	InitedHealthcar Community Plan Physician, Health Care Protessional, Facility and Ancillary Atoministrative Guide	www.uhccommunityplan.com
UnitedHealthcare Community Plan Medicare Advantage Benefit Plans	UnitedHeatmcare Dual Complete Medicare Novider Manual	www.uhccommunityplan.com

Payment Appendix - All Payer

All Payer Fee Information Document: WI 96456/96457

Unless another Payment Appendix to this Agreement applies specifically to a particular Benefit Plan as it covers a particular Customer, the provisions of this Payment Appendix apply to Covered Services rendered by Medical Group to Customers covered by Benefit Plans sponsored, issued or administered by all Payers.

Payment Appendix - Medicare Advantage

Medicare Advantage Fee Information Document: WI 25536/25537

Unless an uner royment Appendix to this digreement applies specifically to a particular Medicare Advantage Benefit Plan as it covers a practular constomer, the provisions of this Payment Appendix apply to Covered Services rendered by Medical Group to Customers covered by all Medicare Advantage Benefit Lens as described in this Agreement.

Payner, Appendix - Wisconsin Mericaio and CHIP

Applicability

This Appendix applies to Covered Services an dered by Medical Group to Customers covered under the following types of Benefit Plans, as described in the Agreement:

Wisconsin Medicaid and CHIP Benex Play

Payment for Cover a Services

ection

- **1.1 Payment.** Medical Group's contract rates for Cover d Services are the lesser of (i) Medical Group's Customary Charges or (ii) the following, in order of a plicability:
 - a) 100% of the primary fee source. The primary fee source is the Wiscom in Medicaid fee schedule as published by the Wisconsin Department of Heath Service or its successor (the "Medicaid Agency").
 - b) For certain CPT/HCPCS codes, United may pay an amount higher than the amount listed in clause (a) above. In the future, United may reduce that higher amount paid for those CPT/HCPCS codes pursuant to this clause (b), but not less than the amount payable in clause (a) above.
 - c) In the event a fee source listed above in clause (a) or (b) does not publish a specific fee amount, then United will pay 40% of Medical Group's Customary Charges for Covered Services.

The actual payment amount is also subject to matters described in this Agreement, such as Payment Policies.

Medical Group will submit claims using a CMS 1500, its successor form or its electronic equivalent. All claims submitted under this Appendix must use CPT Codes, HCPCS Codes, ICD-9 codes or its successor and other codes in compliance with HIPAA standard data set requirements. Claims submitted without HIPAA standard data set requirements may be denied.

If an applicable state or federal program is available to provide items or payment directly to Medical Group for specific Covered Services for Customers subject to this Appendix that would otherwise be payable under this Appendix, the applicable program will apply and not this Appendix. (For example, the Vaccines For Children program currently provides vaccines free of charge, and therefore no amount will be payable under this Appendix for vaccines within the Vaccines For Children program. However, the administration of such vaccine man of payable under this Appendix, if payment is not provided to physicians under the Vaccines For Children program for vaccine administration.)

The contact rates established by this Aprendix the all-inclusive, including without limitation any applicable taxes, for all Covered Services provided on the Customer. Unless specifically indicated otherwise amounts listed in this fee schedule represent global fees and may be subject to reductions based on appropriate and iffers (for example, procession and technical modifiers).

1.2 Routine Mantee ance United routinely undates this fee whedule in response to changes published by the primary fee pures, such as fee amount change. United will use reasonable commercial efforts to implement the fee schedule changes in its systems within 90 days after final publication and make them exective in our sistem on the exective day of the change by the primary fee source.

United also routinely updates this fee schedule in response to coding changes and escribed in this Agreement. When implementing coding odates, United will apply the same percentage(s) as set forth above in section 1.1 and the then current value of the molished code to determine the contact rate. United will use reasonable commercial efforts to implement such changes within 90 days, from the date of publication. However, claims already processed p for to the change being implemented by United will not be reprocessed unless otherwise required by law.

1.3 Medicaid Agency Payment Changes. If the Medicaid Agency changes the manner in which it reimburses or changes the applicable Medicaid primary fee source such that United is required to make significant programming or platform changes in order to implement the Medicaid Agency changes, United will make commercially reasonable efforts to implement the Medicaid Agency 's official correspondence to United or is otherwise formally communicated by the Medicaid Agency to United. Medicaid Group agrees that, in such case, it will accept the current payment and wroth in this Appendix until such a time as United can implement the Medicaid Agency change. At such time as United is able to implement the change, United will communicate the change and the effective date of the change via a copy of a new payment appendix. From that effective date forward, the contract rate will be calculated based on the new Medicaid Agency payment.

If United is unable, through commercially reasonable efforts, to incorporate the Medicaid Agency payment changes in their entirety, United will so notify Medical Group within 90 days from the date the change is published in the Medicaid Agency's official correspondence to United, or otherwise formally communicated by the Medicaid Agency. The parties will then negotiate in good faith for a period of up to

60 days to amend the Agreement to replace this Appendix with a new appendix and stated effective date for the new contract rates. If the parties have not reached an agreement upon such an amendment within the aforementioned 60 day period, either party may initiate Dispute Resolution according to this Agreement.



Payment Appendix Fee Information Document

Fee Schedule Specifications: as of 10/01/2014 Report Date: 09/26/2014

Fee Schedule ID: WI 96456 - NonFacility

ź

Linked Fee Schedule ID: WI 96457 - Facility

Type Of Service	Primary Fee Source	Pricing Level
	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
EVALUATION & MANAGEMENT EVALUATION & MANAGEMENT - NEONATAL	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
EVALUATION & MANAGEMENT - NEONATAL EVALUATION & MANAGEMENT - PREVENTIVE	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
EVALUATION & MANAGEMENT - NURSING FACILITY SVCS	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - INTEGUMENTARY	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - MUSCULOSKELETAL	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - RESPIRATORY	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - CARDIOVASCULAR	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - HEMIC & LYMPH	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - MEDIASTINU A DIAPH. GM	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - DIGESTIV	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - URINARY	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - MALE GENTAL	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - FEMALE ENITAL	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - MATERNIA & DELIVERY	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - ENDOCRIN	2013 CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - NERVOUS	CMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - EYE & OCULAR ADNEY	20. SMS RBRVS Carrier Locality (0095100)	135.000%
SURGERY - AUDITORY	2013 S RBRVS Carrier Locality (0095100)	135.000%
RADIOLOGY	2013 CM RBRVS Carrier Locality (0095100)	130.000%
RADIOLOGY - BONE DENSITY	2013 CMS RBRVS Carrier Locality (0095100)	100.000%
RADIOLOGY - CT	X 3 CMS RL /S Carrier Locality (0095100)	124.000%
RADIOLOGY - MAMMOGRAPHY	20 CMS RBA Carrier Locality (0095100)	135.000%
RADIOLOGY - MRI	2013 CMS RBRV partier Locality (0095100)	124.000%
RADIOLOGY - MRA	2013 CMS RBC 5 C ier Locality (0095100)	124.000%
RADIOLOGY - NUCLEAR MEDICINE	2013 CMS RBF 4 C, Ciar Locality (0095100) 2013 CMS RBF 4 C, Ciar Locality (0095100) 2013 CMS PL VS Carrier Locality (0095100) 2013 CMS 2 RVS Carrier Locality (0095100)	124.000%
RADIOLOGY - PET SCANS	2013 CMS arrier Lecality (0095100)	105.000%
RADIOLOGY - RADIATION THERAPY	2013 CL RBRVS Carrier cality (0095100)	130.000%
RADIOLOGY - ULTRASOUND	2013 US RBRVS Cart Locality (0095100) 2013 CMS RBRVS Coner Local 0095100)	130.000%
LAB - PATHOLOGY	2013 CMS RBRVS Conter Local 2095100)	100.000%
OFFICE LAB	2013 CMS Clinical Sched	100.000%
CLINICAL LABORATORY	2013 CMS Clinical 24b Software 1 2013 CMS RBUS Carrie 26a (0095100) 2013 CMS RBUS Carrie 26a (0095100)	60.000%
MEDICINE - OPHTHALMOLOGY	2013 CMS RB, /S Carrie Local (0095100)	135.000%
MEDICINE - CARDIOVASCULAR	2013 CMS RBRVS Cattor Local (00951	135.000% 135.000%
MEDICINE - ALLERGY & CLINICAL IMMUNOLOGY	2013 CMS RBRVS offier Local / (009050) 2013 CMS RBRVS Carrier Local y (005100)	135.000%
MEDICINE - CHIROPRACTIC MANIPULATIVE TREATMENT		100.000%
MEDICINE - PHYSICAL MED AND REHAB - MODALITIES		100.000%
MEDICINE - PHYSICAL MED AND REHAB -THERAPIES&OTHER	003 CMS RBRVS Carrier Los (00957) 013 CMS RBRVS Carrier Los (009500)	135.000%
MEDICINE - ENTERAL FORMULA	2013 CMS RBRVS Carrier Locality (00 100)	135,000%
MEDICINE - OTHER MEDICINE - IMMUNIZATION ADMINISTRATION	2013 CMS RBRVS Carrier Locality (00 100)	100.000%
MEDICINE - CHEMO ADMINISTRATION	2013 CMS RETVS Carrier Locality (00 100)	135.000%
OBSTETRICS - GLOBAL	2013 CMS LRVS Carrier Locality (009-00)	135.000%
IMMUNIZATIONS	2013 CMS conversion RVS Carrier Locality (009-00) UHC Impenization Fee Schedule	100.000%
INJECTABLES/OTHER DRUGS	CMS and Pricing	100.000%
INJECTABLES - ONCOLOGY/THERAPEUTIC CHEMO DRUGS	CMS/rug Pricing UH/runemotherapy Fee Schedule	100.000%
INJECTABLES - IVIG	2 Jung Pricing	112.000%
INJECTABLES-SALINE & DEXTROSE SOLUTIONS	Drug Pricing	100.000%
DME & SUPPLIES	201 SMŠ DME W	60.000%
DME & SUPPLIES - RESPIRATORY	2013 S DME	60.000%
DME & SUPPLIES - ORTHOTICS	2013 CMS DM VI	60.000%
DME & SUPPLIES - PROSTHETICS	2013 CMS 🔽 🖉 WI	60.000%
DME & SUPPLIES - OSTOMY	2013 CMS DN WI	60.000%
AMBULANCE	2013 CMS Ambs the Schedule - Marin (0, 5100)	135.000%
Default Percent of Eligible Charges: 50.00%		
Professional/Technical Modifier Pricing: Fee Source-Based		
Site of Service: Site of Service applies. CMS Assignment (ASC POS 24 = F)		
Anesthesia Conversion Factor (Based on a 15 minute Anesthesia Tim	ie Unit Value): \$ 30.00	
Calculation of Anesthesia Partial Units: Proration		
Schedule Type: FFS		
••		

Last Routine Maintenance Update: 10-01-2014

Fixed Fees: V5242 - S2500.00 V5243 - S2500.00 V5244 - S2500.00 V5245 - S2500.00 V5246 - S2500.00 V5247 - S2500.00 V5248 - S5000.00 V5249 - S5000.00 V5250 - S5000.00 V5251 - S5000.00 V5252 - S5000.00 V5253 - S5000.00 V5254 - S2500.00 V5255 - S2500.00 V5256 - S2500.00 V5257 - S2500.00 V5258 - S5000.00 V5259 - S5000.00 V5260 - S5000.00 V5261 - S5000.00 V5262 - S2500.00 V5263 - S5000.00 V5263 - S5000.00 V5265 - S2500.00 V5256 - S2500.00 V5257 - S2500.00 V5258 - S5000.00 V5259 - S5000.00 V5260 - S5000.00 V5261 - S5000.00 V5263 - S5000.00 V5263 - S5000.00 V5265 - S2500.00 V5256 - S2500.00 V5257 - S2500.00 V5258 - S5000.00 V5259 - S5000.00 V5260 - S5000.00 V5261 - S5000.00 V5263 - S5000.00 V5263 - S5000.00 V5265 - S2500.00 V5255 - S2500.00 V5256 - S2500.00 V5258 - S5000.00 V5259 - S5000.00 V5260 - S5000.00 V5261 - S5000.00 V5263 - S5000.00 V5263 - S5000.00 V5265 - S2500.00 V5255 - S2500.00 V5256 - S2500.00 V5258 - S5000.00 V5268 - S5000.00 V5265 - S2500.00 V5265 - S2500.00 V5265 - S2500.00 V5258 - S5000.00 V5259 - S5000.00 V5265 - S2500.00 V5265 - S2500.00 V5265 - S2500.00 V5258 - S5000.00 V5265 - S2500.00 V5258 - S5000.00 V5265 - S2500.00 V5265 - S

Fee Amounts listed in the fee schedule are all-inclusive, including without limitation any applicable taxes. Unless specifically indicated otherwise, Fee Amounts represent global fees and may be subject to reductions based on appropriate Modifier (for example, professional and technical modifiers). As used in the previous sentence, "global fees" refers to services billed without a Modifier, for which the Fee Amount includes both the professional component and the technical modifiers). As used in the previous sentence, "global fees" refers to services billed without a Modifier, for which the Fee Amount includes both the professional component and the technical component. Any co-payment, deductible or coinsurance that the customer is responsible to pay under the customer's benefit contract will be subtracted from the listed Fee Amount in determining the amount to be paid by the payer. The actual payment amount is also subject to matters described in this agreement, such as the Payment Policies. No payments will be made for any CMS additional compensation programs, including without limitation incentive or bonus payment programs. Please remember that this information is subject to the confidentiality provisions of this agreement.

Confidential and Proprietary Not for Distribution to Third Parties



Payment Appendix Fee Information Document

Section 1. Definition of Terms

Unless otherwise defined in this document, capitalized terms will have the meanings ascribed to them in the Agreement.

AMA: American Medical Association located at: www.ama-assn.org.

Anesthesia Conversion Factor: The dollar amount that the be used in the calculation of time-based and non-time based Anesthesia Management fees in accordance with the Anesthesia Payment Policy. Unless of the cally stated otherwise, the Anesthesia Conversion Factor indicated is fixed and will not change. The Anesthesia Conversion Factor is based on an anesthesia material value of 15 minutes. In the event that any of United's claims systems cannot administer a 15 minute anesthesia time unit value the Anesthesia Conversion factor will bally stated otherwise.

[(Value of 15 n and Anex esia Conversion Factor / 15) * anesthesia time unit value]

For example, an Austhesia Conversion Factor of \$60.00 (based on a 15 nutle austhesia time unit value) would be calculated to an Anesthesia Conversion Factor of \$40.00 (based on a 1-minute anesthesia time unit value).

x uple: [(\$ 00 / 15 10 = \$40.00]

Anesthesia Management. The increase ment of aresthesia services instead to meneral, surge all or scopic procedures, as described in the current Anesthesia Management Codes list attached to the settlesia ment Policy located at we can dhealthe regulated.

Calculation of Anesthesia Partia

Proration: Partial time units will be proved and elculate to one decimal place bunded to the near the provement. For example, if the anesthesia time unit value is based on 15 minutes and if 17 minutes of actual time is so mittee on a claim, then the 17 minutes will be readed a 15. The resulting figure of 1.1333 will be rounded to the nearest tenth and the total time units for the claim if the 1.1 time time.

In the event that any of United's claims systems cannot administer the capulation of partial uses as indicated power, odifferent calculation method will be used until such time as the appropriate system enhancements can be programmed and implemented. That different calculation that will result in a Fee Amount that is no less than the Fee Amount that would apply under the Proratic method described at we.

CMS: Centers for Medicare and Medicaid Services located

Conversion Factor: A multiplier, expressed in dollars per relative value un which converts relative values inter for pasis amounts

CPT/HCPCS: A set of codes that describe procedures and services, including supplier and materials, performed provide by physical s and other health care professionals. Each procedure or service is identified with a 5 digit code. We use of cr/HCPCS simplifies the reporting of services.

is.gov

CPT/HCPCS Description: The descriptor associated with each CPT/HCPC

Default Percent of Eligible Charges: In the event that a Fee Basis amount is no sourced by either a primary or alternates. Fource, such as services submitted using unlisted, unclassified or miscellaneous codes, the codes are subject to correct coding onew and will be priced at the contracted percentage indicated within this document.

ode.

Expired Code: An existing CPT or HCPCS code that will be expired by the entity that public of the control of the AMA).

Fee Amount: The contract rate for each CPT/HCPCS. The calculation of the Fee Amount is impacted by a weight of factors explained within this document including, but not limited to, Professional/Technical Modifier Pricing, Carrier Locality, CMS year, Place of Service and Pricing Local.

Fee Basis: The amount published by the Fee Source upon which the Pricing Level will be applied to dere the P Amount.

Fee Schedule ID: United's proprietary naming/numbering convention that is used to identify the specific fee the dule which apports the terms of the contractual agreement. This is the fee schedule for services performed in nonfacility Places of Service.

Fee Schedule Specifications: Documentation of the underlying calculation methodology and criteria used to derive the Fee Amounts contained within the fee schedule.

Fee Source: The primary or alternate entity or publication that is supplying the Fee Basis.

Fixed Fees: Fee Amounts that are set at amounts which do not change. The Fee Amounts listed are intended for pricing purposes only and are subject to other matters



described in this Agreement, such as the Payment Policies.

Future Payment Terms: The general description of any pricing terms which will be implemented on a scheduled future effective date.

Last Routine Maintenance Update: The effective date on which this fee schedule was most recently updated. Please refer to the Routine Maintenance section of this document for more information about Routine Maintenance updates.

Linked Fee Schedule ID: United's proprietary name, but being convention that is used to identify the specific fee schedule for each specific contractual agreement. This is the fee schedule for services performed infacility Plans of Service.

Modifier: A Modifier provides the means to report or in certainate arvice to procedure has been altered by some specific circumstance but not changed in its definition or code.

Place of Service The facility or nonfacility setting in which the service performed. This may also be referred to by CMS as Payment Type.

Pricing Level: The commerced percentage amount that will be ultiplied best the repay or alternate Fee Basis amount in order to derive the Fee Amount.

amount for deriving the Fee Amount within each Type of Service fee Bas Primary Fee Source (Caller Lo alif : The main Fee Source used supply t ri ab, u dity is a pplying particular Pricing Level to the CMS Resource-Based Relative Value category of codes are derive category. For instance, if the Fee Amou, for a gi enated to e Prim e Source. The Carrier L ndicate the exact CMS geographic region upon which the Fee Scale (RBRVS), then CMS RBRVS is Amounts are based.

Professional/Technical Modifier Priving: Fe Source-Based: Fee Amounts or Modifiers (are example, -TC or -26 Modifiers) are derived using the Fee Basis amount as published by the primary or alternate Fe Source.

RVU: Relative Value Unit as published by CMS. nited es the RVU at is used by CMS, or example, if MS uses a transitional RVU, then United will as well.

Replacement Code: One or more new CPT or HCPC, hodes that are the consame services or description and will replace one or more Expired Codes within the same Type of Service category.

Report Date: The actual date that this document was produced.

Representative Fee Schedule Sample: A representative listing with most comparing used CPT/HCPCS compared to s, along with other relevant pricing information, for each specific Fee Schedule ID. The Fee Amounts listed are intended or pricing purposes only and are supect to other matters described in this Agreement, such as the Payment Policies.

Schedule Type: FFS: This is a fee-for-service fee schedule. Unless stated otherwise, the Fee product indicated will be and to calculate payment to you as further described within this document.

Site of Service Price Differential: Site of Service applies. CMS Assignment (SC POS 24 =F): this fee schedule follows CMS guidelines for determining when services are priced at the facility or nonfacility fee schedule (with the exception of services perform that An ulatory Surgery Centers, POS 24, which will be priced at the facility fee schedule). CMS guidelines can be located at: www.ems.hhs.gov.

In the event that any of United's claims systems cannot administer the calculation of Site of Service, offerents, pricing as indicated above, a different calculation method will be used until such time as the appropriate system enhancements can be programmed and applemented. The any cent calculation method will result in a Fee Amount that is no less than the Fee Amount that would apply under the method described above.

Type of Service: A general categorization of related CPT/HCPCS codes. Type of Service categories are interned to closely a provide the CPT groupings in the Current Procedural Terminology code book (as published by the AMA) and the HCPCS groupings (as published by CMS).

The Office Lab Type of Service category represents those lab tests, as determined by United, in which the lab test resume cessary to make an informed treatment decision while the patient is in the office.

A partial or complete crosswalk mapping of CPT/HCPCS to Type of Service categories is available to you upon request.

Section 2. Alternate Fee Sources

In the event the Primary Fee Source contains no published Fee Basis amount, alternate (or "gap fill") Fee Sources may be used to supply the Fee Basis amount for

Payment Appendix Fee Information Document

deriving the Fee Amount. For example, if a new CPT/HCPCS code has been created within the Type of Service category of codes described above, and CMS has not yet established an RBRVS value for that code, we use one of the Fee Sources that exist within the industry to fill that gap, such as but not limited to Ingenix Essential RBRVS. For that CPT/HCPCS code, we adopt the RBRVS value established by the gap-fill Fee Source, and determine the Fee Amount for that CPT/HCPCS code by applying to the gap-fill RBRVS the same Conversion Factor and Pricing Level that we apply to the CMS RBRVS for those CPT/HCPCS codes that have CMS RBRVS VS value for that CPT/HCPCS code, we would begin using the Primary Fee Source, CMS, to derive values. At such time in the future as CMS publishes its own P the Fee Amount for that code and no longer use the alternation Source.

More information about all of our Fee Sources can be ate

RBP Medicaid Services (CM) Centers for Me are and Fee Schedules: www.cms.hhs.gov

at:

- nvate Centers for sease Contro nd Prevention (CD ctor Ş Price: www.cdc.gov/vaccines/programs/vfc/cdc-vac-price-list.htm
- Thomson R iters Red Boo. <u>www.micromedex.com</u>
- RJ Health S ems: www.reimbursep.entcodes.com
- -----lixonline.com
- sth jolo ts: <u>www.asahq.org</u> American Society of

Section 3. Routine Updates

at prevised informatic created by the Fee Spree, and as described below, to update the Fee Amounts in. United routinely updates its fee schedure. (As to stay current with applicable coding practices; (2) in mediantions; and (3) to remain in computance with HIPAA requirements. United will not generally attempt Routine updates occur when United my har ally i orporat calculated in accordance with this Fee Internation ocur e medic response to price changes for immunizations and i ect to communicate routine updates of this nature.

The types of routine updates, and their respective effective dates, are des ibed below.

a. Annual Changes to Relative Value Units, Cont rsion Factors, at Rate Fees

o RV fore, the the 2010 flat rate fees (non-RVU based fees such Comprision Factors by CMS may affect this fee This fee schedule follows a "stated year" construction meth The 2 the 2013 Conversion F .or, a XVUs a as DME fees) will be locked in as the basis for deriving Fee Amounts. nual publication l Cor as DME fees) will be locked in as the basis for deriving Fee Amounts. They fore, the annual publication en CVUs a 1 Contension Factors by CMS may affect this fee schedule. Generally, any RVU, Conversion Factor, or flat rate fee choices provided in subsequent years by the Primar nee Sources will not be reflected in this fee schedule except, for example, to add Fee Amounts for new codes or replace a smate Fe Basis amounts. Unite not use incomato commercial efforts to implement the updates in its systems on or before the later of (i) 90 days after the discrive date of new modification made by see Feer furce or (ii) a days after the date on which the Fee Source initially places information regarding such modification in the public omain (for example, when CMS dir fibutes processed prior to the change being implemented by United will not be reprocessed unless otherwise required by late in the event that MS does not publish complete at of Fee Basis amounts for a specific code (for example: Global, -TC, and -26 fees) and the code contains a star code of "Condicating the code is carrie to fee Basis amounts for all modifiers associated with the code bases in fee information available and publish by the local fiscal intermediary of the code public of the code bases in a fee information available and published by the local fiscal intermediary in the local fiscal intermediary in the local fiscal intermediary modifiers associated with the code bases in a fee information available and published by the local fiscal intermediary in the local fiscal intermediary in the local fiscal intermediary discrimentation available and public by the local fiscal intermediary in the local fiscal intermediary discrimentation available and published by the local fiscal intermediary discrimentation available and published by the local fiscal intermediary discrimentation available and published by the local fiscal intermediary discrimentation available and published by the local fiscal intermediary discrimentation available and published by the and by fiscal intermediaries from other locations.

b. Quarterly Updates in Response to Changes Published by Primary Fee Sourt

us, deletons, and changes to CPT codes by the AMA or United updates its fee schedule in response to changes published by Primary Fee Sources as a reall g di. aedules HCPCS codes by CMS and any subsequent changes to CMS' annual update. United updates its fee new CPTHCPCS codes using the applicable ent RVU of the r Conversion Factor and Pricing Level of the original construction methodology along with the then-cu ed CPT/HCPCS code. The effective date of the updates described in this subsection b, will be no later than the first day of the next calendar quart after final dicata by the Fee Source, except that if that quarter begins less than 60 days after final publication, the effective date will be no later than the first day the g dar quarte ollowing the next calendar quarter. riective no lat For example, if final publication by the Fee Source is on April 10, the fee update under this subsection b. will be than July 1, and if final publication by the Fee Source is on June 10, the fee update under this subsection b. will be effective no later than October 1. In IS does not publish a complete set of cating the code is carrier priced), United "C" 6 Fee Basis amounts for a specific code (for example: Global, -TC, and -26 fees) and the code contains a status code will use reasonable commercial efforts to establish Fee Amounts for all modifiers associated with the code based on tec-mormation available and published by the local fiscal intermediary and by fiscal intermediaries from other locations.

However, in the event that the code source has expired a CPT/HCPCS code and replaced it with a Replacement Code, United will crosswalk the fee from the Expired Code to its Replacement Code as further described below:

Based on information published by the code source (AMA Current Procedural Terminology and The HCPCS Level II), when one Expired Code is replaced by one

Payment Appendix Fee Information Document

Replacement Code, United will apply the Expired Code's Fee Amount to the Replacement Code; provided, however, if the Expired Code's Fee Amount was determined by an alternate Fee Source and a Primary Fee Source becomes available, the Replacement Code's Fee Amount will be determined using the Primary Fee Source.

Based on information published by the code source (AMACurrent Procedural Terminology and The HCPCS Level II) and United's claims data, when several other are replaced by one Replacement Code, United will apply the sum of these Expired Code's Fee ne Expired Code's Fee Amount was determined by an alternate Fee Source and a Primary Fee Source Expired Codes that are always done in conjunction with Amounts to the Replacement Code; provided, hower becomes available, the Replacement Code's Fee our will be determined using the Primary Fee Source.

modology as described above: PCS codes, ot included in our direg The following types of crose

- ich as G, K, Q, and Temporary
- Temporary PT codes, such as Category III codes
- codes, such as CPT Category II codes Information
- y hospitals HCPC-C Cod are only
- · Codes categorized as zatio and injectables

re, United reserves the right to make a crosswalk determination at that time. If any types of codes not currently lister the ex ons above are developed .ne 1.

c. Price Changes for Immunizati is a . Inj tables

review and evaluate the drug prices that and ublished by the Fee Sources. In addition, United's Executive United routinely updates the Fee Amount 1 respo e to p sis be used in each quarterly update. The EDPF may address topics Drug Pricing Forum (EDPF) meets on a quarterly turnel to banges, and special circum rances (for example, H1N1 vaccine). Based on supporting information s EDPF in v elect to establish a the Amount or therride a Fee Amount, as published by the Fee Source, in the for a part ular vaccine or drug. These Fee to count prolates will be effective as described below. actur including pricing for emerging drugs, anticipated , Unit provided by the drug manufacturer or the Fee So favor of a Fee Amount that is more appropriate and rease

t, United applies the d. The Fee Ba For Injectable Oncology/Therapeutic Chemotherapy Dru HC Chemotherapy F Sched ich uses a third party vender as the Primary Fee and and onts are calculated as follo 18% of a berge Sales Price. Source to determine the acquisition cost information prov.

- For J codes for which there is no generic available, the Fee basis will sh J code will be calculated using th oll

- For J codes for which there is no generic available, the Fee Basis for ach as - For J codes containing a branded and generic drug, the Fee Basis for ach as Wholesale Price (AWP) for the brand product, plus the lower of either; the acq. ing formula: 20% multiplied by the Average vition corfor the brand product, le ave isition cost of the generic products in the J code.

ms & Payme unitedhealthcareonline.com >> C > Fee Schedule Lookup > More information about the UHC Chemotherapy Fee Schedule can be lo ted at: <u>wy</u> Related Links "Acquisition Cost List"

For Immunizations, United applies the UHC Immunization Fee Schedule. The Conters for Disea the Primary Fee Source used to obtain the Fee Basis amounts. In the event that more than one **Control and Prevention** iyate S or Selling Price (CDC PSSP) is e Basis amount is published CDC PSSP for a specific CPT/HCPCS code, an average of the published amounts will be used.

More information about the UHC Immunization Fee Schedule can be located at: www.unix Claims & Payments > Fee Schedule Lookup > healthcareon .co. Related Links "UHC Immunization Fee Schedule"

The effective date of updates under this subsection c. will be no later than the first day of the next condar qua er after final publication by the Fee Source, except that first day of the b. stion c. wi¹¹ if that quarter begins less than 60 days after final publication, the effective date will be no later than t ar quarter following the next calendar e effec quarter. For example, if final publication by the Fee Source is on April 10, the fee update under this sub-stion c. v publication by the Fee Source is on June 10, the fee update under this subsection c. will be effective no late than e no later than July 1, and if final ober 1.

d. Other Updates

•

United reserves the right, but not the obligation, to perform other updates as may be necessary to remain consistent amary Fee Source. United also will perform other updates as may be required by applicable law from time to time. United will use reasonable commercial efforts to implement the updates in its systems on or before the later of (i) 90 days after the effective date of any modification made by the Fee Source or (ii) 90 days after the date on which the Fee Source initially places information regarding such modification in the public domain (for example, when CMS distributes program memoranda to providers). United will make the updates effective in its system on the effective date of the change by the Fee Source. However, claims already processed prior to the change being implemented by United will not be reprocessed unless otherwise required by law.

Payment Appendix Fee Information Document

. . .

For More Information United is committed to providing transparency related to our fee schedules. If you have questions about this fee schedule, please contact Network Management at the address and phone number on your contract or participation agreement or you may use our fee schedule look-up function on the web at: <u>www.unitedhealthcareonline.com</u> or contact our Voice Enabled Telephonic Self Service line at (877) 842-3210.



Wisconsin Regulatory Requirements Appendix

In addition to our understandings in the agreement between you and us, there are certain additional items which Wisconsin laws and regulations require us to include in our contract. This appendix sets forth those items and is made part of the agreement between you and us.

These requirements apply to all products or benefit plans sponsored, issued or administered by or accessed through us to the extent such products are regulated under Wisconsin laws.

We each agree to be bound by the terms and conditions contained in this appendix. In the event of a conflict or inconsistency between this appendix and any term or condition contained in the agreement between you and us, this appendix shall control except with regard to benefit contracts outside the scope of this appendix. For the purpose of his appendix, "enrollee" or "member" shall mean our customers who are enrolled in benefit cort act insured or administered by us or any participating entity.

Provisions applyable to beneficientry is replated under Wisconsin HMO laws:

1. **Provider Disclosure.** We each agree that noming in the agreement between you and us shall be construed to limit out onlity to disclose information, four on behalf of an enrollee, about the enrollee's medical contation. You may discuss, with the on behalf of an enrollee, all treatment options and any other information the you do armine to be in the lest interest of the enrollee and within the scope of your professional litense. We may not penalize you nor terminate the agreement between you and us because you make referrals to other participating preciders or disclose medically necessary or appropriate care with or on behalf of an increase. We may not retaliate agrees you for advising an enrollee of treatment options that are not covered benealts under the enrollee's benealt contract with us.

2. Acknowledgment of Releipt of Notice and Agreements of to all of Exemption from Wisconsin Statute Section 609.91. We acknowledge receipt of the notice in the form attached as Exhibit A, required by Wisconsin Statute decken 609.94(1). You agree that we shall not exercise the right under Wisconsin Statute, Section 60.92 to dect to be exempt from Visconsin statute, Section 609.91(1)(b) for the purpose of recovering health care costs arising from health are furnished by you. You acknowledge that your agreement not to exercise this right shall mean that you shall remain subject to the restrictions on recovery of health care costs bund in Wisconsin Statute, Section 609.91. In the event that you are not subject to the restrictions on re-over of health care costs found at Wisconsin Statute, Section 609.91(1)(a), (am), or (b), you agree to eact to be subject to such restrictions pursuant to Wisconsin Statute 609.925 and any applicable regulations, and shall arom, by take such action as is necessary to implement such election.

3. Continued Provision of Health Services after Termination. In the event the agreement between you and us is terminated by you for any reason or in the event the agreement between you and us is terminated by us for any reason other than you no longer practice in our geographic service area or misconduct on your part, you agree to continue to provide health services to provide so the following periods:

(a) <u>Enrollee Care.</u> If an enrollee is receiving care from you under a prescribed treatment plan and you are not a primary care physician, you are obligated to continue the provision of health care services to that enrollee until (i) the completion of the treatment; or (ii) a period of ninety (90) days after the effective date of your termination, whichever is shorter, except that the continuation of health care services is not required to extend beyond the end of the current plan year, for an enrollee who has coverage under a contract with us that has no open enrollment period; or the end of the plan year for which it was represented that you were, or would be, a

٠i

1

provider participating in our products for an enrollee with an open enrollment period. You agree to accept and we or a participating entity are obligated to pay the amounts established by the agreement between you and us for covered health care services rendered according to this section after termination of this agreement.

(b) <u>Maternity Care.</u> If an enrollee is receiving maternity care from you and the enrollee is in her second or third trimester of pregnancy, you are obligated to continue the provision of health care services to that enrollee until the completion of the postpartum care. You agree to accept and we or the participating entity, as applicable, are obligated to pay the amounts established by the agreement between you and us for covered health care services rendered after termination of this agreement.

(c) <u>Primary Care Physician.</u> If you are a primary care physician, you are obligated to continue the provision of halth care services until the end of the current plan year for an around with no oper enrollment protod; or until the end of the plan year for which it was represented that you were on world be oparticipating provider for an enrollee with an open enrollment period. You agrees a accest we or the participating entity, as applicable, are obligated to pay the amounts established by one agreement for covered health care services rendered after termination of his agreement.

Additionally, in the vient year erminate the agree vent between you and us for any reason, you shall, within 30 days prior to the termination or 15 days following our reseipt of the termination notice, whichever is later, por a not fication of such termination in your case. This notice requirement applies only if you are a specialist and the do not require a referral.

4. **Grievances.** You must identify complaints and grievances is a time y manner and forward these complaints and grievances to us a a timely manner.

Provisions applicable to benefit contract, regulated by the State of Wisson and but not subject to Wisconsin HMO laws:

1. **Provider Disclosure.** We each agree that obthing in the agreement between you and us shall be construed to limit your ability to disclose information, to or or wehalf of an enroluse, about the enrollee's medical condition. You may discuss, with or on benefit of a controllee, all treatment options and any other information that you determine to be in the best interest of the enrollee and within the scope of your professional license. We may not penalize you nor terminate the agreement between you and us because you make referrals to other participating providers or discuss medicary necessary or appropriate care with or on behalf of an enrollee. We may not retaliate again it you her advise g an enrollee of treatment options that are not covered benefits under the enrollee's contract you us

2. Continued Provision of Health Services after Termination. It the event the agreement between you and us is terminated by you for any reason or in the event the agreement between you and us is terminated by us for any reason other than you no longer practice in our geographic service area or misconduct on your part, you agree to continue to provide health services to enrollees for the following periods:

(a) <u>Enrollee Care.</u> If an enrollee is receiving care from you under a prescribed treatment plan and you are not a primary care physician, you are obligated to continue the provision of health care services to that enrollee until (i) the completion of the treatment; or (ii) a period of ninety (90) days after the effective date of your termination, whichever is shorter, except that the continuation of health care services is not required to extend beyond the end of the current plan year, for an enrollee who has coverage under a contract with us that has no open enrollment period or the end of the plan year for which it was represented that you were, or would be, a provider participating in our products for an enrollee with an open enrollment period. You agree to accept and we or a participating entity are obligated to pay the amounts established by the agreement between you and us for covered health care services rendered according to this section after termination of this agreement.

(b) <u>Maternity Care.</u> If an enrollee is receiving maternity care from you and the enrollee is in her second or third trimester of pregnancy, you are obligated to continue the provision of health care services to that enrollee until the completion of the postpartum care. You agree to accept and we or the participating entity, as applicable, are obligated to pay the amounts established by the agreement between you and us for covered health care services rendered after termination of the agreement.

(<u>Primary Cat Phyacian</u> to you are a primary care physician, you are obligated to continue the provision of the ather re-services until the end of the current plan year for an enrollee with no open enrollment period; o until the end of the plan year for which it was represented that you were, or word to a participating provider for an enrollee with an open encomment period. You agree to occept we or the participating entity, as applicable, are obligated to provide any encourter of the participating entity, as applicable, are obligated to provide any encourter of the participating entity as applicable, are obligated to provide the services rendered after termination with a prement.

Additionally, in the event you terminate the agreement between you end us for any reason, you shall, within 30 days prior to the termination or 15 days following our receipt of the termination notice, whichever is later, post a non-cation of such termination in our office othis notice requirement applies only if you are a specialist and we do not require a referral.

3. Hold Harmless. If you provide services to menrollee after transmission of the agreement between you and us pursuant to the "Continue. Provision of Health Service, after Termination" section of this appendix, you may not, for any resion, including but not limited to example, after Termination, seek remuneration or compensation from, file or meater to de with a credit reporting agency or have any recourse against an enroller, or any person acting on their behalf for costs that are covered under the benefit plan issued under the benefit plan of a participating entity.

4. Grievances. You must identify complaints and grievances in a mely runner and forward these complaints and grievances to us in a timely manner.

Exhibit A to the Wisconsin Regulatory Requirements Appendix

NOTICE REQUIRED BY WISCONSIN STATUTE 609.94

NOTICE

THIS NOTICE DESCRIBES RECENTLY ENACTED HOLD-HARMLESS PROVISIONS WHICH AFFECT YOUR ABILITY TO SEEK RECOURSE AGAINST HMO ENROLLEES FOR PAYMENT FOR SERVICES.

Section 609.94, Wis. Stat., requires each health maintenance organization insurer (HMO) to provide a summary concepto all of its participating providers of the new statutory limitations and requirements in Sections 609.91 (609.935, and participation).

Under Wisconsin lav a heal ocare provider may of hold HMO enrollees or policyholders ("enrollees") liable for costs covered under in HMO policy if the provider is subject to statutory provisions which "hold harmless" the enrollees. For nost health care providers application of the statutory hold-harmless is "mandatory" or it applies in less the provider elects to "opt-ort." A provider permitted to "opt-out" must file timely notice with the Wisconsin office of the Compassioner reginsurance ("OCI").

MMA

Some types of provider care are subject to the heal-harmless statutes only of he provider voluntarily "opts-in." An HMO may partially stirfly its regular ry capital and surfaces equivalents if health care providers elect to remain subject to the statutor, hold-karmless provisions.

This notice is only a summary of the law. Deery efform as been made to accurately describe the law. However, if this summary is inconsistent with a precision of the law or incomplete, the law shall control.

HOLD HARN LESS

A health care provider who is subject to the statutory hold-hamle a provision is prohibited from seeking to recover health care costs from an enrollee. The provider may not bill charge, collect a deposit from, seek remuneration or compensation from, file or threaten to file with accredit reporting agency or have any recourse against an enrollee or any person acting on the enrollee behalf, for health care costs for which the enrollee is not liable. The prohibition on recover does not affect the liability of an enrollee for any deductibles or copayments, or for the premiums owed us ler the policy, or certificate issued by the HMO.

A. MANDATORY FOR HOLD HARMLESS

An enrollee of an HMO is not liable to a health care provider for health care costs that are covered under a policy issued by that HMO if any of the following are met:

1. Care is provided by a provider who is an affiliate of the HMO, owns at least 5% of the voting securities of the HMO, is directly or indirectly involved with the HMO through direct or indirect selection of or representation by one or more board members, or is an Individual Practice Association ("IPA") and is represented, or an affiliate is represented, by one of at least three HMO board members who directly or indirectly represent one or more IPAs or affiliates of IPAs; or,

2. Care is provided by a provider under a contract with or through membership in an organization identified in 1.; or

3. To the extent the charge exceeds the amount the HMO has contractually agreed to pay the provider for that health care service; or

4. The care is provided to an an olled medical assistant recipient under a Department of Health and Social Services prepaid health are olicy.

5. The care is required to be covid a uncleater requirements of Wis. Admin. Code, Ins. 9.35.

B. "O.T-OUT" HOLD-HARMLESS

If the conditions described in A do not apply, the provider shall be subject to the statutory hold harmless unless the provider has time delection with OCL a between if the health care meets any of the following:

1. Provided by a hospial an IPA or

2. A physician service, or oner provider services, equipment, supplies or drugs that are ancillary or incidental to such services and an provided under a contract with the HM of or are provided by a provider selected by the HMO; or

3. Provided by a provider, other that a hospital, upper a contract wither through membership in an IPA which has not elected to be exempt. No a that or the IPA may file election to exemplicate provided by its member providers from the statuton hold harmless (See Exemplions and Elections; #4).

C. "OPT-IN" HOLD HARMLESS

If a provider of health care is not subject to the conditions escribed a A o B, the provider may elect to be subject to the statutory hold-harmless provisions by filing a negative ion with the OCI stating that the provider elects to be subject with respect to any specific HMO. Provider maximizate such a notice of election by stating the termination date in that notice or in a separate notification.

CONDITIONS NOT AFFECTING IMMUNIT

An enrollee's immunity under the statutory hold harmless is not affected by any of the following:

1. Any agreement entered into by a provider, an HMO, or any other person, whether oral or written, purporting to hold the enrollee liable for costs (except a notice of election or termination permitted under the statute);

2. A breach of or default on any agreement by the HMO, an IPA, or any other person to compensate the provider for health care costs for which the enrollee is not liable;

3. The insolvency of the HMO or any person contracting with the HMO, or the commencement of insolvency, delinquency or bankruptcy proceedings involving the HMO or other persons which would affect compensation for health care costs for which an enrollee is not liable under the statutory hold harmless;

4. The inability of the provider or other person who is owed compensation to obtain compensation for health care costs for which the enrollee is not liable;

5. Failure by the HMO to provide notice to providers of the statutory hold-harmless provisions; or

6. Any other condition or agreement existing at any time.

Hospials, IPAs, and providers of physician prvice who may "opt-out" may elect to be exempt from the statute v hold harmless and prohibition or recovery or health care costs under the following conditions and with the following protifications:

ONS AND ELECTIONS

1. If the hospite IPA coubter provider has a written contract with the HMO, the provider must within thirty (30) do is after evering into that contract provide a notice to the OCI of the provider's election to be exempt from the state ory hold harmles and recover constantions for care under the contract.

2. If the hospital, IPA, or other provide does not have a contrart with the HMO, the provider must notify OCI that it intends to be exampt with respect to a specific H 10 are 1 ust provide that notice at least ninety (90) days in advance.

3. A provider who submits a notice of electron to be exempt may term that the electron by stating a termination date in the notice or by submitting a separate termination notice to DCI.

4. The election by an IPA to be exempt from the statutor provisions, or the failure of an IPA to so elect, applies to costs of health care provided by any provider, other than a hospital, an aer contract with or through membership in the IPA. Such a provider, other than a hospital may not exercise an election separately from the IPA. Similarly, an election by a clinic to be exempt are restrictions of the failure of the clinic to elect to be exempt are restrictions of health care provided by any provider may not exercise an election to be exempt are restrictions of the failure of the clinic. An individual provider may not exercise an election to be exempt separate from the clinic.

5. The statutory hold-harmless "opt-out" provision applies to physican service only if the services are provided under a contract with the HMO or if the physician is a selected provider for the HMO, unless the services are provided by a physician for a hospital, IPA or clinic which is subject to the statutory hold-harmless "opt-out" provisions.

NOTICES

All notices of election and termination must be in writing and in accordance with rules promulgated by the Commissioner of Insurance. All notices of election or termination filed with OCI are not affected by the renaming, reorganization, merger, consolidation or change in control or the provider, HMO, or other

person. However, OCI may promulgate rules requiring an informational filing if any of these events occur.

Notices to the Office of the Commissioner of Insurance must be written and received at the Office's current address:

Office of the Commissioner of Insurance 123 West Washington Avenue P.O. Box 7873 Madison, WI 53707

. . /

HMC & PITAL AND SECURITY SURPLUS

Each HM ars equired to most minimum equital and surplus standards ("compulsory surplus requirements"). These standards are night of if the HMO has fewer than 90% of its liabilities covered by the stantary hole harmless. Specifically, the compulsory surplus requirements shall be at least the greate of \$750,000 or 6% of the premium eached by the HMO in the last 12 months if its covered liabilities are less than 60%, or 3% of the premiums eached by the HMO in the last 12 months if its covered liabilities are less than 90%.

In addition to capita and carries, an HMO must iso matriain a security surplus in the amount set by the Commission of Insurance.

FINAN LIAL INFORMATION

An HMO is required to file finance, et emerges whethe OCI. You may request financial statements from the HMO. The OCI also maintains files CHMO financial statements that can be inspected by the public.

MEDICARE ADVANTAGE REGULATORY REQUIREMENTS APPENDIX

\$

THIS MEDICARE ADVANTAGE REGULATORY REQUIREMENTS APPENDIX (this "Appendix") supplements and is made part of the network participation agreement (the "Agreement") between United and the physician or provider named in the Agreement ("Provider").

SECTION 1 APPLICABILITY

This Appendix applies to the overed Services Provider provides to Medicare Advantage Customers. In the event of a conflict between this Appendix and other appendices or any provision of the Aprement, the provision of this Appendix shall control except: (1) with regard to Benefit Plans catside the scope of this Appendix; (2) as noted in Section 2 of this Appendix; or (3) as required by appreciable law.

SECTIC 2 VEFIN (ION

For purposes of this appendix, the following terms shall have the meanings set forth below; provided, however, in the event any definition set forth intensis appendix is in conflict with any definition in the Agreement or the same or substantially amilar term, the definition for such term in the Agreement shall control. All other capitalized terms not otherwise defined in this Appendix shall be as defined in the Agreement.

2.1 **Benefit Plan:** A certificate of coverage, summary plan description, or other document or agreement, whether delivered in paper, electronic, or other former, uncer which a Payer is obligated to provide coverage of Coverel Services for a Customer. B nefit Plan may also be referred to as benefit contract, benefit document, plan, or other similar term under the Agreement.

2.2 **CMS Contract:** A contract between the conters for Medicare & Medicaid Services ("CMS") and a Medicare Advantage Organization for the provision of Medicare benefits pursuant to the Medicare Advantage Program under Title XVIII, Part (10) the Social Security Act.

2.3 **Cost Sharing:** Those costs, if any, under a Benefit Plan the are the responsibility of the Customer, including deductibles, coinsurance, and copayments. Cost Sharing may also be referred to as patient expenses or other similar term under the Agreement.

2.4 **Covered Service:** A health care service or product for which a Customer is entitled to receive coverage from a Payer, pursuant to the terms of the Customer's Benefit Plan with that Payer. A Covered Service may also be referred to as a health service or other similar term under the Agreement.

1

2.5 **Customer:** A person eligible and enrolled to receive coverage from a Payer for Covered Services. A Customer may also be referred to as an enrollee, member, patient, covered person, or other similar term under the Agreement.

2.6 **Dual Eligible Customer:** A Medicare Advantage Customer who is: (a) eligible for Medicaid; and (b) for whom the state is responsible for paying Medicare Part A and B Cost Sharing.

2.7 **Medicare Advantage Benefit Plans:** Benefit Plans sponsored, issued or administered by a Medicare Advantage Organization as part of the Medicare Advantage program or as part of the Medicare Advantage program to there with the Prescription Drug program under Title XVIII, Part C and Part D, respectively, if the Social Security Act (as those program names may change from time to time).

2.8 Medicate Advantage Curtome of MA Customer: A Customer eligible for and enrolled in a Medicare Advantage Pener. Plan in which Provider participates pursuant to the Agreement.

2.9 Medicare Advantige Organization or VA Organization: For purposes of this Appendix, MA Organization is either United or Payer.

2.10 **Payer:** An entire obligated on a Customer to provide reimbursement for Covered Services under the Customer's Benefit Plan and authorized by United to access Provider's services under the Agreement. A Payer manual so be referred to as a payor, participating entity or other similar term under the Agreement.

2.11 United: UnitedHealthcare Instructer Company and/or one of more of its at liates.

SECTION 3 PROVIDER R. QUIPUMENTS

3.1 **Data.** Provider shall submit to MA Organization all rise adjustment data as defined in 42 CFR 422.310(a), and other Medicare Advantage program-calated information as may be requested by MA Organization, within the timeframes specified and a form that meets Medicare Advantage program requirements. By submitting data to MA Organization, Provider represents to MA Organization, and upon MA Organization's rangest Provider shall certify in writing, that the data is accurate, complete, and truthful, based on Provider's best knowledge, information and belief.

3.2 **Policies.** Provider shall cooperate and comply with MA Organization's policies and procedures.

3.3 **Customer Protection.** Provider agrees that in no event, including but not limited to, nonpayment by MA Organization or an intermediary, insolvency of MA Organization or an intermediary, or breach by United of the Agreement, shall Provider bill, charge, collect a deposit

2

from, seek compensation, remuneration or reimbursement from, or have any recourse against any MA Customer or person (other than MA Organization or an intermediary) acting on behalf of the MA Customer for Covered Services provided pursuant to the Agreement or for any other fees that are the legal obligation of MA Organization under the CMS Contract. This provision does not prohibit Provider from collecting from MA Customers allowable Cost Sharing. This provision also does not prohibit Provider and an MA Customer from agreeing to the provision of services solely at the expense of the MA Customer, as long as Provider has clearly informed the MA Customer, in accordance with applicable law, that the MA Customer's Benefit Plan may not cover or continue to cover a specific service or services.

In the event of MA Organization's or an intermediary's insolvency or other cessation of operations or termination of MA Organization's contract with CMS, Provider shall continue to provide Covered Services to an MA Customer through the later of the period for which premium has been paid to MA Organization on beneficient to MA Customer, or, in the case of MA Customers who are hospitalized as a such period or date, the MA Customer's discharge.

This provision shou be construed in favor of the MACSustomer, shall survive the termination of the Agreement regardless of the reason for termination, including MA Organization's insolvency, and shall supersede any control agreement, oral to written, between Provider and an MA Customer or the representative of an MA Customer if the contrary agreement, is inconsistent with this provision.

For the purpose of this provision an extermediary" is operson opentity authorized to negotiate and execute the Agreement of behalf of Provider or on behalf of a network through which Provider elects to participate.

3.4 **Dual Eligible Customers.** Provide agrees that in no event, including but not limited to, non-payment by a state Medicaid agence or other applicable regulator mutherary, over state source, or breach by United of the Agreement, thall Provider bill, charge, collect a deposit from, seek compensation, remuneration or reimbursementation, or have any recourse evaluation on Dual Eligible Customer, person acting on behalf of the Dua Eligible Customer, or MA Organization (unless notified otherwise) for Medicare Part A and B Cost charing. Instead, Provider will either: (a) accept payment made by or on behalf of MA Organization of payment in full; or (b) bill the appropriate state source for such Cost Sharing amount. If Provider imposes an excess charge on a Dual Eligible Customer, Provider is subject to any lawfunction, that may be imposed under Medicare or Medicaid. This provision does not prohibit Provider and a Dual Eligible Customer, as long as Provider has clearly informed the Dual Eligible Customer, accordance with applicable law, that the Dual Eligible Customer's Benefit Plan may not cover or commute to cover a specific service or services.

3.5 Eligibility. Provider agrees to immediately notify MA Organization in the event Provider is or becomes excluded from participation in any federal or state health care program under Section 1128 or 1128A of the Social Security Act. Provider also shall not employ or contract for the provision of health care services, utilization review, medical social work or administrative services, with or without compensation, with any individual or entity that is or becomes excluded from participation in any federal or state health care program under Section 1128 or 1128A of the Social Security Act.

3.6 **Laws.** Provider shall comply with all applicable federal and Medicare laws, regulations, and CMS instructions, including but not limited to: (a) federal laws and regulations designed to prevent or ameliorate fraud, waste, and abuse, including but not limited to, applicable provisions of federal criminal law, the False Claims Act (31 U.S.C. §3729 et seq.), and the anti-kickback statute (§1128B of the Social Security Act); and (b) HIPAA administrative simplification rules at 45 CFR Parts 160, 162, and 164.

3.7 **Federal Funds.** Provide acknowledges and agrees that MA Organization receives federal payments under the CLOS Contract and that payments Provider receives from or on behalf of MA Organization are, in whole or integrit, from federal funds. Provider is therefore subject to certain aws the are applicable or individual, and entities receiving federal funds.

3.8 **CMS Contract**. Provider shall perform the services set forth in the Agreement in a manner consistent with and in complement with MCOrganization's contractual obligations under the CMS Contract

3.9 **Records.**

(a) <u>Maintenance</u>, <u>Povacy and Confidentiality</u>; <u>Customer Access</u>. Provider shall maintain records and information related to the service provided under the Agreement, including but not limited to MA Customer medical records and other health and enrollment information, accurate and timely manner. Howide shall maintain such records for the longer of the following periods:

(i) in the case of records containing information related to the medical loss ratio information reported to CMS by the MA Organization, including, for example, information related to incurred claims and quality improvement activities, at least ten (10) years from the date such medical loss ratio information is reported to CMS by the MA Organization, or

(ii) in the case of all records, at least ten (10) years from the final date of the CMS Contract period in effect at the time the records were created, or such longer period as required by law.

Provider shall safeguard MA Customer privacy and confidentiality, including but not limited to the privacy and confidentiality of any information that identifies a particular MA Customer, and shall comply with all federal and state laws regarding confidentiality and disclosure of medical records or other health and enrollment information. Provider shall ensure that MA Customers have timely access to medical records and information that pertain to them, in accordance with applicable law.
(b) <u>Government Access to Records</u>. Provider acknowledges and agrees that the Secretary of Health and Human Services, the Comptroller General, or their designees shall have the right to audit, evaluate and inspect any pertinent books, contracts, computer or other electronic systems (including medical records), patient care documentation and other records and information belonging to Provider that involve transactions related to the CMS Contract. This right shall extend through the longer of the following periods:

(i) in the case of records containing information related to the medical loss ratio information reported to CMS by the MA Organization, including, for example, information related to incurred claims and quality improvement activities, at least (10) years from the date such medical loss ratio information is reported to C of by the MA Organization, or

(i) in the case of all records, at least ten (10) years from the later of the final of the CMS Contract verice in effect at the time the records were created or the date of completion of any audit, or longer in certain instances described in the argunerole Medicare Alyantage regrettions.

For the purpose or conducting the pove activities, Provider shall make available its premises, any call acilities and equipment, records plating to MA Customers, and any additional relevant information CMS may require.

(c) <u>MA Organization Access to Records</u>. Frovider shall gont MA Organization or its designees such audit, evaluation, and inspection rights ideal ed in subsection 3.9(b) as are necessary for MA Organization to comply with us a cligations under the CMS Contract. Whenever possible and Organization will give Provider reasonable notice of the need for such audit, evaluation or inspection, and will conduct ach abilit, evaluation or inspection at a reasonable time and made. Provider shall submit medical records of MA Customers to the MA Organization as may be requested, within the timeframes specified, for the purpose of (i) CMS budits or risk adjustment corporation (ii) for other purposes medical records from providers are assed by MA Organization, as specified by CMS. Provision of medical records must be in the name or consistent with HIPAA privacy statute and regulations.

3.10 **MA Organization Accountability; Delegated Activities.** Provider acknowledges and agrees that MA Organization oversees and is accountable to SMS for any functions and responsibilities described in the CMS Contract and applicable Medica and vantage regulations, including those that MA Organization may delegate to Provider or others. If MA Organization has delegated any of its functions and responsibilities under the CMS Contract to Provider pursuant to the Agreement, the following shall apply in addition to the other provisions of this Appendix:

(a) Provider shall perform those delegated activities specified in the Agreement, if any, and shall comply with any reporting responsibilities as set forth in the Agreement.

5

(b) If MA Organization has delegated to Provider any activities related to the credentialing of health care providers, Provider must comply with all applicable CMS requirements for credentialing, including but not limited to the requirement that the credentials of medical professionals must either be reviewed by MA Organization, or the credentialing process must be reviewed, pre-approved and audited on an ongoing basis by MA Organization.

(c) If MA Organization has delegated to Provider the selection of health care providers to be participating providers in MA Organization's Medicare Advantage network, MA Organization retains the right to approve, suspend or terminate the participation status of successful care providers.

(d) Provider a knowledge that MA Organization shall monitor Provider's performance of any doegated activities on an ongoing basis. If MA Organization or CMS determines that Provider has no performed satisfactorily, MA Organization may revoke any or all delegated activities and reporting requirements. Provider shall cooperate with MA Organization regarding the transition of any delegated activities or reporting the transition of any delegated activities or reporting the transition.

3.11 **Subcontracts** of provider has any apagements, no accordance with the terms of the Agreement, with affiliates subsidiaries, or any other subcontractors, directly or through another person or entity, to perform and or the services Provider is obligated to perform under the Agreement that are the subject of this Al bendix, Provider shall insurathat all such arrangements are in writing, duly executed and include all the terms contained at this Appendix. Provider shall provide proof of such to 11. Organization upon request. Provider further agrees to promptly amend its agreements with such subcontractors, in a matter consistent with the changes made to this Appendix by Ma Organization, to meet any contained CMN requirements that may apply to the services.

3.12 **Offshoring.** Unless previously authorized by AA Organization here ting, all services provided pursuant to the Agreement that are subject to this Appendix must be performed within the United States, the District of Columbia, or the United States territories

SECTION 4 OTHER

4.1 **Payment.** MA Organization or its designee shall prompty process and pay or deny Provider's claim no later than sixty (60) days after MA Organization or its designee receives all appropriate information as described in MA Organization's administrative procedures. If Provider is responsible for making payment to subcontracted providers for services provided to MA Customers, Provider shall pay them no later than sixty (60) days after Provider receives request for payment for those services from subcontracted providers.

4.2 **Regulatory Amendment.** MA Organization may unilaterally amend this Appendix to comply with applicable laws and regulations and the requirements of applicable regulatory

. .

authorities, including but not limited to CMS. MA Organization shall provide written or electronic notice to Provider of such amendment and its effective date. Unless such laws, regulations or regulatory authority(ies) direct otherwise, the signature of Provider will not be required in order for the amendment to take effect.



ci) /

WISCONSIN BADGERCARE PLUS AND MEDICAID SSI REGULATORY REQUIREMENTS APPENDIX

THIS WISCONSIN BADGERCARE PLUS AND MEDICAID SSI REGULATORY REQUIREMENTS APPENDIX (this "Appendix") supplements and is made part of the provider agreement (the "Agreement") between UnitedHealthcare of Wisconsin, Inc. ("United") and the provider named in the Agreement ("Provider").

SECTION 1 APPLICABILITY

This Appendix applies wan espect to the provision of health care services that Provider provides directly to Covered Per or under the State of Wisconsin BadgerCare Plus and Midicaid Sol programs (collection), the "State Medicaid Program"), as governed by the State's designated regulatory agencies. In the event of a conflict between this Appendix and other appendices or any provision of the Agreement, the provisions of this Appendix shall control excert with regard to benefit contracts outside the scope of this Appendix or unless otherwise required by law. In the event United is required to amend or supplement this Appendix is recent d or requested by the State, Provider agrees that United shall be permitted to unitable such additions, deletions a modifications.

SECTION 2 EFINITIONS

Unless otherwise defined in the Appendix, ell capitalized terms shall be as defined in the Agreement. For purposes of this appendix, the following terms that have the meanings set forth below; provided, however, in the event any definition are forth in the Appendix or the Agreement is inconsistent with any domitions under the State Medicaid Program, the definitions shall have the meaning set orth under the State Medicaid Program.

2.1 Agreement: An executed contract between United and Provider for the provision of Covered Services to persons enrolled in the Site Medical Program.

2.2 **BadgerCare Plus:** The Wisconsin State program that merger Family Medicaid, BadgerCare, and Healthy Start to form a comprehensive health insurance program for low income children and families.

2.3 **Clean Claim:** A truthful, complete and accurate claim the coses not have to be returned for additional information.

2.4 **Covered Person:** An individual who is currently enrolled with United for the provision of services under the State Medicaid Program. A Covered Person may also be referred to as an Enrollee, Member or Customer under the Agreement.

2.5 **Covered Services:** A health care service or product for which a Covered Person is enrolled with United to receive coverage under the State Medicaid Contract.

2.6 **Department or DHS:** The Wisconsin Department of Health Services.

2.7 **Provider:** A hospital, ancillary provider, physician group, or individual physician who has entered into an Agreement.

2.8 **State:** The State of Wisconsin or its designated regulatory agencies.

2.9 **State Medicaid Contract:** United's contract with the Wisconsin Department of Health Services for the purpose of providing and paying for Covered Services to Covered Persons enrolled in the BadgerCare Plus and Medicaid SSI programs (collectively, the "State Medicaid Program")

2.10 Suite Medicaid Programs The BadgerCare Plus and Medicaid SSI programs. For purposes of this Appendix, State Medicaid Program may refer to the State apency(ies) responsible for administrong the State Medicaid Program.

SECTION 3 PROVIDER REQUIREMENTS

The State Medicaid Program, through federal and State statutes and regulations, requires the Agreement to compare creatin conditions that United and Provider agree to undertake, which are as follows:

3.1 Provider shall follow the State Medicuid Contract's activitions for the coverage of Covered Services. Provider's decision affecting the delivery of acute or chronic care services to Covered Persons shall be made on an individualized asist and in accordance with the following definitions:

(a) <u>Emergency Medical Condition</u>: Emergency redical Condition is based all of the following:

(1) A medical condition manifesting its of the aute symptoms of sufficient severity (including severe pain) such that a protent lapperson who possesses an average knowledge of health and medicine, could reasonably expect the absence of immediate medical attention to result in any of the following:

- (i) placing the health of the individual (or with respect to a pregnant woman, the health of the woman or her unborn child) in serious jeopardy;
- (ii) serious impairment of bodily functions; or
- (iii) serious dysfunction of any bodily organ or part.
- (2) With respect to a pregnant woman who is in active labor:

UnitedHealthcare/AmeriChoice Agreement UHC-AMC/UHN PROVIDER.MEDICAID.REGAPX.02.09.WI

- (i) where there is inadequate time to effect a safe transfer to another hospital before delivery; or
- (ii) where transfer may pose a threat to the health or safety of the woman or the unborn child.

(3) A psychiatric emergency involving a significant risk of serious harm to oneself or others.

(4) A substance abuse emergency exists if there is significant risk of serious harm to a Covered Person or others, or there is likelihood of return to substance abuse without immediate treatment.

(5) Emergence dental care is defined as an immediate service needed to receive the patient from prin, an acute infection, swelling, trismus, fever, or trauma. In all emergence situations, the HMO must document in the Covered Person's dental record the require of the emergency.

(b) <u>Ergence</u> <u>Services</u>: Covered inpatient and outpatient services that are furnished by a provide qualified to furnish hose weight services and that are needed to evaluate or stabilize an Ergency Medical Condition

(c) <u>Medically Necessary</u>: A medical service that neets the definition of Wis. Adm. Code HFS 101.03 nombas may be amended from time to time.

3.2 Provider must be artified by the BadgerCare Plus approximation Medicaid program for services required under the Agreement. DHS reserves the right to withhold retrospectively from the capitation pays ents the monies related to services provided by any non-Medicaid or BadgerCare plus-certifier physicians or providers, at the Medicaid fee-for-service rate for those services.

3.3 Provider shall abide by the terms of the State Medicaid Contract or the timely provision of emergency and urgent care.

3.4 Where applicable, Provider shall follow those procedures for handling urgent and emergency care cases stipulated in any required hospit. Vemergency from Memoranda of Understanding (MOU) signed by United in accordance whether Stree Medicaid Contract.

3.5 Provider agrees to submit encounter data in the format specified by United, so that United can meet the Department's specifications required under the State Medicaid Contract. United will evaluate the credibility of data obtained from external databases to ensure that any patient-reported information has been adequately verified. Provider shall also cooperate with United in its preparation of reports and clinical information required under the State Medicaid Contract including, without limitation, all child and adolescent health check-up reporting, HealthCheck encounters, and cancer screening encounters, as appropriate, and such other reporting for Covered Services as may be required under the State Medicaid Contract. 3.6 Provider shall ensure the confidentiality of family planning services in accordance with the terms of the State Medicaid Contract.

3.7 Provider shall abide by the terms of the State Medicaid Contract regarding appeals to United and DHS for non-payment by United for services rendered to Covered Persons by providers, including:

(a) United must accept written appeals from Provider if Provider disagrees with United's payment/denial determination as long as Provider submits the dispute in writing and within sixty (60) days of the initial payment/denial notice. United has forty-fit (45) days from the date of the receipt of the request for reconsideration to respond to Provider in writing. If United fails to respond writin that time, or if Provider is not satisfied with United's response, Provider may reek a final determination con DHS.

Used must inform Provider in writing of United's payment/denial ninations including:

- A specific explanation of the payment amount or a specific reason for the payment deniel
- (ii) a statement begarding Provider's rights and responsibilities in applaling to United about Uniter's initial determination by submitting a separate letter or form: (a) charly marked "appeal";
 (b) containing the provider's name, that of service, date of billing, date of rejection, the Covered Person's name and BedgerCare Plus and/or Medicaid SSI ID number, and recon(a) the chim merits reconsideration; (c) for each appeal; (d) addressed to the person and/or department of United the handles provider appeals within sixty (60) days of the initial denial or partial payment.
- (iii) A statement advising Proviler of itteragheto appeal to the DHS if United fails to respond to the appeal within forty-five (45) days or if Provider is not satisfied with united's response to the request for reconsideration, and that all appeals to the EHS must be submitted in writing within sixty (60) days of United's final decision or, in the case of no response, within sixty (60, days from the forty-five (45) day timeline allotted United to responde on cases where there is a dispute about United's payment/denial determination and Provider has requested reconsideration, the DHS will hear appeals and make final determinations. The DHS will not exercise its authority in this regard unreasonably. The DHS will accept written comments from all parties to the dispute before making the decision. United and Provider must accept DHS's determinations regarding appeals of disputed claims. If DHS's decision is in favor

Ъ)

of Provider, United will pay Provider within forty-five (45) days of receipt of DHS's final determination.

3.8 Provider shall provide for timely access for Covered Person appointments in accordance with the appointment availability requirements established under the State Medicaid Contract including without limitation, appointments for preventative care, urgent care, routine sick care, and well care.

3.9 Provider shall cooperate with United and provide a Covered Person with continuity of treatment (which may include coordination of care as required under law) in the event Provider's participation with United terminates during the course of a Covered Person's treatment by Provide

3.100 Provider shall not create or ders to access to care by imposing requirements on Covered Persons that are incorportent with the provision of Medically Necessary and covered BadgerCare Plus and or Medicald SSI benefits (e.g., coordination of benefits recover proceedings that delay or prevent care).

shall lok solely to United to payment of Covered Services provided to Provid 3.11 Covered Persons purpoint to the Agreement and he Stars Medicaid Contract and hold the State and Covere Perfors Larmless in the event that Unted cannot or will not pay for such Covered Service, Povider agrees not to bill BargerCare Plus and/or Medicaid SSI Covered Persons for reducing Necessary service covered under the State Medicaid Contract and provided during the Overed Person's period of collment with United, pursuant to Section 1128(1)(d)(1) of the pocial Security Act. relider also agrees not to bill a Covered Person for any classe appointments while the Covered Person is eligible under the BadgerCare Plus - Stendard Plan and/or Medicaid Storograms; provided, however, Covered Persons eligible under the sadgerCare Plus - Buchman Plan (as described in the State Contract) may be billed for missed appointments. In a dition, a Covered Person eligible under the Badge care Plus - tandard Plan as de cribed in the State Contract) or the BadgerCare Plus – Renchmerk Plan may be bined or applicable copayments and/or premiums for Medically Decessary services provided during the Covered Person's enrollment with United. Provider may not bine Medicaid SSI Covered Person for copayments or premiums for Medically Newson services provided during the Covered Person's enrollment with United. This pyrision will main in effect even if United becomes insolvent.

Notwithstanding the foregoing, if a Covered Person agrees in writing pay for a noncovered service, then Provider, a subcontractor, or United can bill the covered Person for the service. The standard release form signed by the Covered Person at the time of services does not relieve Provider, a subcontractor or United from the prohibition against billing a BadgerCare Plus – Standard Plan or Medicaid SSI Covered Person in the absence of a knowing assumption of liability for a non- covered service. The form or other type of acknowledgment relevant to BadgerCare Plus or Medicaid SSI Covered Person liability must specifically state the admissions, services, or procedures that are not covered by BadgerCare Plus or Medicaid SSI. 3.12 Provider shall cooperate with United in the event an immediate transfer to another primary care physician or Medicaid managed care contractor is warranted if the Covered Person's health or safety is in jeopardy, as may be required under law.

3.13 In addition to the amount, duration, and scope of Covered Services to be provided by Provider as specified in the Agreement and the State Medicaid Contract, Provider shall continue to provide Covered Services through the duration of the Agreement including, without limitation, the applicable capitation or premium payment period for which the State has paid to United.

3.14 Provider shall offer to its of operation that are no less than the hours of operation offered to commercial benefit daries or comparable to Medicaid fee-for-service if Provider serves one Medicaid ite-for-service operation operations.

3.5 In the event of transitioning coveres Persons from other Medicaid managed care contractors and their provider, Provide shall work with United to ensure quality-driven health outcomes or such Coveres Persons transference extent required by the State Medicaid Contract or other wise required by law

3.16 Provider chall of packe referrals or designated health services to health care entities with which the Provider or a member of the Provider's family has a financial relationship, pursuant to fideral initi-kickback and physician self-referral laws that prohibit such referrals.

3.17 Provider shall provide information to Covered persons regarding treatment options, including the option of neureautent, in a culturally-competent menner and must ensure that individuals with dischilities have effective contraunitations on making decisions regarding treatment options.

3.18 Provider shall not charge for any service provided to a Covered Lucion at a rate in excess of the rates established by the Agreement in accordance with Section 1128B(d)(1) of the Social Security Act (enacted by Section 47.4 of the Lulanced Budget Act of 1997), as may be amended from time to time.

3.19 Provider shall comply with all non-discrimination equirements a set forth in the State Medicaid Contract, including but not limited to: (i) complying with all applicable federal and State laws relating to non-discrimination and equal employment opportunity, including s. 16.765 Wis. Stats., the Federal Civil Rights Act of 1.004 and regulations issued pursuant to that Act, and the provisions of Federal Executive Order 11246 dated September 26, 1985; (ii) assuring physical and program accessibility of all services to persons with physical and sensory disabilities pursuant to Section 504 of the Federal Rehabilitation Act of 1973, as amended (29 U.S.C. 794); (iii) complying with all requirements imposed by the applicable State and federal regulations (45 C.F.R. part 84) and all guidelines and interpretations issued pursuant thereto; and (iv) complying with the

provisions of the Age Discrimination and Employment Act of 1967 and the Age Discrimination Act of 1975.

3.20 As required under State or federal law or the State Medicaid Contract, Provider shall maintain an adequate record keeping system for recording services, charges, dates and all other commonly accepted information elements for services rendered to Covered Persons. Provider shall comply with all record retention requirements under the State Medicaid Contract and, where applicable, the special compliance requirements on abortions, sterilizations, hysterectomies, and HealthCheck reporting requirements. Provider shall maintain records for a period of not less than ten (10) years from the close of the State Medicaid Contract or such other period as required by law. If records are under review or audit, they arist be retained until the review or audit is complete. United will request and obtain provapproval from Provider for the disposition of records under review or apprection.

3 21 Provider shall safegur a information about Covered Persons in accordance with applicable federational State privilegy large and rules including 42 CFR §438.224 and 42 CFR Part Lt., Support F, as may be amended tool time to time.

3.22 As required oncer State or feder (law of the state Medicaid Contract, Provider shall provide representatives of United as well studied authorized agents or representatives of the Department and the U.S. Department of Health and Human Services, access to its contracts and/or medical records. Provider shall otherwise preserve the full confident ality of medical records in a bordance with the State Medicaid Contract and pursuant to: Charter 19, Subcharter 19, Vis. Stats., Wis. Admin. Code HFS 108.01, and 42 Cause. 42 (Subpart F. Except as otherwise required by law, rule, or regulation, access to such information must be limited by United and the Department to persons who, or a tencies which, require the information is order to perform their duties related to the State Medicaid Contract, including the U.S. Department of Health and Human Services and such others as may be required by the Department.

Provider also agrees to make available to the Department the Department's authorized agents and appropriate representatives of the U.S. Department of Health and Human Services and the U.S. Comptroller General any financial records of Provider that relate to the services performed and amounts paid or payable under the State Medicaid Contract.

3.23 Provider shall abide by all requirements for maintenance and transfer of medical records pursuant to the terms of the State Medicaid Contract. Minimum medical record documentation per chart entry or encounter must conform to Wis. Admin. Code, Chapter HFS 106.02, (9)(b), as may be amended from time to time.

3.24 Provider shall clearly specify referral approval requirements to its providers, if any, and in any sub-subcontracts.

3.25 Within fifteen (15) days of United's request, Provider shall forward to United medical records related to grievances. If Provider does not meet this fifteen (15) day requirement, Provider must explain why and indicate when the medical records will be provided.

3.26 Provider shall abide by United's marketing/informing requirements. Provider shall forward to United for prior approval all flyers, brochures, letters, and pamphlets Provider intends to distribute to Covered Persons concerning its United affiliation(s), or changes in affiliation, and other information that relates directly to the BadgerCare Plus and/or Medicaid SSI population. Provider shall not distribute any such marketing or Covered Person informing materials without the consent of United and DHS.

3.27 Provider shall montron during the term of the Agreement, as applicable, general liable y mourance, professional flat any insurance, and workers' compensation insurance for all employees connected with the devision of services under the Agreement. Such workers compensation insurance sholl comply with State Workers' Compensation Law. Such comprehensive general liable ity insurance and professional liability insurance shall provide contrage in an amount established by United pursuant to the Agreement or as required under the State Medicaid Contract.

3.28 Provider chall adematy and hold be State and revered Persons harmless from and against all claims declages, causes of action, cost or expense, including court costs and reasonable attorney feet, to the extent proximitely caused by any negligent act or other wrongful conduct ausing in connection with the Agreement. This clause shall survive the termination of the Agreement including breach due to insolvency. The State Medicaid Program reserves the aghter wait, this requirement for itself, but not Covered Persons, for damages in excess of the statutory cap on damages for public entities if Provider is a public health entity with statutory immunity. As such waiver, must be approved in writing by the State Medicaid Program.

3.29 Provider shall comply with any oltural competency program chablished by United, consistent with the terms of the State Menacaid Contract.

3.30 Provider shall participate in and contribute regard data to United's Quality Assessment/Performance Improvement programs.

3.31 In the event Provider participates in a physician incentive plan (*) IP"), Provider agrees that all PIPs must comply with 42 CFR 417.479, 42 CFR 4.6.6(h), 42 CFR 422.208, and 42 CFR 422.210, as may be amended from time to time. Neither United nor Provider may make a specific payment directly or indirectly under a PIP to a physician or physician group as an inducement to reduce or limit Medically Necessary services furnished to an individual Covered Person. PIPs must not contain provisions that provide incentives, monetary or otherwise, for the withholding of Medically Necessary care. 3.32 United will provide monitoring and oversight and Provider shall ensure that all licensed medical professionals are credentialed in accordance with the applicable State Medicaid Contract credentialing requirements if United delegates credentialing to Provider.

3.33 If Provider delegates any functions of the Agreement, the subcontract or delegation must include all of the requirements of this Appendix, and applicable requirements of the State Medicaid Contract.

3.34 Provider shall comply with all applicable privacy rule and security rule provisions of the Health Insurance Portubility and Accountability Act of 1996 ("HIPAA"), and associated implementing rest tions, as may be amended from time to time.

3.35 Provider shall comply winter a applicable federal and State statutes and rules and regulations that are in effect when the cate Medicaid Contract is signed, or that come in o effect during the term of the state Medicaid Contract. This includes, but is not limited to Title LX of the Social Security Act. Title XXI, SCHIP, and Title 42 of the CFR.

3.36 Provide is sociect to all State are federal laws and regulations relating to fraud, abuse or waste in neal n care and the State Medicaid Program. Provider shall cooperate and assist the State Medicaid Program and any other State or federal agency charged with the duty of identifying investigating, sanctioning or prosecuting suspected fraud, abuse or waste. Provider shall provide originals and/or copies of any and all information, allow access, wherever Provider maintains such books, to previses and provide records, to the State Medicaid Program's fraud are abuse designee, CluS, the U.S. Department of Health and Human Services, the Eldera Bureau of Investigation (For), or any other unit of State or federal government upor request, are free-of-charge.

3.37 Provider understands that no terms of this Appendix or the Agreement are valid which terminate the legal liability of United.

3.38 Provider shall comply, as applicable, with United's givin Lights Compliance Plan.

3.39 Provider shall comply with 42 CFR 438.214, is may be appended from time to time, which includes, but is not limited to the selection and etention of providers, credentialing and recredentialing requirements and nondiscrimination.

3.40 Provider shall cooperate with any audits or other activities conducted by an External Quality Review Organization ("EQRO") as may be required by the State Medicaid Program.

3.41 Provider shall be subject to all applicable accreditation standards (e.g., National Committee for Quality Assurance ("NCQA") accreditation), as may be set forth in the Agreement and any applicable attachments thereto.

3.42 Provider agrees that all relevant federal and State statutes and rules pertaining to Medicaid Managed Care Organizations apply and, in addition, Provider shall comply with the applicable provisions of 42 CFR 434 and 42 CFR 438.6, as may be amended from time to time.

3.43 Upon termination of the Agreement pursuant to the terms contained therein, Provider shall promptly supply United with all information necessary for the reimbursement of any outstanding Medicaid claims.

3.44 If the Agreement is for an amount in excess of \$100,000, Provider shall comply with all applicable standards, orders or regulations issued pursuant to the Clean Air Act, 42 U.S.C. 7401 et seq., and the Federal Water Pollution Control Act, as amended, 33 U.S.C. 1251 et seq. Are violations shall be reported to the U.S. Department of Health and around Services are the expropriate Regional Office of the Environmental Potection Agency.

3.4. Prohibition Use of Federal Fonds for Lobbying: Provider agrees, pursuant to 31 U.S.C. action 1352 and 45 Cr3 Part 93 as have be amended from time to time, that no federally appropriated funds have been poid or will be paid to any person by or on Provider's behalf for the purpose of influencing or attempting to influence an officer or employee of any igenor, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with obtaining any federal contract, grant or any other a and overed by 31 U.S.C. section 152. If the value of the Agreement exceeds \$100,00, Provider agrees to complete approximit to United the certification required under 31 U.S.C. Section 1352 and so CFLP rt 93.

3.46 Provider represents that neither hanor any of its principals providers with whom it contracts, if any, is debarred, su pended or otherwise excluded from participation in any state or federal health care program or be any state or federal age cy.

3.47 To the extent applicable, Provider scall concerate with United Lube Department informs United that the number of Active Coveral Persons from all populations serviced under the Agreement exceeds capacity limits under the Sote Medicaid Contract. Provider shall cooperate with United to resolve any capacity limit that has been exceeded and will submit to United, upon request, an attestation indicating the number of Active Covered Persons from all populations serviced by Provide as of the date of the request. For the purposes of this provision, "Active Covered Person" means a covered Person who is seen by the same primary care physician, or by a physician assistant or advanced nurse practitioner under the supervision of the primary care physician, at least three (3) times within a calendar year.

SECTION 4 UNITED REQUIREMENTS

4.1 United shall not prohibit or otherwise restrict Provider, when acting within the lawful scope of practice, from advising or advocating on behalf of a Covered Person for

UnitedHealthcare/AmeriChoice Agreement UHC-AMC/UHN PROVIDER.MEDICAID.REGAPX.02.09.WI the following: (i) the Covered Person's health status, medical care, or treatment or nontreatment options, including any alternative treatments that might be self-administered and any treatment or non-treatment options that may not reflect United's position or may not be covered by the Covered Person's benefit plan; (ii) any information the Covered Person needs in order to decide among all relevant treatment options; (iii) the risks, benefits, and consequences of treatment or nontreatment; or (iv) the Covered Person's right to participate in decisions regarding his or her health care, including the right to refuse treatment and to express preferences about future treatment decisions. United also shall not prohibit a Provider from advocating on behalf of a Covered Person in any grievance system, utilization review process, or individual authorization process to obtain necessary health care services

4.2 United shall not distriminate with respect to participation, reimbursement, or indocumentation of a Provider care is acting within the scope of Provider's license or confification under applicable states are, solely on the basis of such license or confication. This provision shall be be onstrued to prohibit United from including providers to the extent necessary to meet the needs of Covered Persons or from establishing any neasure designed to maintrar quality and control cost consistent with these response ities.

4.3 United shall not discriminate against Provider for serving high-risk Covered Persons or if Provider operalizes in conditions requiring cossay treatments.

4.4 United shall pay brovider persuant to the State bredie Contract, applicable State law and regulation, and 42 \times R 447.46, 4. CFic 47.45(d)(2), 42 CFR 447.45(d)(3), 42 CFR 447.45(a)(5) and 42 SFR 447.45(d)(b), an applicable and as may be amended from time to time. If a third party liability exists, payment of claims shall be determined in accordance with federal and/or frate third party locality have anothe terms of the State Medicaid Contract. Unless united otherwise requests assistance from Provider, United will be responsible for mird party collections in accordance with the terms of the State Medicaid Contract.

4.5 To the extent applicable under the State Medicain Contract and in the case of newborns, United shall be responsible for any payment of ed to Provider for services rendered prior to the newborn's enrollment with United

4.6 United shall not be responsible for any payments ower to Provider for services rendered prior to a Covered Person's enrollment with United, even if the services fell within the established period of retroactive eligibility; provided, however, if such person is deemed a Covered Person for such retroactive period, United shall be responsible for such payments.

4.7 If United delegates selection of providers to Provider, United retains the right to approve, suspend, or terminate any provider selected by Provider.

SECTION 5

OTHER REQUIREMENTS

2.1

5.1 To the extent applicable and required by law or the terms of the State Medicaid Contract, any notice of termination by United to Provider shall be furnished to the State or its designated government agencies.

5.2 In addition to its termination rights under the Agreement, United shall have the right to revoke any functions or activities delegated to Provider under the Agreement or impose other sanctions pursuant to the State Medicaid Contract if, in United's reasonable judgment, Provider's performance under the Agreement is inadequate.

5.3 As required under an State Medicaid Contract, United shall perform ongoing monitoring of Provider and hall subject Provider to formal review at least once a year, constant with the requirements of State and federal law and the State Medicaid Contract. As a result of such monitoring activities, United shall identify to Provider any dificiencies or areas for improvement manuated under the State Medicaid Contract and Provide shall the appropriate corrective action.

5.4 All task operformed under the Agreea ent must be performed in accordance with the requirements of the State Medicaid Contract as set forth in this Appendix, applicable provider manual s), and projectly, policies and procedure, that United has provided or delivered to Provider. The applicable provisions of the State Medicaid Contract are incorporated into the agreement by reference. Noting in the Agreement relieves United of its responsibility under the State Medicaid Contract. If any provision of the Agreement is in conflict with provision of the State Medicaid Contract, the terms of the State Medicaid Contract shall contract and the terms of the Agreement in conflict with those of the State Medicaid Contract whole considered waived.

UnitedHealthcare/AmeriChoice Agreement UHC-AMC/UHN PROVIDER.MEDICAID.REGAPX.02.09.WI Confidential and Proprietary

Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: ☐ Open ⊠ Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Other - ARISE HEALTH INSURANCE SOVEREIGN IMMUNITY
	Agenda Header (choose one): New Business/Request
	Agenda item title (see instructions):
	Arise Health Plan Insurance Contra
	Action requested (choose one)
	Information on the second sec
	Action - please describe:
	Approval is a read to option into a group part
	Approval is the enter into agreement.
3.	Justification
	Why BC action is required (reginstructions):
	The language in sections 4.8, 5.5, 1.71, 5.7.2, 5.3 constitute or may conditute waivers of sovereign immunity and must therefore be approved by the OLC, in accord with the Tribel Sovereign Immunity Law.
4.	Supporting Materials Instructions
	□ Report □ Resolution □ Contract heck the box below if signature required)
	□ Other - please list (Note: multi-media presentations fue to Tribal Plerk 2 days prior to meeting)
	1. Provider Agreement Legal Review 3.
	2. 4.
	Business Committee signature required
5.	Submission Authorization
	Authorized sponsor (choose one): Dr. Vir, Division Director/Medical
	Requestor (if different from above): Mari Kriescher, Behavioral Health Manager
	Name, Title / Dept. or Tribal Member
	Additional signature (as needed): Name, Title / Dept.
	Additional signature (as needed):

Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org



The above provisions were at one time acceptable; because they did not constitute an "explicit" waiver of Sovereign Immunity (i.e. the Tribe doe not explicitly consent to the jurisdiction of any court). However, case law has now held that certain non-extra it provisions amount to waivers of sovereign immunity, because they can be energed and enforced in court. Even if the binding arbitration clause in the agreement does not state that the orbitration decision can be enforced in court, a waiver likely still exists as it is binding arbitration conducted in accord with Wisconsin law (with Brown County Circuit Court involvement). Additionally, complaints and grievances are agreed to be determined in accord with Wisconsin a diministrative Code and equitable relief is (injunction) is explicitly consented to in order to for brown prevention. Anytime a provision in an Agreement contains language consenting to jurisdiction, or consenting to have a certain court hear a dispute, a waiver of sovereign immunity exists.

The Agreement involves payments owed to the Tribe for Behavioral Health services coverage. There are no payments from the Tribe to Arise Health Care. Potential risk and liability to the Tribe, approved, is very minimal. Such provisions are standard provisions generally found in third party payer agreements.

ONEIDA LAW OFFICE

CONFIDENTIAL: ATTORNEY/CLIENT WORK PRODUCT

TO: Maria J. Doxtator-Alfaro Oneida Health Center Use this number on future correspondence:

2015-0018

Purchasing Department Use Contract Approved

Contract Not Approved

FROM: Kelly M. McAndrews, Staff Attorney

DATE: January 22, 2015

RE: Arise Health Plan-Provider Agreement

If you have any quest of or comments regarding this review, please call 869-4327.

The attaction greement, contrast, policy and/or guaranty has been reviewed by the Oneida Law Office for legal content only. Clease for the following:

This contract may be impacted by Revultion #BC-09-24-2014-A entitled Continuing Resolution for FY 2015, This contract should be reviewed for compliance with this resolution prior to signing.

- Requires Business for mittee approval pror to execution, pursuant to possible limited waivers of sovereit componity.
 - While the Mode QHP Advendum is attached to the appement, the main document addre ses issues not ouched upon in the Model QHP. For example:
 - Complaint, regrievance procedure is referenced in a cord with Wis. Admin, Code 18, 3.
 - Provider acknowledges WPSLP is required to complete with Vis. Stat.
 609.17 (reports of circiplinary action). Sec. 4.8.
 - Sec. 5.5 agrees that WPLAP shall be initial to obtain injenctive relief or other forms of equitable adjef to prevent breach (of Configuratial Business Information). Relief includes costs and attorney fees, minimally.
 - Sec. 5.7.1 agrees that WPSHP of final dispusition of any disputes between Provider and Members shall be blocking on Provider.
 - Sec. 5.7.2 contains a binding arbitration section. Although the QHP Addendum indicates "IHS shall not be required to sublicit any disputes between the parties to binding arbitration", it is not clear this provision applies to Oneida. The Agreement allows bin ling arbitration in accord with Wis. Stat. Ch. 788, with ability of Brown County Circuit Court to approve the arbitrator. Sec. 5.13 also indicates Brown County WI is the sole venue for mediation/ arbitration and proceedings in court.

Notes:

Sec. 3.2.1 indicates Providers are subject to the Provider Manual, which appears to change periodically. The 2015 Provider Manual was reviewed.

Sec. 3.2.6 indicates subcontracts with licensed persons require components listed in this section. Your division is responsible for implementing this requirement and compliance. (There are two page 6s in the agreement sent to the Law Office).

HEALTH PLAN

PROVIDER ACKEEMENT

BET VEEN

WAS HEALTH PLAN, INC. dba Arix, Health Plan

ONEIDA COMPREHENVIVE PEALTH DI VISION

TABLE OF CONTENTS

1.	Recitals
2;	Definitions
3.	PROVIDER's Rights and Duties
4.	WPSHEs Rights and Duties 8 4.1 Lembers 8 4.2 Logram 8 4.3 Program Information 9 4.4 Marketing 9 4.5 Claims Submit ion an Payment 9 4.6 WPSHP's Right to zon fact 9 4.7 Ingurance 9
	4.7 Insurance
5.	Mutual Covenants95.1Independent Contractors105.2Non-exclusive105.3Provider/Patient Relationships105.4Records105.5Use of Confidential Business Information105.6Indemnification105.7Dispute Resolution115.8Term and Termination115.9Entirety125.10Severability125.11Assignment125.12Waiver125.13Governing Law and Venue135.14Notices135.15Headings and Recitals135.16Continuity of Care135.17Reports of Disciplinary Action13

<u>Exhibits</u>

A: PROVIDER and Credentialed Provider InformationB: Payment

3

AGREEMENT ("Agreement") entered into by and between Oneida Comprehensive Health Division (PROVIDER), and WPS Health Plan, Inc., dba Arise Health Plan (WPSHP), effective, 2015.

- 1. **RECITALS.**
 - **1.1** WHEREAS, WPSHP is a Wisconsin corporation licensed under Wisconsin Statutes Chapter 611 to write health insurance and to arrange for the delivery of health care Covered Services to its Members on a prepaid basis; and
 - 1.2 WHEREAS, PROVIDER is a legal entity which is lawfully established pursuant to the laws of its state of domicile, which has the capacity to provide and/or arrange for the provision of Covered Services and which maintains all licenses, permits and approvals required by federal, state and local laws and egulations to provide such Covered Services, and which is willing to receive as paymer in full for such Covered Services, the compensation set forth in this Agreement; and
 - 1.3 WHEREAS, PROVIDER, is young to assist WPSHP by offering and providing to Members and herear care services and factories; and
 - 1.4 WHERE, The indersigned person an authorized to enter into this Agreement on behalf of the partie

NOW THEREFORE, it is agreed as for lows:

2. Definitions.

- 2.1 Clean Claim means a requestion proment for Covered Services which is accurate, complete and in the manner and industry-standard format prescribed by VetoHP and as to which there is no substantial issue regarding WPS IP's responsibility for payment including but not limited to subrogation or coordination of beneficiessues. Clean Claims must be submitted using the current UB-04 and CMS 1500 forms (or any specessor forms) for paper chims and the current HIPAA standard professional or institutional claim formats for electronic claims, as applicable.
- 2.2 Complaint means any dissatisfaction about WNSHPARIts Participating Providers, expressed by a Member or a Member's authorized representative, to WPSHP.
- 2.3 Confidential Business Information means any confidential and/or proprietary information relating to the disclosing party including, with et limitation, Member lists, customer lists or requirements, sales methods, processes, Program, accidentative systems or software programs, health care service benchmarks or protocols, market research, pricing information, and compensation arrangements.
- 2.4 Covered Services are goods and services that are identified as benefits in a Health Plan's certificate of coverage. This Agreement provides for payment for medically-necessary, Covered Services provided to eligible Members which are billed within sixty (60) days from the date of service for primary claims and sixty (60) days from receipt of primary explanation of benefits (EOB) for secondary claims and comply with WPSHP's Program including any required authorizations.

- Credentialed Provider means any person or entity credentialed by WPSHP to provide Covered 2.5 Services to Members under the terms of this Agreement.
- Grievance means any dissatisfaction with the administration, claims practices or provision of 2.6 services by WPSHP that is expressed in writing to WPSHP by, or on behalf of, a Member.
- Health Plan means a contract or governmental health benefit program, other than in connection 2.7with liability or life insurance policy, which provides payment for health care services and related items.
- Medical Necessity meanservices, treatment, supplies, or facilities that WPSHP determines to 2.8
 - (a) Consistent with and appropriate for the diagnosis or treatment of the Member's illness or in, rry; and
 - (b) Commonly and customerary recognized and generally accepted by the medical profession in the United States as apprepriate and standard care for the condition being evaluated or treated a d
 - a tan ated by clinical do umentation (c)
 - (d) The hast appropriate and cost effective level of care that can safely be provided to the Membr; 2 4
 - (e) Proven be useful likely to be uccessful, you additional information, or to improve clinical outcone and
 - converience or preference of the Member, his or her family, or any (f) Not primary the provider.
- is a b peficiary of a V alth lan Member means an individual w 2.9
- Participating Provider means a physician acility, skilled using acility, home health agency 2.10 or any other duly licensed instruction health professional uncer contract with WPSHP to provide Covered Services to Member
- Primary (Specialty) Care Provider means person duly credentialed by WPSHP contracted or 2.11 otherwise as a primary care (specialty) Provid
- Plan Sponsor means either (a) WPSHP or (b) a poson who has acceed to provide a Health Plan 2.12 to a Member and who has contracted with WPSHP to grange for Covred Services.
- Program means the rules, standards, policies, and proceduit or other internal and external 2.13 requirements for administration of Health Plans, including, but the inited to: accreditation, credentialing and re-credentialing of providers; utilization management; pre-certification requirements; recordkeeping; claims submission and payment; and communications with Members.
- Provider means any physician or other duly licensed health professional or entity that: (1) is 2.14 owned or employed by, or is contractually affiliated with, as applicable, PROVIDER to provide health care services; (2) satisfies each requirement imposed by the Program for Participating Providers, and has been approved as a Participating Provider by WPSHP; (3) agrees to provide Covered Services to Members only within the scope of practice applicable to such Provider; and

4

Page 239 of 340



.

5

(4) is identified as a Provider on Exhibit A, which is attached and made part of, and may be updated from time to time by the parties in writing.

3, **PROVIDER RIGHTS AND DUTIES.**

In addition to any other rights and duties of PROVIDER set forth in this Agreement, PROVIDER and WPSHP agree as follows:

- 3.1 Covered Services. PRCNDER, through its Providers, agrees to provide Members with those Covered Services identified in the certificate of coverage. Providers shall render such Covered Services promptly in a manner consistent with community standards and at all times in the Memor's best interest Prevale 's Covered Services to Members shall be of the same nature and quality is care provided to pur-Memory.
 - **1 Petertry Care Model.** PROVIDER and Providers acknowledge that WPSHP and other Heach Plans may operate under a primary care model requiring Members to obtain referrals/referred authorizations, from their Primary Care Provider or in-network Specially that Provider to obtain Concered Services from an out-of—network specialist or tertiary care provider/facility. PROVIDER at bits Providers shall not make referrals for Covered perforces without prior written concern of Health Plan.
 - **3.1.2** Access. PROVIDER agrees that it provides and will maintain twenty-four (24) hour telephone access to Member and that PROVIDER as procedures in place to respond to Members' calls and equate after normal busines, hours.

3.2 Program Participation.

- **3.2.1** Compliance with Legal are Program dequirements. PROVIDER and Providers agree to participate, cooperate, and comply with Program, in Judi'g quality improvement activities as outlined in the Provider vianual found at www.WeCareforWisconsin.com. PROVIDER agrees to allow WPSH to use Provider performance data, such as but not limited to WCHQ, WHIO, etc., for analysis a concer comparison. Such data may be used to develop and evaluate quality improvement activities. Results may be shared via public reporting methods and other methods, including but not mainted to, web-based tools. PROVIDER shall require each Provider to comply with all applicable federal, state, and local laws and regulations; and to maintain all licenses, perpose and approvals required by federal, state or local law for Provider to offer and provide Covered Services under this Agreement. PROVIDER and Providers shall not discriminate in the provision of Covered Services under this Agreement on the basis of amount, method, or source of payment, sex, age, race, color, religion, national origin, health status, handicap, or any other basis forbidden by law and regardless of benefit limitation.
- **3.2.2** Medicare and Medicaid. PROVIDER and each of its employed and contracted Credentialed Providers shall at all times be certified as providers for Medicare and Medicaid and accept Medicare assignment.

- 6
- Credentialing. In accordance with Section 609.32 Wis. Stats, PROVIDER understands 3.2.3 and acknowledges that Providers must complete a WPSHP Credentialing Application, comply with all credentialing requirements and regulations, and be credentialed by WPSHP as outlined in the Program, as a condition precedent to this Agreement becoming effective and remaining in effect. Credentialing shall include, but is not limited to the following: verification of Credentialed Provider's license or certificate, including the history of any suspensions or revocations, and the history of any liability claims made against Credentialed Provider. Recredentialing, which takes place every 36 months, shall include, but is not limited to the following: updating the previous review criteria, assessing Credentialed Provider's performance on the basis of such criteria as Member clinical outcomes, number of complaints, and malpractice actions. No Credentialed Project will be required to provide services outside the scope of Credentialed *Lor* der's license or certificate.
 - PROVIDER of any on PROVIDER agrees to provide the information in Exhibit A, or subsequent notices as a quered by Section 3.5, for all its Providers who are required by Program to apply the credentials as WPSHP providers. PROVIDER shall inform WESTP whether each applicant is willing to accept Members as new patients. WPSHP may approve, limit, supend, or compate the right of any person to provide Covered Sovices of the this Agreement when, hoits sole discretion, such person has failed to mply with the terms of the Agree ents cluding, but not limited to compliance with the rogran.
- Hospital rivinges. POVIDER agree that any rits Credentialed Provider's who use 3.2.5 hospital facilities, if a y, shall maintain star, princeges at one or more WPSHP Participating provider hospitals or provide an administing plan and shall provide Covered clusively at Participating provider facilities if that can be Services to Menters accomplished in the bost in rests of the Members.
- Subcontracts for Covered Services, PROVIDER agrees that each subcontract with 3.2.6 licensed persons or entities fathe provision of Covered Strvicento Members shall
 - (a)
 - specify WPSHP as a third party to the Agreement; require such subcontractors a nform to be a tams of this Agreement (b) applicable to PROVIDER; and
 - allow WPSHP the right to pre-approve or a sapprove the right of each individual (c) licensed person or entity to provide Overed Series Members.

PROVIDER shall maintain written copies of all uch subgentracts and shall provide WPSHP and/or regulatory agencies access to and obline pies of such subcontracts on five (5) days written notice. PROVIDER agrees to disclose all terms and conditions of any payment arrangement between a provider and PROVIDER or its subcontractors that constitutes a physician incentive plan as defined by CMS rules and regulations. Such disclosure shall be in the form of a certification which provides the information required by such rules and regulations, including (a) whether such arrangements include referral Covered Services, (b) the type of incentive plan and the percentage of any withhold or bonus, (c) the size of the panels and the distribution among various types of coverage, or (d) the amount and type of stop-loss coverage provided or required by law. PROVIDER further guarantees to cooperate in assuring that any stop-loss coverage required by law is

- Credentialing. In accordance with Section 609.32 Wis. Stats, PROVIDER understands 3.2.3 and acknowledges that Providers must complete a WPSHP Credentialing Application, comply with all credentialing requirements and regulations, and be credentialed by WPSHP as outlined in the Program, as a condition precedent to this Agreement becoming effective and remaining in effect. Credentialing shall include, but is not limited to the following: verification of Credentialed Provider's license or certificate, including the history of any suspensions or revocations, and the history of any liability claims made against Credentialed Provider. Recredentialing, which takes place every 36 months, shall include, but is not limited to the following: updating the previous review criteria, assessing Credentialed Provider's performance on the basis of such criteria as Member clinical outcomes, number of complaints, and malpractice actions. No Credentialed Project will be required to provide services outside the scope of Credentialed *P* of der's license or certificate.
 - PROVIDER Information in Exhibit A, or subsequent notices as experied by Section 3.5, for all its Providers who are required by Program to apply the credentials as WPSHP providers. PROVIDER shall inform P whether easy apply ant is willing to accept Members as new patients. WPSHP may approve, limit, supend, or criminate the right of any person to provide Covered Sovices and this Agreement when, hoits sole discretion, such person has failed to mply with the terms of the Agreements cluding, but not limited to compliance with the logr n.
- Hospital rivinges. NOVIDER agree that any rits Credentialed Provider's who use 3.2.5 hospital facilities, if any, shall maintain stree printeges at one or more WPSHP Participating provider hospitals or provide an administ n plan and shall provide Covered elusively at Participating provider facilities if that can be Services to Methods accomplished in the bot in rests of the Members.
- Subcontracts for Covered Services. PROVIDER agrees that each subcontract with 3,2.6 licensed persons or entities for the provision of Covered Strvicento Members shall
 - (a)
 - specify WPSHP as a third party to the Agreement; require such subcontractors and form to get the this Agreement (b) applicable to PROVIDER; and
 - allow WPSHP the right to pre-app ve or sapprove the right of each individual (c) licensed person or entity to provide vered Service to Members.

PROVIDER shall maintain written copies of all uch subcontracts and shall provide WPSHP and/or regulatory agencies access to and obtain pies of such subcontracts on five (5) days written notice. PROVIDER agrees to disclose all terms and conditions of any payment arrangement between a provider and PROVIDER or its subcontractors that constitutes a physician incentive plan as defined by CMS rules and regulations. Such disclosure shall be in the form of a certification which provides the information required by such rules and regulations, including (a) whether such arrangements include referral Covered Services, (b) the type of incentive plan and the percentage of any withhold or bonus, (c) the size of the panels and the distribution among various types of coverage, or (d) the amount and type of stop-loss coverage provided or required by law. PROVIDER further guarantees to cooperate in assuring that any stop-loss coverage required by law is

7

made available. Unless PROVIDER purchases such stop loss coverage, PROVIDER agrees that WPSHP may purchase such coverage on PROVIDER's behalf and deduct the cost of such coverage from PROVIDER's compensation under this Agreement.

3.3 Payment.

- **3.3.1** Payment Schedules. PROVIDER agrees to accept the amounts specified in the payment schedule(s) in Exhibit B as payment in full for Covered Services provided to Members pursuant to this Agreement.
- **3.3.2** Conditions for Payment. As a condition precedent to Plan Sponsor's liability for payment, PROVDER agrees to (1) comply with all WPSHP's pre-authorization requirements and (2) submit Clean Claim for Covered Services rendered and all revisions within sixty (60) days after the date of service for primary claims and sixty (60) days from recept a primary explanation of benefits (EOB) for secondary claims or, with respect to the specific Coursel Service(s) involved, within sixty (60) days after WPSHP requests PROVIE 2R to submit a claim for payment.
- **3.3** Liability for Payment, PROVIDER orees that each Plan Sponsor, whether WPSHP or an other suchsor, is responsible to PROVIDER for any amounts owed to PROVIDER after deauty ing applicable deputible, coparaments, and amounts due from coordination of banefit. ("CCS") provisions of the Plan Sponsor's Health Plans.
- **3.3.4** Payment by Members and COB PROVIECR shall have the sole right and responsibility to collect any applicable coinsurance is opayments or deductibles from Members as well as any amounts due, if any from COB provisions of Member's Health Plans. If deductives an ecopayments required order is there in Plan are determined as a percentage of fees or marger, the amount of the deductive or copayment shall be such percentage times the payment(s) intermined in Exactly F. PROVIDER shall notify WPSHP of any amounts due from COB when submitting the claim or, if unknown at submission, within ten (10) days of discretering the excitence of a third-party health benefit plan which covers Coursed Securces provided to Nember's. At the request of either party, WPSHP and PROVIDER will reconcile amounts due PROVIDER from WPSHP as provided in Section 3...3, with the acounts received or receivable from copayments, deductibles, or COB.
- **3.3.5** Right to Collect from Other Third Parties. Except for COP, Plan Sponsors shall have the right to recover payments for Members for Concered Services, by subrogation or otherwise, from any liable third party. PROVNER shall give Plan Sponsors any requested assistance in exercising such rights.
- **3.3.6** Overpayments. WPSHP may recover overpayments made to PROVIDER by giving written notice including the following information: Member's name and identification number, date of service, service provided, the amount paid, the amount that should have been paid, and a brief explanation of the basis for requesting the refund. PROVIDER shall repay overpayments within sixty (60) days of such notice. Notification of overpayments must be given within eighteen (18) months from the date of payment. WPSHP may offset payments due PROVIDER with other amounts due Plan Sponsor under this Agreement.

- Underpayments. If PROVIDER believes a claim has not been paid correctly, 3.3.7 PROVIDER shall give written notice to WPSHP including the following information: Member's name and identification number, date of service, service provided, the amount charged, the amount paid, the amount PROVIDER believes should be paid, and a brief explanation of the basis for contesting the amount paid. If the claim was not paid correctly, WPSHP shall remit correct payment to PROVIDER within sixty (60) days of written notice from PROVIDER. Notification of incorrect payments must be given within eighteen (18) months from the date the claim was initially paid or denied by WPSHP.
- Notice of Cor of ons Affecting Performance. PROVIDER shall notify WPSHP in 3.4 writing at 1 st firty (30) days in advance of any matter that will materially affect the availability of for services to Members under this Agreement.
- Ten (10) Day Notice PROVDER will notify WPSHP within ten (10) days after 3.4.1 DER becor s aver of Member Complaints, Grievances, disputes, or malaractice claims of other solors or events affecting the licenses, accreditation, guernment and private certice tions, permits, staff privileges, and/or professional c pacit o PROVIDER or a v of italicens d health care Providers. PROVIDER agrees to y ompay regiond to Complaints and G vances filed with WPSHP to facilitate resolution as required in s. Ins. 18.03, W.s. Acan. Code. PROVIDER will also notify WPSHP achine on (10) days of any care ellation or reduction in PROVIDER's insurance policies as required by Section 3.5.5 of this Agreement
- Ninety (90) Date of the PR VIDER shall stify WPSPP in writing not less than 3.4.2 ninety (90) days in advance for changes in:
 - the identity of PLOVIDE 's Credentialed Provides; or (a)
 - the willingness of an Provider or Fredentialed vovider to accept Members as (b) new patients; or
 - the identity or ownership of POVIDER; or (c)
 - the transfer of substantially a PROVIDE cs ssets involved in providing the Covered Services up to this Agreement; or (d)
 - entry into a joint operating agreement wereby PROVIDER and other entity (e) share management and income.

Continuing Rights and Obligations. 3,5

Termination by PROVIDER. In the event PROVIDER is the party terminating this 3.5.1 Agreement, the rights and obligations of the parties shall continue until the renewal date of Health Plans in force during the term of this Agreement or until WPSHP releases PROVIDER from its obligations with respect to such a Health Plan, provided that PROVIDER gives notice of termination in accordance with Section 5.8 of this Agreement. Except, however, that the rights and obligations of the parties shall continue with respect to any Member who, at the effective date of termination, is an inpatient of a licensed health facility or until such Member has been discharged and another Participating Provider has agreed to accept responsibility for the care of such Member.

- **3.5.2 WPSHP Insolvency.** In the event of any insolvency or other cessation of operations of WPSHP which might allow PROVIDER to terminate any part of this Agreement, PROVIDER shall continue to provide Covered Services to Members while they are inpatients in a licensed health care facility; and, in addition, until the earlier of:
 - (a) the end of the contract period for which a premium has been paid on behalf of a Member, or
 - (b) acceptance of a referral of the Member to the care of a licensed health care provider.
- 3.5.3 No Recourse Accinst Members for Payment. PROVIDER hereby agrees that in no event shall PLOT IDER or Providers seek, accept, or have any recourse for payment for Covered Services provided under this Agreement against any Member or any person other than Ploa Sconer acting on Member's behalf. This provision does not prohibit PROVIDER from collegeing topayments, deductibles or amounts due from COB as specified in each Member's Health Plan. PROVIDER or Provider may also collect fees for services not covered by the Member's Health Plan and which Provider delivers on a "fee for-service" basis with Member's prior, written acknowledgment and consent. ProVIDER shall have sole reportibility for collecting such payments.
- **3.5.4** Mathematical Records. PROVIDER agree to maintain all records related to Covered Services belowered under this Agreement of payments made for such Covered Services for the guater of any applicable statute of limitations plus three (3) years.
- **3.5.5 Insurance. I**ROVIDER spaces that Provider and all its Credentialed Providers shall participate, if eligible, inde Whotonsin Patients compensation Fund. PROVIDER, at its sole expense, shall obtain and maintain, or its Providers and their employees and agents shall provide and maintain, such policies of underait and professional liability (malpractice) insurance a shall be necessary to insure PEDVIDER, its Providers and its employees and agents against any claimer claims for samages arising by reason of personal injuries or death occasioned directly or indirectly on connection with the performance or provision of any solvice or supply by PROVIDER or Provider. The amounts and extent of such insurance coverage required hereunder shall be consistent with applicable law and the Program.

At WPSHP's request, PROVIDER shall provide WPSaP with copies of all insurance policies required by this Agreement. If PROVIDER uses not p rchase and maintain such coverage, WPSHP may purchase insurance coverage are PROVIDER agrees to reimburse WPSHP for the cost of same. PROVIDER and the sonotify WPSHP within ten (10) days of any cancellation or reduction in PROVIDER's insurance policies.

3.5.5 Survival. This Section 3.5 shall survive termination of this Agreement, regardless of the cause giving rise to termination and shall be construed to be for the benefit of Members. The provisions of Section 3.5.3 shall supersede any oral or written contrary agreement now existing or subsequently entered into between PROVIDER and Member or any person acting on a Member's behalf.

4. WPSHP'S RIGHTS AND DUTIES.

In addition to any other rights or duties of WPSHP set forth in this Agreement, WPSHP and PROVIDER agree as follows:

- 4.1 Members. Each Health Plan shall have the sole responsibility to determine who is a Member and whether a Member is eligible for any benefits under its policies.
- **4.2 Program**. WPSHP may establish, and may from time to time, revise Program for administration of Health Plans. WPSHP will provide at least sixty (60) days notice of any revisions which materially affect this Agreement.
- **4.3 Program Informatics**. WPSHP shall periodically make available to PROVIDER, in written or electronic format copies of Program requirements, standards, policies, procedures, and manuals. At PROVIDER's requise average of shall make available in written or electronic format copies of Health clan Summary Plant escription.
- 4.4 Marketing, VPSHP may us the nones, addresses, phone numbers and pictures of PROVIDER and ins Providers, identify PrOVIDER. Covered Services and applicable restrictions, and indicate the willightees of PROVIDER and each Provider to accept Members as patients in the normal course of bisiness.
- 4.5 Claims Submission and Perspect. Plan Sponset shall make payment on Clean Claim(s) for Medically Necessary povered Services in a manner consistent with Section 628.46 of the Wisconsin Statutes.
- 4.6 WPSHP's Right to Contract of Durne, the term of this 4 greenent, WPSHP may establish Health Plans and/or enter into contracts to provide Coverd Vervices to (1) beneficiaries of governmental Health Plans of (2) Members of third-party hore overmental Health Plans. Governmental Health Plans may include, but are not limited to, Medic re, Medicaid, and programs which replace or expand partor all of Medicare and Medicaid.
 - **4.6.1** Incorporation of Modifications. WISHP may emend this Agreement as it applies to such a Health Plan by giving PROVIDER writter notice of such amendments at least 60 (sixty) days in advance of the effective cate of the contract.
 - **4.6.2** Limitation on Modification of Payment Scheduler. ROVIDER may refuse to accept new payment schedule(s) applying to such plans by giving written notice of such refusal within thirty (30) days of WPSHP's notice of intent of amend the Agreement in accordance with Section 4.6. PROVIDER may not that its agreement to exclude Members of Health Plans sponsored by WPSHP.
- 4.7 Insurance. WPSHP shall purchase such general liability and worker's compensation insurance coverage as it deems necessary to protect WPSHP and PROVIDER against any claim for damages arising, directly or indirectly, from the negligent performance of duties of WPSHP or its employees. At PROVIDER's request, WPSHP shall give PROVIDER copies of all insurance policies covering the term of this Agreement.
- 4.8 Disciplinary Actions. Pursuant to Section 609.17 of the Wisconsin Statutes, WPSHP is required

to notify the Wisconsin medical examining board or appropriate affiliated credentialing board attached to the medical examining board of any disciplinary action taken against PROVIDER's Credentialed Providers. PROVIDER acknowledges and understands that WPSHP is required to comply with the requirements of Section 609.17 of the Wisconsin Statutes.

- 5. **MUTUAL COVENANTS.** In addition to any other duties or obligations of the parties set forth in this Agreement, the parties agree as follows:
 - 5.1 Independent Contractors. The parties are independent legal entities who are contracting with each other solely for the purpose of effecting the provisions of this Agreement. None of the provisions of this Agreement are intended to create nor shall be deemed or construed to create any other relationship between the parties, nor shall any of their respective employees be construed or deemed to or agents, employees or representatives of the other.
 - 5.2 Non-explusive. WPSLP and ExOVIDER each expressly reserve the right to enter into other or similar grangements with other process.
 - 5.3 **Frovides at ent Relations ips.** Notwithstanding any other term or provision of this Agreement, expressed or implied nothing in this Agreement shall be deemed to change or alter any relationship (onept as to payment of Novered Services) which exists or which may come to exist between RGVIDER or its Providers and their patient(s), who are or become Members. Neither WP HP is r aprone else shall have the right to otherfore with the care or treatment given or prescribed by providered cary compensation for Covined Services from Members or Plan Sponsors except as provided in this Agreement. Nothing in the Agreement shall be construed as an arrangement or an agreement for elicitation of protents of ROVIDER by WPSHP. Provider may freely communicate with others bout treatment options regardless of benefit coverage limit dons.
 - 5.4 Records. The parties agree to bainted those records and recordkeeping systems which are expressly or implicitly required by it's Agreement are customarly used to prudently manage each party's responsibilities, or are required by fineral, state or locar on or regulations. Except as required by law, the parties shall keep continential all Member information and records unless a release is requested by the Member of WPSHP in accordance with applicable legal requirements. Upon request, and subject to appliance state and federal laws governing the confidentiality of patient health records, PROVIDER thall provide access to or copies of Members' health records for the purpose of Program, exims particular access to or copies of Members' health records to WPSHP, and to state and federal laws governing the confidentiality of patient health records PROVIDER shall provide access to or copies of Members' health records to WPSHP, and to state and federal laws governing the confidentiality of care or investigating grievances or complaints. PROVIDER shall provide in assessing quality of care or investigating grievances or complaints. PROVIDER shall provide Member's health records at no charge. Copies of records to be used for any other reason shall be reimbursed at rates no greater than the fees specified in Section 146.83 of the Wisconsin Statutes, or its successor.

5.5 Use of Confidential Business Information for Marketing or Other Purposes.

5.5.1 Non-Disclosure. Each party acknowledges that, in the course of negotiating and performing under the terms of this Agreement, they have received from WPSHP

Confidential Business Information developed at great expense by WPSHP and essential for carrying out WPSHP's business in a highly competitive market. PROVIDER and Providers agree and acknowledge that unauthorized use, disclosure or publication of Confidential Business Information provided by WPSHP could harm WPSHPS' current or potential business interests.

Accordingly, during the term of this Agreement and for a period of two years thereafter, PROVIDER and Providers shall not, without prior written consent of WPSHP, except for use in the course of performing pursuant to this Agreement or as required by applicable law, publish, disclose or authorize any other party to use, publish or disclose any Confidential Business Information provided by WPSHP within the geographical area in which such us publication or disclosure could harm WPSHP's existing or potential business in reas. PROVIDER and Providers will inform its representatives and employee having accept to the Confidential Business Information of, and will require such individe a compary with the terms and conditions herein set forth.

PP of JIDER and Providers acknowledge that the covenants made by and duties imposed hereby are fair, reasonable and an imally necessary to protect WPSHP's legitimate owness increases, and uset such ovenants and duties do not and will not place an undue birden after a PROVIDER of Providers in the event this Agreement is terminated and coverants contained herein the strictly energed. PROVIDER and Providers further acknowledge that any breach of this provision will cause substantial and irreparable harm too WPSHP the which money damages would be an inadequate remedy. Accordingly, WPSHP shall be entitled to obtain injunctive and other forms of equitable relief to prevent such breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including, where to breach and recover from PEC/IDER and/or Providers its costs (including this provision, the addition to any other rights or remedies available by law, in equity or by statute.

Notwithstanding the foregoing and irrespective of benefit coverage, PROVIDER and Providers shall not be prohibited from or penalized for ongoing in provider-patient communications related to treatment alternatives, coverage appeal decisions, reimbursement incentives or any other communications necessary to maintain the provider-patient relationship.

- 5.5.2 Approval of Marketing Use. Except as excressly parafited in Section 4.4, each party shall obtain the other party's written permission prior to using my patented, trademarked, trade-named, service-marked or copyrighted material or property belonging to the other party. The owner shall have the right to review an approve the appearance, content, format, and/or distribution of such use.
- 5.6 Indemnification. WPSHP and PROVIDER each agree to indemnify and hold harmless the other party, its directors, officers, agents and employees from any and all demands, claims, suits, liabilities, losses, damages, or expenses of any kind, including costs and attorneys' fees, which result solely from negligent or willful acts or omissions of the other party, its agents or employees, regarding the duties and obligations of the other party under this Agreement, including the duty to maintain the legal standard of care applicable to PROVIDER. This provision shall not operate to increase PROVIDER's malpractice liability to Members. In the event that PROVIDER's malpractice carrier asserts that this Section 5.6 operates as a contractual

13

assumption of additional malpractice liability that would invalidate or terminate said policy, then this provision shall be inoperative to the extent necessary to preserve PROVIDER's malpractice coverage.

5.7 Dispute Resolution.

(a)

- 5.7.1 Disputes between PROVIDER and Members. WPSHP's final disposition of any disputes between PROVIDER and Members shall be binding on PROVIDER.
- 5.7.2 Disputes Between the Parties. In the event that any dispute between the parties arises out of the Agreement, the parties shall meet and confer in good faith to resolve such dispute. If such corts do not resolve the dispute by the earlier of:
 - thaty (0) departs the first meeting, or
 - (b) sixty days from the date such meeting is first requested in writing,

then either party man, by providing written notice, require both parties to submit the anspite to binding a bitration. Upon such notice, the dispute shall be submitted to binding arbitration pursuant to a scontin Statutes, Chapter 788. The arbitrator(s) shall be require to issue written and in s of fact and conclusions of law in conjunction with deire decision. Punitive and exemplary demages shall not be awarded in binding arbitratio under this Section. Each part scall pay one-half (1/2) of the costs of mediation arbitration. This Section 5 is shall survive termination of this Agreement.

5.7.3 Choice of A pitrators. The arbitrator shall be chosen is follows:

- (a) By mutual agreement when in thirty (30) days of the aotice requiring arteration, or
- (b) by a Circuit Court in Broom County, Wiscomm, a provided in Wisconsing statutes, Section 788.04.
- **5.8** Term and Termination. The term of this Arcement shall begin on 5.2015 and continue until December 31, 2015 and shall be automatically renewer from year to year thereafter unless terminated by one of the parties as provided below.
 - 5.8.1 Non-renewal. Any party may terminate the Agreement regiving written notice of non-renewal to the other party at least one hundred tweet (120, days prior to the end of a contract term, in which event this Agreement should terminate at the end of the then current term.
 - **5.8.2** Default. Either party may terminate this Agreement for a material breach by giving the breaching party sixty (60) calendar days advance written notice of the termination. The breaching party may cure the breach during the sixty (60) calendar day period. If the breach is cured, the Agreement shall remain in effect. If the breach is not cured, the Agreement shall terminate at the end of the sixty (60) calendar day period.
 - **5.8.3** Termination by WPSHP. WPSHP may terminate this Agreement immediately upon written notice if:

- (a) The ownership of PROVIDER is transferred to another legal entity.
- (b) PROVIDER files a petition for relief under the U.S. Bankruptcy Code or a petition for appointment of receiver is filed by or against the PROVIDER.
- (c) PROVIDER fails to maintain or incurs suspension, revocation or loss of licenses, certifications, credentials, permits, Medicare qualifications or other qualifications referred to in this Agreement; or
- (d) PROVIDER discloses Confidential Business Information.
- **5.8.4** Termination by WPSHP of a Particular Provider. WPSHP shall also have the right to immediately remove a particular Provider from responsibility for, or have any involvement details or indirectly with, the services and operations related to this Agreement upon the occurrence of any of the following events without disrupting PROVIDER is other Providers' provision of Covered Services to Members or terminating this Agreement, its entirety if:
 - Such Provider in any way jeopardizes the safety of patients; or

(b)

- Such Provide fails to reactain or upon suspension, revocation, or loss of lightness, certification recent is, permits, Medicare qualifications or other qualifications referred to herein; or
 - ch Provider disclose Confidential Pusiness Information.
- 5.8.5 Termination of specialty Provider/Specialty Provider Group. In the case of a specialist/specialist group termination, WPSF will be responsible for the timely notification of such termination to affected Members the have incurred claims with the Specialty Provider/Specialty Provider group. Yor SEP shall notify affected Members prior to the effective date of the termination.
- 5.9 Entirety. This Agreement and documents expressly referred to herein represent the entire Agreement between the parties on the subject matter hereof and upertede. Il prior discussions, agreements, and understandings between them. No modification of this Agreement shall be effective unless in writing and signed by WPS and PROVIDER.
- 5.10 Severability. If any portion of this Agreement is included or unenforceable for any reason, the remaining portions of this Agreement shall not be severable and remain in full force and effect and the rights and obligations of the parties shall be construed a comforced as if this Agreement did not contain the provision held to be invalid.
- 5.11 Assignment. This Agreement shall be binding on and inut to the benefit of the parties hereto and their respective successors. WPSHP may assign this contract to any successor to all or any part of its business. PROVIDER shall not assign, delegate, or otherwise transfer any of PROVIDER's rights or obligations hereunder without the prior written consent of WPSHP.
- **5.12** Waiver. Either party's failure, at any time, to require performance by the other party of any provisions herein or either party's failure to provide notice to the other party of the other party's breach or violation of any provision shall not operate as a waiver (i) to require strict performance of same or like provisions, or any other provisions hereof at a later time; or (ii) of the other party's ongoing breach of any provision of this Agreement.

- **5.13** Governing Law and Venue. The parties consent to personal jurisdiction in the State of Wisconsin and agree that Brown County, Wisconsin, shall be the sole venue for mediation or arbitration proceedings or for proceedings in a court of general jurisdiction. This Agreement shall be exclusively construed and applied according to the laws of the State of Wisconsin and applicable federal law. Neither any arbitrator nor any court of law shall apply the law of any other state to any dispute among the parties or third party beneficiaries of this Agreement, except as to the amount of Workers Compensation Insurance or Physicians Malpractice Insurance required by the laws of the state in which such employment or practice was located.
- 5.14 Notices. Notices or any other communication in connection with this Agreement shall be in writing and shall be delivered in person or by registered mail, or certified mail, return receipt requested, addressed to



Oneida Compute Health Center Attn: Dora Coanfor Comprehensiv Health Operations Director 525 Airport Coad Oneida, WI 54100

or to such other address as one party may specify by written nonce to the ther party. Notices shall be deemed effective when received.

- 5.15 Headings and Recitals. The recitals in Section 1 are a part of this Agreement. The Section headings are for convenience only and shall not have any legal effect.
- **5.16** Continuity of Care. WPSHP and PROVIDER ag te to comply with the requirements of Section 609.24 of the Wisconsin Statutes with regard to the continuity of Care of Members subsequent to the termination of this Agreement by either PROVIDER of WPSHP WPSHP and PROVIDER agree that the terms of this Agreement, including reimburstment, shall continue in effect until the Member is no longer receiving care from PROVIDER in accordance with Section 609.24, Wis. Stats.
- **5.17** Reports of disciplinary action. In accordance with Section 609.17, Wis. Stats., WPSHP shall notify the medical examining board or appropriate affiliated credentialing board attached to the medical examining board of any disciplinary action taken against a Credentialed Provider who holds a license or certificate granted by the board or affiliated credentialing board.

16
17

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date set forth below.

WPS HEALTH PLAN, INC.

By:			
·	John Trochlell, FSA, MAAA		Date
	Vice President		
ONFI	DĂ COMPREHENSIVE HEALLH DAVION		
OTARI			
By:			
	Debra Danforth	Date	
	Comprehensive Health I reason Beatth Division Operations Director		
	the DI Defait	· · ·	
Printe	d Name: Debra Danforth		
By:			
Dy,	Ravinder Vir	Due	
	Comprehensive Health Division Health Division Medical Director		
Printe	ed Name: Ravinder Vir		
Feder	al Tax Identification Number: 396081138		
10401			

Exhibit A: PROVIDER and Credentialed Provider Information

Credentials **Provider Name** Internal Med Michael Flood MD Podiatry Paul Abler DPM Family Med Jay Kennard MD Family Med Ann M Wells FNP Nurse Practic Susan L Katuin RN APNP Pediatrics Gerald R Verstoppen MD Family Med Lori B Thiry NPC Internal Med Ravin MD mily Med St phanie A berhardy PA Fan Vy Med Pul H Sumnicht MD Dichetic Educator Time A Moura APNP Betty Schwartes D CDE Registere Dia etic Priysi an ssistant Carol M Finuca A C Regis yred Diabetic Boni J Colwitz N C -Register Diabetic Susan Higgs RD Registered Diabrat Alyssa J Hudak RD CD Dentist Lisa Frechette DDs B/GYN Sara Anderson MD OL GYN Karen Tammela MD Cert Nurse Midwife Antoinette Westphal RN CNM OB/ IN Robert D Moyer Jr MD Oometry David G VanDerLoop OD Otometry Steven P Drake OD Optonet Roxann Doyle Keszo OD Optome v Lisa Slaby OD OB/GYN Amy Van Gheem MD Certi Nurse Midwin Libby A Woodard CNM **Physical Therapy** Sidney J White PT **Physical Therapy Assit** Constance S Danforth PTA MS, LPC, CSAC, ICS M. Kriescher M. D. J. Rodriguez M, D, D. Dzubinski M. D. C. Shekar M. D. M. ONeill M. D. V. Patil M. D. H.Wynn PH.D Mike Sayers

Page 255 of 340

L. Metoxen Tina Baeten T. Adkins R. Loberger K. Sayers **Benjamin Cheney** L. Shaw T. Nehring Mary Beth King M. Gore Vale – пеrnder M rtha Brite R. Juhtala Tim compert Dale Rasmuss S. Lalonde M. Agneessen

MSW, SAC-IT MSW, ICS, LCSW, CSAC MSW, LCSW, SAC, CSOT, BCD MSE, LCSW, SAC MSW, LCSW, LMFT MSW, LCSW-SAS MSW, LCSW MSE, LPC, CSAC MSW, LCSW, CSAC MSW, LCSW, MSW, LCSW **AS-MFT** MSW LCSW, M.F. LPC SAC BSW CSA

CSA

MA, SAC

MSN, RN

RN, MSE, LPC

SAC, CS-IT

BS,

Joanne Torres Lois Skrivanie **Dave Paluch**

Oneida Community Health Center 525 Airport Road Oneida, WI 54155

AJRCC (Anna John Resident Centered Care Communi 2901 South Overland Road Oneida, WI 54155

Ka Ni' Kuhli Yo Family Center 2640 W Point Rd Green Bay, WI 54104

EXHIBIT B: PAYMENT

COMPENSATION

Payments to be made to PROVIDER for Covered Services rendered to Members in accordance with this Agreement are as follows:

1. MEDICAL Covered Services shall be paid at the lesser of PROVIDER's billed charges or;

One hundred percent (00%) of the amount attributed to the item or service in the Wisconsin Participating Non-Facility Part B Physician Fee Schedule (Part B Fee Schedule) tutilizing the Posor ce-based Relative Value Scale (RBRVS) identified by the Onter for Medicare and Manuaid Services (CMS) for Wisconsin, the Clinical Diagnostic Laboratory see Schedule (NLAB) for Wisconsin, the Wisconsin Durable Redical actingment, Prometics Orthotics and Supplies (DME POS), Enteral & Posneral PEN) Fee Schedul, and Average cales Price (ASP) calculated by the Center for Medicaid Services (CMS) to Part B drugs and biologics (HCPCS J0120-J1/99) another for the curvet year on Jamery 1.

- 2. BEHAVIORAL HEAL A Concred Services shall be pair at the lesser of Provider's billed charges or:
 - a. MD's (Psychatrist) shall be reimbursed one hundred openent (100%) of the amount attributed to the item or nevice in the Wiscensin cartin pating Non-Facility Part B Physician Fee Scheduler, et B Fee Schedule), in affect on the annuary 1 immediately preceding the date of service;
 - b. Ph.D's, PsyD's and APNN's shall be reimbursed one hundred percent (100%) of the amount listed in the Wiscomen Participating Non-Facility Part & Physician Fee Schedule in effect on the January Limmediately preceding the dat of cryice;
 - c. MSW's and others shall be reimbursed lighty five procest (85%) of the amount listed in the Wisconsin Participating Non-Facility Part Partysica. Fee Schedule in effect on January 1 immediately preceding the date of secure.

These fee schedules shall be updated on January 1st and annually trareafter. The rates in effect as of January 1st will remain effective throughout the calendar year.

WPSHP reserves the right, at any time, to update the reimbursement amounts for new, deleted, or revised CPT or HCPCS codes. Further, WPSHP reserves the right, at any time, to implement new reimbursement amounts for items or services that are new or revised Covered Services. All updates to the reimbursement amounts listed in this Exhibit, as well as any new reimbursement amounts implemented by WPSHP, shall be determined by WPSHP in a manner consistent with applicable payment methodologies described herein. All other services not defined by the reimbursement methodology listed above will be paid at seventy five percent (75%) of PROVIDER's billed charges.

The Member shall be responsible for all WPSHP determined deductible, coinsurance, copayments, and non-covered service amounts.





DEPARTMENT OF HEALTH & HUMAN SERVICES

Centers for Medicare & Medicaid Services

200 Independence Avenue SW Washington, DC 20201

DRAFT Model QHP Addendum for Indian Health Care Providers

1. Purpose of Addendum; Supersession.

The purpose of this Addendum for dian health care providers is to apply special terms and conditions egulations to the network provider agreement by and between necessitated by federal law .d n "Offied Health Plan issuer" and/or "QHP issuer") and nere rein Provider"). To the extent that any provision of the Qualified 's network provider gree ent any other addendum thereto is inconsistent with any Healt Plan issu n of this Addendum, the provision of this Addendum shall supersede all such other provisions. provis

2. Definitio

For purposes of the Questin Health Plan is er's agreement any other addendum thereto, and this Addendum, the following terms ap definitions shall apply:

- (a) "Contract health service has the meaning given in the Ir nan Health Care Improvement Act (IHCIA) Section 4(5), 25 U.S.C. § 16 (5).
- (b) "Indian" has the meaning given in 45 C.T.R. 455.300.
- (c) "Provider" means a health program administered by the Indian Health Service a tribal health program, an Indian tribe or a tribal organization to which funding is provided advant to 25 U.S.C. § 47(commonly known as the "Buy Is tan Ad"), or an urban Indian on a zation treceives funding from the IHS pursuant to Title V of the IHCIA ub. L. 94-437), a amended, and is identified by name in Section 1 of this Addendum.
- (d) "Indian Health Service or IHS" means the agency of that the within the S. Department of Health and Human Services established by the IHCIA ection 7, 25 U.S.C. § 166
- (e) "Indian tribe" has the meaning given in the IHCIA Section 4(14), 25 U.S.C. § 1603(14).
 (f) "Qualified Health Plan" (QHP) has the meaning given in Section 1001 of the Affordable Care Act, 42 U.S.C. § 18021.
- (g) "Tribal health program" has the meaning given in the IHCL Section 4(25), 25 U.S.C. § 1603(25).
- (h) "Tribal organization" has the meaning given in the IHCIA Section 4(26) **5 C**, § 1603(26).
- "Urban Indian organization" has the meaning given in the HCL Section 4(29), 25 U.S.C. § (i) 1603(29).

3. Description of Provider.

The Provider identified in Section 1 of this Addendum is (check the appropriate box):

/ / The IHS.

/ / An Indian tribe that operates a health program under a contract or compact to carry out programs, services, functions, and activities (or portions thereof) of the IHS pursuant to the ISDEAA, 25 U.S.C. § 450 et seq.

- / / A tribal organization that operates a health program under a contract or compact to carry out programs, services, functions, and activities (or portions thereof) of the IHS pursuant to the ISDEAA, 25 U.S.C. § 450 et seq.
- / / A tribe or tribal organization that operates a health program with funding provided in whole or part pursuant to 25 U.S.C. § 47 (commonly known as the Buy Indian Act).
- / / An urban Indian organization that operates a health program with funds in whole or part provided by IHS under a grant or contract awarded pursuant to Title V of the IHCIA.
- 4. Persons Eligible for Items and Services from Provider.
- (a) The parties acknowledge that ligibility for services at the Provider's facilities is determined by federation including the HCA, 25 Corc. § 1601, et seq. and/or 42 C.F.R. Part 136. Nothing in this agreement shall be construct to in any way change, reduce, expand, or alter the eligibility uirements for services through the Provider programs. re
- (b) No term or contact, of the QHP is user's a seemed or any addendum thereto shall be construed to require individuals who are me vible under federal law for services from the Provider. The AP is acknowledges the pursuant 45 C.F.R. 80.3(d), an individual shall not be deemed subjected to discrimination by reason of his/her exclusion from benefits limited by federal law to individual eligible or services from the Provid Provider acknowledges that the nondiscrimination provider of federal law may apply.

5. Applicability of Other Feder I Laws.

d to the following: del, include but are no am Federal laws and regulations affect e Pr

(a) The IHS as a Provider:

- (1) Anti-Deficiency Act, 31 U.S.C. § 341
- (2) ISDEAA, 25 U.S.C. § 450 et seq.;
- (3) Federal Tort Claims Act ("FTCA"), 28 2671-2680; S,C, &
- (4) Federal Medical Care Recovery Act, 42 U.S. (§§ 2651-2653;
 (5) Federal Privacy Act of 1974 ("Privacy Act"), U.S.C. § 22. (5) Federal Privacy Act of 1974 ("Privacy Act"), U.S.C. & 22, 45 C.F.R. Part 5b;
 (6) Confidentiality of Alcohol and Drug Abuse Pathet Rev. 4s, 42 F.R. Part 2;
- (7) Health Insurance Portability and Accountability A of 19, ("HIPAA"), 45 C.F.R. Parts 160 and 164; and
- (8) IHCIA, 25 U.S.C. § 1601 et seq.

(b) An Indian tribe or a Tribal organization that is a Provider:

- (1) ISDEAA, 25 U.S.C. § 450 et seq.;
- (2) IHCIA, 25 U.S.C. § 1601 et seq.;
- (3) FTCA, 28 U.S.C. §§ 2671-2680;
- (4) Federal Medical Care Recovery Act, 42 U.S.C. §§ 2651-2653;
- (5) Privacy Act, 5 U.S.C. § 552a, 45 C.F.R. Part 5b; and
- (6) HIPAA, 45 C.F.R. Parts 160 and 164.

(c) An urban Indian organization that is a Provider:

- (1) IHCIA, 25 U.S.C. § 1601 et seq. (including without limitation pursuant to the IHCIA Section 206(e)(3), 25 U.S.C. § 1621e(e)(3), regarding recovery from tortfeasors);
- (2) Privacy Act, 5 U.S.C. § 552a, 45 C.F.R. Part 5b; and
- (3) HIPAA, 45 C.F.R. Parts 160 and 164.

6. Non-Taxable Entity.

To the extent the Provider is a non-taxable entity, the Provider shall not be required by a QHP issuer to collect or remit any federal, state, or local tax.

7. Insurance and Indemnification

- (a) Indian Health Service. The He is covered by the FTCA which obviates the requirement that IHS carreprive malpractice instrance as the United States consents to be sued in place of federal engloyees for any damages a property or personal injury or death caused by the negligence or wongful act or omission of federal engloyees acting within the scope of their employment. 28 U.S.C. §§ 2671-2620. Nothing in the OHP network provider agreement shall be interpreted to authorize or engage any IHS employee to perform by act outside the scope of his/her employment. The IHS shan not be required to acquire insurance, p ovide indemnification, or guarantee that the QHP will be her narmal of from liability.
- (b) Indian Tribes and a ribal Dry azations. A Provider which is a Undian tribe, a tribal organization, or employee of a tribe or ribal organization shall not be required to obtain or maintain professional liability insurance to the extension Provider is covered by the FTC expursuant to federal law (Public Law 101-512, Title III, § 314, as americal by Public Law 103-38, The III, § 308 (codified at 25 U.S.C. § 450f note); and 250 F.R. Part 900 Subpart M; 250 S.C. § 48aaa-15(a); and 42 C.F.R. § 137.220). Nothing in the QHF every network provider agreement or any adjundum thereto shall be interpreted to authorize or obligate such Provider or any employee of a charovider to operate outside of the scope of employment of such employee, such Provider shall not do required to acquire insurance, provide indemnification, or quarantee that the QHP issuer with be held harmless from liability.
- (c) Urban Indian Organizations. To the extent a Provider that is an urban Indian equilization is covered by the FTCA pursuant to Section 224(g)-(n) of the Public Health Service Act, as amended by the Federally Supported Health Centers Assistance Act, Public Lew 164-73, (codified at 42 U.S.C. § 233(g)-(n)), 42 C.F.R. Part 6, such Provider shall not be required to obtain or maintain professional liability insurance. Nothing in the QHP issuer network provider any employed of such Provider to shall be interpreted to authorize or obligate such Provider or any employed of such Provider to operate outside of the scope of employment of such employee. Such a ovider shall not be required to acquire insurance, provide indemnification, or guarantee that the CHP issuer will be held harmless from liability.

8. Licensure of Health Care Professionals.

(a) *Indian Health Service*. States may not regulate the activities of IHS-operated health care programs nor require that IHS health care professionals be licensed in the state where they are providing services, whether the IHS employee is working at an IHS-operated facility or has been assigned to a health care program of a tribe, tribal organization, or urban Indian organization. The parties agree that during the term of the QHP issuer's agreement, IHS health care professionals shall hold state licenses in accordance with applicable federal law, and that IHS facilities shall be accredited in accordance with federal statutes and regulations.

- (b) *Indian tribes and tribal organizations*. Section 221 of the IHCIA, 25 U.S.C. § 1621t, exempts a health care professional employed by an Indian tribe or tribal organization from the licensing requirements of the state in which such tribe or organization performs services, provided the health care professional is licensed in any state. The parties agree that these federal laws apply to the QHP issuer's agreement and any addenda thereto.
- (c) *Urban Indian organizations.* To the extent that any health care professional of an urban Indian provider is exempt from state regulation, such professional shall be deemed qualified to perform services under the QHP Sponsor's agreement and all addenda thereto, provided such employee is licensed to practice in any state. The parties agree that this federal law applies to the QHP issuer's agreement and any addenda thereto.

9. Licensure of Provider; Eliginity for Payments.

To the extent that the Provider determine from state licensing requirements, such Provider shall not be required to hold a state license to a ceiver any payments under the QHP issuer's network provider agreen and any addendum therets.

In the event of any asput a sing under the QKC issues, network provider agreement or any addendum thereto, the parties agree to neet and confer in good faith to resolve any such disputes prior to resolution of any disputes through any process identified in the network provider agreement. If the Provider is an IHS provider, the laws of the onite state shall apply to any provision or dispute hereunder that cannot be resolved by and between the parties in good faith. Notwithstanding any provision in the provider network agreement, IHS shall not be required to submit any disputes between the trajes to binding arbitration.

11. Governing Law.

The QHP issuer's network provider agreement and alloadenda thereto show be overned and construed in accordance with federal law of the United States. If the event of a conflict between such agreement and all addenda thereto and federal law, federal law shall prever. Nothing in the QLP issuer's network provider agreement or any addendum thereto share subjection Indian tribe, tribation or unization, or urban Indian organization to state law to any greater extent manuate law is already applicable.

12. Medical Quality Assurance Requirements.

To the extent the QHP issuer imposes any medical quality esurance requirements on its network providers, any such requirements applicable to the Provider shall be subject to Section 805 of the IHCIA, 25 U.S.C. § 1675.

13, Claims Format.

The QHP issuer shall process claims from the Provider in accordance with Section 206(h) of the IHCIA, 25 U.S.C. § 1621e(h), which does not permit an issuer to deny a claim submitted by a Provider based on the format in which submitted if the format used complies with that required for submission of claims under Title XVIII of the Social Security Act or recognized under Section 1175 of such Act.

14. Payment of Claims.

The QHP issuer shall pay claims from the Provider in accordance with federal law, including Section 206 of the IHCIA (25 U.S.C. §1621e), and 45 C.F.R., Part 156, Subpart E. The QHP issuer shall be deemed compliant with Section 206 to the extent the QHP issuer and Provider mutually agree to the rates or amounts specified in the QHP issuer agreement as payment in full.

15. Hours and Days of Service.

The hours and days of service of the Provider shall be established by the Provider. Though not required prior to the establishment of such service hours, the QHP issuer and the Provider may negotiate and agree on specific hours and days of service. At the request of the QHP issuer, such Provider shall provide written notification of its hours and the provide.

16. Contract Health Service Keff rral Requirements

The Flovider shall comply with coordination of care and referral obligations of the QHP issuer except only a specific dircumstances in which so a renorals would conflict with federal law or that referral requirements applicable to Contract Head a Services would not be met. The Provider will notify the QHP issuer when such a current stances occur.

17. Sovereign Imm hity.

Nothing in the QHP issuer's network provider agreement or in zer addendum thereto shall constitute a waiver of federal or tribal so renign immunity.

18. Endorsement.

An endorsement of a non-federal optime even, product, service, or enterpolse may be neither stated nor implied by the IHS Provider or IHS employees in the official capacities and rules. Such agency names and positions may not be used to suggest official undersement or preference areatrement and non-federal entity under this agreement.



Date

For the Qualified Health Plan Issuer:

Date

For the Preside

Page 263 of 340

Oneida Business Committee Meeting		
	Agenda Request Form	
	<u>Deadlines</u>	Instructions
	Meeting Date Requested: 03 / 11 / 15 Nature of request Session: 🖾 Open 🔲 Executive - justification required (see instructions.) Choose one:	
	Choose one:	
	Agenda header	
	(choose one): New Business/Request	
	Agenda item title (see instructions)	
	Valley Forge Lobbying	
	Action requested (choose one)	
	Information only	
	Action - please class tibe:	
	Approval of Lob ying of its government or government.	
3.	Justification	
	Why BC action is required (seconstructions):	
	For our elected official to gift calico bags to levislation while on diplomatic meetings.	
4.	Supporting Materials	Instructions
	1. Excerpt Joint Minutes 1-12-15 3.	
	2. Loretta Metoxen's Back up Documentation 4.	
	⊠ Business Committee signature required	
5.	Submission Authorization	
	Authorized sponsor (choose one): Brandon Stevens, Council Member	
	Requestor (if different from above): Cheryl Skolaski, Enrollment Director Name, Title / Dept. or Tribal Member	
	Additional signature (as needed):	
	Name, Title / Dept. Additional signature (as needed):	
	Name, Title / Dept.	

A copy of this document can be saved in a pdf format. Please e:mail this form and all supporting materials to <u>BC_Agenda_Requests@oneidanation.org</u>. <u>Save and e:mail</u>

ONEIDA TRUST DEPARTMENT

COMMITTEE

Carole Liggins, Chairperson Debra Danforth, Vice Chairperson Elaine Skenandore-Cornelius, Secretary Brandon Yellowbird-Stevens, Liaison Linda S. Dallas, Member Norbert Hill, Jr, Member Loretta V. Metoxen, Member Rita Reiter, Member Lois Strong, Member

on<yote>a=ka latiwista>nunha

909 Packerland Dr, Green Bay WI 54304 P O Box 365, Oneida WI 54155 Ph: (920) 490-3935•Fax: (920) 496-7491 DEPARTMENT Susan White, Director Jim Bittorf, Attorney Michelle Mays, Attorney

Michelle Mays, Attorney Jeff House, Financial Analyst Misty Cannon, Research Asst. Carol Silva, Administrative Assistant

MEMORANDUM



March 3, 2015

ley Forge Lobering

Date:

Subject:

In June of 2014 discr sion A brought up concerning e use 🗲 white corn as a lobbying tool at the Federal and State level Lore a Muoxen explained Dueida would esent calico bags of white corn with a gnif note explaining the historica allCt nd impact Oneid nad at V bey Forge. Since that meeting in June there have been several discussions supported by a request to be ing t calico bags back. Attached this **b** added to the BC you will find the minutes from the Stust Enro me Committee requ stin agenda to have calico bags made and the ote e plaining Valley Forge.

Minutes Joint BC/TC Mtg 12 January 2015 Page 2 of 3

Linda Dallas motioned to approve report. Seconded Rita Reiter. Motion carried unanimously.

Debra Danforth motioned to remove item from agenda. Seconded Elaine Skenandore-Cornelius. Oppose Linda Dallas. Motion carried.

B. Valley Forge – Discussion

6-24-2014 Reg Mtg. Discussion or obbying with several U.S. Senator Staffers. Loretta V. Metoxen discussed when they use to present calico bags of white correction note concerning Valley Forge. Put Valley Forge on the 3rd quarterly Joint BC/TC meeting. 1-10 T.S. Status Opdere & Recuest to remove from agenda 1 iscussion: To increase one A muite payment would take an Act of Congress. Loretta V.

Associate of the second descent of the second descent of the second descendence of the second de

Linda Dallas rotion d te place item on hext Busines, committee agenda. Seconded Loretta V. Metoxen. Motice carried animously.

C. Memorandum of Agreement between Trust and Business Countitees

 Status Update & Pequeento amove from agenda
 Linda Dallas motioned to approve MOA and put on Business Committee agenda. Seconded Norbert Hill Jr. Treasurer discussed Page 4.476, 9 A. requeend Trust Committee would consider increasing the amount of contribution over time, to fund the budge 100%. Enrollments stated they do 75% of Tribut work.

Linda Dallas withdraws her motion. Norbert all Jr withdraws his second. Request a formal request from the Business Committee. Debra Canforthenothened to approve MOA and section 9A remains the same. Seconded by Rita Reitar. Deposed Linda Dallas. Motion carried.

IV. Trust Fund Performance – Jeffrey S. House – Needs Approved Discussion on the changes of Minor's Trust Fund payment process. Linda Dallas motioned to approve report. Seconded Elaine Skenandore-Cornelius. Motion carried unanimously.

V. Recent Events/Highlights

A. Sustain Oneida Initiative – Susan White & Cheryl Skolaski – Needs Approval 6-24-14 Discussion on material for the Sustain Oneida anthology. Brandon would like language, culture, history in our surrounding schools as to "What it is to be Oneida," Loretta V. Metoxen stated we need a definition of what it means to Sustain Oneida. 9-30-14 Jennifer Hill-Kelley provided the Status Update. Discussion: Contract to be discussed with HRD Manager. 1-12-15 Status Update

Discussion: Will be putting flyers on the chairs at the GTC meetings to make the members aware of the project. Lisa Summers discussed petition with Michelle Danforth. Debra Danforth motioned to accept status update. Seconded Loretta V. Metoxen. Abstain Linda Dallas. Motion carried.

Carol Silva

From: Sent: To: Subject: Attachments: Loretta Metoxen Wednesday, February 04, 2015 2:07 PM Carol Silva FW: Message from "RNP002673743006" 20150202153318007.pdf

Okay, Carol, Here are the references that I thought I sent you regarding the Revolutionary War.

-----Original Message-----From: Scan_Culture Sent: Monday, February 02, 2015 2:32 PM To: Loretta Metaken Subject: Message from "RuP002673743006"

This E-mail was sent from 1 and 22673743006" A ficio MP C2

Scan Date: 02.02.2015 15:33 7 (-05.4 Queries to: <u>Scan Culture@c.leiri.mat.m.or</u>

Carol Silva

From: Sent: To: Subject: Loretta Metoxen Thursday, February 05, 2015 9:51 AM Carol Silva FW: Cost to make calico bags of com

Carol, Amelia could mean "charge" where she said "provide".

-----Original Message-----From: AMELIA CORNELIUS [<u>mailto:ameliacoryelius@yahoo.com</u>] Sent: Thursday, February 05, 2015 9:46 (W) To: Loretta Metoxer Subject: Re: Coroco makacalico bags of runn

I am not sure energy what you want as far as the wize of the wags. Previously, I made some bags for Public Relations Dept., that were about 4 is ones (quare with a m bon tie and builabout 1/8 cup of white corn in it. I believed I charged the PR dept. \$1 for each bag, a did not provide the lept. for the tag that described that this is a sample of the same corn that was taken by the Oneidal to V and Forge to George Wash agton.

On Wed, 2/4/15, Loretta Metoxen < net xe1@oneidanatio.org> wro

Subject: Cost to make calico bags of corr To: "ameliacornelius@yahoo.com'" <<u>an eliacornelius@yahoo.com</u>> Date: Wednesday, February 4, 2015, 2:09

Amelia,

It seems that the Business

Committee will go with this project of calico bags of corn for lobbying purpose. I need to know cost of materials and cost of the corn. If you do not have this information, do you know who does?

Oneida Indian Nation | Oneidas Brought Corn to Washington's Starving Troops at Valley ... Page 1 of 1

Oneida Indian Nation

Print this article

Oneidas Brought Corn to Washington's Starving Troops at Valley Forge

Originally printed at http://www.onei.a/diannation.com/pressroom/morenews/36439904.html December 10, 2008

Two hundred this y-one years ago this month colonial troops arrived at Valley Forge

During the Americal devolutionary War, General George Washington moved his troops to Valley Force in denisylvania on Dec. 0, 177). That winter was harsh and history reports that about 2, 000 oldiers died. As a lies of the cole sists during the war, the Oneida Indian Nation canced their can from their heinelands several hundred miles to help alleviate the hunge of Washington's starving troop during the winter of 1777-78. Oneida oral history reports that an function woman, folly foot is, staved behind after the corn was delivered to help the troops prepare the while corn which was different than the yellow corn the colonists were similar with. Provide langing the corn to Valley Forge, the Oneidas foughent the battles of Oriskany and Baratoga on the side of the colonists.

e usistere

3

Margu

trens

5

Not

 \odot

r'aahi

- 57 A scheme is, indeed: My interpretation follows that of Fleming, Washing ton's Secret War, 166-73, 192-96.
- 57 "your Ardent Desire": Horatio Gates to Lafayette, January 24, 1778, LAAR 1:249
- 58 "As I neither know": Washington to the Board of War, as quoted in LAMR I:250.
- 58 Writing to Laurens: Lafayette to Laurens, January 26, 1778, LAAR, 1:253
- 58 a letter of January 31: Lafayette to [the President of Congress], January 31 1778, LAAR, 1:267–71.
- 58 a resolution of Congress: LAAR, 1:273.
- 58 "see if some harm can be done": Lafayette to Adrienne, February LAAR, 1:462-63.
- 59 "blunders of madness or treachery": Lafer ate to Sashington, Februe p 1778, LAAR, 1:299.
- 59 "from a precipice": Lafayette to Laurer, February 19, 1778, LAAR, 1:296 59
- "Why am I so far from you": Lafayette Washington, F mary 19, 1778, LAAR, 1:299.

pre

- 50 "However sensibly your ardour": Washington Lafa, up Marcheo, 1778 LAAR, 1:342-43.
- 50 "When a man does all he can": George Washington's ales Ci ility a Decent Behaviour in Company and Conversation, ed. Charles M ton: Houghton Mifflin, 1926), 11.

CHAPTER 6: ALLIANCES

- 51 "with infinite pleasure": George Washington to Henry Laurens, May 1778, PGWRW, 15:5.
- ir "in a transport of joy": David Ramsay, The History of the American Revolution (1789; repr., Trenton: James J. Wilson, 1811), 2:93.
- i "I am myself fit to receive": Lafayette to the president of Congress, Maya 1778, LAAR, 2:40.
- I "that in serving the cause of humanity": Lafayette to Adrienne, June 16 1778, LAAR, 2:401.
- 2 Baron Friedrich Wilhelm von Steuben: See Paul Lockhart, The Drillmaster of Valley Forge: The Baron de Steuben and the Making of the American Army (New York: HarperCollins, 2008), 114-15.
- 2 "must have more than the common quantity": PGWRW, 15:41, note 6.
- 2 "in order that due honour": Letter from George Bryan, vice president of the executive council of Pennsylvania, in Lancaster, Pennsylvania, to Wash ington, May 23, 1778. "As it is apprehended here, that the Marquis-de-la Fayette has been nominated by the Most Christian King Ambassador to the United States of America, and that he may be expected shortly to pass through this borough in his way to Congress, it would highly oblige the

his Lordships Journey could be given by one of the Gentiemen of your Excellencys family, in order that due honour might be done to so respectable a personage by this state, as far as present circumstances may admit." PGWRW, 15:195.

- 62 "refused to listen": Laurens to Washington, July 31, 1778, PGWRW, 16:210.
- 63 "if my compatriots make war": Lafayette to Lazare-Jean Théveneau de Francy, May 14, 1778, LAAR, 2:398.
- 63 forty-seven Oneida warriors Joseph J. Glatthaar and James Kirby Martin, Forgotten Allies. The Oneida Indians and the American Revolution (New York: Hill and Wang, 2006), 205-
- 63 "Young warriots often need advice" "Address to Oneida Warriots," Connecticat Journal 556 (June 10, 1778): 2.
- 64 "be all of one mind": On Anne-Louis de Tousard (1749-1817), who would go on to lose an arm fighting under General Sullivan at Newport in 1778, see Michael A. Burke, "Tousard, Anne-Louis," in American National Biography: Supplement 2, ed. Mark Christopher Carnes (New York: Oxford University Press, 2005), 553-54.
 - "The detachment under your command": Washington to Lafayette, May , 1778, LAAR, 2:54.
- 55 nindead: The number of casualties is given by Washington in Washington to Lawns, May 24, 1778, PGWRW, 15:210.
- 65 "a mel and handsome retreat": "York-Town, May 30," Pennsylvania cket; or, The General Advertiser (June 3, 1778): 2.
- "The correlander of the enemy's party": Ibid.
- 66 "Frenchmerce nies": see, for example, "American News," Morning Chronicle nd I don Advertiser 2864 (July 25, 1778): 2. The British soldier's account opean d in cultiple papers, including "Extract of a Letter from Philadelphia, Jay 5," General Evening Post (London), no. 6948 (July 7-9, 1778): 1; Publi dver r (Lon n), no. 13200 (July 8, 1778): 2.
- 67 "set up the war who ?: "York- wn, May 30," Pennsylvania Packet; or, The General Advertiser (Inc 3, 178).
- the diary of Josep. Phin Martin: James Kirby Martin, ed. Ordinary Courage: The Revolutionur, War Adventures of Joseph Plumb Martin, 3rd ed. (Malden Blackwell, 2008), 71-72. I was directed to this source by Glaubre and Martin 208-16, which gives a full account of the role of the On as Bano, defille
- 67 "s. Indians, puts" atthaan and Martin, Forgotten Allies, photo opp., 179. 68 Lee and Washington: the hostilities between Lee and Washington have received considerable attention. My understanding of the events is particularly indebis to Flemin Washington's Secret War, and Charles Lee, The Lee Papers, 4 vols. New York Historical Society, 1872-75).
- 68 "when my honest quadruped friends": Lee Papers, 4:322.
- 68 "indecision": Lee to the president of the Massachusetts Council, Lee Papers, 2:303.

Page 270 of 340

The Revolutionary War | Oneida Indian Nation | Veterans

Page 1 of 3



http://www.oneidaindiannation.com/history/veterans/The-Revolutionary-War.html

Page 271 of 340

The Revolutionary War | Oneida Indian Nation | Veterans

Page 2 of 3

Polly Cooper was an Oneida woman who according to Oneida oral tradition, walked several hundred miles from her home in Central New York to Vatley Forge in the cruel winter of 1777 -78 to help feed Gen. George Washington's starving troops.

Polly Cooper along will several Oneidas, carried hundreds of bushels of corn to feed the troops. The corn they brought was while corn and different from the yellow version that is prepared simply. By contrast, the white corn requires extended preparation before it can be eaten. The soldiers, however, were desperate for food when Poliy Cooper and her fellow Oneidas arrived, and they tried to eat the corn uncooked. The Oneidas stopped the soldiers, knowing that if they are the raw corn it would swell in their stomachs and kill them.

Polly Cooperlaught the soldlers how to cook the white corn taking them through the preparation process and the lengthy cooking time. She stayed on after the other Oneidas departed for their homeland and continued to help the troops.

After the war, the Colonial Army tried to pay Polly Cooper for her valiant service, but she refused any recompense, stating that it was her duly to help her friends in their time of need. However, she did accept a token of appreciation offered by Martha Washington -- a shawl and bonnet. The shawl has been handed down by successive descendants of the poly Cooper.

The United States Congress in 1777 recognized Onelda contribution to the Revolutionary War stating:

ave expe nced your love, stron 16 Ob and y fidelity, unchangeable as . You have. of fast hold of the and ent co ant-c' preserved it free from .n. a. t and decay, 坑d bright as silver. Like bra men giory you espised danger; you d forth, in the cause of your friends, and ventu a you lives in our battles. While the SHI n contin ve light to the world, e shall lo and res t you. As our trusty friends ect you; and shall at all nali pr es consider yo e as our own.'

See also: The American Rev Alion

Provious Adlets Civil War Velerans

1202

MOST POPULAR

advocating for a new name and mascol that doesn't demean and disparage Mative Americans with the...



Turning Stone Resort Casino Welcomes Ron Ross as New Executive Chef The Onelda Nation's Turning Stone Resort Casino today ennounced the appointment of Ron Ross as Executive Chef. In his new position, Ross will oversee and direct the food and beverage operations at the...



ALLER TRANSPORT

Nexi Arile Veterans Tre

Two Turning Stone Resort Golf Courses ighlighted Among America's 100 reatest Public Courses by Golf Digest or the third consecutive time, Turning Stone Resort Casino' bild-class golf courses Atunyote and were each honored on Golf Digest's Kalu agious A a's 100 Greatest Public Courses t for 2 -2016. 1 uring pro-level play, pîclu que sellings al a long list of ewards and des, Turning Sto acce 's golf courses have disti vished to selves as some of the bast in the nation d in the Ы.

urning Otone RESORT | CASINO FRIDAY NIGHT

Petr Petrov to defend NABA & NABO Lightweight titles against Hank Lundy on Friday, January 16 at Turning Stone Resort Casino on ESPN's Friday Night Fights

PROMOTION

On Friday night, Jenuary 16, a mouthwatering title fight has been added as the second co-main event of a chemplonship highl of boxing at the Turning Stone Resort Casino and live on ESPN's Friday Night...

WEATHER

Local Pollen Reports



1/22/2015

Page 272 of 340

Page 1 of 1



Page 1 of 1



Page 274 of 340

Oneidas Brought Corn to Washington's Starving Troops at Valley Forge | Oneida Indian ... Page 1 of 1

ABOUT HISTORY CULTURE HOME ENTERPRISES IN THE COMMUNITY CAREERS NEWS

LATEST NEWS MORE NEWS RELEASES PRESS KITS MEDIA CONTACTS



TOOLS IENTS EMAIL THIS ARTICLE Like Tw 'RINT 0100 AS ARTICLE

several hundred miles to he ilevie of 1777-78. Oneida oral his y rer . s th corn was delivered to help the troops pre the colonists were familiar with. Prior to a Battles of Output ngi Battles of Orlskany and Saratoga on the

Oneidas Brought Corn to Washington's **Starving Troops at** Valley Forge

Story Created: Dec 19, 2008 at 10:14 AM EST (Story Updaled: May 24, 2011 at 1:33 PM EBT)

Two d thirty-one years ago this month roops arrived at Valley Forge

ng the den. Revolutionary War, ener orge Wa logion moved his troops in Pennsylvania on Dec. 19, to V That win was har and history reports 17 ul 2,500 soldie a Gu s aliles of the that a the On. colonists ring the ia Indian Nation carned the .on. om lheir vomelands unger of Washington's s vinler

ving troop, sturing the sh Osmer, stayed behind at an one a woman, Polly Conner, stayed behin the mile corn which was done and than the y be the corn to Valley Forge, the oneidas four of the monst an Oneilla woman, Polly Co. the re thr ont than the yell at the

> ng Slone Resort Hosis P to Kick-Off AHL AU and in Central New York

Search

MEDIA CONTACTS

Onelda Nation/Onelda Nation Enterprises - Public Affairs OfficePhone: 315-366-9291 Fax: 315-829-8386

Email: kabdo@oneida-nation.org or OneidaNationNews@onelda-nation.org

579A Main Street, Onelda, NY 13421

Follow us on Twitten EsavOnStores ETSGolf EOneida Nation News (OneidaNatnNews) For calls related to Turning Stone Marketing/Advertising, call (315) 381-7685. To reach other departments at the Oneida Indian Nation use the following link: http://www.oneldaindlannation.com/about/addres

MOST POPULAR

© 1994-2013 Onekia Indian Netion acy Policy - ONEIDA - Tenns of Uso Powered by Broadcast Interactive Media.

Page 275 of 340

COLONIAL AMERICAN DIGRESSIONS: LESSER KNOWN DAUGHTERS OF LIBE ... Page 1 of 3



Page 276 of 340

COLONIAL AMERICAN DIGRESSIONS: LESSER KNOWN DAUGHTERS OF LIBE ... Page 2 of 3

Soup-(a-traditional iroquois-meal)."Polly stayed on aller the com was delivered and lended to sick men. When offered payment of money for her services, Polly declined it. However, she showed fondness for a black shawl offered for sale in a nearby village. Money was raised to purchase the shawl and it was given her.



'Rachel Silverthorn's Ride' (Mural painted by W Beauchamp, 1938) Courter of comenhistoryblog

During the Revolutionary War, the come by as between American settlers and the acopte in Lyo ming County Pennsylvania (1.6 valle) along the West Branch of the Susquehanna River). Mainly, the sacher called concopte v. the English, so they ere engines.

dhorn

OUS OF ere committed by bou sides in . Three separate se figt Incidents ned 10 June 1778, include lhe 'Plum ee M sacre.' The Pennsylvania co. overnme couldn't send solo s lo p the area, wing the Quaker stand on no violenc whal few militiamen Per /wanla d already deving AT been sent to the C tiner/

After hearing 🐱 the Ayom y Vaile Massacre,' 3 July 17, and also that wa and Royal Army solours were late, g loward them, the settlers in Lycoming unty decided they needed to evace it ne ve and has it that Robert 🖓 cohoven rode the ridge of Bald Eagle Juntal o wam the people in the wester part of the to take a hose along Muncy Creek and the to take a h valley. Rachel Silverthorn volunteer Wyalusing Path (that connected two ranches of the Susception River) to ware people there. Everyone got out alive, hough all of the were burned (Including Rachel's family home).



Elizabeth Hutchinson Jackson Memorial Courtesy of findagrave.com

Elizabeth and Andrew Jackson emigrated from Ireland to North Carolina In the English North American colonies in 1765, along with their two sons, Hugh and Robert. Andrew died in an accident not quite two years later. Their third son was born three weeks later and was named Andrew in honor of his father.

Eilzabeth's oldest son joined the tocal mitilia unit commanded by Colonel William Davie. Hugh died of heat and exhaustion at the Battle of Stono Ferry, South Carolina, May 1780. Robert and Andrew, Jr. became messengers for Col. Davie and were captured at the Battle of Waxhaw, 29 May. Eilzabeth learned of their being held prisoner and went to Camden to get their release. By then, both boys had contracted smallpox, Robert died only a few days after the three of them reached the house of Eilzabeth's sister in Waxhaw. Elizabeth nursed her only surviving son back to health. After he recovered, she learned of an outbreak of cholera among American soldiers aboard a prison ship in Charles Town harbor. Traveling two hundred miles to the seaport,

LAST: WIELS AND TESTAMENTS: OPENING PARAGRAPHS

COLONIAL ROADS FOR COMMERCE

DAUGHTERS OF LIBERTY CHORES: SOAP-MAKING

SONS OF LIBERTY OCCUPATION: CANDLE-MAKER (CHANDLE...

WHAT'S FOR COLONIAL SUPPER?

WHAT'S FOR COLONIAL DINNER?

WHAT'S FOR COLONIAL BREAKFAST?

1742 AMERICANA: MORE ABOUT SOUTHERN COLONIES

1742 AMERICANA: SOUTHERN COLONIES

1742 AMERICANA: MORE ABOUT THE MIDDLE COLONIES

1742 AMERICANA: THE MIDDLE COLONIES

1742 AMERICANA: MORE ABOUT NEW ENGLAND

1742 AMERICANA: NEW ENGLAND

COLONIAL OCCUPATION: BRICKMAKER

COLONIAL OCCUPATION: ROPE-MAKER

COLONIAL MAIL DELIVERY

LESSER KNOWN DAUGHTERS OF LIBERTY,

ESSER KNOWN D/10000 SOF LIBERTY, ART SEVEN

AERICAN COLONAL CLOTHING, PART FOUR

ONIAL DTHES, PART

COLONIAL CLOTHING, PART TWO

AMERICAN COLONIAL CLOTHING

STILL MORE

E AMERICANA

2011 (80)
 About Ms

1741 M



David Webb Fowler San Bernardino, CA, United States David Fowler stambled

across an odd fact: Benedict Amotd was a Son of Liberty, but George Washington was not. This led to his writing a history: Sons of Liberty-- Tools of Destruction. David can be contacted at 2editordavid@gmail.com, Entries from this website are now available in book form.

Page 277 of 340

Page 1 of 1

https://www.google.com/blank.html

Oneida Business Committee Meeting Agenda Request Form

1.	1. Meeting Date Requested: <u>3</u> / <u>11</u>	/ 15				
2.	2. Nature of request Session: ⊠ Open □ Executive - jus					
	define what is c	onsidered "executive" information, then choose from the list:				
	Agenda Header (choose one): New Business/Request					
	Agenda item title (see instructions):					
	ONSS Contract Personnel Salarier , d E	ONSS Contract Personnel Salarier and Benefits SOP				
	Action requested (choose on					
	☑ Information only					
	Action - please describe:					
	Motion in cept the callSS Contract Pres	onne Salaries and Benefits SOP as an FYI.				
っ	3. Justification					
J.						
	Why BC action is required set instant	ions):				
	Deswart for DC to review ONCC burley of					
	Request for BC to review ONSS	Peren sel Salaries and Bener's SOP.				
4.	. Supporting Materials	Instructions				
	⊠ Memo of explanation with required					
		contract (check the box below if sign turn required)				
	Other - please list (Note: multi-media presentations due to Tribal Clerk 2 days prior to meeting)					
	1. Memo	3. Ochoard Minutes 2/6/15				
	and Benefits 4.					
	Business Committee signature required					
5.	5. Submission Authorization	Submission Authorization				
	Authorized sponsor (choose one): Fawn I	Billie, Council Member				
	Requestor (if different from above): Debbio	e Danforth, School Board Chair				
		Title / Dept. or Tribal Member				
	Additional signature (as needed):	Title / Dept.				
	Additional signature (as needed):					
	Name,	Title / Dept.				

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Page 279 of 340

From:	Fawn J. Billie
Sent:	Friday, March 06, 2015 6:16 AM
То:	BC_Agenda_Requests
Cc:	Lisa M. Summers; Lisa A. Liggins; Fawn L. Cottrell
Subject:	Fwd: BC agenda item re: school board
Attachments:	14 05 02 Memo to ONSS School Board re HR EE Contract Issue.pdf

Importance:

High

Sent from Samsung Mobile.

------ Original message ------From: "Debra J. Danforth" Date:03/05/2015 5:13 PM (GMT-06:00) To: "Fawn J. Billie" Cc: "Cathy L. Bachhuber", "Sharon A. Mousseau", "Sudoanne termon Curry PhD (JC: 19/@. Inlaw.com)" ,"Apache B. Danforth", "Rhiannon R. Metoxen", dell' a containt, "Sylvia S. Corneliu, "Prisci, "E. Déstart" Subject: RE: BC agenda item

The School Board took the official action to approve the SOP based board he Board pourthority to deviate from a policy when the members of the Board concluded that such deviation was require to be consistent with "sound excational administration." The Board's long-standing authority is delegated to it by the General T hal Council (GTC). The GTC Resolution # 1-29-77-A created the School Board to "coordinate existing a d future programs of the Tribe," including activities related to short-term and long-term planning, and to *coordinate in activities with the to siness Committee*. By GTC Resolution #7-9-83-A, on January 8, 1983, the GTC directed that

An agreement be reached between the Oneida Business Committee and the meida wibal School Board regarding the *autonomous administration* of the Oneida Tribal School which yould provide for for involvement of the Oneida Tribal School Board in all personnel matters related to all personnel employed with operation of the Oneida Tribal School.

Pursuant to this authority, the School Board entered into a Memorandum of Agreement (MOAL with the Oneida Business Committee. The GTC adopted and approved a resolution on March 21, 1988, which approved the MC. The MOA specifically provides that decisions related to the school personnel are to be

Based on sound educational administration recognizing the contractual and school year provision for teachin personnel.

The Memorandum of Agreement (MOA) also addresses certain issues in the personnel and contract management ONSS. The MOA provides, in part, the following:

+he

All contracts...related to the operation or planning of the Oneida Tribal School shall be first reviewed and endorsed by the BOARD. Such review and endorsement shall be presented in writing to the COMMITTEE at time consistent with existing Oneida Tribal Policies and Procedures. *The recommendations of the BOARD shall be followed by the COMMITTEE, unless good cause to the contrary is shown.... (JD Joanne Harmon Curry, Fredericks Peebles & Morgan LLP May 2, 2014)*

1

Therefore, based upon the legal advice from the School Board attorney as well as the historical difficulties the ONSS has had over the past year in retaining qualified teaching personnel

and based upon the number of still vacant positions within the ONSS, the SOP was created through the assistance of the School Board's attorney JD Joanne Harmon Curry as well as previous legal opinion of the Board's authority.

At our February 2, 2015 school board meeting, HR was invited to our school board meeting as the Contract discussion was to take place to be in compliance with our contractual obligations for the 2014-2015 contracts. Based upon the 2014-2015 contracts, the ONSS is contractually obligated to notify the contracted percented no later than March 15, 2015 of our intent to offer a 2015-2016 contract. HR was offered a copy of the core of looked at the contract and made no further recommendations and left before the contract discussion and the part of k place. Therefore, based upon the Board's authority as noted above, the School Board took official action beed up h sound educational administration to approve the SOP and the 201 ntracts and are forwarding to the servicess Committee for -010 contracts ve been reviewed by acceptance as information only. All 2015-20 and of Jard of ney and are in our complian date of 03-15-2015. This is bed up previous BC action process to be sent out to personnel to mee in which the Business Committee approved he wage proposal effective October 1, 20 to be ffect for 2015-2016 contracts. I hope this helps to address Scho e Business Committee's concerns regarding t Board's BC item specific to the Contract SOP.

Debbie Danforth Oneida Nation School Board Chair

From: Fawn J. Billie Sent: Thursday, March 05, 2015 4:07 PM To: Debra J. Danforth Cc: Cathy L. Bachhuber; Sharon A. Mousseau Subject: BC agenda item

OK, received more feedback this afternoon and rest of team is requesting a responsibly to norrow aday, or 6 at 9am. Being more specific, at the BC Agenda Review we need a follow-up on agenda item XII. E. Oneider a tion Scho Board's SOP Contract Personnel Salaries & Benefits.

2

6 HRD?

The question: Why is there no signature box for HRD since the SOP is related to Personnel? Whe SOP did not

Sorry in for the inconvenience. Thank you again!

Fawn Billie, Councilwoman Oneida Business Committee Oneida Tribe of Indians of WI Office: (920) 869-4432 fbillie@oneidanation.org https://oneida-nsn.gov/



JOANNE HARMON CURRY*

1900 Plaza Drive Louisville, CO 80027 Telephone: (303) 673-9600 Direct: (303) 815-1727 Fax: (303) 673-9155 E-Mail: jcurry@ndnlaw.com www.ndnlaw.com *Licensed in the State of Wisconsin



This Memorandum is prepare was a response to the concern based by Seraldine R. Danforth, Tribal Human Resources Director, in recent combunications regarding the Oneide Nation School System (ONSS) 2014-2015 Employee Contracts and the "Indirect Combensation Provision. In particular, Ms. Danforth has expressed the polition that the contracts issued by the Oneida Nation School Board (Board) for the ONSS employees must be in compliance with two policies of the Tribal Human Resource Department (HR) related to (1) the mathematication of personal leave that are allowed to accumulate and carry over from one contract over to the next contract year, and (2) the "trade-back for cash" provision.

In an e-mail dated February 4, 2014, regarding the total accumulatechours of personal leave by five (5) ONSS contracted employees, Ms. Danforth stated that bree (3) employees "still remain to be over the max" at the end of January. Ms. Danforth asserted that the three employees are currently over the maximum number of hours of personal leave that an employee is allowed to accumulate and carry over from one year to the next under Tribal employment policy. Ms. Danforth also asserted that the School Board should change its contract language regarding the carry-over of personal leave time to "no payout of unused time, with a use or lose." Nevertheless, Ms. Danforth's e-mail message also stated that current Tribal policy allows for the accumulation of a maximum of 280 hours that may be accumulated under HR policy. This Memorandum focuses on the authority of the elected officials of the Board to deviate from HR policy when the members of the Board conclude that such deviation is required when consistent with "sound educational administration." The Board's long-standing authority is delegated to it by the General Tribal Council (GTC). The GTC Resolution #1-29-77-A created the School Board to "coordinate existing and future programs of the Tribe," including activities related to short-term and long-term planning, and to *coordinate its activities with the Business Committee*. By GTC Resolution #7-9-83-A, on January 8, 1983, the GTC directed that

an agreement be reached between the Oneida Business Committee and the Oneida Tribal School Board regarding the *autonomous administration* of the Oneida Tribal School which would provide for full involvement of the Oneida Tribal School Board in all personnel matters related to all personnel employed in the operation of the Oneida Tribal School. [Emphasis supplice.]

Purport to this autiority, the Scool Board entered into a Memorandum of Agreement (MOA) with the Business Corporates. The TTC adopted and approved a resolution on March 21, 1988, which approved the MOA. One MOA specifically provides that decisions related to school personnel are to be

based on sound educational administration recognizing the contractual and school year provisions for teaching personnel.

The Memorandum of Agreerent (MOA) also addresses certain issues in the personnel and contract management of the ONSS. The MOA provides, in part, the following:

All contracts . . . related to the operation or planning of the Cheida Tribal School shall be first reviewed and endorsed by the BOALD. Such review and endorsement shall be presented in writing to the CC 4MIT EE at times consistent with the assing Oneida Tribal Policies and Procedures. *The recommendations of the BOARD shall be followed by the COMMITTEE, unless good cause to the contrary is shown* [Emphasis supplied.]

The MOA and its resulting authority that the GTC delegated to the elected officials of the Board is unique within the Tribal structure.

The above provision of the MOA remains in force and has been alied on by the Board and the Business Committee for the establishment of the ONSS Laplove. Contracts for many years now. After the Board reviews its current-year personnel contracts, it determines what, if any, revisions are needed in the contracts and then endorses a final contract to be officied to be Employees by Board action. The terms and conditions of the annual ONSS Employee Contracts are determined by the Board, with input from the ONSS Administration, based, in part, on the terms and conditions of public school contracts in the region, as well as other, comparable tribally-operated schools, in order for the ONSS to remain competitive in the market for highly trained, certified, and licensed school personnel. The ONSS must employ teaching personnel who hold degrees and licenses comparable to that required by the State of Wisconsin, Department of Public Instruction because the ONSS receives federal funds that support its teaching mission and must comply with certain minimum federal requirements. This requirement contributes to the unique position of the elected members of the Board to act independently with regard to ONSS employees because the Board has unique considerations for selecting the ONSS employees that are grounded in the "sound educational administration" of the School System. These requirements set the ONSS and the Board apart from other tribal departments and the typical considerations applied by HR when recruiting and hiring other tribal employees. The independence of the Board articulated in the MOA is consistent with federal law governing Tribally-Controlled Grant Schools, such as the ONSS, where federal funding is provided and decisions regarding the schools are to be made at the discretion of the Board.

Notably, the MOA does not provide for HR review of the Board Employee Contract decision—including any authority to overrule, overturn, or prohibit the ONSS elected officials' decisions. The Board is required to present its annual Employee Contract decisions to the Business Committee, as per the above quoted provision. Additionally, even at the Business Committee level, the decision of the Board "shall to followed by the Committee," and cannot be overturned by the Business Committee "unless poor cause to the contrary is shown."

inally, even if the Borch's contractomployee decisions were not required to be made independently of the general HR principal when required for the sound educational administration of the ON S, the GTC Posolution 05-21-11-1 on "Amendments to the Personnel Policies and Procedures on Tade Lack for Cash of Personal and Vacation Time" does not prohibit a Board Employee Contractor provision that provides to the actual trade back for cash for unused personal time. In fact, it specificary states as follows:

WHEREAS, in some instances, limiting the trade back for cash program to those employees who have backed per onal and/or vacation hears and are unable to utilize those hours due to their working conditions, such as a staffing shortage, will be fiscally responsible and will enable those cap, wees to receive compensation in place of time off.

As the Board has explained to HR in he pair, the requirements of the CNSS for the operation of a Tribal <u>School</u> results in working coelitions that require licensed personnel in the classroom and school environment at all times when school is in session. Whenever any such employee takes personal leave during the school day, the ONSE is burchined with the obligation to replace that employee during the personal leave time that has been granted. The ONSS must hire a substitute, licensed individual to replace the ONSS Employee notice classroom. This adds to the fiscal burden on the ONSS budget, not to mention the administrative time required to replace the Employee.

The GTC Resolution allows for this very type of consideration. The Board has determined that the conditions for operating the Tribal School are such that its employees are sometimes unable to utilize their personal time due to the working conditions of a school, are it is "fiscally responsible" to allow the employees the option to trade back their personal leave time for cash, which "will enable those employees to receive compensation in place of time off."

In conclusion, the Board has determined that its decision to provide the ONSS Contracted Employees with a trade-back for cash option, as well as designated accumulated personal leave time, is grounded in the sound educational administration of the School, and it has approved the 2014-2015 Employee Contracts, including the Indirect Compensation provisions at issue here (consistent with previous years, including the current school year). The Board's decision is also consistent with the GTC 2011 Resolution permitting variances to the general policy on employee trade-back for cash options. The Board provided the employees with notice of any changes to the current Employee Contracts for the 2014-2015 school year by March 15, as it is compelled to do under the terms of the 2013-2014 Employee Contracts. The Employees have been notified that the Indirect Compensation terms and conditions for the 2014-2015 school year will remain the same as those offered in the 2013-2014 Employee Contracts.

Therefore, it is recommended that the Board proceed to the Business Committee and present the 2014-2015 Employee Contracts to the Business Committee, which is obligated to accept the Board's recommendations, *unless good cause to the contrary is shown*.



To: Oneida Business Committee
From: Cathy Bachhuber, ONES
Date: March 4, 2015
Re: ONSS Contract Personnel Salaries and Benefits SOP

The School Board is submitting the attached ONSS Contract Personnel Salaries and Benefits SOP as information only for Business Committee review. The School Board approved the SOP at the February 6, 2015 Special School Board Meeting and approved the February 6, 2015 minutes at the March 2, 2015 Regular School Board Meeting.

Requested action:

Motion to accept the ONSS Contrant Persentel Salaries and Benefits SOP as an FYI.

ONEIDA TRIBE OF WISCONSIN	TITLE: ONSS Contract Personnel Salaries and Benefits	ORIGINATION DATE: 02/06/15
ONEIDA NATION SCHOOL BOARD		REVISION DATE:
STANDARD OPERATING PROCEDURE		EFFECTIVE DATE: One week after last signature
SOP NUMBER: 10	APPROVED BY: Oneida Nation Scher Board	DATE: 02/06/2015
PAGENO 1 of 3	REVIÉVED BY : Oneida Busiliess Committe	DATE:

1. PURPOSE

1.1 The General Troal Council provided for the autocomous administration of the Oneida Nation I chool System (ONSS) (formerly known as the Oneida Tribal School) through the elimination of the Oneida vation School Board (formerly known as the Oneida Tribal School Board). By agreement with the Oneida Business Committee in the Melnoranium of Agreement ("MOA") lated March 21, 1988, the Oneida Nation School Board, in its responsibilities related to all personnel matters, is to ensure that personnel decisions are based upon sound educational administration policies.

1.1.1 The MOA also directed that the Perseane Policies and Procedures system (formerly known as the Oneida Aribal Lanagement System) shall be revised to accommodate the decisions made by the Oneida Nation School Board related to personnel.

1.2 The MOA also provides that the Oneida Nation School Board, hereafter referred to as "School Board") has the authority to enter into contracts, including contracts with personnel, as necessary for the sound educational administration of ONSS.

1.2.1 The MOA further directs that all contracts related to the operation or planning of the ONSS shall be first reviewed, endorsed, and approved by the School Board.

1

1.2.2 The School Board's initiation, review, endorsement, and approval of contracts, including personnel contracts, shall be presented in writing to the Oneida Business Committee at times consistent with existing Oneida Tribe of Indians Personnel Policies and Procedures or with the ONSS hiring and/or contract renewal timeline.

1.2.3 The MOA provides that the decisions and actions of the School Board shall be followed by the Oneida Business Committee, unless good cause to the contrary is shown.

1.3 This Standard Operating Procedure is enacted for the purpose of promulgating a rule that provides for the School Board's determination and approval of the salaries and tenents of ONSS contracted personnel.

3.1 The School Boold's decisions for ONSS contracted personnel salaries and benefits shall be back on sould educational administration that considers the professional and educational ceeds of the ONSS, the School Board's ability to regular and hire qualified personnel, and any annual funding constraints of the One aa Tri

1.3.2 the School Board has determined one the salaries and benefits for ONSS personnel, wheneve required to satisfy certain licensing and certification requirements for employment, and the School Board's ability to successfully recruit and hire bersonnel for the ONSS are market down by the public, private, and tribal school within the region and athlene concerning of the ONSS, and that it is fiscally responsible to eview and determine personnel calaries and benefits on an annual basis to remain competitive in the market made from the personnel, consistent with sound exceptional comministration.

2. **DEFINITIONS**

- 2.1 Blue Book Oneida Tribe of Ind. us P sonnel Policies and Recedures
- 2.2 Contracted Personnel/Employee– Ar individual with is issued a contract of employment with the ONSS from the School Pourd and who has accepted employment with the ONSS.
- 2.3 MOA March 21, 1988, Memorandum of Agreement between the Oneida Business Committee and the Oneida Nation School Berard approved by the General Tribal Council
- 2.4 ONSS Oneida Nation School System
- 2.5 Salary and Benefits Financial compensation for work performed under the ONSS personnel contract, such as base salary, and may include, but not be limited to, additional duties compensation, merit pay or bonuses, paid personal leave time, and health insurance, dental insurance, retirement benefits, life insurance, and short- and long-term disability insurance.
- 2.6 School Board Oneida Nation School Board

2

3. WORK STANDARDS

- 3.1 The School Board shall make an annual determination of the nature and type of salary and benefits that will be offered to current and new employees under a contract with the ONSS to be issued by the School Board.
- 3.2 The annual review and determination of the ONSS contracted employees' salary and benefits shall be made in a timely manner and based on sound educational administration practices and market forces related to the recruitment and employment of educational personnel.
- 3.3 The School Board shall revise the then-current employee contracts based on its annual review, at it sole discretion as it deems necessary, and prior to any employee renew reptification that may be required under the then-current employee contracts.

The School Board share stermine the salary and benefits terms and conditions that will be offered to contract personnel, including exceptions to the thencurrent Blue Book.

The for bol Board share adopt the annual employee contract(s), including the arms and conditions for galaries and be efits, by motion at a properly convened meeting of the School Board.

3.57 S pervisors are responsible for menitoring employee vacation/per ional time accruals, scheduling employees' working hours and approving a charging time off requests.

3.5.2 Implyees are responsible for requesting time off utilizing personal, vacation, banked, and don ted hours, if applicable, and hall comply with the procedures established by three NSS Adminicration for documentation of the use of personal leave time, including a proval.

3.5.3 Trade-back or case of unused personal lease time shell be governed by the ONSS Standard Operating Procedure Number CON S Employee Indirection Compensation: trade back for Cash.

3.6 Consistent with the MOA, the que Book shall be revised to accommodate the decisions made by the Oneida Nation School Board related to presonnel, including the ONSS salary and benefit determinations for personnel contracts.
3.6.1 The Oneida Human Resource Department shall process ONSS Contracted Employees' salaries and benefits on strent with the then-current terms and conditions of the ONSS employee contracts.

4. **REFERENCES**

- 4.1 Memorandum of Agreement between the Oneida Busiless Committee and the Oneida Nation School Board, March 21, 1988
- 4.2 Oneida Tribe of Wisconsin, Personnel Policies and Procedures
- 4.3 ONSS Employee Contracts
- 4.4 ONSS Standard Operating Procedure Number 8, ONSS Employee Indirection Compensation: Trade Back for Cash
TIME: 12:15 p.m.

Oneida Nation School Board Special Meeting Minutes

DATE:Friday, February 6, 2015PLACE:ONES Conference Room

The Oneida Nation School Board is committed and accountable to students, parents, families, staff, and community members to provide regulatory oversight for a safe, positive, culturally diverse, holistic, learning atmosphere based on OnAyoté a ka values. We will provide the expectations, resources, and educational opportunities to encourage students to be productive.

PRESENT: Apache Danforth, Debbie Danforth, Dellora Cornelius, Sylvia Cornelius, Priscilla Dessart, Rhiannon Metoxen

EXCUSED: Dewain Danforth

OTHERS: Inaron Mousseau, Arthy Exer indore, Linda Jenkins, Cathy Bachhuber

OPENING Dellora Cornelius

Approval of Asend

CALL TO ORDER: De bie Donforth

TIME: 12:15 p.m.

- II. Special Presentation
- III. Minutes

I.

IV. Tabled Business

V. Old Business

- A. Follow-up
 - 1. Employee Contracts 2015-2016

Motion by Priscilla Dessart to approve the contract with the noted changes, seconded by Sylvia Cornelius. Motion carried unanimously.

Motion by Sylvia Cornelius to adopt the teachers' Scale 22% increase for the 2015-2016 employee contracts and include the 2% for the Superintendent and Administration contract, seconded by Rhiannon Merexen. (Ne vote)

2. Superintendent Contracts 2015-2016

VI. New Business

- A. ONSS Contract Personnel Salaries and Benefits SOP Motion by Dellora Cornelius to approve, seconded by Priscilla Dessart. Motion carried unanimously.
- VII. Reports
- VIII. Executive Session

IX. Recess/Adjourn

Recess/break at 12:50 p.m. (Lack of quorum) Return from recess/break at 1:15 p.m.

Motion by Apache Danforth to recess [at 1:15 p.m.] until Wednesday, February 11, 2015, seconded by Rhiannon Metoxen. Motion carried unanimously.

Oneida Nation School Board Special Meeting Agenda

DATE: Wednesday, February 11, 2015 **PLACE:** ONES Conference Room

PRESENT: Dellora Cornelius, Debie Danforth, Apache Danforth, Rhiannon Metoxen, Sylvia Cornelius, Priscilla Dessart

EXCUSED Dewan Danforth

OTHERS: Artley Skenaptore, Linda Jerlans, Vvette Peguero, Sharon Mousseau, Cathy Bachhuber

CALL TO ORDER: Ve bie Denforth

TIME: 12:05 p.m.

TIME: 12:00 p.m.

Motion by Sylvia Cornelius to obmer ut of recess of 12:05pm. Execonded by Dellora Cornelius. Motion carried unanimously.

V. Old Business

A. Employee (Teacher) Contracts 2021-2016 Excerpt from Feb. 6, 2015. Solid row Princilla Dessart to appropriate the contract with the noted changes, seconded by Sylvia Cornelius. Major corried unanimously.

Motion by Sylvia Cornelius to an of the teachers' Scale D 2% includes for the 2015-2016 employee contracts and include the 2% for the Sure Intendent and Administration contract, seconded by Rhiannon Metoxen. (No vote)

[Vote on teachers' scale motion] Motion arried unapidously.

B. Superintendent Contract 2015-2016 Motion by Rhiannon Metoxen to approve the contract with the hanges noted, seconded by Priscilla Dessert. Motion carried unanimously.

VI. New Business

A. Administrator Contracts 2015-2016

Motion by Apache Danforth to accept the changes for the Administrator Contract and approve the contract, seconded by Dellora Cornelius. Motion carried unanimously.

Motion by Apache Danforth to support the increase to the base of the administrative salaries to 3, to include the other changes to reflect the changes that the Board had made with the contracts specifically under classification II., adding the Principal, K-8 and 9-12, and the days to 260, seconded by Sylvia Cornelius. Motion carried unanimously.

VII. Reports

VIII. Executive Session

IX. Recess/Adjourn

Motion by Priscilla Dessart to adjourn [at 12:42 p.m.], seconded by Dellora Cornelius. Motion carried unanimously.



Page 292 of 340 Oneida Business Committee Meeting Agenda Request Form

2. Nature of request Session: ⊠ Open ☐ Executive - justification required. See instructions for the applicable laws that define what is considered "executive" information, then choose from the list:
define what is considered "executive" information, then choose from the list:
Agenda Header (choose one): New Business/Request Agenda item title (see instructions): Accept quarterly reporting update and direct appropriate follow up Action requested (choose one) Information only Action - please rescribe: 1. Accept quarterly reporting update 2. Direct the espective forting update 2. Direct the espective liaisons/supervisors to follow up on the missing/unlocated quarterly reports. 3. Justification Why BC action is required item instructions): Image: Supporting Materials Imstructions Memo of explanation with required information (see instructions) Imstructions Imstructions Imstructions
Agenda item title (see instructions): Accept quarterly reporting update and direct appropriate follow up Action requested (choose one) Information only Action replease rescribe: 1. Accept charted or only 2. Direct the respective liaisons/supervisors to follow up on the missing/unlocated quarterly reports. 3. Justification Why BC action is required isconstructions):
Accept quarterly reporting update and direct appropriate follow up Action requested (choose one) Information only Action - please bescribe: 1. Accept quarterly reporting update 2. Direct the respective liaisons/supervisors to follow up on the missing/unlocated quarterly reports. 3. Justification Why BC action is required iscentistructions): Memo of explanation with required information (see instructions) Memo of explanation with required information (see instructions) Report Resolution
Accept quarterly reporting update and direct appropriate follow up Action requested (choose one) Information only Action - please bescribe: 1. Accept quarterly reporting update 2. Direct the respective liaisons/supervisors to follow up on the missing/unlocated quarterly reports. 3. Justification Why BC action is required iscentistructions): Memo of explanation with required information (see instructions) Memo of explanation with required information (see instructions) Report Resolution
Action requested (choose one) Information only Action - please rescribe: 1. Accept please rescribe: 2. Direct the respective liaisons/supervisors to follow up on the missing/unlocated quarterly reports. 3. Justification Why BC action is required isconstructions):
□ Information only △ Action-please rescribe: 1. Acceptometrepresentation update. 2. Direct the respective liaisons/supervisors to function on the missing/unlocated quarterly reports. 3. Justification Why BC action is required iscennstructions):
 Action - please rescribe: Accept parterby exorting update. Direct the respective liaisons/supervisors to for own p on the missing/unlocated quarterly reports. Justification Why BC action is required seconstructions): Memo of explanation with required information (see instructions) Report Resolution Contract (check use box below if signature required)
2. Direct the assignment of a forward on the missing/unlocated quarterly reports. 3. Justification Why BC action is required as unstructions): 4. Supporting Materials Section Memo of explanation with required information (see instructions) Section Report Resolution Contract (check we box below if signature required)
2. Direct the assignment of a forward on the missing/unlocated quarterly reports. 3. Justification Why BC action is required as unstructions): 4. Supporting Materials Section Memo of explanation with required information (see instructions) Section Report Resolution Contract (check we box below if signature required)
Why BC action is required seconstructions): Image: Contract Check of Explanation with required information (see instructions) Image: Contract Check of Explanation explanation
Why BC action is required seconstructions): Image: Contract Check of Explanation with required information (see instructions) Image: Contract Check of Explanation explanation
A. Supporting Materials Memo of explanation with required information (see instructions) Report □ Resolution □ Contract (check are box below if signature required)
 Memo of explanation with required information (see instructions) Report Resolution Contract (check the box below if signature required)
 Memo of explanation with required information (see instructions) Report Resolution Contract (check the box below if signature required)
 Memo of explanation with required information (see instructions) Report Resolution Contract (check the box below if signature required)
□ Report □ Resolution □ Contract (check the box below if signatine required)
Other - please list (Note: multi-media presentations due to Tribel Clerk 2 days prior to meeting)
13
2. 4.
Business Committee signature required
5. Submission Authorization
Authorized sponsor (choose one): Lisa Summers, Tribal Secretary
Requestor (if different from above):
Name, Title / Dept. or Tribal Member
Additional signature (as needed): Name, Title / Dept.
Additional signature (as needed): Name, Title / Dept.

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

INTEROFFICE MEMORANDUM

TO: ONEIDA BUSINESS COMMITTEE

FROM: LISA SUMMERS, TRIBAL SECRETARY

SUBJECT: QUARTERLY REPORTING UPDATE

DATE: MARCH 3, 2015

Background

On February 5, 2015, my office requested a review of the FY-2014 and FY-2015 quarterly reports from the Business Complete Support Office. This review was to ensure all reports for FY-2014 and FY-2015 cere submitted to the Business Committee.

On March 2, 2015, Kathlee Pietoren, Frecutive Tribal Clerk, informed my office that the following reports are missing/uplocated for FY-2014:

Area	Missie / Uncecated Quarter	Liaison/Supervisor
Emergency Managemen	FY-2014 C2	OBC Officers
Oneida Gaming Competion	FY-2014 Q4	Brandon Stevens
Human Resources Department	FY-2014 C	OBC Officers
Oneida Land Claims Commission	FY-2014 Q2	Brandon Stevens
Retail Enterprise	F-2014 Q2-Q4	OBOOfficers
All reports Q1 reports for FY-2015	are up to date.	
All reports Q1 reports for FY-2015	are up to date.	$\mathbf{\vee}$

Requested Action

- 1. Accept quarterly reporting update.
- 2. Direct the respective liaisons/supervisors to follow on the mixing/unlocated quarterly reports.

Oneida Business Committee Meeting Agenda Request Form

1. Meeting Date Requested: 03 / 11 / 15

2. Nature of request

Session: 🛛 Open 🛛 Executive - justification required. See instructions for the applicable laws that

define what is considered "executive" information, then choose from the list:

Other - BC Approval to Accept TribalNet Advisory Board Position

Agenda Header (choose one): New Business/Request

TribalNet Advisory Board Approved & Acceptance

Agenda item title (see instructions):

Action requested (choose or

- □ Information or y
- Actio please describe:

Requesting PC approval for Kelly L. Scenandore to represent the Oneida Tribe of Indians of WI by accepting a Triban of Advisory Board Position

3. Justification

Why BC action is required the intractions):

Kelly L. Skenandore would like to accept a position as a TribalNerBoard /lember with the approval of the BC. As a member of the TribalNet, dvisory Bratonit is requested for the board members to be present at the conference. The Conference Fee of \$700 is wanted for TribalNet Bratch tembers.

4. Supporting Materials

Memo of explanation with required information (see inductions)

Instructions

- Report Resolution Contract beck the box below if signate required)
- Other please list (Note: multi-media presentation due to Tribe Nerk 2 days prior to meeting)

1.	,	3.			
2.		4.			

Business Committee signature required

5. Submission Authorization

Cubimosion Autionzution	
Authorized sponsor (choose one):	Debbie Danforth, Division Director/Operations
Requestor (if different from above):	Jeff Carlson, CHD Business Office Manager/Comprehensive Health
	Name, Title / Dept. or Tribal Member
Additional signature (as needed):	
	Name, Title / Dept.
Additional signature (as needed):	
	Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

1

Memorandum

- Oneida Business Committee To:
- Kelly Skenandore, Health Information Systems Specialist CC: Debbie Danforth, Comprehensive Health Division Manager
- Concurrent with mundation Jeff Carlson, Comprehensive Health Division Business Operations Manager From:
- Date: 2/23/2015

TribalNe Soa ce Approval by OBC Acce

OF C Meinbers, Shek

As a 20 p its year employee in the One of a Comprehensive Health Division, I would like to request your surfact a proval for Kelly L. Seenando to accept another year as a Board Member to the TribalNet Conference. The TribalNet Conference is in its and has become a very successful industry organize are information Technology resource for it bal communities in all of Indian Country nationw

As a TribalNet Board fember, Kely will be expected to predicipation h conference calls with other tribe contributors from the board to develop a tent for the ThoalN conference. Content will be specific to Kelly's field of Tribal Health, key may as lend ideas to that to be used and the over ment or General IT tracks of the conference's agenda.

The TribalNet 16th Annual Conference is selectuled for Monday November 2, 2015 hrough Thursday November 5, 2015 and is being held this pear in Austin TX. As a Triba let Board Member, Kelly will be requested to attend the conference and partake as either moderato, and/or presenter for 1-3 agenda track items of the conference.

She is very honored to be a 5 year TribalNet Board Member and regresent our Oneida Nation in the areas of health & government while serving on this toar it, with great enthusiasm that I ask for you to support and approve her continued efforts to service on the Tribal Conference, once again as a TribalNet Board Member.

Thank you & Respectfully,

Jeff Carlson, Comprehensive Health Division Business Operations Manager

TribalNet 16th Annual Conference- 11/2/15-11/5/15 – Renaissance Austin, Austin, TX

"Bringing Technology & Tribes Together" - www.tribalnetonline.com

Page 296 of 340

TribalNet

Advisory Board Member Duties and Calendar- 2015

What are the duties of an advisory board member?

To make recommendations and/or provide key information for the Annual TribalNet Conference.

This is typically fulfilled through the following actions

- Participate in pre-scheduled conference calls to collaboratively develop content for the TribalNet conference.
- Attend the TribalNet conference as a moderator and/or presenter (FREE conference registration).
- Partake in reviewing and deciding on the recipient of the Annual Tribal Technology Leadership Award.
- Board member bios and pictures will be included on the TribalNet website and related marketing materials.
- In 2015, board members will be asked to create session summaries and/or develop key learnings for up to 3 sessions on the conference agenda that fall into their area of expertise and influence.

Advisory Board Member Structure:

Board Chairman/Director- This position we over see the direction of both boards- 2015- Michael Day Board Coordinator ans esition will have be converted to meetings and related materials- 2015- Shannon Bouschor Gaming & Hospinity Boar

Consists of selered members with a background attered in the gaming and hospitality.

-A minimum of Available member seats, terms are 1, ea. with renewal by invitation

Tribal Government,alth Brand

Consists of selected menuber with a background/interest in tribal sover ment and/or membership and health services. -A minimum of 7 available menuber seattherms are 1 year with renewal by invitation Industry Liaison Committee-

TribalNet may appoint up to 3 individual as Liabons to the tribal technology in estry for this committee. These individuals are appointed volunteers and will not be required to participate in board meeting calls, however may be asked to join on some calls for input. Committee members will not take part in the moustry award decising, and like board members, will not be responsible for making any fiduciary decisions.

> Board members and committee members are appoint a volunteers, will be reportsible for any fiduciary decisions and may here or be smove. From their position of any ime.

Tentative Meeting Calendar:

Board Meeting- 4/30/15: Introduction of 2015 board, Cerall idease eview '14 feedback confirm yearly eeting calendar Board Meeting- 5/21/15: Potential topics, speakers, tracks key tes, workshops, themes, in ustry war process review Board Meeting- 6/25/15: More session/topic/speaker brainstern, ratings evolutions decided, two of inference week ideas Board Meeting- 7/16/15: Drill down on sessions and speaker idea 50 agenda Board Meeting- 8/6/15: Drill down on sessions and speaker ideas- 75 agenda Board Meeting-9/17/15: 90-100% Agenda- session summaries finalized Board Meeting- 10/15/15: Confirm all moderating sheets and checklist, deck on for houstry ward finalized Annual Conference- 11/2/15-11/5/15: Onsite participation-Austin, TX

We realize that there is potential for meeting conflicts, however we simply \mathbf{k} that by \mathbf{k} cepting the volunteer appointment that board and committee members make their best effort to participation and efform the duties as requested. You have been invited to serve as an advisory board member because of your valued industry knowledge and experience.

Please select if you accept/decline the invitation to serve in 2015- DUE 3/30/15

I am honored to participate on the 2015 TribalNet Advisory Board! :)

Thank you, but I decline the invitation to participate on the 2015 TribalNet Advisory Board :(

kingene Date: 2 Name: Signature:



Page 297 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 3 / 11 / 15						
2.	Nature of request Session: 🗵 Open 🛛 Executive - justification required. See instructions for the applicable laws that						
	define what is considered "executive" information, then choose from the list:						
	N/A						
	Agenda Header (choose one): New Business/Request						
	Agenda item title (see instructions):						
	Dissertation Research Review – Cotteell "The Enduring Alterity of American Indians in German Museums"						
	Action requested (choose one)						
	Action - please, escribe:						
	Motion a support the dissertation recearce by Courtney Cottrell ternatively titled "The Enduring Alterity of American Indians i German Museums" and tern quest that a copy of the approved dissertation be made available to the Oneida Nation Museum and to the Oneida Community Library.						
3.	3. Justification						
	Why BC action is required seconstructions):						
	Community support letters are typically required of Internal Review Braids of Universities when dissertation research includes Tribil communates						
4.	Supporting Materials Image: Memo of explanation with required information (see instructions)						
	□ Report □ Resolution □ Contract (check the box below if signatine required)						
	☑ Other - please list (Note: multi-media presentation, due to Trike) Clerk 2 days prior to meeting)						
	1. Draft Correspondence 3.						
	2. 4.						
	Business Committee signature required						
5.							
	Authorized sponsor (choose one): Jo Ann House, Chief Counsel						
	Requestor (if different from above):						
	Name, Title / Dept. or Tribal Member						
	Additional signature (as needed): Name, Title / Dept.						
	Additional signature (as needed):						
	Name, Title / Dept.						

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Page 298 of 340

JO ANNE HOUSE, PHD CHIEF COUNSEL JAMES R. BITTORF DEPUTY CHIEF COUNSEL REBECCA M. WEBSTER, PHD SENIOR STAFF ATTORNEY **ONEIDA LAW OFFICE**

N7210 SEMINARY ROAD P.O. BOX 109 ONEIDA, WISCONSIN 54155 PATRICIA M. STEVENS GARVEY CAROYL J. LONG KELLY M. MCANDREWS MICHELLE L. MAYS

(920) 869-4327

FAX (920) 869-4065

MEMORANDUM

TO: Oneida Business Committee

FROM: Jo Anne House, Chief Counsel

DATE: March 5, 2015

SUBJECT: Dissertation Lessurch Review × Cottrell ÙThe Enduring Alterity of American adians in German Mass amsU

Courteev Cottrell is a doctoral caudidate at the University of Michigan. Her dissertation proposal looks at Comman masseums and Triba museums projentation of Tribal contemporary and historical culture. The dissertation focus is on itentifying whether museums have continued the propensity to view datives merican culture to having a historical and non-contemporary existence. The conflusion of this dissertation could result hermodifications or improvements to museum management and resultation of Native American culture, and more specifically a third party review of the Tribel adsertation could result.

Ms. Cottrell proposes interviews of Tribal employees, a visit to the truseum, photographic documentation of the museum and exhibit. This work would be repeated on at least one other Tribal or tribal related museum as well-as suscents in Germany.

A review of the proposal for the dissertation identifies no financial or confidential information of the Tribe will be released. Further, the interviews of Tribe employees to not effect the youth or elders as a protected population under NIH research gradelines. However, terrically Tribal governments are requested to provide community support to there types of research projects under those same guidelines.

I have included a draft community support letter for Ms. Cattrell is include with her application to the University's Internal Review Board as supporting and authorizing heresearch.

If you have further questions, please contact me.

Oneida Tribe of Indians of Wisconsin

Post Office Box 365



Oneidas bringing several hundred bags of corn to Washington's starving army at Valley Forge, after the colonists had consistently refused to aid them.



Oneida, Wi 54155



UGWA DEMOLUM YATEHE Because of the help of this Oneida Chief in cementing a friendship between the six nations and the colony of Pennsylvania, a new nation, the United States was made possble.

March 5, 2015

Courtney Cottrell Doctoral cancilate University of Michigan Anthropology Department Via e-meil en ccottract emich.edu

RE: Dissertation Research neview – Convellence Encyring Alterity of American Indians in German Museums

Dear Ms. Cottrell:

The Oneida Business Committee has reviewed your request for a community support letter and approval of the research project involving the Oneida Nation Museum and interviewing Tribal employees. We understand the goar of your research project is to identify trends in museum exhibits regarding Native American culture, which portray a non-common approximation of tribes and ongoing tribal communities. You have also indicated that you will provide a review of the museums included in the study and a discussion of how museums can recognize this trend in an attempt to avoid incorporating it into ongoing and future exhibits.

We believe that this study will have a positive impact on Indian country, and by including the Oneida Nation Museum, can help the Oneida Tribe of Indian of We possin to recognize where we can make improvements in our own museum exhibits. We also believe that this study can result in a positive impact on museum management across the United states as well as abroad.

On March ____, 2025, the Oneida Business Committee, the elected government of the Oneida Tribe of Indians of Wisconsin, met in regular session and approved the following motion. Motion to support the dissertation research by Courtney Cottrell tentatively titled UThe Enduring Alterity of American Indians in German MuseumsUand to request that a copy the approved dissertation be made available to the Oneida Nation Museum and to the Oneida Community Library.

Please note, our support and authorization of this research project requires that the Oneida Law Office review the draft dissertation in order to identify errors or unclear references to the Oneida Tribe of Indians of Wisconsin. Our past and current government is a long and complex

relationship with the federal government, Tribal governments and the surrounding communities. This review will not reflect upon the research, findings or recommendations made in the dissertation as those are purely the responsibility of yourself.

We look forward to seeing the end result of your research and wish you strength in your educational endeavors. If you need further assistance, please contact me.

Sincerely,



Page 301 of 340 Oneida Business Committee Meeting Agenda Request Form

 2. Nature of request Session: ⊠ Open ☐ Executive - justification required. See instructions for the applicable laws define what is considered "executive" information, then choose from the I 	
Agenda Header (choose one): Follow Up Agenda item title (see instructions): Accept Self-Funded Health Insurance Rate Financial Impact 1st Quarter Report Action requested (choose one) Information e dy Action-please rescribe:	st:
Agenda item title (see instructions): Accept Self-Funded Health Insurance Rate Financial Impact 1st Quarter Report Action requested (choose one) Information only Action replease rescribe: 3. Justification Why BC action is required iscentistructions): Follow-up nell bee handout Requested reporting per BC meeting 10/22/14 4. Supporting Materials Memo of explanation with required information (see instructions) X Report Resolution Contract (check in e box below if Supartine required) Other - please list (Note: multi-media presentation due to Tribh Clerk 2 days prior to meeting	
Agenda item title (see instructions): Accept Self-Funded Health Insurance Rate Financial Impact 1st Quarter Report Action requested (choose one) Information only Action replease rescribe: 3. Justification Why BC action is required iscentistructions): Follow-up nell bee handout Requested reporting per BC meeting 10/22/14 4. Supporting Materials Memo of explanation with required information (see instructions) X Report Resolution Contract (check in e box below if Supartine required) Other - please list (Note: multi-media presentation due to Tribh Clerk 2 days prior to meeting	
Accept Self-Funded Health Insurance Rate Financial Impact 1st Quarter Report Action requested (choose one) Information only Action replease rescribe:	
Accept Self-Funded Health Insurance Rate Financial Impact 1st Quarter Report Action requested (choose one) Information only Action replease rescribe:	
Action requested (choose one) ☑ Information only △ Action-please rescribe: ☑ 3. Justification Why BC action is required accountstructions): Follow-up toll beta handout Requested reporting per BC meeting 10/22/14 4. Supporting Materials □ Memo of explanation with required information (see instructions) ☑ Report □ Resolution □ Other - please list (Note: multi-media presentation due to Trikel Clerk 2 days prior to meeting	
Action-please rescribe:	
3. Justification Why BC action is required beginstructions): Follow-up of bea handout Requested reporting per BC meeting 10/22/14 4. Supporting Materials Memo of explanation with required information (see instructions) X Report Resolution Other - please list (Note: multi-media presentation due to Trible Clerk 2 days prior to meeting	
Why BC action is required secunstructions): Follow-up will be handout Requested reporting per BC meeting 10/22/14 4. Supporting Materials Imstruction Memo of explanation with required information (see instructions) Imstruction Imstruct	
Requested reporting per BC meeting 10/22/14	
4. Supporting Materials Instru □ Memo of explanation with required information (see instructions) Instru ☑ Report □ Resolution □ Contract (check the box below if signature required) □ Other - please list (Note: multi-media presentation due to Trikel Clerk 2 days prior to meeting)	
 Memo of explanation with required information (see instructions) Report Resolution Contract (check the box below if signature required) Other - please list (Note: multi-media presentations due to Trikel Clerk 2 days prior to meeting 	
 Report Resolution Contract (check the box below if signature required) Other - please list (Note: multi-media presentations due to Trikel Clerk 2 days prior to meeting 	tions
1.	
2. 4.	
Business Committee signature required	
5. Submission Authorization	
Authorized sponsor (choose one):	
Requestor (if different from above): Larry Barton, Chief Financial Officer]
Name, Title / Dept. or Tribal Member	
Additional signature (as needed): Name, Title / Dept.	
Additional signature (as needed): Name, Title / Dept.	

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Oneida Business Committee Meeting **Agenda Request Form**

1. Meeting Date Requested: 03 / 11 / 15

2. Nature of request

Session: 🖾 Open 🛛 Executive - justification required. See instructions for the applicable laws that

define what is considered "executive" information, then choose from the list:

Agenda Header (choose one): Travel Report

	Agenda item title (see instruction	s):
	Travel Report -BYS-	
	Action requested (choose on ,	
	Information only	
	🖾 Action - please describe:	
	Approve 22 Approve rep	ort
•		
3.	Justification	
	Why BC action is required a e	instantions):
4.	Supporting Materials	Instructions
	\boxtimes Memo of explanation with rec	uired information (see instructions)
	Report Resolution	Contract (check the box below if signature required)
	☐ Other - please list (Note: multi	-media presentation rate to Tribr Clerk 2 days prior to meeting)
	1.	3.
	2.	4.
	Business Committee signatur	e required
5.	Submission Authorization	
	Authorized sponsor (choose one):	Brandon Stevens, Council Member
	Requestor (if different from above):	
		Name, Title / Dept. or Tribal Member
	Additional signature (as needed):	Name, Title / Dept.
	Additional signature (as needed):	
	U	Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Page 303 of 340



ONEIDA TRIBE OF INDIANS OF WISCONSIN PO BOX 365 ONEIDA, WI 54155 THE OFFICE OF: COUNCILMAN STEVENS OFFICE: 920-869-4378 EMAIL: BSTEVENS@ONEIDANATION.ORG RMETOXE2@ONEIDANATION.ORG

MEMO

TO:	ONEIDA BUSINESS COMMITTEE
FROM:	COUNCILMAN BRANDON STEVENS
SUBJECT:	TRAVEL REPORT -BYS NCAI-
DATE:	MARCH 11 TH , 2015

National Congress of Arter Lan Indians (NCAI) 10/27-31/2015

During the NCAI conference or as able to obtain a meeting with the acting Chairman of the National Indian Gamma computerion (NIGC), Jonodev Chaudhuri on Tuesday October 28, 2014. Also in attervance was NIGC staff attorney John Hay. The purpose of non-eeting was to discuss the receive change in reporting requirements of Oneida's financial qudits. NIGC is requering that one la submit individual consolidated financial statements are each aming location. Convertly Oneida submits one financial statement with accompanying revenue schedule for each location. I shared two concerns; the undue financial but len that the propose request for consolidated financial statements for each location would inport to the tribe and the imposition the proposed request would have on the way Cheida manages their operation. I shared that the proposed request would alter the way Oreida manages because Oneide would have to make changes to their current way of managing to be acte to track and report the specific costs that NIGC is requesting. I also suggested that crease hable allocation firmed a could be used by Oneida with minimal cost and intrusion in to Oneida's management as e way to resolve the issue.

Overall, this meeting was much more productive that the prior phone conferrace with NIGC staff. In the previous phone conference with One ida, NIGC staff was rigid about the regulation and did not seem willing to work with One ida or understand the possible burdens the change would impose on One ida. At this meeting, Chairman Characteri expressed a willingness to work with One ida to resolve the issue and understood the possible burdens not could be imposed on One ida. NIGC will review One ida's suggestion of using a reason of e all cation formula and will send formal correspondence stating NIGC's position.

Page 304 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15
2.	Nature of request Session: 🗵 Open 🔲 Executive - justification required. See instructions for the applicable laws that
	define what is considered "executive" information, then choose from the list:
	Agenda Header (choose one): Travel Request
	Agenda item title (see instructions):
	CACIC Conference - Carlton MN - Arx, 27-30, 2015
	Action requested (choose one)
	Action - please escribe:
	Approve to travel contest for Lisa commers to attend the Crimes Against Children in Indian Country
	(CACIC) Certer Incer- Carolton, MN April 27, 0, 2015
3.	Justification
	Why BC action is required seconstructions):
	Business Committee travel
4.	Supporting Materials
	Instructions I Memo of explanation with required information (see instructions)
	□ Report □ Resolution □ Contract (check the box below if signatione required)
	☑ Other - please list (Note: multi-media presentation due to Tritch Clerk 2 days prior to meeting)
	1. Travel Authorization 3.
	2. 2015 CACIC Agenda 4.
	Business Committee signature required
5.	Submission Authorization
	Authorized sponsor (choose one): Lisa Summers, Tribal Secretary
	Requestor (if different from above):
	Name, Title / Dept. or Tribal Member
	Additional signature (as needed): Name, Title / Dept.
	Additional signature (as needed):
	Name, Title / Dept.

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

INTEROFFICE MEMORANDUM

TO:ONEIDA BUSINESS COMMITTEEFROM:LISA SUMMERS, TRIBAL SECRETARYSUBJECT:TRAVEL REQUEST – CACIC CONFERENCEDATE:MARCH 3, 2015

This memorandum serves as request to attend the Crimes Against Children in Indian Country (CACIC) Conference in Carlton, MN.

From the conference website (w ...ncjtc.org/cacic):

"Notice American your are impact to y unique and challenging threats that affect their physical and notional development and rel'oeur Ofter tribal communities do not have the same access to the necessary resources for addressing these threas. The Crimes Against Children in Indian Country (CACIC) Sonference was formed in response to be often unmet needs of Native youth. The advisory/planning committee is and up of representations from a wide wave of tribal and non-tribal law enforcement, government, one social service agencies. In implement goal of the conference is to strengthen relationships between variou agences tribes, and states, to promote a multi-disciplinary, multi-jurisdictional approach to serving Nation young pupple."

As liaison to both the One le Chile 1 resective Board and the One in Police Commission, this travel fits with those responsibilities.

Event Name: Location: Dates: Estmated Cost: Cimes Against Children in Indian Country (CACIC) Conference Carlton, Ma April 23, 30, 2015 \$667.73

Requested Action

1. Approve the travel request for Lisa Summers to attend the brimes Against Children in Indian Country (CACIC) Conference – Carolton MN/ April 7-30, 2015

ONEIDA TRIBE OF INDIANS OF WISCONSIN TRAVEL AUTHORIZATION REQUEST

General Travel Information

Name of Traveler	Lisa Summers				
Please list name as it appears on Travelers Driver's License or WI State ID					
	Employee # Date of Birth				
Destination	Carlton, MN				
Departure date	April 27, 2015 Return date April 30, 2015				
Purpose of travel	Crimes Against Children in Indian Country Conference				
Charged GL Account	001-4272000-004-701000-000				

GSA Rate Information for the destination

s 46.00	Lodaina	rate ner dav	/ \$	83.00	
Ψ	Lodging		γ Ψ]
			Dave/		
	Rate	Factor	Miles	Total	
	46.00	75 %	1	\$ 34.50	
	_{\$} 46.00	100 %	2	\$ 92.00	
	_{\$} 46.0	75 %	1	\$ 34.50	
				\$	
taxes ^{6.875%}	\$ 52.00		3	\$ 166.73	taxes
	\$			\$	
	\$			\$	
or Tribal Vehice	\$ 115.00			\$ 115.00	
	¢			\$	
Yes / No	225.00			_{\$} 225.00	
Total Cost Estim	nate re-opr	ov. is rec	ured	\$ 200 00-	
	_			\$667.73	
	v taxes ⁷ 6.875% or Tribal Vehice	Rate 46.00 \$ 46.00 \$ 46.00 \$ 46.00 \$ 46.00 \$ 52.00 \$ 52.00 \$ 115.00 \$ 115.00 \$ 225.00 Total Cost Estimate re-tope	Rate Factor 46.00 75 % \$46.00 100 % \$46.00 75 % \$46.00 75 % \$46.00 75 % \$46.00 75 % \$46.00 75 % \$52.00 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500 \$ 500	Rate Factor Days/ Miles 46.00 75 % 1 \$46.00 100 % 2 \$46.00 100 % 2 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$46.00 75 % 1 \$52.00 3 3 \$ \$ 5 \$ \$ 5 \$ \$ 5 \$ \$ 5 \$ \$ 5 \$ \$ 5 \$ \$ 5 \$ \$ 5 \$ \$ 5 \$ 5 5	Rate Factor Days/ Miles Total 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 1 \$ 34.50 \$ 46.00 75 % 3 \$ 166.73 \$ 52.00 \$ \$ \$ \$ 115.00 \$ 115.00 \$ 115.00 \$ 225.00 \$ 225.00 \$ 225.00 \$ 225.00 \$ 2200.00=

I understand this advance will be deducted from my claim for reimbursement of actual travel expenses. I also understand that if this advance in not cleared within 10 calendar days after my travel return date, I shall be held responsible for the full amount advanced and that I may be reprimanded in accordance with the Personal Policies and Procedures for my failure to clear this matter within the time so allotted. Further, in the event that the advance payment is not cleared within 10 calendar days of my return, and I have not filed a formal written dispute as to the amount due, by signing below, I am hereby making a knowing and voluntary wage deduction for the entire amount outstanding.

Signatures / Approvals

	Signature	Date	Contact Phone #
Traveler			
Program Director			Not needed
General Manager			Not needed

axes included



April 28-30, 2015 | Carlton, Minnesota

10th Annual Multi-Disciplinary Approaches to Prevent **Crimes Against Children in Indian Country**

8:00 aı		Tuesday, April 3		
	m – 4:30 pm	Conference Registration and Information	esk Open	Otter Creek Ballroom
8:30 aı	m – 9:00 am	Opening Ceremony Mistress of Ceremonies ~ Janell Rasmussen ~ Director, Criminal Justice Training & Education Bureau of Criminal Apprehension Minnesota Department of Public Safety Fond du Lac Band of Lake Cerior Chippewa Presentation of Colore Prum ~ Traditional Prayer	9:00 am – 9:20 am	 Welcoming Remarks Karen Diver ~ Chairwoman, Fond du Lac Band of Lake Superior Chippewa Andrew Luger ~ United States Attorney, State of Minnesota (Invited) Mark Dayton ~ Governor, State of Minnesota (Invited)
9:20 ar	m – 9:30 ar	Break		
9:30 ar	m – 10:20 m	Chil. ren Exposed to Violance a ceature a Clin David Raasch ~ Chief Judge (retired), Pockbridge-M This man will discuss the new to reinvig that the off pros. cultures that need to be rescaded, nurture a we Protecting our York, by Preserving our Cultures	unsee Community, Band tures of Native America, cu	Itures that have lain dormant for hundreds of
10:20 a	am – 10:30 am	Break		
10:30 a	am – 11:20 pm	Workshor #1		
A	Steve Del Negi This presenta technology u pornography Synthetic D	ing & Investigator Chile ron ography o ~ Sergeant (retiree), Dignal Evidence Media Section/I tion will provide prosecutor and investigators with an unders sed, the scope of the problem as it pertains to the ternet, the offender will also be discussed and rugs ~ Forensic Scientist, Minnesota Bungau of Cryminals of	tanding of and port ap e impact nat it he one vi 	bhy and the effects of Child sexual abuse. The
с	Session Summary Sex Trafficking – Safe Harbors David Pinto ~ Assistant Ramsey County Atty & Director, Safe Nerbor Training and Protocol Development, MN House of Representatives Session Summary			
11.20 ;	am – 11:30 am	Break		
	12.20 pm	Workshop #2		
	Steve Del Negi This presenta technology u	ing & Investigating Child Pornography (contin o ~ Sergeant (retired), Digital Evidence Media Section/I tion will provide prosecutors and investigators with an unders sed, the scope of the problem as it pertains to the internet, the offender will also be discussed	CAC Task Force of assact tanding of child poor ograp e impact that it has on ovin	by an the effects of Child sexual abuse. The and the risk levels associated with the child
11:30 a	Understand Steve Del Negu This presenta technology u pornography Synthetic Di	ing & Investigating Child Pornography (contin to ~ Sergeant (retired), Digital Evidence Media Section/I tion will provide prosecutors and investigators with an unders sed, the scope of the problem as it pertains to the internet, the offender will also be discussed. rugs (continued) ~ Forensic Scientist, Minnesota Bureau of Criminal App	CAC Task Force chassac tanding of child polyograp e impact that it has on win	chuset State Police why and the effects of Child sexual abuse. The child and the risk levels associated with the child

	Tuesday, April 3	28, 2015 (con't)
1:30 pi	n – 2:20 pm Workshop #3	
Α	Threat Assessments BCA Fusion Center ~ Session Summary	Otter Creek Ballroom
В	Drugs – Case Study Amy Granlund ~ Forensic Scientist, Minnesota Bureau of Crimina Session Summary	Fond du Lac Creek Hall al Apprehension
с	engaging and elingency. Consequently, many south we a status in the current research egarding this unique copulation, and so iss bo of measurable outcodes. Specifically, the session will be high bottra	
2:20 pr	n – 2:30 pm <i>Break</i>	\wedge
2:30 pi	n – 3:20 pm Workship #4	
А	Threat Assessments (communit) BCA Fusion Center ~ Session Summary	Otter Creek Ballroom
В	Drugs – Case Study (continued) Amy Granlund ~ Forensic Scientist, Minnesota Burgar Scrimin Session Summary	Fond du Lac Creek Hall
с	Dual Status Youth Project (continued) Elizabeth Raile ~ Intervention Program Specialist, Beltrami Area Dual Status Youth: Working Together to Improve Outcomes for Youth Research has shown that children with a history (or present) of Child B engaging in delinquency. Consequently, many youth have a status in the current research regarding this unique population, and discuss bo of measurable outcomes. Specifically, the session will highlight Beltra already proven to positively impact youth and families, as well as prof	Revesented in Multice Systems Protection/Child Worare Systems involvement are at an increased risk of both the social convession of venile justice systems. This session will look at the national approach forts to increase engagement and improve a number mi County's shally-Involved Your register, which though in its infancy, has
3:20 pr	n – 3:30 pm <i>Break</i>	
	m – 4:20 pm Tribal Youth Police Academy – Reachin Warren Warrington ~ Master Sergeant, Menon The Tribal Youth Police Academy (TYPA) provided and other criminal justice careers. Classroom and serve as mentors, and answer questions about cri	ninee Tribal Police Department, Menominee Tribe of Wisconsin an excellent opportunity for Native American youth to explore law enforcement hands-on learning connected students and practitioners. These professionals minal justice careers. Students participated in a large scale police scenario ug to end. Academy students heard from tribal leaders and tribal police officers
5:00 pr	m – 7:00 pm Reception	Otter Creek Ballroom
Menor Traditi	r of Ceremonies ~ Warren Warrington ~ Master Sergeant, ninee Tribal Police Department, Menominee Tribe of Wisconsin onal Prayer Youth Police Academy Slide Show	<u>South of the Border Buffet includes:</u> ~ Flour Tortillas and Crispy Corn Taco Shells ~ Seasoned Chicken & Beef Spanish Rice ~ Freshly Baked Corn Bread ~ Fry Bread

		Wednesday, April 29, 2015	
8:30 ar	m – 5:00 pm	Conference Registration and Information Desk Open	Outside Otter Creek Ballroom
	m – 10:50 am	Native Mob Case Study and Recruiting Juveniles into Gangs <i>Ricky Wuori</i> ~ Special Agent, Minnesota Bureau of Criminal Apprehension <i>Jerry Wilhelmy</i> ~ Investigator, Minnesota Department of Corrections This session will cover the history of how and why Native gangs originated in Minnesota gangs. Specifically the instructors will address how juveniles are targeted by gangs, the and touch on the newer gangs that are being established in Minnesota. The session will Native Mob and how laws surrounding Racketeer Influenced and Corrupt Organizations Attendees will be taken from arrest, to trial and final outcomes. The session will conclue Native Mob as a result of this case.	hierarchy of different gangs in Minnesota II also include a recent case involving the 5 Act (RICO) brought down the gang.
	am – 11:00 am	Break	
11:00 a	am – 11:50 am	Workshop #5	
Α	Technology Karina Hedinge Session Summ	r ~ Training and Education Coordinator, Bureau of Criminal Apprehension, Minne ary	Otter Creek Ballroom sota Department of Public Safety
В	Heroin U a Bryan Koncelic This sussion w	Ind Leafficking in the Michael Spect Agent, Division of Criminal Investage ion, Wisconsin Department of Justic ill provide a basic overview of hervin, which includes astruction on the different types of h didwest, treatment options, safety concerns, is public safety personal, and heroin overdose	eroin, the effects on the user, its prevalence
С	Commercial Patina Park ~ E In August 2009 Native Americ affects the per	Victimes Volue: Combating Divenile Sex Tradicking and Sexual in Joitation in the Native American Population xecutive Director of these and an Woman's Resource Center 9, the Minu Sotan diate Vomen's Resource Center released the Shattend Hearts Report in an population oppresented by their clients. This work hop will identify the oppact of historic ristence of commercial sector exploitation/trafficking is our youth and how to work with cl erability to exploitation	
11:50 a	am – 1:00 pm	Lunch (On Your Own	
1:00 pi	m – 1:50 pm	Workshop #6	
Α	Technology (Karina Hedinge Session Summ	$m{r}$ ~ Training and Education Coordinator, Burrau of Criminal Apprehension $m{M}$, me	Otter Creek Ballroom second artment of Public Safety
В	Bryan Kastelic ⁻ This session w	nd Trafficking in the Midwest (continued) Special Agent, Division of Criminal Investigation, Wisconsty Department of Just Ill provide a basic overview of heroin, which includes instruction on the different types of he use in the Midwest, treatment options, safety concerns for upbic safety personal, and hero	Fond du Lac Creek Hall erony, the effects on the user, its bin overdose death investigations.
с	Commercial Patina Park ~ E In August 2009 Native Americ affects the per	Victim's Voice: Combating Juvenile Sex Transking are (consued) Sexual Exploitation in the Native American Population executive Director, Minnesota Indian Women's Resource Center 9, the Minnesota Indian Women's Resource Center released the Shatten Hearts Report in an population represented by their clients. This workshop will identify the repact unistorier resistence of commercial sexual exploitation/trafficking in our youth and how to bork with cl erability to exploitation.	on ting the impact of trafficking in the cal turns on the population in how it
1:50 pi	m – 2:00 pm	Break	

	Wednesday, April 29, 2015 (con't)				
2:00 pi	m – 2:50 pm Workshop #7				
Α	 How Much does your phone know about you? Lee Reed ~ Officer (Retired), City of Abilene Police Department; Consultant, Team Adam This session will discuss the operation of a cellphone/smartphone and to learn the proper procedure in obtaining the necessar much does your phone know about you? 	Otter Creek Ballroom y information. How			
В	Native American Gangs and their Connections to National Gangs Fond du Lac Creek Hall Bryan Kastelic ~ Special Agent, Division of Criminal Investigation, Wisconsin Department of Justice Richard Van Boxtel ~ Chief of Police, Oneida Tribal Police Department, Oneida Tribe of Wisconsin Since 2007, the Native American Drug and Gang Initiative (NADGI) Task Force has addressed the crime associated with gang and drug activity in the Wisconsin Native American communities. The NADGI covers 9 10 Tribal Reservations spanning most of the State of Wisconsin. Although the gangs on the Reservations hold many elements associated with Native American culture, they hold strong ties to larger gangs in urban areas. These affiliations have led to a steady trade in drugs, meapons and gang violence human trafficking. The presentation will provide a look into the unique character of Native American gangs in Wiscon and their relationships with other gangs in the large metropolitan areas. Past and present criminal investigations will be referenced to illustructure tase points and also offer strategies that have had success in combating this activity. NADGI has formed many partnerships throughout ascers in with Eederal, State, County, local, and Tribal partners to include programming such as Drug Endangers and (DEC).				
с	Construction Stoneybrook Hall Cary Wa banascum Project Specialist, National Criterinal Justice Training Center of Fox Valley Technical College This set for explores the impact of historical and introgene otional hauma on suicide in Native American/Alaska Native communities. We will discuss signature trisk for use cluding substance to use, prior spicide at empts, mood and anxiety disorders and access to lethal means. Participants will be a low to reduce the risk of suicide by eliminating may be element of the Deadly Triad - alcohol, firearms and distress.				
2:50 pr	m – 3:00 pm Break				
3:00 pi	m – 3:50 pm Work: op 13				
Α	How Much does your phone know about you? (continued) Lee Reed ~ Officer (Retired), City of fullence once a partment; Consultant, Team Ada This session will discuss the operation of a sulphone/smitphone and to leave the proper vocedure in obtaining the necessar much does your phone know about you?	Otter Creek Ballroom y information. How			
В	Native American Gangs and their connections to national Gangs (Initianed) Fond du Lac Creek Ha Bryan Kastelic ~ Special Agent, Division of Criminal westigation, Wisconsin Department of Junce Richard Van Boxtel ~ Chief of Police, Oneida Triba Police Department, Oneida Tribe of Venonsin Since 2007, the Native American Drug and Gang Initiative (NADGI) Take orce has addressed the arime a sociated wite gang and drug activity in the Wisconsin Native American communities. The NADGI cover 9 10 Total Reservations spanning most of the State of Wicconsin. Although the gangs on the Reservations hold many elements associated with Native American culture wey hold strong ties o large mangs in urban areas. These affiliations have led to a steady trade in drugs, weapons and gang tiolence hung a trafficking. The presenction of provide a look into the unique character of Native American gangs in Wisconsin and their relations as with ther gangs in the large metropowan areas. Past and present criminal investigations will be referenced to illustrate these points and also offer to ategies that have bed success in combating this activity. NADGI has formed many partnerships throughout Wisconsin with Federal, State, Courte, local, and a local partners to include programming such as Drug Endangered Children (DEC).				
с	C Stoneybrook H Cary Waubanascum ~ Project Specialist, National Criminal Justice Training Centry of Fox Vally, Technical College This session examines suicide myths and facts, warning signs and a 3-step emergency response to ecopie in the sist. Participants will learn how to apply the QPR (Question, Persuade, Refer) method, an evidence-based life-saving technique.				
3:50 pr	m – 4:00 pm Break				
4:00 pı	 m – 4:50 pm Using Social Media in Crimes Against Children Investigations Jerry Jones ~ Consultant, National White Collar Crime Center, Portland Police Department (Retired Social Media is an invaluable intelligence tool to assist with crimes against children investigations. In this see the usage of several social media mining tools that search multiple sites, harnessing the power of social media mining tools are that are free and attendees can start using them immore the tools are that are free and attendees can start using them immore the search multiple sites. 	ession we demonstrate dia for investigations.			
5:00 pr	m Dinner (On Your Own)				

	Thursday, April 30, 2015	
8:30 am – 12:00 pm	Conference Registration and Information Desk Open	Otter Creek Ballroor
9:00 am – 9:50 am	 The Path of the Native American Runaway Child Lee Reed ~ Officer (Retired), City of Abilene Police Department; Consultant, Team Adam This program will give an overview of the runaway issue in America and Indian Country. It will also Ide and the personality traits that a chronic runaway will exhibit and effective ways to use intervention in 	
9:50 am – 10:00 am	Break	
10:00 am – 10:50 am	Child Sex Trafficking and Exploitation in Indian Country Jim Walters ~ Program Administrator-Amber, National Criminal Justice Training Center of Fox Human trafficking and exploitation is one of the fastest growing forms of victimization facing tribal con- technology, man camps associated with natural resources and increase in tribally operated casinos and contributed to the growth of this problem. This course will provide the participant with the background tribal communities in the United States as well as issues of generational trauma which facilitate contin Participants will learn about the use of technology in victimization as well as investigative and prevent children in Indian Count	mmunities. The expansion of d travel plazas have all nd of sexual exploitation in nued victimization.
10:50 am – 11:00 am	Preak	
11:00 am – 11:50 am	Cas Study – Bobler ao's Surney Home Grat Snyder ~ Sergeant, Criminal Levestiga fon Division, Minneapolis Police Department Session Summary	Otter Creek Ballroom
11:50 am – 12:00 pm	Closin ceremony and Conference coclusio. Mistrix of five conies ~ Janell Rasmissen ~ Greector, wiminal Justice Training & Educati Bureau of ariminal Approximation, Minnesota Department on Public Safety Fond du Lac Bald of take Superior Chippew Retire Colors ~ Marm ~ Martine Prayer	Otter Creek Ballroon

Page 312 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15	
2.	Nature of request Session: 🛛 Open 🛛 Executive - justification required. See instructions for the applicable laws that	
	define what is considered "executive" information, then choose from the list:	
	Agenda Header (choose one): Report	
	Agenda item title (see instructions):	
	Organizational Development Specialist 1st Quarter Report	
	Action requested (choose one)	
	Action - please escribe:	
	Desweet of March 25, 2015 marchine the week of March 0.12	
	Request determine Narch 25, 2015 a sendardue or pacation the week of March 9-13	
3.	Justification	
	Why BC action is required seconstructions):	
	Regular quarterly reporting.	
4	Supporting Materials	
4.	Supporting Materials Memo of explanation with required information (see instructions)	
	□ Report □ Resolution □ Contract (check is box below if signative required)	
	Other - please list (Note: multi-media presentation due to Trike) Clerk 2 days prior to meeting)	
	1. 3.	
	2. 4.	
	Business Committee signature required	
5.	Submission Authorization	
	Authorized sponsor (choose one):	
	Requestor (if different from above): Melanie Burkhart, Organizational Development Specialist	
	Additional signature (as peeded):	
	Additional signature (as needed): Name, Title / Dept.	
	Additional signature (as needed):	
	Name, Title / Dept.	

- Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Oneida Business Committee Meeting Agenda Request Form

- 03/11/15 1. Meeting Date Requested: 73 / 10 / 15 2. Nature of request Executive - justification required. See instructions for the applicable laws that Session: X Open define what is considered "executive" information, then choose from the list: Quarterly Report Agenda Header (choose one): Agenda item title (see instructions): SG Quart Action requested (choose on X Information only Action - please describe: 3. Justification Why BC action is required se instructions): Requested 4. Supporting Materials Instructions Memo of explanation with required information (see instructions) X Report Resolution Contract (check the box below if supative required) Other - please list (Note: multi-media presentation due to Tribal Clerk 2 days prior to meeting) 1. 2 Business Committee signature required 5. Submission Authorization Melinda Dartaton, Vice-Chan. Authorized sponsor (choose one): Requestor (if different from above): 56 Name, Title / Dept. or Tribal Member Additional signature (as needed): Name, Title / Dept. Additional signature (as needed): Name, Title / Dept.
- 1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Oneida Tribe of Indians of Wisconsin

Post Office Box 365



Oneidas bringing several hundred bags of corn to Washington's starving army at Valley Forge, after the colonists had consistently refused to aid them.



Phone: (920) 869-2214

Onelda, Wi 54155



UGWA DEMOLUM YATEHE Because of the help of this Oneida Chief in cementing a friendship between the six nations and the colony of Pennsylvania, a new nation, the United States was made possble.

Quarterly Report to the Oneida Business Committee March 10, 2015

> Christopher Johns Gr. et ance Coordinator

FY 2015 Budg

Congress finally approved appropriation of FY2015. While there were modest increases in some areas, the Tribe's share did not significantly increase. Congress once again failed to address the cost to tribes of the sequest at in the triber cently occurred, and this failure means that all tribes continue to fall behind in their ability to keep pace with the rising costs of herein care.

President's FY 2016 Budget Requise

Following are some of the more agnificant items contained in the residue. Budget Request for FY 2016.

→ Mandatory Contract Support Costs: To abbilize long-term funding and address programmatic concerns with CSC, are 2016 budget also puts forth a proposal to reclassify these costs to mandatory funding beyoning in fisch treat 2017. Beginning the reclassification in 2017 will allow time for tribal consultation in 2.16 on operational details. The budget proposes to adjust the discretionary budget caps to reflect the reclassification. The current estimate for projected BIA program grows, above the discretionary cap amount, totals \$105 million for fiscal years 2017. 2019 and would be treated as a PAYGO cost for the authorizing legislation. New 1C estimates will be provided on a three-year cycle as part of the reauthorization process renacted, mandatory funding for contract support costs will help stabilize this vital funding for tribes and further self-governance and self-determination efforts. Additionally, Indian Affairs will continue to work with and consult with tribes to strengthen administrative capacity and program management. The FY 2016 request also will fully fund contract support costs, based on the most recent BIA and IHS analyses.

→ Significant Increases: The budget proposes an overall increase of 12 percent for BIA over the FY 2015 enacted level, the largest increase in more than a decade

(excluding Recovery Act funding). The Indian Health Service would receive a nine percent increase.

 \rightarrow Public Safety: The budget includes \$417.4 million for the Department of Justice (DOJ) public safety initiatives in Indian Country, which is a \$102 million increase compared to the FY 2015 DOJ enacted total for Indian Country.

→ The President's budget proposes \$4.0 million to establish a "One-Stop Tribal Support Center" to support Tribes in accessing hundreds of services across the Federal government.

→ The FY 2016 budget includes \$4.5 million to establish an "Indian Energy Service Center" to facilitate vital energy development in Indian Country.

→ In the BIA, a data initiative of \$12.0 million is proposed to establish an Office of Indian Affairs Policy, Plor am Evaluation, and Data which will help the Interior Department collect, analyse, and the evidence to support effective policy making and ogram implementation. The true as also will assist the Department in working with fribes to improve Interior and all duca quality and availability and will support efforts with the Census Bureau to Identify and address data gaps in Indian Country.

→ The BiA budget builds to the "Twahe Family) Initiative": a comprehensive and integrated approach to address the attended problems of poverty, violence, and substance a user accel by Indian communities. The FY1cbudget would provide \$15 million to expand the Tiwome Initiative, \$6 million more for Social Services, \$4 million more for law enforcement for an enables to incare ration, and \$5 million more for aid to tribal family course.

→ "Generation holigenous" is a cinitiative in the budge to address Native youth issues. The Generation Indicances, or Gen-3", initiative to test comprehensive approach to help improve the lives of and opportunities for Nation yourd. The initiative crosses multiple agencies, including the Denortments of the Interior Education ED), Housing and Urban Development (HUD), wealth and Human Services (HHS) Agriculture (USDA), Labor (DOL) and Justice (DOJ). Increases include: (1) \$34.2 m Jon of DOI to extend broadband internet and computer access to all BIE-funded schools and dormitories; (2) \$10 million at HUD and \$8 million at DOI to accrete teacher housing needs; (3) \$50 million at HHS to provide youth-focksed incluvioral mental health, and substance abuse services; and (4) \$53 million for Native Youth Community Projects at ED to support comprehensive strategies to improve collider and concernitations of Native youth.

→ Tax provisions: Treasury includes a proposal to exclude from income student loan forgiveness and certain scholarship amounts for participation in the IHS health professions program; a modification of the adoption tax credit to allow Indian Tribal Governments to make a status determination of a "child with special needs"; modifications of Tax Exempt Bonds for Indian Tribal Governments that include the repeal of the "essential government function" for tax exempt bond financing, and new flexibility for Tribal Economic Development Bonds. → The HHS Tribal Behavioral Health Grant (TBHG) program would receive \$30 million, including \$15 million in the Mental Health appropriation and \$15 million in the Substance Abuse Prevention appropriation as part of Generation Indigenous. With the expansion of the TBHG program, SAMHSA aims to reduce substance use and the suicide among Native youth and address conditions which impact learning in BIE schools. The TBHG program will support mental health promotion and substance use prevention for high-risk Native youth and their families, enhance early detection of mental and substance use disorders among Native youth, and increase referral to treatment.

→ The proposed Budget includes a \$70 million increase for the Purchased/Referred Care (formerly Contract Health) program to cover rising health care costs and to expand services provided through this important program, which funds care outside of IHS and tribal facilities when it is not available at an IHS or tribal facility.

→ **Carcieri**: Language to address the Carcieri Supreme Court decision is again included in the Department of Interior general provisions of the President's budget.

FY 2017 udget

The FY 2017 Budget Formulation or cless has begun. Regional meetings have been held. Each Region with make a cress intation to the sibal-Interior odget Council (TIBC) in March.

Contract Support Cons (CS / Laims Update

The Tribe has beeeved the settlement amount for this main, for unpaid CSC funds under the Self-Governance Compact with the indicative. There has not yet been a resolution to the Salazar v. Ramah Navajo Chapter class action suit for unpaid CSC runds under the SG Compact with the Department of Interior's Bureau of Indian Affairs.

Office of the Special Trustee (OST) Automated Audi Pilot Project

The OST Office of Trust Review and Judit (CLRA) is in the process of designing a "Tribal Methodology" system which would greatly simpley the statuterily-required judit distributions programs assumed by SG Tribes. This would open te as a voluntary, self-reporting computer-based system that would greatly reduce the cost of on-site visit for the OST and standardize report data for each participating tribe which should make the process more efficient. One da has agreed to be one of several sites for the testing phase of the Project which should be up and running by October 1, 2015.

Self-Governance Advisory Committee (SGAC)

The SGAC most recently discussed three concerns with Kevin Washimm, the Assistant Secretary -- Indian Affairs (AS-IA) and Tommy Thompson of the Bureau of Indian Affairs Office of Management and Budget (BIA-OMB) including speeding up the release of funding to Tribes, changes to the internal financial processes, and the budget formulation process for DOI. Self-Governance Tribes continue to urge BIA/DOI to work more quickly to get Tribal money out more quickly, especially CSC and programs outside the BIA. BIA indicated that they had made significant progress regarding funding distribution internally but when Congress fails to provide an entire year of appropriations, it leads to impediments in the timely distribution of funds to Tribes. The BIA also reported that anticipated changes to the current financial system will improve the speed in which the agency can get money out to Tribes. As a result, SGAC requested that DOI consider the impact changes to their finance system will have on Tribes. Tribes are also concerned that the current regional formulation process is not working well for Self-Governance Tribes and that often the priorities from TIBC are not represented in the President's Budget request. The AS-IA agreed that the current formulation process does not allow for maximum Tribal participation at the regional level and would be willing to consider changes.

Indian Health Service Director

Congress has yet to confirm the President's nominee - Dr. Yvette Roubideaux - as the Director of the Indian Health Service. There is a statutory limit on the amount of time a nominee can serve in an "Acting" position for which they are nominated. As a result, HHS Secretary Burwell has appointed Dr. Roubideaux as a Senior Advisor to the Secretary and appointed Robert McSwain, the current IHS Deputy Director, to serve as Acting Director. Mr. McSwain has previously served as the Director.

Bemidji Area Office Director

Mr. Keith Longie has been named the new Director of the Bemidji Area. He is an enrolled member of the Nortle Mountain, and of mippowa Indians. Mr. Longie is a Commissioned Officer in the U.S. Public Health Service and has held a variety of positions in the Portland and Phoenix Areas and served to IHS Chief Information Officer at the UHS' Headquarters in Rockville, MD.

Upcoming Events

Following an dates and locations of upcoming meetings and conferences as requested.

March 5-6, 2015	ſ
March 22-26, 2015	5
April 26-30, 2015	
May 20-21, 2015	T
July 20-24, 2015	Se
August 5-6, 2015	T
October 5-8, 2015	Se

rib rior Budget Cour Washington, DC f-Governance Advisory Commit es Mashington, DC nnual SG congultation Conference eno, NV al-Interic But ret Council /ashington, DC elf-Governalise Advisory Committees Valuington, DC ribal-Interior Budget (Juncil our arque NM elf-Governince Advisory Committees War ington, D

MRO/BIA and BAO/IHS Representatives

TIBC Representatives:

Primary:	Darrell Seki, Treasurer, Red Lake Band of a lopew Indians
Alternate:	Jimmie Mitchell, Natural Resources Dector, title Riter Band of Ottawa Indians

SGAC Representatives:

Primary:	Derek Bailey, Tribal Councilor, Grand Traverse	and
Alternate:	Vacant	

Technical Committee Members:

John Mojica, Mille Lacs Dave Connor, Red Lake Jessica Burger, Little River Barb Brodeen, Bois Forte Christopher Johns, Oneida

TSGAC Representatives

Primary:Derek Bailey, Tribal Councilor, Grand Traverse BandAlternate:Vacant

Technical Committee Members:

John Mojica, Mille Lacs Jessica Burger, Little River Christopher Johns, Oneida

Federal Of	ficials
Bureau of Indian Affairs	Indian Health Service
Kevin Washburn, Assistant Secretary Mike Black, Director	Mr. Robert McSwain, Acting Director Dr. Yvette Roubideau, Nominated
Diane Rosen, Director, Midwester Gron Tammi, Poitra, Duputy Director - Truck Scott Suctional, Deputy Pirector - Services	Mr. Keith Longie, Director, Bemidji Area
Kimberly Bouchard, Surerintendent, Greatinkes Gerald Walhovd, Dervity – unit Services	

Oneida Business Committee Meeting Agenda Request Form

1. Meeting Date Requested: 03 / 11 / 15

	OAHC 1st Quarter Report	
	Action requested (choose on	
	☑ Information only	
	□ Actic 1 - please describe:	
3.	Justification	
	Why BC action is required so i	nstantions):
A	Currenting Materiala	
4.	Supporting Materials	uired information (see instructions)
	\boxtimes Report \square Resolution	Contract (check the box below if signature required)
		media presentation aue to Tribal Clerk 2 days prior to meeting)
	1.	3.
	2.	4.
	Business Committee signature	e required
5.	Submission Authorization	
	Authorized sponsor (choose one):	Trish King, Tribal Treasurer
	Requestor (if different from above):	Janice Skenandore-Hirth
		Name, Title / Dept. or Tribal Member
	Additional signature (as needed):	Name, Title / Dept.
		Name, nue / Dept.

Additional signature (as needed):

Name, Title / Dept.

1) Save a copy of this form in a pdf format.

2) Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

Oneida Airport Hotel Corporation Radisson Hotel & Conference Center Quarterly Report For the quarter ended: December 31, 2014

Narrative Section

Business practice, market overview, place within market:

- STR report shows for the Quarter that we are down in Occupancy YOY -0.6% index, down in ADR YOY -1.0% index with a result of -1.6% index YOY for RevPar; group rooms had shortage in New mber due to funding getting cut from a large group & we discounted transient received up the loss of occupancy.
- Salo com Booking act for Yest Q1 includes an increase of +\$3.61 ADR per room; an increase of 4,888 room 20 Y an increase of room revenue of \$511,262 and F&B revenue increase of \$77,596 for a overall Revenue Booking Pace increase from prior year of +\$588.551.

Competitive and sis:

• Casino food or deta ponti de to be a source of composition for F&B revenue

Strategies for improved value:

- Developed packaging in Corporate/use ciation Meanger four & Travel, Wedding, Travel Agents and Transient markers to gein short term bookings for Q1 and other need months
- Working with Radisson Corporate in all Stress markets to gain leverage and recognition
- Working closely with Green Bay CVB in all Sales markets to keep top of mind and participate in all sales initiatives
- Conducted a Client Appreciation event in Notember with our corporate clients
- Conducted a Holiday/Corporate Meeting sale, blitz in Cetover; Sales team visited over 100 companies in the Green Bay area cross promoting Wingate as well

Material changes or developments in market/business:

• KI Convention Center Expansion began Dec 6, 2013, adding 23,00° square ft of meeting space total of 80,000 when complete, due to complete September 1, 2015

Market growth:

- Northland Hotel to start construction on 147 room hotel downtown to compliment the KI Convention Center expansion, due to open fall of 2015
- Former Clarion Hotel downtown, has started construction to become a Hampton Inn due to open fall of 2015 with 146 rooms.
- Residence Inn 100 room property set to build and open 2016 in Green Bay

• 77 room Staybridge Suites will be under construction and attached to Brett Favre's Steakhouse

Pending legal action:

• One (1) Personnel issue.



Oneida Airport Hotel Corporation Three Clans Airport, LLC Quarterly Report For the quarter ended: December 31, 2014

Business practice, market overview, place within market:

- Ranking for the 1st quarter from the STR Report the Wingate is 4 out of 5 based on REVPAR and 3 out 5 based on Occupancy. Wingate is continuing to gain and maintain fair market size the within the competitive set.
 For the start of the
- For the start of the conjunct of the radius of the radiu

The Vingate averages a 60.2% coupancy for the 1st quarter. This was a 3.44% increase year over year. Due to the consistent high occupancy the Wingate is notforming at, the strategy for the second half of the 1st quarter was to increase rates, w that there is base beginss.

Competitive anal sig

- The competitor rest a PEUPAR of \$93.70 to 2 ingate's \$60.03 for the quarter.
- The competitive set 1 ad an ocupancy of 6.1% to the Wingate's 69.2%.

Strategies for improved value

- The Wingate's strategy to continually grow rate is to yie drate soones especially on high demand dates and to effer less neavily discounted rates on slov dates. Discounted rates will still be offered for slower lates but not us much as they have been in the past.
- The Wingate continues to use all 3rd party pooking channels to increase bookings on low demand dates.
- The Wingate continues to run last minute deals or Dipeida and hotels.com over the weekend to increase last short term booking.
- The Radisson Sales team continues to cross sell the hotel. We have begun to see new leads and new bookings from this joint effort.

Material changes or developments in market/business:

- We continued to see an increase in the corporate business traveler over the 1st quarter. The corporate travelers are now booking their negotiated rates and not buying down to the deeply discounted rates.
- We were able to negotiate rates with some new local companies to add to the Wingate's preferred list.

Market growth:

- For the 1st Quarter, the Wingate ended with 64.1% REVPAR Index. The Wingate was able to grow 3.4% year over year for REVPAR Index.
- For the 1st Quarter, the Wingate ran an average of 69.2% which is an increase of 6.1% year over year.
- Revenues for the first quarter were \$436708 which were up from prior year by \$17757. This was achieved better within our competitive set and the Green Bay Market.



Oneida Business Committee Meeting Agenda Request Form

1	Meeting Date Requested: 03 / 11 / 15		
Ζ.	Nature of request Session: 🗵 Open 🛛 📋 Executive - justification required. See instructions for the applicable laws that		
	define what is considered "executive" information, then choose from the list:		
	Agenda Header (choose one): Report		
	Agenda item title (see instructions):		
	OGE 1st Quarter Report		
	Action requested (choose on		
	⊠ Information only		
	Action - please describe:		
•			
3.	Justification		
	Why BC action is required (see instantions):		
4.	Supporting Materials		
	☐ Memo of explanation with required information (see instructions)		
	🗵 Report 🛛 Resolution 🔲 Contract (check the box below if signatur, required)		
	□ Other - please list (Note: multi-media presentation, due to Triba Clerk 2 days prior to meeting)		
	1. 3.		
	2. 4.		
	Business Committee signature required		
5.	Submission Authorization		
	Authorized sponsor (choose one): Trish King, Tribal Treasurer		
	Requestor (if different from above): Janice Skenandore-Hirth		
	Name, Title / Dept. or Tribal Member		
	Additional signature (as needed): Name, Title / Dept.		
	Additional signature (as needed):		
	Name, Title / Dept.		

Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org
Oneida Golf Enterprise Corporation

Fiscal year-To-Date Reporting for period Ending:

December 30, 2014

The Oneida Golf Enterprise Corporation (OGEC) is a corporation of the Oneida Tribe of Indians of Wisconsin established to oversee and manage the business known as Thornberry Creek at Oneida.

Narrative Report

Business Practice, Market Over ew Place Within Market:

- Thorn erry Creek at the day, CO) to 27 hole premier golf course settled in a rural prestigious housing development on the Oneida reservation.
- O is the pricial Golf Course of the Green Bay Packers.
- TCL is the of 10 Official PGA Pumily golf courses in Wisconsin offering tees for kids based upon skill evel an ability to further mjoy be game.
- TCO include a balquet acility, driving range, pro shap and sports pub & grill.
- TCO is considered a frader in charitable golf event nosting and a leading choice for future brides.
- TCO hosted 37 special events in the first quarter.

Competitive Analysis:

GOLF

- Area competitors in this market for public golf conses with 9 holes is Vienge Green in Howard; 27 holes includes Mid Vallee in De Pere; 18 holes include Brown County and Crystal Springs.
- Outside our 10 mile radius are other courses such as usual St. Patrick's, Ledgeview, The Woods, and Northbrook. Courses outside the area out stin considered competitors include The Bull in Sheboygan and Blackwolf Run/Whistling St. pits in Kinler.
- Our nearest competitor, Brown County Golf Course, had to apporary givens due to winter damage and have renovated and rebuilt all their greens to be reader or opening in spring 2014. New greens opened May 15, 2014 at Brown County and their business has had a negative impact on our rounds.

Banquet

• Thornberry Creek at Oneida has 7,000 square feet of flexible banquet space with large windows and spectacular views. This space will accommodate 300 guests at round tables.

We also future a porch room that accommodates up to 40 people. Included in this space is the Cupola available for smaller wedding ceremonies and intimate gatherings.

- Thornberry Creek at Oneida also offers a full range of conference equipment and amenities for the corporate client.
- Our large bar and grill offers full coverage on all 24 HDTV flat screen televisions for all sporting events and seats up to 160 customers.
- Area competitors in this market include Rock Gardens and the Radisson.

Strategies for Improved Value:

• *Golf Digest* has continue a to speak about considering TCO for "Best in State" honors to be upplished in 2017 ratio g would occur over next two years.

TCO vas featured on the cover of *themiere Golf Destinations Magazine*, February 2015 issue. TCO will also be featured inside the publication with a 3 page editorial.

- Publication or aches 4 surrounding states and will be available in all travel centers, are so, image circulation is 96, 20. Debugs at the Cities Golf Expo
- Thank You N.E.Y romo generated a ajon traffic, 450 tee times reserved in 12 hours
- Our "Taske got the observe" event saw lose to 100 potential clients; well received
- Packer Radio Shows ontinue to grow in Attendary and evenue
- Thornberry Fall assignme ag in sold out to 1.8 players in a cold rainy day in November
- Halloween Cash Bash had over 00 in attendance and vas y well received
- Ugly Sweater Party has over 100 as undance and the room intum continues to grow
- New strategic pricing has been purinto place to offer access to multiple budgets different times of the day
- Season pass prices have been recuced to encourage purchases and drive traffic: as of December 31st we've grown pass safe by 15% over last year
- New tee signs, tee markers, trash recepticles practice tee necessities and benches have been received and will be unveiled at the surt of our 20 is reason.
- Our social media presence continues to grow and gran teres, 283 new users engaged in the 1st quarter.
- New signage has been added to the entryway allowing for granter visual existence
- Using the Packer partnership we have grown revenues the ugh ticket promotions by over \$20,000
- Working with Joint Marketing on new events and ideas that include the Radisson and Oneida Casino. We are committed to working with our partners for the good of all.
- Started working on 2015 Marketing Calendar to be used with our partners to generate more interest in our special events
- Was the featured facility in *Wedding Magazine* showcasing our grounds and reception venues

- New website was being built and anticipated launch in early 2015; focus on ease of access and create a more visually appealing product
- Spending necessary marketing dollars and are starting to gain recognition in multiple markets

Material Changes or Developments in Market/Business:

- Facility improvements of tinue to occur including the parking lot and additions in signage and guardrail systems rise anticipated to be installed in 2015
- we winter mend released or using more on creating a consistent food product but also shrinting the total oriening to all w for assistance in food cost reductions
- Debuted at the "Friends of Thumberry" event, we launched a few new signature items such as the tracho's, Cheese Steer pizza and Cheken Parmesan sandwich
- New Service Training 101 is being developed to be trained to all staff regarding our expectations and what "service" is
- Old equipment is being sold and creating additional secure to the facility
- Our new truck a choice string good use at the facility and at the Wingate
- We've made an effor to bid or additional plow jobs in the area
- We have started interviewing for the position of Ner-Busines Development Manager, this is a new position who's exponsibility will be solely generating new ausiness to the facility
- Our new website is almost complex, and will be launched in party 2015
- Special event menus have been finalized and the increases will have a significant financial impact on the facility in 2015
- Verbiage for the new bronze plaques ound on the course is under by zod will be installed in the Spring of 2015

Market Growth:

- Our social media presence continues to grow, up 26% since May of 2014 and up 12% in the first quarter of 2015
- We have added 7 new golf outings to the 2015 schedule
- Weddings continue to grow and only 1 Saturday remains in inventory for the 2015 peak season
- Weddings for 2016 are being booked and showings for 2017 have started
- We have begun exploring the rebranding of the restaurant which will be necessary for its growth and sustained success in the future

- Josh Doxtator and Zach Knight have become members of Current YP in Green Bay
- They have attended a few events and have already booked a few events from the contacts made within the group and a major event in August for the YP group
- Mark Becker was named as a "Master Kids Teacher" by US Kids Golf, their highest designation for teaching juniors
- Communication has begun with local hotels to offer Thornberry as their preferred golf and restaurant space



Page 329 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	. Meeting Date Requested: 03 / 11 / 15	
2.	Nature of request Session: 🗵 Open 🛛 Executive - justification required. See instructions for the applicable laws that	
		lered "executive" information, then choose from the list:
		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
	Agenda Header (choose one):	
	Agenda item title (see instructions):	
	Bay Bancorporation Inc.	
	Action requested (choose one)	
	Action - please, escribe:	
3.	3. Justification	
	Why BC action is required seconstructions	
_		
4.	4. Supporting Materials Follow-up will ⊠ Memo of explanation with required info	
		ac. (check he box below if signature required)
		entation due to Trikel Clerk 2 days prior to meeting)
	1.	3.
	2.	4.
	Business Committee signature required	
5.	5. Submission Authorization	
	Authorized sponsor (choose one):	
	Requestor (if different from above): Jeff Bowm	an, President
		/ Dept. or Tribal Member
	Additional signature (as needed): Name, Title	/ Dept
	Additional signature (as needed):	
	Name, Title	/ Dept.

From:Mary C. GravesSent:Friday, March 06, 2015 9:17 AMTo:Jeff BowmanCc:Chad A. WilsonSubject:RE: BC Agenda Review: Follow-up (Due 9:00 am Friday) - Bay Bancoporation Inc - 1st
Quarter Report (Oct to Dec 2014) Not Sumbitted

Will do Jeff.

From: Jeff Bowman [mailto:jeff.bowman@baybankgb.com] Sent: Friday, March 06, 2015 9:12 AM To: Mary C. Graves
Subject: RE: BC Agenda Review: Follow-vo (Duct 90 am Friday) - Bay Bancoporation Inc - 1st Quarter Report (Oct to Dec 2014) Not Symbittee
Hi Mary,
We'll put the package logs the right away. I checked with Stave or CFO, and he never received the reporting calendar for 2015.
I thought that a reporting deacane was cruning up, but I didn't have the capadar for 2015 either.
Can you have someone send that to me mease?
We'll get a package ready by the end of today, or Moraday morning at the very latest.
Sorry for the delay.
Jeff
From: Mary C. Graves [mailto:MGRAVES@oneidanation.org] Sent: Friday, March 06, 2015 8:54 AM To: Jeff Bowman (jeff.bowman@baybankgb.com) Subject: FW: BC Agenda Review: Follow-up (Due 9:00 am Friday) - Bay Bancorp.ration Iru - 1st Quarter Report (Oct to Dec 2014) Not Sumbitted

From: BC_Agenda_Requests
Sent: Thursday, March 05, 2015 4:13 PM
To: Cristina S. Danforth
Cc: Mary C. Graves
Subject: BC Agenda Review: Follow-up (Due 9:00 am Friday) - Bay Bancoporation Inc - 1st Quarter Report (Oct to Dec 2014) Not Sumbitted

Good afternoon:

Tina,

I wish things are going along good.

At BC Agenda Review we need a follow-up on agenda item XV.B. Baybancorporation Inc.

Could you please research where the Bay Bancoporation Inc - 1st Quarter Report is at? And let me know if it is going to be deferred or forward me the report.

Please submit your findings to me by 9:00 a.m. tomorrow, Friday March 06, 2015.

Thanks,

Chad W.



Page 332 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 14	
2 .	Nature of request Session: ⊠ Open □ Executive - justification required. See instructions for the applicable laws that	
	define what is considered "executive" information, then choose from the list:	
	Agenda Header (choose one):	
	Agenda item title (see instructions):	
	Defer Oneida Total Integrated Enterroises Report to the 03/25/15 BC meeting	
	Action requested (choose one)	
	Action - please escribe:	
	Motion to defer to me March 25, 201. Business Coopmittee Regular Meeting	
	Wotion to denote a the watch 20, 20 he business committee Regular Meeting	
3.	Justification	
	Why BC action is required seconstructions):	
	Regular quarterly reporting	
4.	Supporting Materials	
	Memo of explanation with required information (see instructions)	
	 Report Resolution Contract (check the box below if stypet the required) Other - please list (Note: multi-media presentation, due to Trike) Clerk 2 days prior to meeting) 	
	1.	
	2. 4.	
	Business Committee signature required	
5.	Submission Authorization	
	Authorized sponsor (choose one): Tehassi Hill, Council Member	
	Requestor (if different from above): Wilbert "Butch" Rentmeester	
	Additional signature (as needed):	
	Name, Title / Dept.	
	Additional signature (as needed): Name, Title / Dept.	

From:	Wilbert Rentmeester <wrentmeester@otie.com></wrentmeester@otie.com>	Sent: Tue 2/24/2015 8:51 AN
Го:	BC_Agenda_Requests	
Cc	Heather Cotey; Nick Ni; Ronald W. Hill; Jennifer A. Webster; Lisa A. Liggins	
Subject:	First Quarter report	
Dear Or	One ida Business Committee,	
	espectfully request that the first quarter report scheduled for the March 3, 2010 and tharch 1, 2015 Business Committee M ess Committee meetings. OTIE will present the \$200,000 payment to the Tribe at the Worch 25, 1, 215 meeting	Meetings be deferred until the March 24, 2015 and March 25,2015 Oneida
Sincerel		
and the second second second	rt Rentmeester	
	tive Vice President	
	la Total Integrated Enterprises (OTIE)	
and the second se	Packerland Drive	
	n Bay, WI 54313	
	34.3966 direct	
	i4.5598 mobile	
	14.3989 fax	
www.ot	<u>otie.com</u>	
Wildlage	OTIE 25	ſ.
677	DELIVERING CUSTOMER SERVICE	
they want		▲
Enginee	eering, Science and Construction	
4		

Page 334 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	1. Meeting Date Requested: 03 / 11 / 15	
2.	Nature of request Session: 🛛 Open 🛛 Executive - justification required. See instructions for the applicable laws that	
	define what is considered "executive" information, then choose	
	Agenda Header (choose one):	
	Agenda item title (see instructions):	
	Defer Oneida Engineering Science & Construction Group, LLC to the 03/25/15 BC meeting	ıg
	Action requested (choose one)	
	Information only	
	Action - please escribe:	
	Motion to defer to the March 25, 201. Business Committee Regular Meeting	
2	3. Justification	
э.	Why BC action is required seconstructions):	
	Wity DC action is required be institutions).	
	Degular guartarly reporting	
	Regular quarterly reporting	
4.	4. Supporting Materials	Instructions
	 Memo of explanation with required information (see instructions) Report Resolution Contract (check he box below if superior required) 	ine d)
	 Report Resolution Contract (check de box below if signature requinance) Other - please list (Note: multi-media presentation due to Trikel Clerk 2 days prior to the second details of the seco	
	1 3	
	2. 4.	
	Business Committee signature required	
5.	5. Submission Authorization	
	Authorized sponsor (choose one):	
	Requestor (if different from above): Wilbert "Butch" Rentmeester	
	Name, Title / Dept. or Tribal Member	
	Additional signature (as needed): Name, Title / Dept.	
	Additional signature (as needed):	
	Name, Title / Dept.	

From:	Wilbert Rentmeester < WRentmeester@otie.com>
Sent:	Friday, March 06, 2015 9:46 AM
То:	Lisa A. Liggins; BC_Agenda_Requests; Ronald W. Hill; Jennifer A. Webster
Cc:	Nick Ni; Heather Cotey
Subject:	FW: First Quarter report

I inadvertently forgot to include that the OESC Quarterly report also be deferred until the March 24 and 25 OBC meeting. Please allow the requested deferment to apply to both the OTIE and OESC report.

Sincerely,

Wilbert Rentmeester *Executive Vice President* **Oneida Total Internace Enterprises** (1920.884.3966 dir et www.otie.com

From: Wilbert Rentmeers . Sent: Tuesday, February 2, 1915 8:51 AM

To: 'BC_Agenda_Requests@meidar.ar on.org' Cc: Heather Cotey (<u>hcotey@ptie.com</u>) Nick Ni; 'rhill7@on idanation.org: \iwebste1@oneidanation.org'; 'Lisa A. Liggins' Subject: First Quarter report

Dear Oneida Business Committee,

OTIE respectfully request that the first quester report who buled for the March 3 (2015 and March 4, 2015 Business Committee Meetings be deferred until the March 4+, 2015 and March 25, 2015 Cheir's Business Committee meetings. OTIE will present the \$200,000 payment to the Tribecht the March 25, 2015 Treema

Sincerely,

Wilbert Rentmeester *Executive Vice President* **Oneida Total Integrated Enterprises (OTIE)** 2555 Packerland Drive Green Bay, WI 54313 920.884.3966 direct 920.664.5598 mobile 920.884.3989 fax www.otie.com



Engineering, Science and Construction

	Page 336 of 340		
	Oneida Business Committee Meeting		
	Agenda Request Form		
	03 / 11 / 15		
1.	Meeting Date Requested: 2== 7 25= 7 15=		
2.	Nature of request		
	Session: 🛛 Open 🛛 Executive - justification required. See instructions for the applicable laws that		
	define what is considered "executive" information, then choose from the list:		
	Agenda Header (choose one): BCC Report		
	Agenda item title (see instructions):		
	Land Claims Commission BC Quarterly Report: 2014 4th Quarter Report October-December, 2014		
	Action requested (choose one)		
	☑ Information cely		
	Action - please escribe:		
2	Justification		
υ.	Why BC action is required seconstructions):		
	with be action is required be anisotrolity.		
4.	Supporting Materials Instructions		
	Memo of explanation with required information (see instructions)		
	Report Resolution Contract (checkede box below if streature required)		
	Other - please list (Note: multi-media presentation due to Trik Clerk 2 days prior to meeting)		
	13		
	24		
	Business Committee signature required		
5.	Submission Authorization		
	Authorized sponsor (choose one): Brandon Stevens, Council Member		
	Requestor (if different from above): Jennifer Stevens, LCC Executive Assistant		
	Name, Title / Dept. or Tribal Member		
	Additional signature (as needed): Name, Title / Dept.		
	Additional signature (as needed):		
	Name, Title / Dept.		

ONEIDA LAND CLAIMS COMMISSION QUARTERLY REPORT

October 1 to December 31, 2014

Submitted by Jennifer M. Stevens, LCC Executive Assistant I 490-3955

LAND CLAIMS COMMISSIONERS:

Chair-Amelia Cornelius, Vice-Chair-Loretta V. Metoxen, Secretary/Treasurer-Rita Summers, Newly Elected Commissioners: Dakota Webster and Michael Hill. New BC appointed Commissioner; Donald McLester. Brandon Stevens/Jenny Webster, BC Chair and LCC Liaison. (Note: Need to do Internal Elections when we have a full board-waiting for another BC appointment to be completed.)

PURPOSE:

"Make recommendations to the Business Committee on ways to foster General Tribal Council participation in the decision making process regarding the settlement whe Oneida land claims in New York State. Article I-Authority 1-b ~LCC By-Laws

• Special Note: The Land Claims Compassion is an educational resource and liaison for the Oneida community regarding the New York Land Claims can all history. The Land Claims Commission strives to provide educational opportunate to learn more thout the Upston New York in order for the G.T.C. to make better informed and a second ecisions. Although there is a sensitivity and confidentiality to our case, the Land Claims Commission continue to struct to assist G.T.C. there is a sensitivity and better informed in the New York Land Claims case to the least of our knowledge and resourced provided for us.

OBJECTIVES:

The Oneida Land Claims Commission is an elected governing body are will still strive to meet the needs of the GTC. The Land Claims Commission lince has been invited to atter the to Executive Sessions. Jennifer Stevens completed her training with Mick Osterburg October 7, 2014 and developed a New York Land Claims On-Base Project Manual-waiting for the LCC a wew-first draft completed December 18, 2014. Donald McLester was given a BC appointment to be on the LCC at their 6th commission of a context waiting for the 7th BC appointed commissioner.

MEETINGS:

Land Claims Commission has been meeting once a month of a regular basis as scheduled, the third Thursday of every month at 5PM.

FOLLOW UP:

Questions asked during our September 2014 BC Quarterly Report:

Who is the LCC Liaison and who is the current New York Land Claims Assistant who remaced Diane House?

- Answer: Lisa Summers alternate: Jenny Webster (This needs clarified-it-suys Brandon Stevens/alternate, Jenny Webster.
- LCC requests a BC and LCC Meeting: Agenda-New York Land Claims Update.
- Lisa Summers, Tribal Secretary will be in contact with the LCC to set up NYLC Update Meeting. (There is no follow-up on this request.)

Page 338 of 340 Oneida Business Committee Meeting Agenda Request Form

1.	Meeting Date Requested: 03 / 11 / 15	
	Nature of request Session: ⊠ Open □ Executive - justification required. See instructions for the applicable laws that	
	define what is considered "executive" information, then choose from the list:	
	Other - type reason	
	Agenda Header (choose one): BCC Report	
	Agenda item title (see instructions):	
	Land Commission 1st Quarter Report	
	Action requested (choose one)	
	Action - please, escribe:	
	Accept Report	
3.	Justification	
	Why BC action is required seconstructions):	
	It is on the reporting schedule	
4.	Supporting Materials Instructions Image: Memo of explanation with required information (see instructions) Image: Memo of explanation with required information (see instructions) Image: Memo of explanation with required information (see instructions) Image: Memo of explanation with required information (see instructions) Image: Memo of explanation with required information (see instructions) Image: Memo of explanation with required information (see instructions) Image: Memo of explanation with required information (see instructions) Image: Memo of explanation (see instructions)	
	□ Other - please list (Note: multi-media presentation, due to Trib) Clerk 2 days prior to meeting)	
	1.	
	2. 4.	
	Business Committee signature required	
5.	5. Submission Authorization	
	Authorized sponsor (choose one): Patrick Pelky, Division Director/EHS	
	Requestor (if different from above): Lori Elm, DOLM Office Manager	
	Name, Title / Dept. or Tribal Member	
	Additional signature (as needed): Name, Title / Dept.	
	Additional signature (as needed):	
	Name, Title / Dept.	

- Save a copy of this form in a pdf format.
 Email this form and all supporting materials to: BC_Agenda_Requests@oneidanation.org

FIRST QUARTER IMPRESSION: This quarter has been filled with expectations and information on how to achieve goals in an efficient and effective manner.

New Member; Bart Cornelius, is a great asset to the commission bringing in new vibrant qualities. Overall, expectations are quite high for FY 2015 Land Commission.

The moratorium is still in place until April, 2015. Procedural exceptions have been granted by the OBC and the Land Commission will continue to move forward in acquiring land to benefit the *Tribe* and Tribal members.

Oneida Lang commission Membe

Officer

Amelia Cometius Char Rae Skenandore, Wsc Lloyd Powless, Secr. (ar

Members:

Donald McLester Sherrole Benton Bart Cornelius Vacant Position

BC Liaison: Ron "Tehassi" Hill, Jr., OBC Alternate: Jennifer Webster, OBC

LAND ACQUIRED

- 12.77 acres 1ST quarter
- 8 DREAM homes (1 Reacquired)
- 1 OHA home
- 2.04 Vacant land
- Currently Own 38.3% Or 25,072.48 Acres Of Original Reservation 65,400 Acres

MISSION "The Land Commission is an r ected body of policy makers for the Division of Land Nanage at nt, overseeing Land acquisition, Leases, Loan approvale, Lond use, probates and at Lother tribal Land issues "

M eting

Regult's Meetings hold the 2nd Monday of each month; setuisition Meetings held the 4th Monday of each month

Strategy Meetings were help once a month for 3 months

8 Meetings were held:

- 3 Regular
- 2 Acquisition
- 3 Strategy
- 0 Hearings

Land Commission-(Oct.Dec, 2014)

BC Collaborations

Next Joint meeting with Business Committee is scheduled for Thursday, March 19, 2015 at the BCCR

POINTS OF INTEREST

- strategies for I and acquisition-met with; Oneida Housing Authority, Development Division, Division of Land Management, Retail, gaming, Oneida Nation Farms, internal services and Environmental to assist in developing strategies
- Developing specific considerations for acquisitions ۲
- Developing an SOP or meeting cancel lations ٠
- Scheduling ummit to review a categies for the new York properties
- Review Land Commission Py-Laws to ensure they accurately reflect duties and sponsibil itie
- Reviewing processes and agreements to ensure that decisions are made in the best interest of the tribe
 - acquisition Process & considerations Lease agreements and processes $\sqrt{}$
 - √
- Contributing to the amendments and development of;
 - ✓ Real Property Law
 - ✓ Leasing Law

Land commission oversight

- **Residential Leases**
- **Commercial Leases**
- **Agricultural Leases**
- Land Use Agreements
- Easements
- Mortgage Loans
- Home Equity Loans
- AND all other Land issues that may arise

Communication to Community



- External Funding Received for Demolition
- \$47,732 from Brownfield Grant for total cost of Demolitions
- \$14,258 received for recycling materials Costs Covered Include:
- \$21,424 for Tribal staff & equipment use
- \$47,732 tank Removal & disposal, consultant fees & fencing

BC Action Needed: Request BC to accept the Land Commission 1st Quarter report